#### FRANCHISE DISCLOSURE DOCUMENT



Sylvan Learning, LLC, a Delaware limited liability company 2350 Airport Freeway, Suite 505, Bedford, TX 76022 (410) 843-8000 franchise@sylvanlearning.com www.sylvanfranchise.com

**Sylvan Learning:** The franchisee will operate a Sylvan-branded learning center with a system designed for specialized assessment and teaching of individualized educational programs for children in the principal areas of reading, mathematics, writing, and test preparation.

**SylvanSync:** The franchisee will operate a portable SylvanSync and Sylvan-branded learning environment individualized for children, using proprietary SylvanSync computer systems and the Internet, alongside the Sylvan Learning Center business. In addition, the franchisee may carry the SylvanSync system outside of the Sylvan Learning Center to provide services at one or more additional approved short-term or permanent locations in the protected territory granted by Sylvan. The principal areas of instruction are reading, mathematics and writing.

The total investment necessary to begin operation of a Sylvan and SylvanSync business is \$100,752 to \$226,842. This includes \$36,900 that must be paid to us or an affiliate.

We may offer to enter into a multi-territory incentive plan agreement to establish and operate two to three Sylvan franchised businesses at specific locations under individual franchise agreements. The total initial investment necessary under the development agreement for two to three Sylvan centers ranges from \$201,504 to \$680,526. This includes \$56,900 to \$76,900 that must be paid to us or an affiliate.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.** 

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact Joshua Wall, Sylvan Learning, LLC at 2350 Airport Freeway, Suite 505, Bedford, TX 76022, or Franchising@UnleashedBrands.com.

The terms of your contract will govern your franchise relationship. Don't rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1.877.FTC.HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at <a href="https://www.ftc.gov">www.ftc.gov</a> for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: March 8, 2024, as amended May 10, 2024.

# **How to Use This Franchise Disclosure Document**

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits, or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit H.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit G includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Sylvan Learning business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a Sylvan Learning franchisee?	Item 20 or Exhibit H lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

# What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

**Business model can change**. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

**Renewal**. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

When your franchise ends. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

# **Some States Require Registration**

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit F.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

# Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The franchise agreement requires you to resolve disputes with the franchisor by arbitration and/or litigation only in Texas. Out-of-state arbitration or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to arbitrate or litigate with the franchisor in Texas than in your own state.
- 2. <u>Mandatory Minimum Payment</u>. You must make minimum royalty or advertising fund payments, regardless of your sales levels. Your inability to make the payments may result in termination of your franchise and loss of your investment.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

# NOTICE REQUIRED BY THE STATE OF MICHIGAN

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU:

- (a) A prohibition on the right of a franchisee to join an association of franchisees.
- (b) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (c) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) The term of the franchise is less than 5 years and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- (e) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- (g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
  - (i) The failure of the proposed transferee to meet the franchisor's then current reasonable qualifications or standards.
  - (ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
  - (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.

- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- (i) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to: Department of the Attorney General, Consumer Protection Division, 670 Law Building, 525 West Ottawa Street, Lansing, Michigan 48933, 517-335-7622.

\* NOTE: NOTWITHSTANDING PARAGRAPH (f) ABOVE, WE INTEND TO, AND YOU AGREE THAT WE AND YOU WILL, ENFORCE FULLY THE PROVISIONS OF THE ARBITRATION SECTION OF OUR AGREEMENTS. WE BELIEVE THAT PARAGRAPH (f) IS UNCONSTITUTIONAL AND CANNOT PRECLUDE US FROM ENFORCING THE ARBITRATION PROVISIONS.

# TABLE OF CONTENTS

<u>ITEM</u>		<b>Page</b>
Item 1: The	Franchisor And Any Parents, Predecessors And Affiliates	1
	ness Experience	
	ation	
_	Kruptcy	
Item 5: Initi	al Fees	g
Item 6: Othe	r Fees	10
Item 7: Estir	nated Initial Investment	
Item 8: Rest	rictions On Sources Of Products And Services	20
Item 9: Fran	chisee's Obligations	22
Item 10: Fin	ancing	24
Item 11: Fra	nchisor's Assistance, Advertising, Computer Systems, And Training	24
	ritory	
Item 13: Tra	demarks	40
Item 14: Pat	ents, Copyrights, And Proprietary Information	43
	ligation To Participate In The Actual Operation Of The Franchise Business	
	strictions On What The Franchisee May Sell	
Item 17: Rei	newal, Termination, Transfer And Dispute Resolution	47
	olic Figures	
	ancial Performance Representations	
Item 20: Ou	ttlets And Franchisee Information	54
Item 21: Fin	ancial Statements	63
Item 22: Con	ntracts	63
Item 23: Re	ceipts	63
STATE-SPEC	FIFIC APPENDIX	
<b>EXHIBITS</b>		
Exhibit A	License Agreement for Sylvan and SylvanSync, Attachments, and State-S <sub>I</sub> Amendments	pecific
Exhibit B	Sylvan Edge Package License Agreement	
Exhibit C	ACE IT! Tutoring System License Agreement	
Exhibit D	Multi-Territory Incentive Plan Amendment	
Exhibit E	Operations Manual Table of Contents	
Exhibit F	List of State Administrators and Agents for Service of Process	
Exhibit G	Financial Statements	
Exhibit H	Franchises in Operation, Franchisees Who Have Signed But Not Yet Opened,	
	Franchisees Who Have Left the System, and Affiliate-Owned Locations	
Exhibit I	Sample Form of General Release	
Exhibit J	Franchise Disclosure Questionnaire	
Exhibit K	State Effective Dates	
Exhibit L	Receipts	

#### ITEM 1: THE FRANCHISOR AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language of this disclosure document, "Sylvan," "us" or "we" means Sylvan Learning, LLC, the franchisor. "You" means the person, corporation, partnership, or other entity who buys the franchise. We use "corporation" to refer to any of a corporation, partnership or limited liability company, and "shareholder" to refer to any of a shareholder, partner, or member. We call our franchise agreement the "License Agreement" and we call our franchisees "licensees." We use both the "franchisee" and "licensee" terms interchangeably throughout this disclosure document. Our Sylvan Learning License Agreement for the Sylvan System and the SylvanSync System is Exhibit A to this disclosure document.

This disclosure document describes the offer of franchises to operate using Sylvan's proprietary Sylvan System and SylvanSync System at Sylvan Learning Centers (each a "Center" and collectively the "Centers") and other approved locations. The SylvanSync System and Sylvan System are separate licenses (which are licensed under the same agreement) and systems, but products and services of both are offered alongside each other from the Centers and other approved locations. These terms and concepts are further described later in this disclosure document.

#### The Franchisor, Our Parents and Our Affiliates

Sylvan is a Delaware limited liability company converted from Sylvan Learning, Inc. on June 18, 2015, and does business as Sylvan Learning Centers, Sylvan Learning, SylvanSync, and Sylvan. Our principal business address is 2350 Airport Freeway, Suite 505, Bedford, TX 76022. Our agents for service of process are disclosed in Exhibit F.

Our direct and indirect subsidiaries include Educate Corporate Centers Holdings, LLC ("ECCH"), Learning System of the Future, LLC ("LSF"), Learning Partnerships, LLC ("LP"), Maryland Learning Centers, LLC ("MLC"), and Sylvan In-Home, LLC ("SIH"). ECCH is MLC's parent company. MLC operates the type of Sylvan System and SylvanSync System businesses that you will operate. LSF has invested in and is developing the Learning System of the Future platform (the "LSF Platform") used in the SylvanSync System. LSF licenses the LSF Platform to Sylvan, which in turn licenses the SylvanSync System to franchisees. LP has invested in and is developing the Sylvan Edge Packages (the "EDGE Packages") used in delivering Sylvan Edge robotics, computer programming, engineering and mathematics learning programs and other licensed intellectual property. LP licenses the EDGE Packages to Sylvan, which in turn licenses the EDGE Packages to franchisees. SIH operates an online marketplace (the "Marketplace Platform") to enable parents to seek tutors specializing in fields of study where their students may need additional assistance. SIH has never operated a Center, and it has never offered franchises in any line of business. The principal business address of each of the entities above is 350 Airport Freeway, Suite 505, Bedford, TX 76022.

On February 16, 2024, we became a wholly owned subsidiary of Unleashed Brands, LLC ("Unleashed Brands"). Unleashed Brands' parent company is Leviathan Intermediate Holdco, LLC, which is owned by UA Holdings, LLC ("UA Holdings"). Unleashed Brands Foundation, the charitable affiliate of Unleashed Brands, is a Texas based nonprofit corporation which conducts certain charitable activities. Unleashed Brands, Leviathan Intermediate Holdco, LLC, Unleashed Brands Foundation, and UA Holdings share a principal business address at 2350 Airport Freeway, Suite 505, Bedford, Texas 76022.

Neither Sylvan nor its affiliates or predecessors have offered or offer franchises in other lines of business. However, we have affiliates that offer franchises in other lines of business. All of the affiliates listed below share a principal business address at 2350 Airport Freeway, Suite 505, Bedford, Texas 76022:

UATP Management, LLC ("UATP") offers URBAN AIR ADVENTURE PARK franchises, which are
venues for recreational activities, birthday parties, and other group events featuring the Adventure
Park Attraction package. UATP began offering franchises in May 2013 and had 179 franchises as
of December 31, 2023. UATP Canada Franchising, Ltd., a British Columbia corporation, offers

these franchises in Canada.

- TLGI, LLC (formerly The Little Gym International, Inc., "<u>TLGI</u>") offers THE LITTLE GYM franchises which provide physical fitness, recreational gymnastics, motor skills development, and other programs for children under The Little Gym name and trademarks. TLGI began offering franchises in September 1992 and had 185 franchises as of December 31, 2023.
- Snapology, LLC ("Snapology") offers SNAPOLOGY franchises, which provide curriculum-based courses, events and hands-on learning experiences using LEGO® brand bricks, K'Nex® brand toys, and other building toys, robotics, animation, coding and engineering techniques. Snapology began offering franchises in March 2015 and had 102 franchises as of December 31, 2023. Snapology International, LLC, a Pennsylvania limited liability company, offers these franchises outside of the USA. Our affiliate Snapology IP, LLC is the owner of certain trademarks and intellectual property associated with the SNAPOLOGY franchises.
- Premier Franchising Group, LLC ("PMA") offers PREMIER MARTIAL ARTS franchises, which are
  martial arts studios for self-defense and character development. PMA began offering franchises in
  April 2018 and had 213 franchises as of December 31, 2023. Our affiliate PMA IP, LLC is the
  owner of certain trademarks and intellectual property associated with the PREMIER MARTIAL ARTS
  franchises.
- Class 101 Franchise, LLC ("Class 101") offers CLASS 101 franchises, which provide advice, guidance and training to high school students and their parents in preparing for, selecting, applying to, and paying for college. On April 11, 2022, Class 101 acquired the assets of Class 101, Inc., which began offering franchising in June 2007 and had 55 franchises as of December 31, 2023. Our affiliate Class 101 Franchise IP, LLC is the owner of certain trademarks and intellectual property associated with CLASS 101 franchises.
- XP League Franchise, LLC ("XPL") offers XP LEAGUE franchises which follow traditional youth sports formats delivering values and life skills learned in coach-led athletics in an esports format, for elementary and middle school aged children. On April 21, 2022, XPL acquired certain assets of XP League, LLC, which began offering franchises in August 2020, and had 41 franchises as of December 31, 2023. XP League Franchise, LLC is the owner of certain trademarks and intellectual property associated with XP League franchises.
- Our affiliate, Unleashed Services, LLC ("Unleashed Services"), provides executive management services to us, UATP, TLG, Snapology, PMA, Class 101, and XPL, but Unleashed Services does not offer franchises in any line of business.

You will not conduct business directly with UATP, TLG, Snapology, PMA, Class 101, or XPL (each an "Affiliated Brand"), unless you decide to co-brand with an Affiliate Brand. If you decide to co-brand the premises of your Sylvan franchise with an Affiliated Brand, you will be offered a separate franchise disclosure document of your desired Affiliate Brand and will be required to sign a separate franchise agreement for that Affiliate Brand. Other than the above, we do not have any affiliates which offer or sell goods or services to our franchisees, and no other parent, predecessor, and affiliates offer franchises in this or any other lines of business.

## The Sylvan System: Franchise History

Sylvan began offering Sylvan franchises in June 2003 as Sylvan Learning, Inc. Sylvan's predecessors operated or offered licenses under the Sylvan System from February 1979 to June 2003, as follows: Sylvan Learning Corporation, an Oregon corporation (February 1979 until March 1984), which merged into Sylvan Learning Corporation, a Washington corporation (March 1984 until May 1985), which merged into Sylvan Learning Corporation, a Delaware corporation (May 1985 to February 1991), which entered into a general partnership with KEE Incorporated to form Sylvan KEE Systems (February 1991 to January 1993). KEE

Incorporated acquired Sylvan KEE Systems in January 1993 after the partnership dissolved, and changed its name to Sylvan KEE Systems, Inc. On September 2, 1993, Sylvan KEE Systems, Inc. changed its name to the more descriptive Sylvan Learning Systems, Inc. Sylvan Learning, LLC acquired substantially all of our assets from Sylvan Learning Systems, Inc., which later changed its name and line of business, on June 30, 2003. From June 30, 2003 to June 18, 2015, we were a Delaware corporation known as Sylvan Learning, Inc. On June 18, 2015, we converted our entity from a Delaware corporation to a Delaware limited liability company, and accordingly changed our name from Sylvan Learning, Inc. to Sylvan Learning, LLC.

# The Sylvan System and SylvanSync System Franchise Program

The Sylvan System and SylvanSync System franchise opportunity is comprised of two separate licenses that Sylvan currently offers together in a joint License Agreement for convenience:

- (1) a license to use certain of Sylvan's proprietary programs, systems, teaching and management techniques, individualized diagnostic tests, and academic and prescriptive educational courses or programs which are designed to be personally taught, supervised or administered by trained instructors (the "Sylvan System") and to be delivered in Sylvan-branded learning centers ("Sylvan Learning Centers" or "Centers") under the Sylvan brand name; and
- (2) a license to use certain of Sylvan's owned and licensed proprietary technology, digital assets, systems, business methods, techniques and methodologies that are designed to be used in connection with portable proprietary computer systems and the Internet to provide a digital learning environment incorporating assessment results, instruction, and the management of student educational programs (the "SylvanSync System"), to be delivered at locations within a defined geographical area.

Your right to operate a Sylvan Learning Center, use the Sylvan System and the SylvanSync System, and deliver products, services and programs to customers and students using Sylvan's trademarks and service marks ("Licensed Marks") is governed by the License Agreement.

Sylvan began offering Sylvan System franchises in June 2003. In March 2012, we began offering the SylvanSync System as an optional license alongside a Sylvan Learning Center License Agreement. We first began offering the Sylvan System and the SylvanSync System jointly in one License Agreement for convenience in March 2016. Effective January 1, 2019, Sylvan updated its required operational standards and specifications to make the use of the SylvanSync System mandatory.

We refer to a franchisee's Sylvan System and SylvanSync System business in a territory as the "Licensed Business."

Each franchisee delivers the Sylvan System and the SylvanSync System within a primary Sylvan-branded brick-and-mortar Center in its territory. In addition, if approved by Sylvan, each franchisee may carry the SylvanSync System, and other programs and services that Sylvan approves, outside of the Center to provide services at one or more additional approved short-term or permanent locations in its territory, such as community centers, libraries, schools, and short-term and/or small retail locations. We refer to these locations as "Satellites." Further, franchisees may open additional Centers in their territories after their first Center (the "Primary Center") is fully and continuously operating, subject to Sylvan's approval.

Commencing in March of 2020, Sylvan granted franchisees the temporary right to deliver Sylvan and SylvanSync System programs to students virtually, through live, online platforms (the "Temporary Virtual License"). In January of 2022, Sylvan elected to offer franchisees the right to deliver Sylvan and SylvanSync System programs on a longer-term basis through the expiration of the term of their License Agreements, pursuant to an amendment to the License Agreement (the "Virtual License Amendment"). As of the date of this disclosure document, the right to deliver virtual services is set forth by the License Agreement and included within the Licensed Business; hence, it is also a part of Sylvan's 2024 franchise

offering that is described in this disclosure document (so there will no longer be a separate "Virtual License Amendment").

The principal areas of instruction are reading, mathematics and writing; study skills instruction and test preparation are also available. The Sylvan System is operated, and provided to students, under the Sylvan brand name. The SylvanSync System is operated, and provided to students, under the SylvanSync and Sylvan brand names.

The Sylvan System and SylvanSync System are licensed to franchisees under our Sylvan License Agreement. A copy of our current License Agreement as of the issuance date of this disclosure document is attached as Exhibit A.

Sylvan's affilate, Educate Corporate Centers Holdings, LLC, operates Centers and Satellites in the United States directly and as a member of subsidiary limited liability companies. This entity may operate the type of Sylvan System and SylvanSync System business offered to you.

Sylvan developed the SylvanSync System under rights Sylvan received from LSF, which is a related company, and from third parties and in cooperation with third parties. You must operate the SylvanSync System in your Licensed Business in compliance with Sylvan's standards and specifications.

# **Optional Additional Product Licenses**

# Sylvan Edge Program and Product License

In September 2014, Sylvan began offering licensees the option to purchase Sylvan Edge Packages so that they may offer Sylvan Edge robotics, computer programming, engineering, and mathematics learning programs (the "Sylvan Edge Program") to students at their Centers and at other approved locations in their territories. The Sylvan Edge Packages consist of teaching materials and guidelines; specifications for the technology needed to provide instruction in robotics, computer programming, engineering, and mathematics; a license to use certain Sylvan and third-party marks and trade dress; and other methods, techniques and knowhow related to teaching the basic principles of robotics, computer programming, engineering, and mathematics to students. Courses offered under Sylvan Edge Program are personally taught by trained instructors and may be delivered at locations we approve in your Territory (which is further described in Item 12), such as Centers and Satellites, as well as un-branded locations, such as schools and community centers, under the Sylvan Edge brand name.

Licensees have the option to offer the Sylvan Edge Program, which is not part of the Sylvan System or the SylvanSync System. If you wish to offer the Sylvan Edge Program, you must sign the Sylvan Edge Package License Agreement, successfully complete the required training program, and deliver the Sylvan Edge Program in accordance with the Sylvan Edge Package License Agreement, a copy of which is attached to this disclosure document as Exhibit B and amends the License Agreement.

# ACE IT! Program and Product License

Sylvan also offers licensees the option to offer the ACE IT! Program, as defined in this paragraph, in accordance with the ACE IT! Tutoring System License Agreement. The "ACE IT! Program" uses certain proprietary programs, systems, teaching and management techniques to provide academic programs to students under contracts funded by third parties, such as school districts, other government entities and not-for-profit organizations, ("Third-Party Contracts"). In addition, with Sylvan's separate approval, you may provide the ACE IT! Program via retail services that you offer directly to consumers. These programs, systems, teaching and management techniques are personally taught by trained instructors and may be delivered at locations we approve in your Territory, such as Centers and Satellites, as well as in un-branded locations, such as schools and community centers, under the ACE IT! TUTORING® brand name (the "ACE IT! System").

Licensees have the option to offer the ACE IT! Program, which is not a part of the Sylvan System or the SylvanSync System. If you wish to offer the ACE IT! Program, you must sign the ACE IT! Tutoring System License Agreement, successfully complete the required training program, and use the ACE IT! System in accordance with the ACE IT! Tutoring System License Agreement, a copy of which is attached to this disclosure document as Exhibit C and amends the License Agreement.

Sylvan offered the ACE IT! Program as a separate franchise opportunity, with a separate Franchise disclosure document, from January 2006 until January 2016. As of the date of this disclosure document, Sylvan offers the ACE IT! Program as an optional product license under the ACE IT! Tutoring System License Agreement, and only as an add-on or adjunct tutoring service for an existing Sylvan and SylvanSync license. Sylvan does not offer the ACE IT! Program as a separate franchise opportunity with a separate franchise disclosure document.

# Sylvan In-Home and mySylvan Marketplace+

SIH operates an online Marketplace Platform that enables parents to seek in-person or virtual tutors in a designated geographic area who specialize in fields of study where their students may need additional assistance. At its inception in 2017, the Marketplace Platform was referred to as the Sylvan In-Home Tutoring Marketplace; as of the issuance date of this disclosure document, Sylvan has changed the name of the Marketplace Platform to "mySylvan Marketplace+". Once a parent finds a tutor that fits the needs of his or her student, the Marketplace Platform allows the tutor and parent to schedule tutoring sessions at a mutually convenient location, whether in-person or online, and facilitates online payment processing. As of the issuance date of this disclosure document, tutors visible to parents on the Marketplace Platform are independent contractors of SIH and they pay a fee to SIH for the ability to be featured on the Marketplace Platform. SIH tutors have the option to offer and deliver SylvanSync programs to students, though they are not required to do so. SIH reserves the right to modify elements of its business model, service terms, or terms of engagement with tutors from time to time. As of the issuance date of this disclosure document, SIH is advertising the Marketplace Platform across the United States, as the Marketplace tutors are permitted to deliver virtual tutoring to any student who does not live within fifty (50) miles of a Sylvan Learning Center that is not participating in the Marketplace Platform offerings. SIH reserves the right to advertise the Marketplace Platform and allow affiliated tutors to deliver services in licensed territories. As of the date of this disclosure document, new licensees will be required to enroll their territories in the Marketplace Platform, subject to the terms and conditions of the mySylvan Marketplace+ Franchisee Participation Pilot Agreement, attached as Attachment L to the License Agreement. Sylvan does not guarantee that it will continue to allow franchisee participation in the Marketplace Platform beyond the pilot phase. Sylvan aims to roll out the Marketplace Platform across the United States, and Sylvan anticipates making certain portions of the Marketplace Platform available to all franchisees, pursuant to certain fees, training, and other requirements, as determined by SIH and Sylvan at their discretion. In particular, Sylvan contemplates enabling the customer self-service booking feature of the Marketplace Platform for franchisees who choose to opt in, and providing those franchisees access to the tutors who provide services through the Marketplace Platform. The in-person and online tutoring business which takes place through the Marketplace Platform is not currently part of a franchisee's Licensed Business.

As of the date of this disclosure document, new licensees are required to enroll their territories in the Marketplace Platform, subject to the terms and conditions of the mySylvan Marketplace+ Franchisee Participation Pilot Program Agreement attached as Attachment L to the License Agreement.

#### **Bright Horizons**

Sylvan entered into an agreement with Bright Horizons Family Solutions, LLC ("Bright Horizons") whereby Sylvan and its franchisees may deliver services to Bright Horizons' Clients' employees as part of the Bright Horizons Back-Up Care Program. As of the date of this disclosure document, new licensees are required to participate in the Bright Horizons Program. The Bright Horizons Participation Agreement is attached to this disclosure document as Attachment K to the License Agreement.

## Industry-Specific Laws and Regulations; Marketplace and Competition

The general market for the Sylvan System and SylvanSync System franchise opportunities offered to you consists primarily of parents who wish to enhance their school-age children's academic performance.

Some states may have laws or regulations that permit only a state-certified teacher to perform certain activities in your Licensed Business, or that may require a license for operating a Center or a Satellite. You must also operate your Licensed Business in compliance with Sylvan's standards and specifications.

In some states, a Sylvan Learning Center may be subject to licensing requirements or may be required to have a state-certified instructor as a director or on the premises when instruction is being offered. For instance, in Pennsylvania a Sylvan Learning Center is subject to regulations issued by the State Board of Private Academic Schools. These require private schools to apply to the Board for a license to operate. A school applying for a license must include with its application evidence that it has financial resources or forms of surety adequate to reimburse any unearned tuition in case it ceases to operate. Only teachers holding a valid Pennsylvania professional teacher certification may teach in a licensed private school. A Pennsylvania certified specialist in each subject must oversee the instructional process of that specialty area (Reading-certified for Reading programs, Math-certified for Math programs). Similar requirements may apply in your state.

Sylvan and SylvanSync franchisees compete with other supplemental instruction programs offered to school-age children and adults, through a variety of media as well as in association with public and private schools. Your ability to compete will depend on various factors, including the geographical area of your facility, your specific site location, general economic conditions, accessibility of your facility, and your particular capabilities.

## **ITEM 2: BUSINESS EXPERIENCE**

#### SYLVAN LEARNING, LLC

<u>Susan Valverde - President</u>: Ms. Valverde is our President and has held that position since May 2024 in Towson, Maryland. From March 2017 to April 2024, served as our Chief Franchise Operations Officer in Towson, Maryland.

<u>John McAuliffe – Brand Ambassador</u>: Mr. McAuliffe is our Brand Ambassador and has held that position since May 2024 in Towson, Maryland. He has been with Sylvan since January 2010, and from February 2016 to April 2024, served as our Chief Executive Officer in Towson, Maryland.

Amy Przywara - Chief Marketing Officer: Ms. Przywara has been with Sylvan since January 2009, and since June 2015, has served as our Chief Marketing Officer in Towson, Maryland.

<u>Jacob Jones - Chief Financial Officer</u>: Mr. Jones has been with Sylvan since December 2011, and since January 2022, became Sylvan's Chief Financial Officer in Towson, Maryland.

# UNLEASHED SERVICES

Michael Browning, Jr. – Chief Executive Officer: Michael Browning, Jr. has been the Chief Executive Officer of both Unleashed Brands and Unleashed Services since July 2021 in Bedford, Texas. He is one of co-founders of UATP and has served as UATP's Chief Executive Officer from its inception in May 2013 to June 2021 in Bedford, Texas. Michael also served as the Chief Executive Officer of UA Attractions, LLC from May 2018 to October 2021 in Bedford, Texas. Previously, he served as the Manager of Southlake Urban Air, LLC from March 2011 to December 2018 in Southlake, Texas; Mansfield Urban Air, LLC from January 2013 to September 2020 in Mansfield, Texas; Frisco Urban Air, LLC from May 2013 to February 2019 in Frisco, Texas; Garland Urban Air, LLC from March 2015 to July 2020 in Garland, Texas; Coppell Urban Air, LLC from March 2015 to July 2020 in Coppell, Texas; and Fort Worth Urban Air, LLC since August 2016 in Bedford, Texas. Michael has been a Manager of UATP Holdings, LLC since 2015 in Bedford, Texas.

<u>Stephen Polozola – Chief Legal Officer</u>: Stephen Polozola has served as the Chief Legal Officer of Unleashed Services since July 2021 in Bedford, Texas. Stephen is one of the co-founders and has served as the Executive Vice President and General Counsel of UATP since its inception in May 2013 to June 2021 in Bedford, Texas. He has served as a Manager of UATP Holdings, LLC since July 2015 and has served as a Vice President of UATP IP, LLC since October 2013 in Bedford, Texas. Stephen has served as President of Adventis Insurance, Inc. since March 2020.

<u>Joe Luongo – Chief Operating Officer</u>: Joe Luongo has served as the Chief Operating Officer of Unleashed Services since April 2022 in Bedford, Texas. Since June 2019, he also serves as Chairman of the Board for WellBiz Brands and previously served as Executive Chairman from October 2017 to June 2019 in Englewood, Colorado.

<u>Scott Perry – Chief Financial Officer</u>: Scott Perry has served as the Chief Financial Officer at Unleashed Services since July 2021 in Bedford, Texas. Before this position, he served as the Chief Financial Officer and Executive Vice President of UATP from March 2019 to June 2021. Previously, he was a Member of Laguna Woods Consulting, LLC from September 2018 to March 2019 in Austin, Texas.

<u>Jessica Correa – Chief Marketing Officer</u>: Jessica Correa has served as Unleashed Services' Chief Marketing Officer since July 2021 in Bedford, Texas. Previously, she served as the Chief Marketing Officer from August 2019 to June 2021. Prior to joining UATP, Jessica served as Head of Marketing for Planet Fitness in Hampton, New Hampshire from November 2014 to May 2018.

<u>Josh Wall, CFE – Chief Growth Officer</u>: Josh Wall has been the Chief Growth Officer of Unleashed Services since July 2021 in Bedford, Texas. From June 2019 to June 2021, Josh Wall served as UATP's Executive Vice President and Chief Franchise Officer. Previously, Josh served as the Chief Development Officer for Christian Brothers Automotive Corporation in Houston, Texas from January 2018 to June 2019.

<u>Chris Andrews – Chief Information Officer</u>: Chris Andrews has been the Chief Information Officer of Unleashed Services since May 2022 in Bedford, Texas. Previously, from May 2019 to May 2022, he was the Chief Information Officer of Smoothie King in Coppell, Texas. From November 2018 to May 2019, he served as the Chief Information Officer at Pei Wei in Dallas, Texas.

<u>Diane Sanford, SHRM-SCP – Chief People Officer</u>: Diane Sanford has served as the Chief People Officer at Unleashed Services since March 2023 in Bedford, Texas. Previously, she was the Chief People Officer at Local Favorite Restaurants in Dallas, Texas from May 2022 to March 2023. Before this role, she served as the Chief People Officer at On the Border Mexican Grill & Cantina from December 2014 to April 2022 in Irving, Texas.

Ryan Slemons – Chief Development Officer: Ryan Slemons has served as the Chief Development Officer of Unleashed Services since April 2023 in Bedford, Texas. From July 2021 to April 2023, he served as Vice President, Global Real Estate and Development at Game Stop in Dallas, Texas. Previously, from September 2014 to July 2021, he held various positions with Amazon, most recently serving as Head of Real Estate – Amazon Go, Amazon Style, and New Concepts in Dallas, Texas.

Eric Schechterman, CFE – Vice President of Franchise Finance: Eric Schechterman has served as the Vice President of Franchise Finance of Unleashed Services since April 2023 in Bedford, Texas. Previously, from April 2011 to February 2023, he held several positions with Benetrends Financial, most recently serving as Chief Development Officer from April 2017 to February 2023 in Philadelphia, Pennsylvania. He currently also serves as Senior Advisor to Lander Analytics, and has held that position since January 2014 in New York. New York.

<u>Josh Barker – Vice President of Franchise Recruitment</u>: Josh Barker has served as Unleashed Services' Vice President of Franchise Recruitment since August 2021 in Houston, Texas. Previously, he served as the Vice President of Franchise Development at Neighborly in Waco, Texas from October 2020 to August

2021. At Christian Brothers Automotive in Houston, Texas, he served as the Franchise Development Manager from June 2017 to April 2018 and Director of Franchise Development from April 2018 to October 2020. From June 2015 to June 2017, he was the General Manager at Air Power Services Inc. in Houston, Texas. Josh serves in his present capacities in Bedford, Texas.

#### **ITEM 3: LITIGATION**

# PENDING LITIGATION AGAINST FRANCHISEES TO COLLECT ROYALTY PAYMENTS

<u>Sylvan Learning, LLC v. Renryder, LLC and Federico Sandoval</u> (Cause No. 4248-21-C, District Court for Hidalgo County, Texas).

# LITIGATION AGAINST FRANCHISOR AFFILIATES

In the Matter of Snapology Community Programs, L.P. and its successor Snapology, LLC, Administrative Proceeding Before the Securities Commissioner of Maryland, Case No. 2015-0429. As a result of an investigation into the franchise related activities of Snapology Community Programs, L.P. and its successor Snapology, LLC, the Maryland Securities Commissioner ("Commissioner") concluded that grounds existed to allege that Snapology violated the registration and disclosure provisions of the Maryland Franchise Law in relation to the offer and sale of a Snapology franchise. In responding to inquiries from the Maryland Securities Division, Snapology acknowledged that, during the time it was not registered to offer and sell franchises in Maryland, it entered into two separate License and Training Service Agreements in Maryland that the Commissioner concluded constituted the sale of two franchises. Snapology represented that it has offered rescission to one of those franchisees. On January 15, 2016, the Commissioner and Snapology entered into a consent order whereby Snapology, without admitting or denying any violations of the law, agreed to: immediately and permanently cease from the offer and sale of franchises in violation of the Maryland franchise law; complete registration of its franchise offering in Maryland; and, offer rescission to the remaining franchisee who was sold a franchise in Maryland while Snapology was not registered with the State.

UATP Management, LLC v. Leap of Faith Adventures, LLC, District Court of Tarrant County, Texas Case No. 017-300796-18. On July 9, 2018, UATP filed this lawsuit ("Petition") against the defendant Leap of Faith Adventures, LLC ("LOFA"), that, at the time of filing, was a distributor and installer of attractions used in Urban Air Adventure Parks. UATP claimed that LOFA had stopped paying UATP rebates on revenue LOFA received from selling attractions to UATP franchisees, alleging breach of contract and tortious interference and have since added a fraud and fraudulent inducement claims. UATP is seeking compensatory damages in excess of \$6.5 million on its various claims, attorneys' fees, and costs. LOFA answered UATP's Petition on August 13, 2018 and filed a counterclaim on October 31, 2018. LOFA alleged, among other things, conversion, breach of contract, interference with business relationships, violation of the Texas Theft Liability Act, and theft of trade secrets arising primarily from UATP's alleged interference with LOFA's contracts with its franchisees and relationships with other entities, all for the supposed purpose of bringing in house, to the exclusion of LOFA, the installation of attractions at Urban Air Adventure Parks. LOFA seeks unspecified compensatory and exemplary damages, equitable relief, and attorneys' fees. On March 29, 2019, the Court granted UATP's motion to dismiss certain of LOFA's counterclaims, in particular the trade secrets claim. After UATP appealed the Court's order, the Court of Appeals on May 4, 2021, dismissed additional claims asserted by LOFA, leaving only claims for, among other things, interference with contracts and business relationships with UATP's franchisees, conversion, breach of contract, and violation of the Texas Theft Liability Act. Before the Appellate Court's ruling, LOFA filed its own new petition on September 10, 2020, against certain of UATP's affiliates and principals, including Michael Browning, Jr. and Stephen Polozola, which was consolidated with the lawsuit described in this paragraph. In an effort to resolve the matter and bring it to a final conclusion, the case was dismissed with prejudice following the execution of a confidential settlement agreement, wherein UATP Management, LLC and Leap of Faith Adventures, LLC released all claims against each other without admission of any liability in exchange for a one-time payment of five million dollars to Leap of Faith Adventures, LLC.

The Commissioner of Financial Protection and Innovation v. Premier Franchising Group, LLC doing business as Premier Martial Arts International and/or Premier Martial Arts. On November 18, 2021, PMA entered into a consent order with the California Commissioner of Financial Protection and Innovation related to four licensees of PMAI. The Commissioner found that PMAI offered and sold at least four franchises in California without being registered with the Commissioner or exempt, in violation of Section 31110 of the California Franchise Investment Law. The Commissioner further found that we and PMAI willfully omitted to state in subsequent franchise registration applications the material fact that PMAI had at least four California studios, in violation of Section 31200 of the California Franchise Investment Law. Pursuant to the consent order, we agreed to (1) refrain from violating Sections 31110 and 31200, (2) pay a \$10,000 administrative penalty, (3) file a post-effective Amendment updating PMA's current registration to include the consent order, and (4) disclose the existence of each and every California studio in Item 20 and in the exhibit list of current and former franchisees in any PMA disclosure document filed with the Commissioner moving forward.

Unleashed Services, LLC vs. Tom Pabin vs. Josh Wall, pending in the 48<sup>th</sup> Judicial District of Tarrant County, Texas, bearing Cause No. 48-346174-23. On September 18, 2023, Unleashed Services, LLC ("Unleashed") filed its Original Petition against Thomas Pabin ("Pabin") requesting the issuance of a Declaratory Judgment to construe the terms of Mr. Pabin's employment agreement and the rights, duties, status, and legal relations of Unleashed and Pabin under such employment agreement. On February 22, 2024, Pabin, in response to Unleashed's Motion to Dismiss the claims asserted in Pabin's Counterclaim and Third-Party Petition, filed his First Amended Counterclaim against Unleashed and Third-Party Petition against Josh Wall. Within this Counterclaim and Third-Party Petition, Pabin has alleged (1) Unleashed breached the employment agreement and Pabin is entitled to a severance payment; (2) breach of contract/promissory estoppel against Unleashed and Wall related to the subject asset purchase agreement ("APA") and payment of a potential earnout; (3) fraud as against Unleashed and Wall related to the calculation of the earnout set forth in the APA; (4) indemnification as against Unleashed and (5) requested a declaratory judgment interpreting the rights, duties, status, and legal relations of the parties under the APA and employment agreement. Unleashed strongly denies all allegations asserted against it and Wall and believes no sums are due Pabin under the applicable agreements.

Other than these actions, no litigation is required to be described in this Item.

## **ITEM 4: BANKRUPTCY**

No bankruptcies are required to be disclosed in this Item.

#### **ITEM 5: INITIAL FEES**

#### **License Agreement**

The initial license fee is \$36,900, which includes the licenses for the optional Sylvan Edge Package and ACE IT! Program. All initial license fees are non-refundable and payable in full when you sign the Sylvan License Agreement, EDGE Package License Agreement, and the ACE IT! Tutoring System License Agreement.

## **Multi-Territory Incentive Plan**

If we award you multi-unit development rights, you must sign our "Multi-Territory Incentive Plan," which is an amendment to the License Agreement and attached to this disclosure document as Exhibit D, and pay a development fee (the "Development Fee") for the respective number of territories:

Number of Territories	<b>Total Development Fees</b>
-----------------------	-------------------------------

One Territory	\$36,900
Two Territories	\$56,900
Three Territories	\$76,900

The Development Fee you must pay under the Multi-Territory Incentive Plan (a) includes the initial license fees payable by you for the license agreements to be developed, (b) will be due in a lump sum payment upon the signing of the Multi-Territory Incentive Plan, and (c) is fully earned and non-refundable in consideration of administrative and other expenses we incur in entering into the Multi-Territory Incentive Plan, and for our lost or deferred opportunities to enter into the Multi-Territory Incentive Plan with others. The Development Fee is uniform for all developers.

#### **Incentives**

Sylvan offers a 5% reduction of the Initial License Fee for existing Sylvan franchisees seeking to expand their Sylvan Learning Center businesses by acquiring and developing additional territories. Sylvan also offers a 5% reduction of the Initial License Fee for current franchisees of Affiliated Brands in good standing when they purchase a Sylvan franchise.

Sylvan may also reduce the Initial License Fee by 5% for qualified veterans of the United States military, first responders, or social workers seeking to become new franchisees.

Sylvan may offer certain additional incentives to franchise candidates with educational experience, such as a reduction of the Initial License Fee by 5%. A "Qualified Educator" is a person with teaching and/or school administrative experience who owns or controls 50% or more of the franchisee entity.

Sylvan may modify or discontinue incentive programs at any time. In 2023, Sylvan charged initial license fees ranging from \$12,500 to \$36,900 for the License Agreement, including the EDGE Package License Agreement and the ACE IT! Tutoring System License Agreement. Unless otherwise stated, all fees listed above in this Item 5 are uniformly imposed.

**ITEM 6: OTHER FEES** 

TYPE OF FEE <sup>1</sup>	AMOUNT	DUE DATE <sup>4</sup>	REMARKS
Royalty	16% of monthly gross revenues, with a total Minimum Annual Royalty of not less than \$32,000, derived from the License Agreement. <sup>2, 3, 4</sup> Optional programs to License Agreement:  12% of monthly gross revenues derived from the Sylvan Edge Program amendment, subject to a minimum annual royalty of \$2,100 assessed in the first calendar year and \$4,200 assessed in the remaining calendar years <sup>3</sup> 12% of monthly gross revenues	Payable by the 15th day of each month.	A portion of the royalty may be directed by Sylvan toward Sylvan Advertising under the License Agreement. For each calendar year, your monthly royalties paid to us must total at least the Minimum Annual Royalty. Non-refundable. <sup>3</sup>

TYPE OF FEE <sup>1</sup>	AMOUNT	DUE DATE <sup>4</sup>	REMARKS
	derived from the Ace It! Program amendment, subject to \$1,200 minimum annual royalty <sup>3</sup>		
Territory Advertising Minimum for Local Advertising and Co-op Expenditures <sup>5</sup>	8% of monthly gross revenues, capped at a maximum of \$40,000 per territory per calendar year. <sup>2, 3</sup> Optional programs:  3% of monthly gross revenues derived from the Sylvan Edge Program	Payment liability accrues monthly, up to the cap.	Paid to third-party vendors you select and/or your co-op. Local advertising expenditures are always required, and you will incur them either through advertising activities you undertake directly and/or through a local or regional advertising co-op. We can request that you contribute 100% of your required local advertising expenditure to an advertising co-op. Non-refundable.
1-800- Educate or Other Toll- Free Number	Currently \$0.20/minute	Payable by the 25th day of each month.	Invoicing for these fees commences after 30 seconds have elapsed, and fees accrue in sixminute increments. These fees are billed on franchisees' monthly billing statement following the month that the charges are incurred. Non-refundable.
Hub Technology Platform	Currently \$114/Center per month	Payable by the 25th day of each month.	Sylvan requires full utilization of the Hub Technology Platform in all Centers. Hub fees are included on franchisees' monthly billing statements for each Center. Non- refundable.
Audit <sup>6</sup>	Cost of audit, plus late fee on deficiency. <sup>6</sup>	Upon billing.	This charge applies only if the audit discloses that you failed to pay 3% or more of the fees actually due to us during the period covered by the audit. Non-refundable.
Contact Center Program Fee <sup>7</sup>	\$700 flat fee per month plus \$10/inquiry (Model A) or \$350 flat fee per month plus \$30/inquiry (Model B).	Payable by the 25th day of each month.	Payable for call center services that are provided in connection with Sylvan's "Contact Center Program." Participating franchisees select their payment model based upon their anticipated call volume. This charge does not apply until you

TYPE OF FEE <sup>1</sup>	AMOUNT	DUE DATE <sup>4</sup>	REMARKS
			commence operations and utilize the Contact Center Program services. Monthly required payment will be billed by Sylvan and based upon the prior month's inquiries received. Non- refundable.
Transfer Fee <sup>8</sup>	50% of our then-current initial license fee if the Center transferred to a new approved franchisee; or  25% of our then-current initial franchise fee Center is transferred to an approved existing franchisee.	Before transfer.	Payable before transfer of your Center if you request and we consent to transfer. There is no charge if you transfer your franchise to a corporation that you wholly control. Non-refundable.
Initial Training Fee	\$1,000	Upon invoice.	Payable only if you or your Managing Principal does not complete Sylvan's initial training program within 180 days of the effective date of the license agreement. Non-refundable
Additional Training Fee	Currently \$500 per day, plus reimbursement of our out-of-pocket costs	Upon invoice	Payable if Sylvan requires you or your employees to attend additional training or if you request Sylvan to provide additional training. Non-refundable.
Taxes and Income Taxes	Varies.	Upon billing.	You will be responsible for paying Sylvan for all taxes that Sylvan must pay on the amounts you pay to Sylvan, unless the tax is credited against Sylvan's payment. Taxes vary and cannot be estimated in advance.
National Conference Fee	Currently \$900 to \$1,100 per attendee	Upon invoice	In addition to payment of the National Conference Fee, you are responsible for wages, travel, lodging, and other fees and costs for you, your employees, or other personnel to attend such conferences. Non-refundable
Renewal Fee	\$6,000	Before renewal.	Non-refundable.

TYPE OF FEE <sup>1</sup>	AMOUNT	DUE DATE <sup>4</sup>	REMARKS
Additional Center Fee	\$1,000 for each additional Center opened in existing territory.	Upon invoice	Non-refundable.
Dashboard Access License Fee	\$10 per month per license	Monthly upon invoice	This fee is payable to us and may be increased by Microsoft from time-to-time, which is a pass- through fee and does not include any markup or rebate.
Territory Split Fee <sup>9</sup>	\$5,000 per territory split.	Upon invoice.	Payable if you desire to split your current territory into two or more territories, pursuant to a separate then-current franchise agreement. For example, if you split your Territory into three total territories, the total Territory Split Fee is \$10,000. Non-refundable.
Resale Assistance Program	Up to 5% of the purchase price, and may include a \$500 administrative fee payable upon enrollment.	Before transfer.	Voluntary program and fees are payable only if you elect to participate in our optional Resale Assistance Program in connection with the sale of the Licensed Business to an approved transferee of the License Agreement. Non-refundable.
Liquidated Damages	Average monthly royalty over preceding 36 months multiplied by the number of months until the injunction issues, or by 24 or such lower number of months remaining of the License Agreement term.	Upon demand by, or notice from, Sylvan.	Applies to the period from breach until injunction issues for breach of Sections 3, 12.1 or 12.5 of the License Agreement; applies to lost future profits if you cease operation of the Center prior to expiration or renewal of the License Agreement; applies for two years from the date of Center closure if closure occurs prior to the expiration of the then-current term of the License Agreement, or such shorter period remaining on the term. Non-refundable.

**NOTE 1:** Unless otherwise stated, all fees are non-refundable and are imposed by and payable to Sylvan (unless we direct you to pay another person). These fees are uniformly imposed by Sylvan.

**NOTE 2:** "Gross revenues" means the total gross amount of all revenues that you (or anyone else) actually receive in operating your Licensed Business or otherwise at your Center, or using Sylvan's Licensed Marks, including any amounts that you receive if you are in violation of the terms of the License Agreement. Gross

revenues do not include sales tax or other tax receipts or documented credits or refunds you make in good faith to your customers. We may require you to pay your royalty fees by electronic funds transfer, preauthorized auto-draft arrangement or other method we specify.

NOTE 3: Beginning in the sixth year of the initial term of your License Agreement, we may increase the Minimum Annual Royalty and the cap on Local Territory Advertising for a Cost-of-Living Adjustment based on changes in the Revised Consumer Price Index for Urban Wage Earners and Clerical Workers: All items (1982-1984 = 100) ("Consumer Price Index") published by the Bureau of Labor Statistics, United States Department of Labor. The base index ("Base Index") shall be the Consumer Price Index for January 1 of the calendar year of the Effective Date. Sylvan may adjust the Minimum Annual Royalty and the cap on Local Territory Advertising amounts indicated in the table above (and other amounts, as detailed in the License Agreement) so that the adjusted flat dollar amount for a given year shall be equal to the product of (a) a fraction, the numerator of which is the Consumer Price Index for January of the year of such adjustment, and the denominator of which is the Base Index, multiplied by (b) the flat dollar amount set forth in such paragraph. No adjustment based on the CPI will be made to the percentage of gross receipts charged as monthly royalty (16%) or Territory Advertising Minimum (8%).

As further described in Item 12, if you fail to open a Primary Center by the original Required Open Date, and Sylvan extends the Required Open Date as an accommodation to you, Sylvan reserves the right to enforce the Minimum Annual Royalty requirement as if you had opened the Primary Center on time, according to the original requirement.

**NOTE 4:** If you do not pay a fee or expense on its due date, then you must pay a late charge of 10% of the amount due and interest at the rate of 1.5% per month, which begins to accrue as of the due date. No late fee will exceed the maximum late fee permitted by law.

**NOTE 5:** We may require you to spend up to 100% of your local advertising obligations through local or regional advertising cooperatives we designate. There are approximately 60 such cooperatives in the United States and Canada. Sylvan-owned Centers are required to contribute to such cooperatives on the same basis as franchised Centers, and their voting power is determined on the same basis as any other Center in the same cooperative.

**NOTE 6:** You are required to make monthly reports of your gross revenues and income and expenses, as well as any information we request concerning services you provide on a scholarship or reduced-fee or nofee basis. These reports are due at the same time as your monthly royalty payments to us. We may require you to submit your monthly reports electronically. You must accurately record gross revenues in the Computer System, as defined in the License Agreement, on the day they are received, or in such other manner and frequency that we prescribe. We may inspect the premises of your Licensed Business, your books of account and other materials during reasonable times of the business day with or without notice. We will use reasonable efforts not to disrupt the operations of your Licensed Business when we conduct an inspection. We may perform this inspection remotely. If we perform this inspection remotely, you are required to make available immediately upon request all materials that you are required to maintain electronically. You are required to provide any other materials by the delivery date we designate, which shall be not less than two calendar days from our request. If there is a deficiency of 3% or more of fees you were required to pay for the relevant period, you must pay the reasonable cost of the audit. The cost of the audit includes the cost of an independent accountant and the travel expenses, room, board and compensation of the accountant and our agents involved in the conduct of the audit. The late fee applicable to the deficiency is the same as the interest rate described in Note 3.

**NOTE 7:** Sylvan requires that licensees participate in the Contact Center Program for the first twelve months that they operate a Center (from the date the Center opens for business until the first anniversary of that date). After such twelve-month period from Center opening has elapsed, the licensee may elect to continue participation or to terminate its Contact Center Franchisee Participation Agreement pursuant to its

terms. If a licensee does not provide adequate notice of termination of the Contact Center Franchisee Participation Agreement, it may be subject to an early termination fee.

Franchisees that participate in the Contact Center Program select one of the two payment models described in the chart above (either low variable per inquiry cost with a higher monthly flat fee, or a lower monthly flat fee with a higher variable per inquiry cost). Payment for Contact Center Program usage is billed by Sylvan on franchisees' monthly billing statements; payment is currently due on the 25<sup>th</sup> day of each calendar month, based upon the prior calendar month's usage. Sylvan then pays the third-party vendor responsible for operating the Contact Center directly. Sylvan retains a portion of the fee paid by the franchisee for services that it performs, including having in place certain designated Sylvan staff members to train Contact Center staff, provide support to participating franchisees, and serve as a liaison with the Contact Center. Contact Center Program fees are subject to change from time to time based upon changes in the third-party vendor fees charged to Sylvan.

**NOTE 8:** In order to qualify as an "existing Sylvan licensee" for purposes of paying the reduced transfer fee of \$3,000, you must have (1) completed all initial training requirements, and (2) operated the Licensed Business at a Primary Center for at least six months prior to the date of the contemplated transfer. In addition, you must be in good standing under your existing License Agreement(s) to qualify.

**NOTE 8:** With our written consent, you may divide your territory into two separate territories in order to continue developing Centers. For each newly created territory, you will execute our then-current form of license agreement and pay to us \$5,000 (the "Territory Split Fee"). Our approval to subdivide your territory is subject to conditions, which may include successful completion of training by a new manager (if we require you to have a second manager), your compliance under all license and franchise agreements and other agreements with us and our affiliates, and the subdivided territory independently meeting our then-current standards for a territory. Upon an approved transfer, we may require you to split the territory into two or more territories (each of which meet the then-current standards for a protected area); if you choose to develop one of the split territories that does not contain your transferred Center, you shall pay the Split Territory Fee instead of the then-current initial franchise fee.

# ITEM 7: ESTIMATED INITIAL INVESTMENT YOUR ESTIMATED INITIAL INVESTMENT

TYPE OF EXPENDITURE	AMOUNT	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Initial License Fee	\$36,900 <sup>1</sup>	Lump sum.	Upon signing the License Agreement.	Sylvan
Travel and Living Expenses While Attending Initial Training	\$1,400 - \$2,4002	As incurred.	Before opening.	Airlines, hotels, restaurants, etc. <sup>9</sup>
Initial Three Months' Rent and Security Deposit	\$4,000 - \$15,500 <sup>3</sup> ,	Per lease terms. <sup>3</sup>	Per lease terms. <sup>3</sup>	Landlord <sup>10</sup>
Real Estate Improvements and Site Preparation	\$15,000 - 75,000 <sup>4</sup>	Lump sum or per lease terms.	Before opening or per lease terms.	Landlord and/or contractors <sup>8, 10</sup>

TYPE OF EXPENDITURE	AMOUNT	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Initial Inventory and Instructional Materials	\$2,750 - \$3,050 <sup>5</sup>	Lump sum.	Before opening.	Approved vendors <sup>8</sup>
Specialized Furnishings	\$6,000 - \$15,000 <sup>6</sup>	Lump sum.	Before opening.	Vendors <sup>8</sup>
Other Furniture and Miscellaneous Supplies	\$3,400 – \$4,500	Lump sum.	Before opening.	Vendors <sup>8</sup>
Signage	\$2,000 - \$8,000	Lump sum.	Before opening.	Vendors <sup>8</sup>
Marketing During Initial 3-Month Period When Primary Center First Opens <sup>8</sup>	\$10,000 <sup>7</sup>	As incurred.	Across the 3-month period: before opening, the month of opening, and the month following the opening of your Primary Center.	Vendors <sup>8</sup>
Computers and other hardware, Telephone, Internet Access & Software <sup>11</sup>	\$7,500 - \$13,93011	Lump sum.	Before opening.	Vendors <sup>8</sup>
FOA Annual Membership Fee for Initial Period <sup>12</sup>	\$33012	Lump sum.	At the time you sign a License Agreement.	FOA <sup>8</sup>
Miscellaneous Supplies	\$15 - \$150 <sup>13, 14</sup>	Lump sum.	As incurred.	Vendors <sup>8</sup>
Hub Technology Platform	\$34215	As incurred.	After opening.	Sylvan <sup>8</sup>
Contact Center Participation	\$1,480 - \$2,93016	As incurred.	After opening.	Sylvan <sup>8,16</sup>
QuickBooks or Equivalent Accounting Software	\$135 - \$600 <sup>18</sup>	As incurred.	As incurred.	Vendors <sup>9, 10</sup>
Optional: Sylvan Edge	\$0 - \$3,21019	As incurred.	Before opening.	Vendors
Additional Funds – Three Months	\$9,500 - \$35,000 <sup>17</sup>	As incurred.	As incurred.	Employees, suppliers, utilities <sup>8</sup>

TYPE OF EXPENDITURE	AMOUNT	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
TOTAL <sup>19</sup>	\$100,752 - \$226,842			

**NOTE 1:** The Initial License Fee is not refundable under any circumstances. The Initial License Fee may be modified as described in Item 5 if you agree to a Multi-Territory Incentive Plan. Sylvan currently provides initial training free of charge.

**NOTE 2:** As of this issuance date of this disclosure document, Sylvan delivers the first portion of the initial training program in person at Sylvan's facilities in Towson, Maryland or another designated facility over the course of five days. The second portion of the initial training program is delivered online via 9 live webinars. Sylvan estimates that the estimated cost for travel, food, and lodging would range from \$1,000 to \$2,500 per person in attendance (including between \$100 and \$800 round trip travel, \$150 per day for lodging, and \$25 per day per person for meals for the five-day initial training program).

**NOTE 3:** If you do not own adequate space, you must buy or lease space for your Center. Strip malls are the typical locations for Centers. Your Center must be a minimum of 1,000 square feet. We anticipate that, under the business model in the License Agreement, we will approve Center sizes of 1,000 to 2,000 square feet. Center size across the current Sylvan Network varies, with most Centers having between about 1,500 and 2,400 square feet. The amount of rent that you pay will vary greatly depending on your geographic location, the type of location in which your Center is located, the size of your Center, and general economic conditions, among other factors.

Some franchisees choose to operate at SylvanSync Satellite sites in addition to their Primary Centers; doing so is not required by Sylvan. Sylvan does not recommend that you purchase dedicated space for Satellites. You may wish to lease space on a short-term or temporary basis. School libraries and community centers are some possible locations for Satellites. You must obtain Sylvan's approval of a location before you may deliver services there. Square footage will vary greatly by number of students you are serving. The amount of temporary rent that you will pay will vary greatly depending on your geographic location, the type and size of your service site, and general economic conditions, among other factors, but will typically range from \$250 to \$3,000, and many Satellite locations are used at no cost.

**NOTE 4:** The costs of improving your lease site for your Center are negotiated with the landlord as part of your negotiating your lease. If the landlord provides the improvements at its cost and those costs are financed as part of the lease, you will substantially reduce your initial costs, but your monthly rental costs will likely be higher.

**NOTE 5:** Sylvan will specify the initial inventory and instructional materials that you must obtain from third parties; this includes educational materials and manipulatives to help you deliver Sylvan and SylvanSync programs. You may source these from any third party, so long as the materials meet our specifications. This amount may change from time to time throughout the year. The amounts do not include all shipping costs and taxes, which must be paid by you. Depending on your location and applicable state taxes, your shipping costs should be between approximately \$200 and \$500, and your taxes will vary. At the end of your first year of operations, you will be required to purchase additional Learning Materials for an additional cost.

**NOTE 6:** You must purchase certain specialized furnishings for your Center that meet Sylvan's specifications. You may purchase these from any vendor that can produce furnishings that meet Sylvan's specifications (see Item 8). You may use vendors recommended by Sylvan or any vendor that you choose, but if you source vendors outside of Sylvan's list of recommended vendors, then you must seek and obtain

approval of the furnishings you wish to purchase in advance of purchasing them, as further described in Item 8. You are responsible for shipping costs and the set-up and installation of the specialized furnishings.

**NOTE 7:** Across the three-month period beginning in the month before the Primary Center first opens for business in your Territory, you are required to spend a total of at least \$10,000 on advertising and marketing approved by Sylvan to promote your Primary Center. We will recognize these expenditures toward your 8% Monthly Territory Advertising Minimum for the month in which you spend the money and the month(s) that follow until we have recognized your whole expenditure. This requirement applies only to the first time the Primary Center opens for business, and does not apply if you purchase or relocate an already-operating Primary Center. See Item 6 for information about your 8% Monthly Territory Advertising Minimum.

**NOTE 8:** This fee or payment, to Sylvan or a vendor, is non-refundable.

**NOTE 9:** The refundability of this fee depends on the provider.

**NOTE 10:** This fee is usually non-refundable but varies by landlord.

**NOTE 11:** The estimated costs indicated in the chart include the costs for three to five computer workstations with operating software at \$600 - \$1,000 each, 12-21 iPad (7th generation or later) devices at \$349 each, 12-21 covers/screen protector kits for the iPads at \$10 each, \$700 total for a printer and copier (combined or separate), and three months of internet and telephony services at \$100 - \$230 per month and a small office/business firewall and wireless access point at \$350 - \$700. You must purchase or lease a minimum of ten Apple iPad (7th generation or later) devices for each Center, unless we grant prior approval for a lower inventory. You must have a sufficient number of iPads, accessories, and peripherals (including but not limited to keyboards and headphones for programs that require them) to serve all students enrolled in SylvanSync programs. We provide certain of the other equipment and licensed software at no additional cost to you. See the Computer Systems section of Item 11 for more information on required materials and equipment we provide and materials and equipment you must obtain from third parties. Computers and other equipment may often be leased over several years. Equipment leases typically require you to pay approximately 10% of the "purchase" price at the outset. To the extent that you are able to lease your equipment, you will substantially reduce your initial costs.

NOTE 12: Except as described below in this Note 12, Sylvan's current policy is that all new franchisees must join the Sylvan Franchise Owners Association ("FOA") and must maintain membership in the FOA for the remainder of the calendar year during which they joined plus the subsequent calendar year (the "Initial Period"). The FOA is an independent franchisee organization that is administered by the Board of Directors of the Sylvan Franchise Owners Association, Inc. ("SFOA"). Sylvan does not own or control any part of the FOA or of the SFOA, and is not a member of the FOA. You must pay the Annual Membership Dues to the FOA during the Initial Period after you sign a License Agreement; the FOA collects your dues. FOA annual membership runs on a fiscal-year basis from July 1 to June 30. Annual Membership Dues are prorated for licensees who first join during a fiscal year. FOA Annual Membership Dues are based on the number of Centers that a member is licensed to operate. FOA Annual Membership Dues for the Initial Period are as stated in the table in this Item 7 for licensees who are licensed to operate one Center and who join the FOA upon the execution of the License Agreement, as required (the FOA provides a 50% discount for the first two years of membership to franchisees who join immediately, as required). If you are licensed to operate more than one Center, the FOA Annual Membership Dues are as follows: 2-3 Centers: \$390 and 4+ Centers: \$480. See Item 20 of this disclosure document for FOA contact information. As an exception to the requirements described above, new franchisees are not required to join the FOA during a given calendar year if fewer than 25% of active licensees in the United States and Canada were members of the FOA as of December 31 of the preceding calendar year. Sylvan may modify, or repeal, its policy that all new franchisees must join FOA from time to time, at its option.

- **NOTE 13:** This estimated amount is \$5 per month per active core student. Your expenses for miscellaneous supplies (including, without limitation, markers, buckets, pens, pencils, paper, instructor name tags, incentive gifts for motivational programs, etc.) may differ depending on the number of students you enroll.
- **NOTE 14:** You may have additional costs if you reimburse your employees to travel to a Satellite location. Travel reimbursement depends on how you pay your employees (e.g., on an hourly or on a per-session basis), among other factors. Sylvan does not set compensation or reimbursement policies for franchisees' employees. We recommend that you consult your professional advisor.
- **NOTE 15:** Sylvan requires that all franchisees use the Hub Technology Platform for its intended uses (for example, recording contacts, scheduling customer appointments, enrolling and scheduling students, tracking attendance, billing customers, and reporting revenue). The Hub Technology Platform Fee is \$114 per Center per month and is included on franchisees' monthly billing statements, due and payable by the 25<sup>th</sup> day of each month. The Hub Technology Fee is subject to change from time to time as it is based upon the license fees and costs that Sylvan is required to pay to the third-party licensor of the system.
- **NOTE 16:** Sylvan requires new licensees to participate in the Contact Center Program for the first twelve months of operation of their first Center (from the date the Center opens for business until the first anniversary of that date). Contact Center Program fees are not charged until you open your first Center. The range included in the chart above includes estimated Contact Center Program fees over a three-month period, and is based upon actual monthly fees paid by franchisees who participated in the Contact Center Program during 2023. There are two different payment models for the Contact Center Program: the first model is a higher flat fee per month rate with a lower per-inquiry variable rate (\$700/month, plus \$10/inquiry), and the second model is a lower flat fee per month rate with a higher per-inquiry variable rate (\$350/month, plus \$30/inquiry). Sylvan currently bills participating franchisees each month based upon their Contact Center Program usage in the prior month. Sylvan then pays the third-party vendor directly. Contact Center Program fees are subject to change from time to time based upon changes in the third-party vendor fees charged to Sylvan.
- **NOTE 17:** This estimates your initial start-up expenses for the initial three-month period. These expenses include business licenses, utility deposits, initial local marketing, payroll costs and required insurance, which is discussed in Item 8. These figures do not include payroll for the owner. These figures are estimates, and Sylvan cannot guarantee that you will not have additional expenses starting the business. We recommend that, in addition to the capital you estimate you will need to reach a break-even point in your operations, you should have available sufficient capital as a contingency to fund the business for a full year. This estimate is based on information provided to us by franchisees, and from our experience in opening and operating company-owned Centers.
- **NOTE 18:** Sylvan requires that licensees use QuickBooks or equivalent or better accounting software in the operation of the Licensed Business. Any software or service used for accounting or bookkeeping must be approved by Sylvan.
- **NOTE 19:** You are required to obtain certain materials to operate your Sylvan Edge Program, and your cost for these materials will vary on the number of sets of materials you chose to purchase. The range provided is the cost of the recommended number of materials you should obtain before providing services under your Sylvan Edge Program. There are no additional materials costs associated with your ACE IT! Program.
- **NOTE 20:** Unless otherwise indicated in the notes above, the fees listed are not refundable, unless you and the vendor agree differently. Neither we, nor our affiliates, finance any part of your initial investment.

# YOUR ESTIMATED INITIAL INVESTMENT – MULTI-TERRITORY INCENTIVE PLAN (Additional Costs to the Above Per Unit Initial Investment)

Type of Expenditure	Amount		Method of	When Due	To Whom
	Low	High	Payment		Payment is Made
Development Fee <sup>1</sup>	\$56,900	\$76,900	Lump Sum	On signing the Multi-Territory Incentive Plan	Sylvan
Legal, Accounting and Other Fees <sup>2</sup>	\$1,000	\$2,000	Lump Sum	As Incurred	Third parties
TOTAL <sup>3</sup>	\$57,900	\$78,900			

**NOTE 1.** You must develop two to three Centers under the Multi-Territory Incentive Plan. The Development Fee due under the Multi-Territory Incentive Plan is based on the additional initial franchise fees for the second and the third Center.

**NOTE 2.** You may incur additional legal, accounting, and other fees for reviewing the Multi-Territory Incentive Plan.

**NOTE 3.** For each Center you develop under the Multi-Territory Incentive Plan, you will also incur the expenses in the table above in this Item 7 (except for the initial franchise fee). The cost for opening multiple centers under the Multi-Territory Incentive Plan is the same as opening one center multiplied by the number of centers the Licensee has been granted to open, which is subject to then-current costs of opening subsequent Centers after the first Center

The amounts in the above charts are estimates only and specific amounts vary depending upon various local conditions that are outside of our control. We relied on our experience in this business to compile these estimates. You should review these figures carefully with the business advisor of your choosing before you purchase a franchise. We do not offer direct or indirect financing for your initial investment.

## ITEM 8: RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

# Products and Services You Must Source from Us

You must obtain SylvanSync Materials from us or our affiliate. The SylvanSync Materials include the then-mandatory SylvanSync instructional programs, and may vary from time to time. As of the issuance date of this disclosure document, these were SylvanSync Reading, SylvanSync Mathematics, SylvanSync Academic Writing, and SylvanSync Advanced Study Skills. SylvanSync proprietary programs are delivered to your server electronically. Core, mandatory SylvanSync programs are covered by your royalty.

The software to use the SylvanSync System is available only from us. We will license to you the required software under the terms of the Software License Agreement and Terms of Use and under the royalty at no additional charge. The standard form of Software License Agreement and Terms of Use in effect as of the issuance date of this disclosure document is attached as Attachment G to the License Agreement.

#### Products and Services You Must Source from Sylvan-Approved Vendors

Before opening any Center, you are required to purchase from a Sylvan-approved vendor an initial inventory of Sylvan Materials. You must at all times maintain a supply of Sylvan Materials sufficient to serve customers and students at your Center and to maintain the quality standards of the Sylvan System and the SylvanSync System. These include materials for the instructional programs that are then mandatory for all Sylvan Learning Centers and that are not already provided in the SylvanSync Materials described in this

Item 8, or by third parties. As of the issuance date of this disclosure document, these materials were various physical teaching tools, called "manipulatives." The price of these materials is subject to change.

You must also purchase certain services and commercially available equipment, software, point-of-sale systems, merchant processing systems, insurance carriers or brokers, marketing services, and other products, services, and materials required by us in the Sylvan Confidential Operations Manual or otherwise in writing for use in a Licensed Business, which we may modify from time to time. We will designate the approved sources of such services and equipment and other materials in the Confidential Manual. You are free to choose in accordance with the approved sources list the wholesale or retail distributor from whom you purchase this equipment and materials.

You may request that we add certain suppliers to the list. There is no charge for our determination of whether to do so. If you nominate a supplier, we will inform you within 30 days whether that supplier qualifies as an approved supplier. We determine whether or not to include a supplier on our list of approved suppliers based on whether its products are compatible with and meet the general quality and service standards for the Sylvan System. These criteria are available to you and approved suppliers upon your request. We may add to the list of approved suppliers, or remove suppliers from the list if they no longer meet the general quality and service standards for the Sylvan System.

We currently receive, and in 2023 received, rebates from some vendors of promotional items and branded merchandise equal to or less than 1% of the gross revenues earned by the vendors from sales to licensees. Other than these rebate programs, we do not receive any payments or other consideration from approved suppliers for your purchases from them or for our designating them as approved suppliers. There are no approved suppliers in which any of our officers own an interest. We do not provide material benefits (e.g., renewal or granting additional franchises) to our franchisees based on a franchisee's purchase of particular products or services or use of particular suppliers. We have not negotiated discounts, group purchasing discounts or special commercial arrangements from vendors for the benefit of licensees. There are no purchasing or distribution cooperatives required or available for Sylvan franchisees. We do not have any vendor contracts for franchisees to purchase or lease equipment. Vendors who participate in our vendor display hall at our Annual Conference pay us a fee to do so. These fees are not based on any purchase(s) by franchisees.

# Products and Services You May Source from Any Vendor

Before beginning operations in your Licensed Business, you must purchase certain initial instructional materials that we designate in our Confidential Operations Manuals or otherwise, and that meet our specifications. In addition, you are also required to purchase certain furnishings (the "Specialized Furnishings") that we designate in our Confidential Operations Manuals or otherwise, and that meet our specifications. You may buy the initial instructional materials and Specialized Furnishings from any vendor as long as the furnishings they supply meet our specifications. The initial instructional materials will be identified by title, author, year of publication, name of product or other identifying data. We base our specifications for the Specialized Furnishings on maintaining the quality of the Sylvan System and the SylvanSync System and a proper environment for providing the services offered at a Sylvan Learning Center. Upon your request, we will provide you with the specifications for the Specialized Furnishings so that you may supply these to your selected vendor.

You are also required to lease or purchase certain commercially available computer equipment, business equipment, supplies and materials, and Internet service that meet our specifications and as we require in the Confidential Operations Manual or otherwise. See Item 11 for more information about our specifications. Upon our request, you must install, update, or replace any equipment (including computer equipment) and software designed to be used in the operation of your Center, and utilize this equipment and software in the manner we specify. This equipment includes computer hardware and software that we require and that is compatible with and accessible to our central accounting system (see Item 11). Certain equipment, supplies, materials, and software for use in your Licensed Business are currently available only

from us, as described above in this Item 8 and in the Confidential Operations Manual. You may source all other items from any vendor so long as the products and services supplied meet our specifications.

Upon our request, you must refurbish, modify, and upgrade your Center to meet our then-current standards, specifications and procedures.

You are required to obtain and maintain insurance as specified in the Confidential Manual in amounts that we require, including insurance for worker's compensation, all risk protection with replacement lost coverage on equipment, furnishings, fixtures, and the building, and comprehensive public liability insurance. We must be named as an insured under the comprehensive public liability insurance policies.

#### **Revenue Disclosure**

For the fiscal year ending December 31, 2023, Sylvan's total revenue from franchisees' required purchases or leases of Sylvan System and SylvanSync System products and services was \$1,155,788, or approximately 3% of Sylvan's total revenues of \$34,594,784.

The purchase or lease from Sylvan of products and services specified by Sylvan is estimated to represent approximately 1% to 3% of a franchisee's total initial investment. The purchase or lease from third parties of materials and products specified by Sylvan is estimated to represent approximately 11% to 12% of a franchisee's total initial investment. The purchase or lease of products and services specified by Sylvan, from Sylvan and from other sources, is estimated to represent up to 6% of a franchisee's annual expenses to operate a Center.

## **ITEM 9: FRANCHISEE'S OBLIGATIONS**

This table lists your principal obligations under the License Agreement and other agreements. It will help you find more detailed information about your obligations in these agreements and in other Items of this disclosure document.

	OBLIGATION	SECTION IN LICENSE AGREEMENT	DISCLOSURE DOCUMENT ITEM
a.	Site selection and acquisition/lease	Section 1.5 of License Agreement	Items 7, 11, and 12
b.	Pre-opening purchases/leases	Sections 7.1, 7.13, and 8 of License Agreement	Item 8
c.	Site development and other pre- opening requirements	Sections 1.6 and 7.6 of License Agreement	Items 7 and 11
d.	Initial and ongoing training	Sections 5, 7.10, and 7.11 of License Agreement	Item 11
e.	Opening	Section 1.6 of License Agreement	Item 11
f.	Fees	Sections 4.1.3, 4.1.4, 6, 7.8.1, and 7.8.2 of License Agreement	Items 5, 6, and 11
g.	Compliance with standards and policies/Operations Manual	Sections 1.7, 1.8, 2, 5, 6, 7, and 8 of License Agreement	Item 11
h.	Trademarks and Proprietary Information	Sections 3, 8.5, and 12.1.2 of License Agreement	Items 13 and 14

	OBLIGATION	SECTION IN LICENSE AGREEMENT	DISCLOSURE DOCUMENT ITEM
i.	Restrictions on products/services offered	Sections 1.8.3 and 7.1 of License Agreement	Item 16
j.	Warranty and customer service requirements	Sections 6.10 and 7.4 of License Agreement	Item 11
k.	Territorial development and sales quotas	Sections 1.5, 6.2.2, 7.7, and 11.2.9 of License Agreement	Item 12
1.	Ongoing product/service purchases	Sections 7.1, 7.4, and 8.1 of License Agreement	Item 8
m.	Maintenance, appearance, and remodeling requirements	Sections 7.1 and 7.9 of License Agreement	Item 11
n.	Insurance	Section 14.2 of License Agreement	Item 8
0.	Advertising	Sections 7.3, 7.5, 7.8, and 8.4 of License Agreement	Items 6, 7, and 11
p.	Indemnification	Section 14.1 of License Agreement	Not applicable
q.	Owner's participation/ management/staffing	Sections 4.1.3, 5.1, 7.2.1, 7.4, 7.10, and 11.2.7 of License Agreement	Items 11 and 15
r.	Records/reports	Sections 6.4, 6.5, 6.6, 7.9, 7.12, 8.1.5, and 8.2 of License Agreement	Item 6
S.	Inspections/audits	Sections 6.6, 7.9, 7.12, 8.1.5, and 8.2 of License Agreement	Item 6
t.	Transfer	Section 9 of License Agreement	Item 17
u.	Renewal	Section 10.2 of License Agreement	Item 17
v.	Post-termination obligations	Section 12 of License Agreement	Item 17
w.	Non-competition covenants	Sections 7.3, 8.4, and 12.5 of License Agreement	Item 17
х.	Dispute resolution	Sections 7.9.4, 20, and 21 of License Agreement	Item 17
y.	Other Sylvan Edge Package License Agreement	Exhibit B to the disclosure document	Items 1, 5, and 12
	ACE IT! Tutoring System License Agreement	Exhibit C to the disclosure document	Items 1, 5, and 12

OBLIGATION	SECTION IN LICENSE AGREEMENT	DISCLOSURE DOCUMENT ITEM
Contact Center Franchisee Participation Agreement	Attachment F to the franchise agreement	Items 6, 7, and 15
Bright Horizons Participation Agreement	Attachment K to the franchise agreement	Item 1
mySylvan Marketplace+SM Franchisee Participation Pilot Agreement	Attachment L to the franchise agreement	Items 1, 12

#### **ITEM 10: FINANCING**

Sylvan does not offer any direct or indirect financing. Sylvan does not guarantee any note, lease, or other obligation. We do not have agreements to refer you to third parties for financing.

You may, however, obtain financing for your equipment purchases and purchases of educational materials from third parties.

# ITEM 11: FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, Sylvan is not required to provide you with any assistance.

# **Pre-Opening Obligations**

Before you open your business, we will provide you with the following assistance on the same basis that we provide this assistance to other Sylvan licensees:

- (1) We will designate your Territory. (License Agreement—Section 1.3).
- (2) We will assist you in the selection of a location for your Primary Center and make recommendations and suggestions regarding the layout and décor of the Primary Center to assist you in complying with our standards. (License Agreement—Section 4.1.1). Franchisees generally lease premises from third-party landlords. We do not generally own and lease premises to franchisees.

In order for you to operate any Center, you must first obtain our approval of the premises and location from which your Center will be operated (License Agreement—Section 1.5) and of your Center floor plans and signage (License Agreement—Section 4.1.1). We will consider the condition of the building and layout, the size and age of the population within your Territory, accessibility of the facility to that population, existence and proximity of other supplemental instruction programs, the proximity of the facility to your Territory boundary and other Sylvan locations, the extent to which the neighborhood is commercial, industrial, or residential, and available parking in the area. We assign values for each of these factors and use the values to evaluate the premises. We also make determinations based upon our Territory Boundary Policy, which we may modify from time to time (see Item 12). We typically notify franchisees of whether or not their site has been approved within two weeks after the franchisee provides a completed Site Approval Package to us.

Our License Agreement requires you to open your Primary Center within 180 days of the effective date of the agreement (or another date on which you and we agree) (see Item 15). As noted above in Item 1, you may commence delivery of virtual services prior to opening the brick-and-mortar location of your Primary Center, provided that you and your staff have completed all initial training requirements in advance and that you have obtained Sylvan's approval to do so. If we do not approve the proposed site that you submit

to us or if you do not open your Primary Center within this 180-day timeframe, we may cancel your License Agreement (License Agreement—Section 1.5). Sylvan may cancel your License Agreement based upon such failure regardless of whether you are delivering services to students virtually in your Territory at such time. Sylvan reserves the right to charge a fee in connection with any extension of the required opening date for the Primary Center in your Territory if Sylvan, in its sole discretion, approves your request for the extension. In our Confidential Operations Manual, we will provide you specifications for materials, equipment and furniture required for opening and operating your Primary Center, additional Centers, and Satellites (see Item 8).

(3) We will give you access to our Confidential Manual.

The Confidential Manual is confidential and proprietary and remain our property. Sylvan may modify the Confidential Manual, and you must at all times adhere to the then-current contents of the Confidential Manual. (License Agreement—Section 1.8.1, 1.8.2, 1.8.3, 2.1, 4.1.2, 7.1, 7.2, 7.9, 7.11 and 8). You must keep the Confidential Manual and its content confidential. (License Agreement—Section 2.1.3). Exhibit E to this disclosure document includes a copy of the Table of Contents for the Confidential Manual, which has a total of 205 pages.

(4) We will provide initial training to up to three members of your staff (including yourself). (License Agreement—4.1.3 and 5). The training that Sylvan provides to you and to your employees does not constitute employment, direction, or supervision of you or your employees by Sylvan. You shall be fully and solely responsible for all employment actions and decisions related to your employees, including hiring and firing your employees, and providing any additional training you deem necessary in your day-to-day oversight of your Licensed Business.

You must complete your initial training before you may open your Primary Center. If you wish to commence virtual delivery ahead of opening your Primary Center, you must complete your initial training before doing so. Franchisees are typically required to open their Primary Centers within 180 days of signing a License Agreement. Therefore, you must complete your initial training within 180 days after you sign a License Agreement. For new franchisees who purchase a new territory, the three total initial training slots are at no cost to you as long as you complete them all within 180 days of the Effective Date of your License Agreement; for new franchisees who purchase an existing territory through a resale transaction, the three total initial training slots are included within the Supplemental Training and Administrative Fee that Sylvan may elect to charge. You are required to complete this training to our satisfaction as a condition of opening your Center. Sylvan reserves the right to require in-person training, which would take place at Sylvan's facilities in Towson, Maryland or a Sylvan Learning Center location designated by Sylvan.

8). We will also assist you by providing certain required supplies (as further described in Item 8). We will provide to you electronically instructional materials we consider appropriate for instruction of your personnel in conducting our instructional programs. (License Agreement—Section 4.1.4). We will also provide certain software that you must purchase from us or our affiliate, which we may deliver to you electronically and help you install remotely. (License Agreement—Section 4.2.7). Further, you must also purchase certain services and commercially available equipment and other materials that we identify in the Confidential Manual, or otherwise in writing, as required for use at Centers. (License Agreement—Section 7.1). We will designate the approved sources of such services and equipment and other materials by maker or author, publisher, and title in the Confidential Manual. You are free to choose in accordance with the approved sources list the wholesale or retail distributor from whom you purchase this equipment and materials. Other initial purchases, such as some initial instructional materials and Specialized Furnishings, may be purchased from any vendor as long as the items they supply meet our specifications. We reserve the right to charge separate fees for additional, specialized, or supplemental training.

# Opening – Primary Center

Licensees typically open their Primary Center three to four months after they sign a License Agreement and pay the Initial License Fee. You must complete the first portion of required training before you may open your Primary Center. You must open your Primary Center no later than 180 days after you sign your License Agreement. The factors affecting the time to open are the ability to purchase or lease a site location, whether it is necessary to make substantial improvements to the site, obtaining financing, complying with zoning and local ordinances, weather conditions, shortages and delayed installation of equipment, and fixtures and signs.

If you receive advance approval from us, you may begin offering and delivering virtual services in your Territory, or operating one or more Satellite locations in your Territory, at any time during the 180 days before you are required to open your brick-and-mortar Primary Center, so long as you already have completed satisfactorily, and caused your staff to complete satisfactorily, initial training.

# **Post-Opening Obligations**

During the operation of your franchise, we will provide you the following assistance on the same basis as we provide this assistance to other Sylvan franchisees:

- (1) As we develop products and services you will offer to your customers and otherwise improve and develop the Sylvan System, we will provide you pertinent new and different standards, procedures, and techniques that are developed and recommended by us, and such changes to the Confidential Manual. We reserve, to the fullest extent permitted by then-applicable law, the right to establish policies and programs regarding pricing of products and services, including, but not limited to, establishing the maximum and minimum retail prices, recommending retail program prices, advertising specific retail prices for some or all products or services sold at your Licensed Business, and developing and advertising price promotions or package promotions. We may compel you to observe, honor, and participate in any such policies or programs we establish. (License Agreement Section 2.1.4)
- (2) We will assist you in developing an action plan for operating your Licensed Business. The plan will include a marketing strategy and an estimate of expenses and sales goals. (License Agreement—Section 4.2.1)
- (3) We will maintain a toll-free number you may use to contact our employees at our headquarters. (License Agreement—Section 4.2.2).
- (4) We will provide you certain advertising, marketing and/or promotional materials in forms and media determined by us at our sole discretion, which we shall have approved and updated at least every other year. (License Agreement—Section 4.2.3).
- Reviews. We will assist you in resolving operating problems through periodic Quality Assurance Reviews. We will maintain a procedure and checklist for the review of all aspects of your operation of your Licensed Business ("Quality Assurance Review") whereby we can determine the extent to which your operations are in compliance with our standards. Approximately 120 days after you first open your Primary Center, we will conduct a Quality Assurance Review of your Primary Center through remote review of materials or by having a Sylvan representative conduct an on-site review. At your request, we will have a business consultant or quality assurance manager conduct a Quality Assurance Review at your Licensed Business once in every two-calendar year period if we have not ourselves initiated one, provided that you are in compliance with all financial and reporting obligations to us, and that you are not subject to default or termination under the License Agreement. If you are not in compliance or are subject to default or termination, you agree to reimburse us for our travel, lodging, meal and other costs incurred in providing an on-site Quality Assurance Review. At your request, we will provide you with the materials and forms used in the Quality Assurance Review process for your use in reviewing your operations. (License Agreement—Section 4.2.4).

- (6) If a Sylvan business consultant or quality assurance manager has not already conducted two visits to your Licensed Business, then we will at your request cause a Sylvan representative to provide assistance and offer suggestions and recommendations, either via up to two one-day visits each calendar year or via other customary means of providing advice, at our sole discretion. (License Agreement—Section 4.2.5).
- (7) We will attempt to keep you abreast of the most recent and advanced knowledge concerning instructional equipment and new reading and mathematics or other instructional techniques. (License Agreement—Section 4.2.6).
- (%) We will lend or otherwise provide to you certain Sylvan computer hardware and software ("Sylvan Hardware" and "Sylvan Software") that you must use in operating the SylvanSync System in your Licensed Business. We identify the Sylvan Hardware and Sylvan Software in our Confidential Operations Manual; we may change the items we designate from time to time. The Sylvan Hardware and Sylvan Software remains our property. (License Agreement—Section 4.2.7).
- (9) We will make available to you at no charge each assessment that we deem mandatory in any mandatory Sylvan System or SylvanSync System program. You may also use these assessments at no charge to you in other optional and mandatory Sylvan System or SylvanSync System programs where we approve you to use the assessments. (License Agreement—Section 4.2.8).
- (10) From time to time, we may conduct a regional training program and provide ongoing training as we deem necessary. (License Agreement—Sections 4.4 and 5.1). We describe our current training programs below. The training that Sylvan provides to you and to your employees does not constitute employment, direction, or supervision of you or your employees by Sylvan. You shall be fully and solely responsible for all employment actions and decisions related to your employees, including hiring and firing your employees, and providing any additional training you deem necessary in your day-to-day oversight of your Licensed Business.
- (11) At least once every two years, we will hold a conference of not more than five days at a location we select. We will determine the topics and agenda for each conference to serve the purpose of updating licensees on new developments and exchanging information between licensees and us regarding Sylvan System and SylvanSync System operations and programs. The conference may take place in person or virtually, at our option. We may charge a fee for attendance at this conference. You must bear all travel, room, and board expenses for you and your personnel attending these conferences. (License Agreement—Section 4.3.1).

Except as described above, the License Agreement does not require us to provide any other assistance or services to you during the operation of the Licensed Business. For example, we are not required to establish and use administrative, bookkeeping, accounting, or inventory control procedures.

#### **Computer Systems**

You are also required to lease or purchase certain commercially available computer equipment, business equipment, supplies, materials, and Internet service that meet our specifications and as we require in the Confidential Operations Manual or otherwise. As an example, and not as a limitation, the Apple iPad (7<sup>th</sup> generation or later) is the only device we designate for use with SylvanSync app as of the issuance date of this disclosure document. Sylvan recommends using Apple iPad 9<sup>th</sup> generation or later tablet computers; Sylvan does not support Apple iPad devices that are 5<sup>th</sup> generation or earlier. We estimate the costs for three to five computer workstations with operating software at \$600 - \$1,000 each, 12-21 iPad (7<sup>th</sup> generation or later) tablet computers at \$349 each, 12-21 covers/screen protector kits for the iPads at \$10 each, \$700 total for a printer and copier (combined or separate), and three months of internet and telephony services at \$100 - \$230 per month and a small office/business firewall and wireless access point at \$350 - \$700. You must purchase or lease a minimum of ten Apple iPad (7<sup>th</sup> generation or later) devices for each Center, unless we grant prior approval for a lower inventory. You must have a sufficient number of iPads,

accessories, and peripherals (including but not limited to keyboards and headphones for programs that require them) to serve all students enrolled in SylvanSync programs. See below in this Item 11 for more information about our specifications. Upon our request, you must install, update, or replace any equipment (including computer equipment) and software designed to be used in the operation of your Licensed Business, and utilize this equipment and software in the manner we specify. This equipment includes computer hardware and software that we require and that is compatible with and accessible to our central accounting system (see below in this Item 11). Certain equipment, supplies, materials, and software for use in your Licensed Business currently are available only from us, as described above in Item 8 and in the Confidential Operations Manual. You may source all other items from any vendor so long as the products and services supplied meet our specifications.

You must own or lease (as applicable), install, use, and maintain the Computer System we specify. Our minimum Computer System specifications as of the issuance date of this disclosure document are as follows:

#### Hardware and Software

When acquiring new workstation computers, you must purchase systems installed with Windows 10 Professional or better. At least one computer in the Center must have a current copy of Microsoft Office or equivalent software installed. We require 64-bit operating systems to run our software suite. You must have at a minimum:

- i. depending on Center size, from 3-10 personal computers/workstations with an Intel i3 or equivalent processor or better (i5 or equivalent recommended), memory of at least 8.0 GB RAM minimum (16.0 GB RAM recommended) and a 100/1000 Ethernet NIC network card or highspeed wireless adapter;
- ii. 17-inch LCD or better monitor, with 1280 x 1024 resolution or better;
- iii. keyboard, mouse, and multimedia sound card;
- iv. required software, as loaned to you by Sylvan. This is part of the Sylvan Software defined earlier in this Item 11;
  - o Hub Center management software;
  - o EdAdmin software;
  - SylvanSync and SylvanSync Connect software;
  - Assessment software;
- v. Ethernet or wireless-enabled laser printer;
- vi. copier (a combined printer-copier unit is acceptable);
- vii. telephone system;
- viii. service, equipment, or software with ability to record outgoing telephone greetings and incoming telephone messages;
- ix. at least 10 Apple iPad (7<sup>th</sup> generation or later) tablet computers, each with Wi-Fi capability and at least 16 gigabytes of memory;
- x. a dedicated email address for your Center, and a separate dedicated email address for your director, which we may assign to you; and
- xi. other equipment and specifications as we determine. As an example, and not a limitation, if you do not have a scanner for sending documents via email, you must have a fax machine.

You must install the most current version of either the Google Chrome or Microsoft Edge browser on every workstation that you use to conduct assessments.

Only the following Windows operating systems are currently supported:

- Windows 11 Professional (recommended)
- Windows 10 Professional (end of support October 14<sup>th</sup>, 2025)

The SylvanSync Software is used in the delivery of assessments and offering the Sylvan instructional programs at your Center.

One of your personal computers/workstations is to be used for your full-time employee (Center Director and Director of Education; if you have two full-time Directors, each should have a workstation), and at least two workstations are to be used for testing and instruction. The personal computers must have a sound card, hard drive(s) with sufficient memory, an operating system that meets at least our minimum criteria above, or other operating system we specify, an Internet browser compatible with our systems, and other software. You must have a sufficient number of tablet computers to serve all students enrolled in SylvanSync.

After the initial installation of your computer hardware, you must plan for the ongoing upgrade and replacement of hardware every 3-5 years to maintain current compatibility with the hardware specifications supported by Sylvan Software.

# Internet Service, Email, and Telephony

You must have a business-class Internet connection through a national or local internet service provider (ISP). In some areas, T1 or fractional T1 lines may be the only options to support the required bandwidth and stability.

For optimal performance, the recommended Internet speeds in-Center are 30 Mbps download and 3 Mbps upload speed (or greater). Minimum required speeds are 15 Mbps download and 3 Mbps upload speed.

Mobile broadband solutions are not acceptable, except as a backup.

You must have an email service. See below in this Item 11 concerning email addresses you must use.

You must have reliable telephony services. We strongly recommend using Voice over Internet Protocol (VoIP) services; we do not recommend using analog services.

You must purchase and maintain wired and wireless network services in your Center, including a firewalled internet services gateway, wireless access point(s), and wired networking switch(s). The minimal equipment needs will vary based on the location's square footage and floorplan. The equipment purchased should meet the requirement of supporting a minimum of 25 users/devices. These include workstations, laptops, iPads, VoIP phones, printers, and network attached storage devices.

You are responsible to us for our proprietary software and all other software we loan to you (collectively, the "Sylvan Technology"), and you must maintain it in good condition and repair, normal wear and tear excepted. You are responsible for loss of, or damage to, the Sylvan Technology due to fire, theft, or other catastrophe, and you must replace any damaged or lost components of the Sylvan Technology at your cost. You must keep the Sylvan Technology insured against catastrophic loss. In some instances, we may elect to replace hardware or software in the case of software failure or corruption. We will provide support for the Sylvan Technology software and hardware via phone during advertised hours as posted on our intranet, SLCLink, or as we otherwise specify, provided that you are current on your payments to us of SylvanSync billings.

You must provide us unimpeded independent access to your Computer System and all required software. There are no contractual limits on Sylvan's ability to access your data. The hardware and software you use

in your business must be compatible with and accessible to our central accounting system through the Internet.

None of the suppliers of the above computer hardware or software, including without limitation Sylvan, has an obligation to provide you with ongoing maintenance, repairs, upgrades, or updates, unless required by your contracts with them or with the wholesalers or retailers from whom you purchase these items. You are responsible for hiring a local technician in your area to assist you in maintaining the integrity of your hardware and networking components and to provide you support with installation and troubleshooting.

Because changes to technology are dynamic and cannot be predicted within the term of your License Agreement, and in order to provide for inevitable but unpredictable changes to technological needs and opportunities, we have the right to establish, in writing, reasonable new standards to address new technologies, whether published in the Confidential Manual, or otherwise in writing, and we have the right to implement those changes in technology into the Computer System and to add or modify fees accordingly, subject to the maximum required expenditures identified in the License Agreement.

You must implement upgrades and updates to the Computer System and all other technology aspects of the Licensed Business at such times, and in the manner, as we specify. We anticipate that you will need to update the Computer System every 3-5 years, if not sooner, for certain elements of the Computer System. The estimated cost to upgrade the Computer System is likely to be about \$3,000 to \$7,000, but Sylvan may not require you to spend more than \$5,000 per Center on computer upgrades in any given year.

#### **Email Addresses**

In running the Licensed Business, you must use the email address(es) and email domain name(s) that we specify. These specifications are subject to change by Sylvan. Your dedicated email addresses must be used only in operating your Licensed Business and not for any other purpose.

#### Websites

We shall arrange for maintenance of a website for the benefit of Sylvan and the Sylvan brand and Sylvan Network generally if we draw Advertising Allocations to fund same. We shall provide one or more references to your Center within such Sylvan website that we fund. You must ensure that all information concerning your Center is accurate and up to date. (License Agreement—Section 7.5.3)

You may not establish, use, or authorize others to establish or use, any separate website that relates to your Center or refers to our Licensed Marks except with our prior written approval, which we may withhold at our sole discretion. If we approve your request to use a separate website, you must comply with our standards and requirements for the website and for advertising in general. (License Agreement—Section 8.4)

#### Advertising

We may direct some amount (the "Advertising Allocation") of your royalty toward funding advertising and marketing programs and activities that are intended to increase the public's awareness of the Sylvan Network, the Sylvan System, and the SylvanSync System generally ("Sylvan Advertising"). We determine the amount and frequency of the Advertising Allocation and may change the Advertising Allocation at any time. We have sole discretion to determine what advertising, marketing, and public relations programs, and activities we undertake within Sylvan Advertising, and we have sole discretion over the creative concepts, materials, and endorsements used in such Sylvan Advertising and the geographic, market, and media placement and allocation of advertising and marketing materials in connection with Sylvan Advertising. The License Agreement does not require that Sylvan Advertising benefit licensees directly or on a pro rata basis, so we are not required to spend any amount on advertising in your Territory. We may, at our sole discretion, modify, decrease, suspend, and/or increase expenditures and/or operations of Sylvan Advertising from time to time. You are required to participate in all Sylvan Advertising activities and programs that Sylvan designates, but you are not required to make expenditures outside of the

Advertising Allocation and/or your local advertising expenditure requirements and the limits on those in the License Agreement. (License Agreement—Section 7.5.2). The amounts collected in connection with the Advertising Allocation are not audited, however, unaudited financial statements are available to franchisees upon request. The amount of your Advertising Allocation may be determined from time to time based on an "alternative funding formula" ("AFF"). Licensees who signed License Agreements prior to March 2016 are on various earlier forms of Sylvan License Agreements. We call those earlier forms "Legacy License Agreements." Certain Legacy License Agreements provide for an AFF that sets the level of those licensees' spend on the types of activities pursued in Sylvan Advertising. For the purposes of setting your Advertising Allocation, if an AFF is proposed to licensees in the United States under the terms of the Legacy License Agreements, you shall be deemed to have agreed to the AFF, and the Advertising Allocation to Sylvan Advertising from your royalty shall be in the amount set forth in the AFF for the term of the AFF, if the AFF is approved by licensees owning at least two-thirds (2/3) of outstanding License Agreements in the United States. In all other cases, Sylvan will determine the Advertising Allocation amount as we describe in the first paragraph of this "Advertising" section of Item 11.

In the third quarter of 2016, we established and administered a Licensee Advertising Advisory Committee (the "Committee") that consults with us concerning the selection and placement of Sylvan Advertising. Sylvan appoints three licensees to sit on the Committee for a one calendar year term; members may be reappointed for additional terms. Each licensee member must be in good standing under all of its License Agreements with Sylvan during such licensee member's term on the Committee; otherwise, Sylvan may remove them from the Committee. Licensee members do not have voting power in Sylvan's decisions. Sylvan meets with the Committee as appropriate up to four times per calendar year. All meetings are held by telephone unless Sylvan designates an alternate means of meeting. (License Agreement— Section 7.5.1.) We have the power to form, change, dissolve, or merge the Committee, and any other licensee advisory council.

We will report to licensees annually about advertising, marketing, and public relations activities we conducted and propose for the future, and about the amount that we expend on Sylvan Advertising.

Sylvan-owned Centers are not contractually obligated to contribute an Advertising Allocation, but in the past have contributed to funding advertising on the same basis as licensee Centers.

Most licensees on Legacy License Agreements make their advertising contributions through a National Advertising Fund (the "Fund"). Contributions by licensees on Legacy License Agreements to the Fund are administered by SLC National Advertising Fund, Inc. ("SNAF") and its four directors, two of whom are appointed by Sylvan, and two of whom are appointed by the FOA. We do not have any voting power within FOA. The monies in the Fund were spent in the 2023 fiscal year as follows (based on unaudited figures):

	Percentage of Total
Advertising Production	4%
Media Placement	36%
CRM Enterprise Solution & Website	18%
Local and Social Marketing and Public Relations	13%
General and Administrative	
(including internal resources dedicated to marketing effo	erts) 29%
Total	100%

The main Sylvan company website features a link to a microsite with information about becoming a Sylvan franchisee. Other than these assets, SNAF has never used any of the Fund for solicitation of new licensees.

In addition to the contributions you are required to make via your Advertising Allocation, you are required to expend certain amounts on local advertising (see Item 6). We will provide you with advertising materials. You may not use any other advertising materials unless we give our prior approval.

We may require you to contribute up to 100% of your local advertising expenditures to an advertising cooperative. An Acknowledgment of Request to Contribute to Advertising Cooperative (the "Acknowledgment") is attached to the License Agreement as Attachment E. We will require you to sign the Acknowledgment when you sign the License Agreement. Membership in advertising cooperatives is usually based upon market areas defined by media rating agencies such as Arbitron or Nielsen. Cooperative members administer the advertising cooperatives. Sylvan advertising cooperatives are not required to operate from written documents, but most do, and these should be available for your review by contacting the cooperative membership. Many Sylvan advertising cooperatives prepare annual financial statements that should be available for your review by contacting the cooperative membership. We may require any advertising cooperatives to be formed, changed, dissolved, or merged. Sylvan is not required to participate in any cooperative for its company-owned Centers, but may choose to do so. We will require you to contribute to your local advertising cooperative on the same basis as a majority of the other members contribute, for up to three years.

We must approve your signs that contain our Licensed Marks before you may display them to the public.

#### **Training**

We provide you and your staff certain training through our "Sylvan University" department and through "Sylvan University Now" ("SUN"), our proprietary online learning management system for Sylvan franchisees and staff. The Confidential Operations Manual contains the detailed requirements and features of our training programs for the respective system at any given time. Sylvan currently provides the Initial Training Program free of charge for up to three members of your staff (including you) who must complete it within 365 days after you sign your License Agreement, as described below. For all other training, we may charge you our then-current additional training fee plus reimbursement of our out-of-pocket expenses.

We may offer training via instructor-led in-person sessions, group webinars, and online via SUN and online discussion boards. We may change our training requirements (including without limitation programs and fees) from time to time via changes to our Confidential Operations Manual, or otherwise in writing, and you and your staff must comply with such changed requirements. The "Training Programs" chart later in this Item 11 provide details of training programs we currently offer as of the issuance date of this disclosure document.

#### Our Initial Training Program; Your Initial Training Requirements

We provide to licensee an Initial Training Program, which we describe later in this Item 11. If you are licensing an undeveloped territory, our License Agreement requires you to open your Primary Center within 180 days of the effective date of the agreement (or another date on which you and Sylvan agree). You must ensure that the following initial training requirements are met in connection with opening your Primary Center or commencing delivery of Sylvan and SylvanSync System programs at any other site, whether in person or online:

TRAINING PROGRAMS

SUBJECT	HOURS OF IN-PERSON TRAINING	HOURS OF IN- CENTER TRAINING	TRAINING DESCRIPTION	LOCATION
Initial Training Program:  Franchisee Certification Part I and II (Note 1)	40	32	Part I: Approximately 3 hours of online self-directed activities, 8 hours of live webinars, 2 hours of on- demand training, followed by one week (40 hours) of live, in-person, training in a Sylvan Learning Center designated by Sylvan.  Part II: A series of 9, 1.5- hour live webinars over a three-week period, plus approximately 5.5 hours of independent activities. Several personal	Online in SUN  Live Webinars via Zoom  In a Sylvan Learning Center of Sylvan's choosing  Participants will secure their own travel and lodging arrangements, as necessary.
			consultations are also required and conducted throughout this training. (Duration varies.)	
Optional Training Courses on SUN (Note 2)	0	0	Approximately 100 hours of online self-directed activities and training in addition to the other programs in this table.	Online in SUN
In-Center Training (Note 43	Approximately 6	0	0	Varies
Annual Conferences (Note 4)	Approximately 24	0	0	In-person; varies
Total	70	32	100 (Optional)	

All training may be held in-person, online, or at the location that we designate (which may be Sylvan's office in Towson, Maryland or at a Center), at our option. We may modify the location of training or the format (e.g., utilizing Zoom or another virtual interface) at any time.

#### **NOTE 1:** Initial Training Program: Franchisee Certification Parts I and II

This Initial Training Program consists of two sequential components that licensee (you or, if the licensee is a corporation, one of its shareholders) must complete in connection with opening your Center to the public, as described above. We provide Franchisee for up to three members of your staff (including yourself).

Curriculum: Our Franchisee Certification program begins with a conference call with a Franchise Business Consultant. During that consultation, the recommended sequence of training courses for you and your stakeholders as well as scheduling details will be discussed. Among other subjects, the Franchisee Certification program covers an overview of a Center's educational services, the customer lifecycle, marketing, sales, operations, management, Center technologies, education, SylvanSync, Satellites, and online reputation management.

Fees: Currently, you receive up to three seats in our Franchisee Certification program at no charge under your Initial License Fee, as long as the seats are claimed within the first 180 days after you sign your License Agreement. The licensee (you or, if licensee is a corporation, one of its shareholders) must complete the Initial Training Program, Franchisee Certification Part I, within 180 days of signing the License Agreement. The licensee (you or, if licensee is a corporation, one of its shareholders) must complete the Initial Training Program, Franchisee Certification Part II, within 180 days of opening the Sylvan Learning Center or completing Franchisee Certification Part I, whichever comes first. We may charge you an extra fee if you wish more than three people to participate in the Franchisee Certification component. If it becomes necessary, after you open your Primary Center, to send a new staff member to participate in our Franchisee Certification component of the Initial Training Program because you are replacing a member of your staff, or if you complete the Initial Training Program after the one-year anniversary of the Effective Date of your License Agreement, we may charge you our then-current additional training fee plus require reimbursement of our out-of-pocket expenses.

When Offered: We offer Franchisee Certification from time to time in connection with new licensees entering into License Agreements and subject to the availability of Sylvan's instructors.

## **NOTE 2:** Optional Trainings Available on SUN

SUN also contains approximately 100 hours of additional single-subject and curriculum trainings. All trainings are self-directed and offered online. There is no additional fee for any of these trainings so long as you are current on payment of your royalty. Our then-current online trainings are available online at all times.

#### **NOTE 3:** In-Center Instruction

We may, but are not obligated to, offer trainings to you and your staff on an ad hoc basis at your Center, or at another location we select at our discretion.

#### **NOTE 4:** Annual Conferences

At least once every two years, Sylvan will hold an Annual Conference at a location we determine (though we refer to these as "Annual Conferences," we may choose to offer them only every other year). Annual Conferences will last no more than five days. We will determine the topics and agenda for Annual Conferences, which generally include updating our licensees on new developments affecting them and exchanging information between our licensees and our personnel concerning the operations and programs of Centers. We may charge you a reasonable fee for you and your staff to participate in any Annual Conference, and you must pay all costs and expenses you or your staff who participate incur, including travel and living expenses and wages for you and your staff. The current fee for the Annual Conference is

approximately \$900 - \$1,100 and is subject to change from time to time based on Sylvan's costs related to the Annual Conference and in Sylvan's sole discretion.

The purpose of the training programs described in this Item 11 and in Item 15 is to apprise senior Center personnel of Sylvan brand standards, so that the senior personnel may deliver, and train the Center's staff to deliver, consistent, high value service under the "Sylvan" and "SylvanSync" marks from all Sylvan Learning Centers. The training that Sylvan provides to you and to your employees does not constitute employment, direction, or supervision of you or your employees by Sylvan. You shall be fully and solely responsible for all employment actions and decisions related to your employees during training and during their employment in the Licensed Business.

#### Instructors: Sylvan System and SylvanSync System

Erin Conery is the Senior Director of Training and Communication. She holds a B.A. in Comprehensive English Education from University of Wisconsin–Eau Claire and an M.S. in Instructional Design from Walden University. After working in franchised Sylvan Learning Centers for 3 years, Erin joined the Sylvan Franchise Services team in 2006 and began training SylvanSync in November 2011. Erin has over 20 years of training, teaching, and facilitation experience.

Leslie Poteet is the Vice President of Operations. Leslie holds a B.A in English from Suffolk University and an M.A in Classical Studies from Boston University. After working in both company-owned and franchised Centers for five years, Leslie joined the Sylvan Franchise Services team in 2011 and began training on contract services and SylvanSync. Leslie has 18 years of training, operations, business development, and product management experience.

Many other instructors will be involved in training licensees. Many instructors are experts in education generally and Sylvan in particular. They have backgrounds in one or more areas including education, instructional design, technology, marketing, business development, and management.

#### **Additional Programs**

Sylvan has developed additional programs and may continue to develop other additional educational programs and packages, such as the Sylvan Edge Program and the ACE IT! Program, which are not part of the core Sylvan System or the SylvanSync System but Licensee may choose to participate by signing the appropriate agreement, which will amend the License Agreement. If Sylvan develops additional education programs, Sylvan may, but is not obligated to, offer the additional programs to licensees. These additional programs are optional. In order to participate in the additional programs, you may be required to sign a separate license agreement, pay additional initial and ongoing fees, and satisfactorily complete additional training relating to the program.

#### **ITEM 12: TERRITORY**

You will be assigned a protected territory (your "Territory") by your License Agreement, and you will be licensed to deliver the Sylvan System and SylvanSync System programs and services in your Territory. Using demographic data, Sylvan will work with you to determine the territory boundaries to promote growth of the Licensed Business. A typical territory is an area with a radius of three to five miles, depending upon the density of the population in the territory. However, a territory in a metropolitan area might be much smaller, and a territory in a rural area might be much larger. Each territory is delineated by geographical or political boundaries, latitude and longitude, streets, or an area within a specified radius from an identified point.

We have a Territory Boundary Policy (the "Policy"), which we may modify from time to time. Under our current Policy, each Center and Satellite must be located within your Territory. Each Center should be centrally located within your Territory at a site chosen based upon population density and demographic data. You may not open a Center which is less than one quarter of a mile from any border of your Territory. Sylvan reserves the right to withhold its approval of any Center site proposed to be located less than two

miles from your Territory boundary, based solely on the proximity of such proposed Center site to a bordering territory, regardless of whether such bordering territory is licensed by Sylvan at the time of your proposal. Sylvan may allow you to open a Center between one quarter of a mile and two miles inside the border of your Territory if you satisfy the Policy's then-current population density requirements and other guidelines. If you open a Center close to the boundary of your Territory, you risk that a neighboring franchisee may open another Center similarly close to the same border within its own territory. Sylvan expressly disclaims any and all liability which may result from its approval of multiple Centers located in close proximity to one another. If and when applicable, Sylvan may notify neighboring franchisees of your intention to open a Center within two miles of your Territory border; alternatively, Sylvan may require you to do the same in writing. You may operate one or more Satellites within your Territory with Sylvan's prior approval, which may offer services in compliance with Sylvan's Confidential Operations Manual inperson or virtually. If Sylvan approves any Satellite location which is less than two miles from your Territory border, Sylvan may notify the appropriate neighboring franchisee or require you to do so in writing. We must approve all locations in advance; you must not commence operations in any Center or Satellite location (including a virtual Satellite) without first obtaining Sylvan's prior written approval. We may make exceptions to the Policy from time to time at our sole discretion.

If you are unable to find a suitable site within your Territory and wish to shift the boundaries of your Territory to accommodate your Primary Center site proposal, Sylvan has no obligation to allow such a modification. Sylvan reserves the right to deny your request or to charge a \$1,000 fee in connection with the amendment to your Territory.

Your license permits you to operate one Primary Center within your Territory. If you desire to open an additional Center in your Territory, you must obtain our written approval and pay us \$1,000 for each additional Center to compensate us for our expenses in site review and plan approval. For all Center openings, we require you to comply with the Territory Boundary Policy described above. You are not required to pay us a fee to open additional Satellites. You may only operate in Centers and Satellites that we approve in advance in writing. You may relocate any existing Center within your Territory with our prior written approval, and subject to our then-current Territory Boundary Policy and standards and procedures for approval.

Your Territory is protected, which means that, except as we provide in the License Agreement and in this Item 12, we will not operate or license to others the right to operate a Sylvan Learning Center utilizing the Sylvan System within your Territory so long as your License Agreement remains effective, and you are in compliance with its terms and conditions. We reserve all other rights. As an example, and not a limitation, among other examples, Sylvan may sell, provide, and deliver, directly or through affiliates or third parties, Sylvan System programs pursuant to school system or local, State or Federal government-sponsored or funded contracts or corporate contracts, which require the delivery or performance of Sylvan System services within your Territory. Further, and for the avoidance of doubt, any third-party licensee, franchisee, or a corporate affiliate of Sylvan may provide virtual services to any student, wherever he/she is located, including students who reside within your Territory.

We may sell, operate directly and license and franchise others to sell and use within your Territory under Sylvan's trademarks or under different names or trademarks programs and products that primarily emphasize vocational training and skills in the adult workplace or the delivery of computer-based testing. We may also use the Licensed Marks and other aspects of the Sylvan System in conjunction with activities not contemplated by Sylvan as being conducted solely through Centers; and to offer, distribute, use and/or sell components of the Sylvan System through courses or programs offered and conducted by methods or other channels of distribution or through facilities other than Centers, within your Territory or outside your Territory, including but not limited to: (i) the Internet; (ii) CD-ROM, books, audio or video tapes; (iii) sales for home use by students; or (iv) other concepts currently in existence or as may be developed in the future, for example, through advancements in technology. Sylvan franchisees do not have the right to use other channels of distribution or methods or facilities outside of their territories.

As described in Item 1, our affiliate, SIH operates an in-person and online tutoring initiative utilizing the Marketplace Platform developed by Sylvan and SIH. As of the issuance date of this disclosure document, SIH is advertising the Marketplace Platform across the United States, as the Marketplace tutors are permitted to deliver virtual tutoring to any student who does not live within fifty (50) miles of a Sylvan Learning Center that is not participating in the Marketplace Platform offerings. SIH reserves the right to advertise the Marketplace Platform and allow affiliated tutors to deliver services inside and outside of your Territory. As of the date of this disclosure document, new licensees are required to participate in tutoring offerings by SIH and the Marketplace Platform, pursuant to the mySylvan Marketplace+ Franchisee Participation Pilot Agreement, attached as Attachment L to the License Agreement. Sylvan does not guarantee that it will continue to allow franchisees to participate beyond the pilot phase of the program, and Sylvan is not required to do so. If Sylvan elects to allow participation by franchisees beyond the pilot phase, participation will be optional and governed by a separate product agreement between Sylvan and each participating franchisee. Participation will be subject to certain fees, training, and other requirements, as determined by SIH and Sylvan at their discretion.

We and our affiliates currently operate and may open and operate Centers outside of your Territory. We have the right to license third parties to open and operate Centers outside of your Territory. Also, we may conduct and operate businesses, and offer and sell services, programs, and products, within or outside of your Territory, that do not comprise the Sylvan System and that use the Sylvan, or other, trademarks and names.

We have the right to offer products and services under the "SylvanSync®," "Sylvan Edge®" and/or "ACE IT!®" brands and marks within your Territory and to license others to do so. We have not done so.

As described above, your territorial protection and exclusivity is limited by certain rights reserved to us or other licensees, regarding the use of and/or offer of the SylvanSync System, Sylvan Edge Program or ACE IT! Program in your Territory. Therefore, you will not receive an exclusive territory regarding the offer of the SylvanSync System, Sylvan Edge Program, or ACE IT! Program products and services. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

However, our rights with respect to the SylvanSync System, the Sylvan Edge Program, and the ACE IT! Program will not affect your territorial rights and exclusivity within your Territory with respect to your operation of the Sylvan Learning Center under your License Agreement.

If you sign the optional Sylvan Edge Package License Agreement for your Territory, then, during the term of that agreement, so long as you are in material compliance with all of the terms of that agreement and of the Sylvan License Agreement, we will not use, nor license any third party to use, the EDGE Package in your Territory. We may, however, use the Licensed Marks and certain aspects of the Sylvan Edge Program in conjunction with activities not contemplated by Sylvan as being conducted solely through Centers; and to offer, distribute, use and/or sell components of the Sylvan Edge Program through courses or programs offered and conducted by methods or other channels of distribution or through facilities other than Centers, within your Territory or outside your Territory, including but not limited to: (i) the Internet; (ii) CD-ROM, books, audio or video tapes; (iii) sales for home use by students; or (iv) other concepts currently in existence or as may be developed in the future, for example, through advancements in technology. Sylvan franchisees do not have the right to use other channels of distribution or methods or facilities outside of their territories.

If you sign the optional ACE IT! Tutoring System License Agreement for your Territory, then, during the term of that agreement, so long as you are in material compliance with all of the terms of that agreement and of the Sylvan License Agreement, we will not use, nor license any third party to use, the ACE IT! System in your Territory to serve Third-Party Contracts.

You may not promote, market, offer, use, or deliver any portion of Sylvan Edge Program or the ACE IT! Program at or to any location, or to any recipient or device, outside of your Territory. While you may

accept inquiries from customers outside of your Territory, you may not sell, offer, or deliver products or services, or conduct operations of the Sylvan Edge Program or the ACE IT! Program outside of your Territory.

As noted above, we and our affiliates may offer certain products and services through other channels of distribution, including through the Internet. If we offer or sell products or services through the channels of distribution, we are not obligated to pay any fees or other compensation to franchisees if sales are made to customers in their territories.

We have not offered SylvanSync, Sylvan Edge Program or ACE IT! Program products or services under Sylvan's marks or under different trademarks or marks to customers within a franchisee's territory.

You must generate at least a certain amount of gross revenues each year in your Licensed Business and pay to Sylvan a Minimum Annual Royalty for each calendar year in order to maintain your rights under the License Agreement. For each calendar year, you are required to generate sufficient gross revenues in your Licensed Business so that the total royalties that you pay to Sylvan for each calendar year, after reductions to and refunds of gross revenues, will total at least the Minimum Annual Royalty. The Minimum Annual Royalty at the Effective Date of your License Agreement is \$32,000, and you must meet this level each year during the initial term and each renewal term of your License Agreement, beginning in the calendar year that follows your Required Open Date stated in the License Agreement. The Required Open Date is typically 180 days after the Effective Date of your License Agreement. If your Required Open Date falls in the second half of the calendar year, your Minimum Annual Royalty for your first year is reduced to \$20,000. If you fail to open a Primary Center by the original Required Open Date, and Sylvan extends the Required Open Date as an accommodation to you, Sylvan reserves the right to enforce the Minimum Annual Royalty requirement as if you had opened the Primary Center on time, according to the original requirement. The Minimum Annual Royalty may be increased for cost-of-living adjustments beginning in the sixth year of the initial term of your License Agreement. See Item 6 for how the cost-of-living adjustment is calculated.

If you do not meet the gross revenue level sufficient for your royalties to us to be at least the Minimum Annual Royalty for a given calendar year, you can cure the first two such failures during a given term of your License Agreement by paying to Sylvan the difference ("Shortfall") between your actual royalty generated and paid, and the Minimum Annual Royalty level. To cure these breaches, you must pay the Shortfall to Sylvan at the same time that your monthly royalty payment is due in the January immediately following the calendar year for which the gross revenue level did not meet the required level.

If you fail more than two times (each a "Subsequent Failure") to generate sufficient gross revenues for your royalty to meet the Minimum Annual Royalty during a given term of your License Agreement, you cannot cure the third or Subsequent Failure through payment. Instead, Sylvan may consider the third and any successive failure to be an incurable material default of your License Agreement, and Sylvan will have the right to terminate your License Agreement after giving you at least 20 days to appeal, but Sylvan has this right only during the six months beginning in the January immediately following the calendar year for which the gross revenue level did not meet the required level. If Sylvan does not terminate your License Agreement during this six-month period, then Sylvan loses the right to terminate your License Agreement for that particular Subsequent Failure, and you are not required to pay any Shortfall for that Subsequent Failure.

You have the right to appeal Sylvan's intended termination of your License Agreement for a Subsequent Failure. See the License Agreement for the full appeals process. If you appeal according to the process described in the License Agreement, the termination is stayed until Sylvan rules on your appeal in writing. To appeal, you must send to Sylvan your appeal in writing within the appeal period that Sylvan sets out to you in writing, and you must commit to paying the Shortfall within 15 days of Sylvan's ruling on your appeal. Sylvan is required to give you at least 20 days to appeal. Sylvan has the right to deny or accept your appeal. If Sylvan accepts your appeal, and you pay the Shortfall to Sylvan within 15 days of Sylvan's

ruling on your appeal, then Sylvan will rescind the termination for Subsequent Failure. If Sylvan accepts your appeal, but you do not pay the Shortfall to Sylvan within the 15-day period, then Sylvan may terminate your License Agreement immediately. If you do not appeal, Sylvan may terminate your License Agreement at the end of the appeal period.

You must exercise your best efforts to cultivate and develop and expand the market in your Territory.

During the initial term of your License Agreement, we will not modify your territorial rights, except with your agreement. However, if you are in good standing under your License Agreement, we may choose to give you the option of modifying your Territory in one of two ways. To exercise either of these options, you must pay us the fee specified in the Confidential Operations Manual. We may change the fees stated below from time to time.

#### Affiliates' Businesses

Certain products or services from our affiliates Class 101 and Snapology, whether currently existing, in research and development, or developed in the future, may be distributed in your Territory by our affiliates, or its franchisees, licensees, or designees, in the manner and through such channels of distribution as our affiliates determine at its sole discretion without compensation to you. Class 101, whose principal business address is the same as Sylvan's principal business address, currently offers franchises for CLASS 101 franchised businesses, which provide advice, guidance and training to high school students and their parents in preparing for, selecting, applying to, and paying for college; Class 101 may also provide certain ACT, SAT, and other testing preparation services under the CLASS 101 name. Snapology, whose principal business address is the same as Sylvan's principal business address, currently offers franchises for SNAPOLOGY franchised businesses, which provide curriculum-based courses, events and hands-on learning experiences using LEGO® brand bricks, K'Nex® brand toys, and other building toys, robotics, animation, coding and engineering techniques; Snapology may also provide certain STEM classes and camps under the SNAPOLOGY name. Class 101 and Snapology may also offer other services and products that may overlap with Sylvan's services and products. While the principal business addresses for Snapology, Class 101, and Sylvan are the same, Class 101 and Snapology each maintain training facilities that are physically separate from Sylvan. Snapology and Class 101 franchisees may solicit or accept customers near your Licensed Business and in your Territory. We do not expect any material conflicts between Sylvan franchisees and Snapology and Class 101 franchisees regarding territory, customers, or support because the principal products and services offered by each do not materially overlap. However, we intend to use reasonable efforts to resolve any conflicts that might arise in the future.

#### **Territory Expansion**

The first option for modifying your Territory is to expand it. If we permit you to expand your Territory, you must agree to open and keep open at all times at least two Centers in your Territory within 180 days of the date of executing the Amendment. We currently charge \$10,000 for territory expansions. However, this fee may be modified from time to time, and is specified in the Confidential Operations Manual. Upon our approval, we will terminate your current License Agreement and you must sign a then-current form of Sylvan License Agreement to take advantage of this expansion option with the newly awarded territory.

#### **Territory Split**

We also offer a second option to modify your Territory. If you are in good standing under your License Agreement and the demographics of your Territory will support two Centers in accordance with our thencurrent criteria for each Center, you may split your Territory and operate two Centers, one within your original territory and the other within the newly created territory. Our approval to subdivide your Territory is subject to conditions, which may include successful completion of training by a new manager (if we require you to have a second manager), your compliance under all license or franchise agreements and other agreements with us and our affiliates, and you must sign our then-current form of Sylvan License Agreement to create the new territory and open the second Center, or create the new territory, commence

operations therein, and sell it. The fee to exercise this second option is specified in the Confidential Operations Manual and may change from time to time. We currently charge \$5,000 to split a territory.

In certain transfers, we may require that the Territory be split into two or more smaller protected areas (each a "<u>Fractional Area</u>"), each of which meet the then-current standards for a Territory. The Fractional Area that contains the Primary Center shall serve as the Territory in the transferee's franchise agreement and will be the only territory transferred under the purchase agreement between you and transferee. We may offer the remaining Fractional Areas to transferee or you under a separate then-current franchise agreement if you or transferee meet our then-current standards for multi-unit development. If the transferee or you undertake development of the remaining Fractional Areas, you or the transferee will pay the Territory Split Fee for each Fractional Area in lieu of an initial franchise fee.

If you expand your Territory, we subdivide your newly expanded territory into two sub-territories, each of which is required to have at least one Center. One sub-territory will contain your operating Center. You are required to open a Center in the other sub-territory within 180 days of our agreeing to allow you to expand your Territory. If you fail during that 180-day period to open a Center in the sub-territory that does not have one, or if at any time you abandon a Center (which includes closing a Center for a period of 10 or more consecutive days) without our prior written approval, and the result is that a sub-territory does not have a Center located within it, then the rights to that sub-territory may, at our sole discretion, revert to us. If this happens, your license will be terminated as to the sub-territory that has no Center, and your revised Territory will cover only the sub-territory that has at least one Center located within it. The reduction of your Territory becomes effective upon our providing you written notice and a revised description of your Territory. If you lose a sub-territory, we do not refund the fee that you must pay to expand your Territory.

Except with our consent and to the extent some forms of media advertisement cannot be restricted to your Territory, you may not place advertisements for the Sylvan System in another licensee's territory or in any territory where a Sylvan-owned Center operates. For forms of media advertisement that cannot be restricted to your Territory, you must comply with our guidelines set forth in the Confidential Operations Manual or otherwise in writing.

### Optional Programs: Sylvan Edge Program and ACE IT! Program

If you choose to participate in the Sylvan Edge Program and/or the ACE IT! Program, your Territory for those optional programs will be the same as that which is licensed to you under your License Agreement. Under both programs, you will deliver products and services at your Center and at other approved locations within your Territory in accordance with the respective license agreement for each program.

#### **ITEM 13: TRADEMARKS**

We grant you the right to operate your facility under the name Sylvan Learning Center. You also may use our other principal trademarks listed below to identify your facility and services provided at your facility. The term "principal trademarks" means primary trademarks, service marks, names, logos, and commercial symbols used to identify the franchise. You will operate your franchise under these primary trademarks and any other name or mark that we designate. You may also use our other current or future marks to identify your franchise, as we designate.

# **United States Registered Marks**

We or our predecessors have registered the following principal trademarks with the United States Patent and Trademark Office (USPTO). All trademarks are registered on the Principal Register unless otherwise indicated:

Mark	Registration Number	Registration Date
SYLVAN	1410891	September 23, 1986
SYLVAN LEARNING	3732803	December 29, 2009
Sylvan	3735313	January 5, 2010
Sylvan Learning	5329939	November 7, 2017
SYLVAN INSIGHT	4227443	October 16, 2012
SylvanSync	4467950	January 14, 2014
SYLVAN ADVANCED STUDY SKILLS	4732189	May 5, 2015
SylvanSync	4756178	June 16, 2015
SYLVAN LEARNING CENTER	5335718	November 14, 2017
SYLVAN TUTORING	4883227	January 5, 2016

Mark	Registration Number	Registration Date
Sylvan	5087116	November 22, 2016
Sylvan	5,142,208	February 14, 2017
ACE IT!	3392255	March 4, 2008
ACE IT! TUTORING	3402237	March 25, 2008
SYLVAN EDGE	5088525	November 22, 2016
Sylvan	5088526	November 22, 2016

We have filed all required affidavits and renewal documentation for all registered marks listed above.

We do not know of any presently effective determinations of the United States Patent and Trademark Office, the trademark trial and appeal board, the trademark administrator of any state or any court, any pending infringement, opposition, or cancellation proceeding or any pending material litigation involving these trademarks which is relevant to their use in any state in which your franchise is to be located.

There are no agreements currently in effect which significantly limit our rights to use or license the use of these trademarks in any manner material to your license.

You must follow our rules when using our trademarks. You cannot represent that you have acquired any ownership rights in any of our trademarks. You cannot infringe upon our trademarks. You also cannot contest or help others to contest the validity of our right to use our trademarks or take any other actions in

derogation of our use of these marks. You cannot commit any act causing an incurable tarnishment of our trademarks or the reputation or goodwill associated with the Sylvan System or the SylvanSync System.

You must monitor the use of our trademarks. You must notify us promptly of any claim that we may have as a result of any person's unauthorized attempt to use any of our trademarks. At our request and expense, you must assist us in taking appropriate action to stop these activities. You cannot, however, take any action or incur any expenses on our behalf without our prior written approval. If we defend or prosecute any claims concerning our trademarks, you must sign any documents and take any actions that our legal counsel deems necessary. We shall have the right to any damages, settlement proceeds, or benefit from a third party's infringement, challenge, or claim.

You must operate and advertise your Licensed Business only under the trademarks that we designate. You must use our trademarks only in the manner we prescribe, and you must not use our marks to perform any activity or to incur any obligation that may subject us to liability. You must observe all laws concerning the registration of trade names and assumed names and include in any application for a trade name or assumed name a statement that your use of our trademarks is limited. You must provide us with copies of these applications and other registration documents. You also must observe the requirements for our trademark registrations, including affixing "SM," "TM" or "®" adjacent to all our trademarks and using other appropriate notice of ownership and registration.

We may designate one or more modified or replacement marks for use by our licensees and we may require you to use these different marks. If this happens, you will have 120 days from receipt of notice from us to make any changes we require in signage or outdoor advertising, and you must pay all costs and expenses associated with the use of the modified or replacement marks. If we require you to change your signage or outdoor advertising more than once during the initial or any extended term of your License Agreement, we will reimburse you the reasonable costs of altering your signage or outdoor advertising.

You are not entitled to any compensation if you lose your right to continue to use our trademarks.

We do not know of any infringing uses of our trademarks which could materially affect your use of our trademarks.

#### ITEM 14: PATENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION

There are no patents or registered copyrights material to the franchise, but we claim copyright protection in many elements of the Sylvan System including the Confidential Operations Manual, the design elements of the Licensed Marks, our software, signage our advertising and promotional materials, and the content and design of our website.

You must keep our proprietary information confidential and disclose it only to your employees and others affiliated with your Center who have a need to know the information and who have signed employee nondisclosure agreements (see Item 15).

If we decide to add, modify, or discontinue the use of an item or process covered by a patent or copyright, you must also do so. This might result in additional costs to you.

You do not receive the right to use any item covered by a patent related to the Sylvan System. You may use the proprietary information in our Confidential Manual, instructional materials, SylvanSync Software, Sylvan Insight Assessment Suite, Hub software, STS/SAAS software, EdAdmin, and other software that we may make available. Sylvan has not filed an application for a copyright registration for any of these materials, however, we claim copyrights on these materials, and they are proprietary. We may charge you a fee for licensing and support of Hub, STS/SAAS, EdAdmin, and other software.

You also receive the right to use the methods and equipment utilizing processes that are the subject of the pending and issued patents that are described in the table below. We hold licenses from the patent owners to use and to license others to use each of the processes below through December 31, 2030. All applications and patents are owned by Learning System of the Future, LLC ("LSF"), an affiliate company of Sylvan,

unless another owner is identified below. LSF holds a license from the owners of the other applications and patents to use and to license others to use them. Patent-owner Laureate Education, Inc. operates businesses that concern international education in the college and post-secondary sectors.

Our right to use or license these patents is not materially limited by any agreement or known infringing use. There are no currently effective determinations of the United States Copyright Office or any court, nor any pending litigation or other proceedings, regarding any copyrighted materials or patents.

You must tell us immediately if you learn about an infringement or challenge to our use or license of any of these patents. We will take action that we think is appropriate. The License Agreement does not address patent infringement claims; however, we intend to protect the patent rights and to defend you against infringement claims arising from your use of patented items in accordance with our standards and specifications. You must also agree not to contest our interest in these or our other trade secrets.

# ITEM 15: OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

Your Center must at all times meet franchisee management, educational staffing, and certification requirements. Our Confidential Operations Manual provides the details of our Center franchisee management, educational staffing, and certification requirements. These requirements are subject to change from time to time.

#### Franchisee Management Requirement

Sylvan does not permit absentee ownership by the franchisee. To satisfy the management requirement, you must be fully involved in the operation of your Center(s). The Sylvan business model requires that the franchisee provide hands-on leadership, oversight, and support to employees who provide services to customers. If the franchisee entity is a partnership, corporation, or other business entity, at least one of the owners must be designated to fulfill the management requirement. We refer to this individual as your "Managing Principal." At all times, at least one owner shall be fully involved as the Managing Principal in whichever full-time role benefits the business most.

An individual owner/franchisee is not required to fill the role of Director, but you may do so. It is common for franchisees to hire a Director to manage the day-to-day operations of the Center so that the Managing Principal can take on business growth responsibilities, such as community involvement, fiscal management, local marketing, and awareness building of the brand. It is required that all Managing Principals complete Franchisee Certification training, and Managing Principals who do not serve as Directors should be prepared to step in to mitigate risk in the event of staff turnover in such key position. In addition, Managing Principals must be fully engaged with representatives from the Sylvan Corporate staff; this includes, without limitation, regular and ongoing communication with the Franchise Business Consultant, participation in training activities, and attendance at Town Hall calls.

Each Owner of the franchise or the franchisee entity must sign an Undertaking and Guaranty substantially in the form of <u>Attachment C</u> to the License Agreement to personally guarantee to us that you will perform all obligations under the License Agreement in a timely manner according to the respective terms of the License Agreement. Each Owner (regardless of the limitation of their ownership percentage in the Franchised Business), Managing Principal, Director, any supervisors, and other key employees having access to our Confidential Operations Manual and proprietary information must sign a Confidentiality and Non-Competition Agreement substantially in the form of <u>Attachment D</u> to the License Agreement.

"Owner" means you if you are an individual, or each individual or entity holding more than a ten percent equity interest in you if you are a business entity (regardless of voting rights), and the franchisee individual(s) or entity(ies) that enter into the License Agreement if you are a business entity. It includes all officers, directors, and shareholders of a corporation, all managers and members of a limited liability company, all general and limited partners of a limited partnership, and the grantor and the trustee of the

trust. If any Owner is a business entity, then the term "Owner" also includes the Owners of that business entity.

If an owner retains another business interest or full-time job, Sylvan may require that a Director or other key staff member in the business be added as an owner in the License Agreement to fulfill the Managing Principal role. Alternatively or in addition, Sylvan may, at its discretion, also require additional staff members to complete Franchisee or Director Certification training.

#### **Educational Staffing and Certification Requirements**

Each of your Centers must have at least one primary Center Director on an ongoing basis once you begin to offer services. The Center Director is not required to have an ownership interest in the Center. A Center with more than 100 students enrolled in mandatory programs also must have a second Director (either a Director of Education, or a second Center Director) or a lead teacher or program manager.

The primary Center Director must work full-time in the Center. The other Director(s), lead teachers, and program managers may be full-time or part-time.

You may be a Director, but you are not required to be a Director. You are required to complete the Franchisee Certification curriculum regardless of whether you plan to act as a Director.

An individual may not serve as Center Director for more than one brick-and-mortar Center location. Directors of Education may be the Director of Education in multiple Centers and Satellites.

All Directors must be Director-certified within six months of starting to work at your Center. In addition, each teacher that delivers SylvanSync must have an effective SylvanSync Certification. Sylvan will provide Director Certification to Directors who meet the following criteria:

Director of Education
State or provincial teaching certification
Demonstrates mastery of the educational
skills necessary to administer diagnostic tests
and individualized learning progressions
Satisfactorily completed Sylvan's required
training on Sylvan and SylvanSync with a
material mastery score of 80% or better
A four-year post-secondary degree is
recommended

Center Director
Demonstrates appropriate business management skills
Satisfactorily completed Sylvan's required training with a material mastery score of 80% or better

#### **Center Certification Requirements**

Your Center must have all of the following Certifications in effect from the time you begin to operate:

- 1) **Sylvan Franchisee Certification**. Sylvan will confer this Certification on a Center if the licensee (or, if the licensee is a corporation, at least one of its shareholders) has successfully completed Sylvan's Franchisee Certification curriculum.
- 2) **Sylvan Director Certification**. Sylvan will confer this Certification on a Center if licensee (or, if licensee is a corporation, at least one of its shareholders) has successfully completed Sylvan's Director Certification curriculum. If you are serving as the Center Director, then the Franchisee Certification will meet this requirement.

3) **Assessment Proctor Certification**. Sylvan will confer this Certification on a Center if all Directors and staff who deliver assessments have successfully completed Sylvan's required Assessment Training and passed the associated online assessments.

The details of each of the learning paths and Certification assessments are described in our Confidential Operations Manual. These requirements are subject to change from time to time at Sylvan's discretion.

During any period where your Licensed Business is operating without someone meeting our management or educational services certification requirements, we may enter and manage or offer educational services at your Licensed Business on an interim basis and in return for a reasonable management fee.

Sylvan may from time to time offer a Contact Center Program. Sylvan currently has in place a Contact Center Program, and Sylvan has partnered with a third-party vendor to offer contact center services to franchisees the terms of which are in Attachment F to the License Agreement. As of the issuance date of this disclosure document, new franchisees are required to participate the Contact Center Program for the first twelve months of their Primary Center's operation (from the date the Center opens for business until the first anniversary of that date). See Item 6 and Item 7 for more information concerning fees charged by Sylvan in connection with the Contact Center Program.

You must execute nondisclosure agreements with all employees and managers at your Center who have or could have access to our proprietary material to maintain the confidentiality of the proprietary information described in Item 14. We will be a third-party beneficiary of these nondisclosure agreements and must approve them in writing. You cannot modify or terminate these agreements without our prior written consent.

You, and if you are a corporation, your directors, officers, and shareholders, must not, during the term of the License Agreement, engage as an owner, operator, or manager in any educational business except your Center. For your noncompetition requirements after termination or expiration of your License Agreement, see Item 17.

The minimum staffing requirements and the hours of operation described in the License Agreement and Confidential Operations Manuals have been established exclusively for the purpose of preserving Sylvan's brand standards. You will have sole authority and discretion regarding all employment matters pertaining to your personnel, including, without limitation, hiring, firing, discipline, promotion, compensation, benefits, and scheduling. Those decisions and actions are yours to make, as the independent owner and operator of your Licensed Business, and such decisions and actions will not be, nor be deemed to be, a decision or action of ours, even if we provide any advice or recommendations on any issue.

#### ITEM 16: RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

The courses of instruction and services that you may offer in your Licensed Business are limited to (1) the educational programs we license to you pursuant to the License Agreement; and (2) any other programs that we specifically approve in writing. You are not required to offer all of the educational programs we license, but you are required to offer those indicated as mandatory in this document, in the Confidential Manual, or otherwise in writing. We have the right to change the types of programs (or other goods and services) that you are authorized to sell at the Center during the term of the License Agreement, and there are no contractual limitations on this right.

The License Agreement prohibits persons closely connected with your Licensed Business from engaging in any type of notorious behavior that would tend to bring the reputation of your Licensed Business or Sylvan's instructional techniques into disrepute. By implication, this could prohibit those persons from selling certain types of goods or services or could restrict them from publicly dealing with certain classes of businesses or individuals.

We impose no other restrictions that restrict either the goods or services you may offer or the customers to whom you may sell.

# ITEM 17: RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

	PROVISION	SECTION IN LICENSE AGREEMENT	SUMMARY
a.	Term of the License Agreement	Section 10.1	Term is 10 years.
b.	Renewal or Extension of Term	Sections 10.2 and 10.3	If you are in good standing, you may elect to continue operating the franchise for two additional, consecutive five-year successor terms. You must pay us a renewal fee of \$6,000. A month-to-month renewal is also possible, subject to 30 days' notice of termination by you or Sylvan.
c.	Requirements to Renew or Extend	Section 10.2	Comply with terms and conditions for granting renewals at that time, including notice at least 180 days before expiration, executing a general release, never failing during last three years of term to timely cure a curable default, never committing an incurable material default, executing Sylvan's then-current License Agreement (provided that if the monthly royalty fees in Sylvan's then-current License Agreement are less favorable than those in your current License Agreement, you may retain the monthly royalty fees set forth in your current License Agreement), and modernizing your Center. When renewing, you may be asked to sign a contract with materially different terms and conditions than your original contract.
d.	Termination by You	No provision	Not applicable.
e.	Termination by Sylvan— Without Cause	No provision	Not applicable.
f.	Termination by Sylvan— With Cause	Section 11.1	Sylvan may terminate only for cause following incurable material breach or failure to cure a curable default.

	PROVISION	SECTION IN LICENSE AGREEMENT	SUMMARY
g.	"Cause" Defined— Curable Defaults	Section 11.2	You have at least 10 days to cure nonpayment of financial obligations and at least 30 days to cure nonmonetary defaults that are curable. Curable defaults include noncompliance with any law or regulation, or breach of the License Agreement or Confidential Manual, including failure to pay obligations or provide reports when due, first two failures during a term to achieve gross revenues sufficient to achieve the Minimum Annual Royalty, offering unapproved programs, making unapproved guarantees of results of study programs, not providing adequately trained instructors, notorious behavior, absentee management, improper advertising, failure to open your Center on time and failure to maintain insurance.
h.	"Cause" Defined— Incurable Defaults	Section 11.3	Non-curable defaults include insolvency or bankruptcy, violation of transfer and assignment provisions, repeated breaches even if cured, violation of confidentiality provisions, criminal charge or conviction or harm to Sylvan's reputation, abandonment, maintaining books and records or submitting reports which are in Sylvan's reasonable judgment intentionally false, submitting reports which underestimate gross revenues or are otherwise substantially incorrect (and are not corrected by you before Sylvan gives you written notice), third and subsequent failures during a term to generate gross revenues sufficient to achieve the Minimum Annual Royalty (if you do not appeal or if Sylvan declines your appeal), breaching the terms of the License Agreement four or more times in a twelvemonth period, and failure to attend two out of three mandatory regional or national conferences.
i.	Your Post- Termination Obligations	Section 12	Obligations include complete de-identification with Sylvan, payment of all amounts due, return of Confidential Manual and other property of Sylvan, transfer of all telephone numbers, email addresses, social media accounts, Internet domain names and the like, and advertisements, removal of all signs indicating a connection with Sylvan, settlement of outstanding contracts with customers, and may, in the event of breach, require payment of liquidated damages (see Item 6).
j.	Assignment of Contract by Sylvan	Section 9.1	Sylvan may transfer its interest to any party that agrees to assume its obligations under the License Agreement.

	PROVISION	SECTION IN LICENSE AGREEMENT	SUMMARY
k.	Transfer by You	Section 9.2	Includes sale, assignment, transfer, conveyance, give away, pledge, mortgage, or other encumbrance of any interest in your License Agreement or that reduces the ownership share of your initial principals to less than 51%.
1.	Sylvan's Approval of Transfers	Section 9.3	Sylvan must give its prior written approval of all transfers. Sylvan will not unreasonably withhold its approval.
m.	Conditions for Sylvan's Approval of Transfer	Section 9.3	Your obligations to Sylvan fully satisfied, current disclosure document provided to and executed by new franchisee, transfer fee of \$6,000 (\$3,000 if transfer is to existing franchisee) paid, general release signed, new franchisee travels to Sylvan headquarters and qualifies, transferee's obligations to Sylvan fully satisfied, quality review of Center by Sylvan, if required, and any deficiencies corrected, new franchisee completes training and, if required by Sylvan, executes a new License Agreement (also see I above). If you elect to participate in our resale assistance program, then you must also comply with our then-current resale program requirements, execute a then-current resale program agreement, and pay the then-current resale program fee.
n.	Sylvan's Right of First Refusal to Acquire Business	Section 9.6	Sylvan has right of first refusal to purchase any interest in the sale or transfer of your License Agreement on the same or better terms as those offered by a third party. This right of first refusal does not apply to a transfer to (i) any existing holder of a minority interest in licensee which results in the transferee continuing to hold a minority interest in licensee or (ii) an heir of the first degree of licensee or the holder of a majority interest in licensee (to be exercised only once).
0.	Sylvan's Option to Purchase Business	Not applicable	
p.	Your Death or Disability	Section 9.5	Upon death or incapacity of anyone upon whose skill and experience Sylvan relied in granting the franchise, Sylvan may assist you or your heirs in providing a substitute to assume control of your Center. Sylvan may manage your Center on interim basis for compensation. Your heirs may transfer License Agreement on same terms you could.

	PROVISION	SECTION IN LICENSE AGREEMENT	SUMMARY
q.	Non-Competition Covenants— During Term	Section 12.5.1	No involvement in any educational business by you or, if you are a corporation, by your shareholders.
r.	Non-Competition Covenants—Post- Termination	Section 12.5.2	No involvement in competing educational business for two years in your Territory or within 50 miles of it, or within 25 miles of any other Center, by you or, if you are a corporation, by your shareholders.
S.	Modification of Agreement	Sections 1.8.1, 1.8.2, 1.8.3, 2.1.1, 7.9.3, 8.1.1, 25, and 26.10	No amendments or modifications to License Agreement except by mutual writing. Sylvan standards, policies, and Confidential Manual subject to change.
t.	Integration/ Merger	Section 30.9	Only the terms of the License Agreement are binding (subject to state law). Any representations or promises made outside of the disclosure document and License Agreement may not be enforceable.
u.	Dispute Resolution	Sections 7.9.4, 23	Except for certain claims, all disputes must arbitrated in Texas unless contrary to applicable state law.
v.	Choice of Forum	Section 21	Under the License Agreement, if you sue Sylvan, litigation must be in Texas, and if Sylvan sues you, litigation must be in the state of your Territory. This provision is subject to state law.
w.	Choice of Law	Section 21	Under the License Agreement, Texas law applies. This provision is subject to state law.

Our right to terminate your License Agreement in the event of your bankruptcy may be restricted by federal bankruptcy law (11 U.S.C. Section 101 et seq.).

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within three years after the franchise related to the claim is sold.

#### **ITEM 18: PUBLIC FIGURES**

Sylvan does not use any public figures to promote our franchise.

#### ITEM 19: FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

With respect to the Sylvan System, as of December 31, 2023, Sylvan had 421 licensed territories in the United States, including a total of 459 Centers, which provided revenue reports to us. Of these, 307 licensed territories operated continuously during the 24 calendar months of 2022 and 2023 under the same or similar form of joint Sylvan and SylvanSync License Agreement which is referenced by and included in this disclosure document. This Item 19 does not include revenue from any licensed territory which operated during 2022 or 2023 pursuant to an earlier form of License Agreement which did not include the SylvanSync System offering; all 307 territories that are the subject of this Item 19 used SylvanSync in 2023. This Item 19 also does not include revenue from any territories that closed or opened during 2022 or 2023, and therefore were not open for the full 24 months of 2022 and 2023. See Item 20 for openings, closings, and other unit changes during each of 2021, 2022, and 2023.

The Table below sets forth certain historic performance information for United States licensed territories that operated continuously throughout the 24 calendar months of 2022 and 2023 pursuant to a joint Sylvan and SylvanSync License Agreement. All such licensed territories are included in the following calculations. The term "licensee" as used in this Item 19 refers to a person or entity that entered into a joint Sylvan and SylvanSync License Agreement. The financial performance of company-owned territories is not included. The figures included in this Item 19 have not been audited.

#### Table: 2023 Sylvan Territory Gross Revenues by Quartile

This Table reflects the total gross revenues from the Sylvan businesses in the 307 reporting territories for 2023. The Table divides the group into quartiles, based upon gross revenue by territory. In addition, the Table reflects the gross revenues for each of the top 10 territories within each quartile, as well as the bottom 10 territories within each quartile.

Top 25% of Territories by Gross Revenue	Mid-Upper 25% of Territories by Gross Revenue	Mid-Lower 25% of Territories by Gross Revenue	Bottom 25% of Territories by Gross Revenue
	Category Avera	ge Gross Revenues	
\$815,079 1	\$432,048 2	\$293,283 3	\$169,133 4
	Category Median	Gross Revenues	
\$688,803 5	\$433,156 <sup>6</sup>	\$294,502 <sup>7</sup>	\$179,103 8
(	Gross Receipts of the Top 10	Territories in Each Quarti	le
\$2,286,964	\$523,730	\$339,622	\$240,172
\$1,609,142	\$522,473	\$338,812	\$238,881
\$1,598,190	\$515,234	\$336,716	\$238,777
\$1,491,633	\$512,973	\$336,585	\$237,653
\$1,483,158	\$511,493	\$336,553	\$236,509
\$1,439,560	\$511,152	\$333,031	\$236,411
\$1,393,252	\$511,071	\$331,516	\$234,926

Top 25% of Territories by Gross Revenue	Mid-Upper 25% of Territories by Gross Revenue	Mid-Lower 25% of Territories by Gross Revenue	Bottom 25% of Territories by Gross Revenue
\$1,383,225	\$508,481	\$329,172	\$233,694
\$1,344,690	\$506,097	\$329,066	\$225,008
\$1,227,392	\$505,256	\$327,326	\$224,877
G	Fross Receipts of the Botton	n 10 Territories in Each Qua	artile
\$545,006	\$367,505	\$256,838	\$114,566
\$543,258	\$364,885	\$253,245	\$113,326
\$541,544	\$363,940	\$251,231	\$111,405
\$541,458	\$362,043	\$248,146	\$90,212
\$541,337	\$361,375	\$246,754	\$90,002
\$538,999	\$357,993	\$245,378	\$82,722
\$536,997	\$347,019	\$243,548	\$54,516
\$532,641	\$344,397	\$243,316	\$49,942
\$528,215	\$341,833	\$242,759	\$42,721
\$527,644	\$341,766	\$240,325	\$40,616

#### Notes to Table:

- 1. Of the 77 territories in this group, 24 territories, or 31% of this group, met or exceeded the group's average gross revenue of \$815,079.
- 2. Of the 76 territories in this group, 39 territories, or 51% of this group, met or exceeded the group's average gross revenue of \$432,048.
- 3. Of the 77 territories in this group, 41 territories, or 53% of this group, met or exceeded the group's average gross revenue of \$293,283.
- 4. Of the 77 territories in this group, 42 territories, or 55% of this group, met or exceeded the group's average gross revenue of \$169,133.
- 5. Of the 77 territories in this group, the median gross revenue was \$688,803, and the highest and lowest numbers in the range were \$2,286,964 and \$527,644, respectively.
- 6. Of the 76 territories in this group, the median gross revenue was \$433,156, and the highest and lowest numbers in the range were \$523,730 and \$341,766, respectively.
- 7. Of the 77 territories in this group, the median gross revenue was \$294,502, and the highest and lowest numbers in the range were \$339,622 and \$240,325, respectively.
- 8. Of the 77 territories in this group, the median gross revenue was \$179,103, and the highest and lowest numbers in the range were \$240,172 and \$40,616, respectively.
- 9. Gross revenue numbers in this Table include Sylvan System and SylvanSync System gross revenues, as well as Sylvan Edge gross revenues and ACE IT! gross revenues, if any, for the given territory. Not all franchisees delivered the Sylvan Edge Program, which is optional. System-wide Sylvan Edge Program gross revenues in 2023, as reported by franchisees, were \$993,984. Sylvan offered the ACE IT! Program as a separate franchise opportunity from January 2006 until January 2016; Sylvan began to offer the ACE IT! Program as an optional product license in 2016. Not all franchisees delivered the ACE IT! Program, which is optional. System-wide ACE IT! Program gross revenues in 2023, as reported by franchisees, were \$5,632,957.

- 10. Of the 307 total territories included, 91% operated one Center within the territory, 8% operated two Centers, and 1% operated more than two Centers within the territory.
- 11. Some licensed territories operated Satellites in the territory as part of the Licensed Business during this period. Satellites are optional. As described in Item 1, a "Satellite" may operate from one or more additional short-term or permanent locations in its territory, such as community centers, libraries, schools, and short-term and/or small retail locations. Satellites typically do not offer the full range of Sylvan programs. Generally, licensed territories which operate one or more Satellites generate more revenue than those which operate only one Center. You are not required to operate any Satellite in your Territory.

# Some Centers have earned this amount. Your individual results may differ. There is no assurance that you'll earn as much.

Sylvan offered substantially the same services to all of the 307 licensed territories included in the foregoing tables and substantially all of the licensed territories offered the same instructional services in reading, math, study skills, writing, and SAT/ACT test prep.

Some licensees open and operate more than one Center in their territory. Some licensees open and operate one or more Satellites in their territory.

You are urged to speak to existing licensees about their individual experiences. A list of our existing licensees can be found in Exhibit H to this disclosure document.

We do not require our licensees to report fixed and variable expenses, and accordingly, we cannot estimate the average "break-even" sales volume of the reporting territories. We expect that our licensees' fixed and variable expenses vary widely from territory to territory. As a result, even if your gross revenues are similar to the 2023 average or median revenues reported by our licensees, there is no guarantee that your Territory will be profitable. The data received from reporting territories was accumulated using a uniform method that included direct reports to us by licensees. We have not independently verified the figures given by licensees. We do not require our licensees to utilize a uniform accounting method and therefore cannot confirm whether their revenue figures were compiled in accordance with generally accepted accounting principles.

Written substantiation for the financial performance representations in this Item will be made available to you upon your reasonable request; however, we will not disclose the identity of any specific licensee or the location of the territories in any category.

We do not make any representations about the future performance of any franchisee's territory or any company-owned territory.

You are urged to consult with appropriate financial, business, and legal advisors and existing Sylvan franchisees in connection with the use of any of the information contained in this Item 19.

Other than the preceding financial performance representation in this Item 19, Sylvan does not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an operating business in an existing territory, however, the selling franchisee and/or we may provide you with the actual records of that territory. If you are buying an operating business in an existing territory from a selling franchisee and the selling franchisee makes financial performance representations, such financial performance representations are those of the selling franchisee and not of Sylvan. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Josh Wall, Chief Growth Officer, Sylvan Learning, LLC, 52350 Airport Freeway, Suite 505, Bedford, TX 76022, the Federal Trade Commission, and the appropriate state regulatory agencies.

# ITEM 20: OUTLETS AND FRANCHISEE INFORMATION

# <u>Table No. 1</u> Systemwide Outlet Summary For Years 2021 to 2023

Outlet Type	Year	Outlets Operating at the Start of the Year	Outlets Operating at the End of the Year	Net Change
	2021	466	474	+8
Franchise	2022	474	468	-6
	2023	468	474	+6
	2021	6	6	0
Corporate	2022	6	5	-1
	2023	5	5	0
	2021	472	480	+8
Total	2022	480	473	-7
	2023	473	479	+6

Table No. 2
Transfers of Outlets from Franchisees to New Owners (Other than the Franchisor)
For Years 2021 to 2023

State	Year	Number of Transfers
	2021	0
Alabama	2022	0
	2023	4
	2021	0
Arizona	2022	0
	2023	1
	2021	5
California	2022	3
	2023	5
	2021	0
Colorado	2022	2
	2023	1
	2021	3
Florida	2022	1
	2023	6

State	Year	Number of Transfers
	2021	1
Georgia	2022	0
	2023	0
	2021	0
Illinois	2022	0
	2023	2
	2021	5
Kansas	2022	0
	2023	0
	2021	0
Massachusetts	2022	0
	2023	2
	2021	4
Michigan	2022	0
	2023	4
	2021	1
Mississippi	2022	0
	2023	0
	2021	0
New Hampshire	2022	0
	2023	2
	2021	0
New Jersey	2022	1
	2023	0
	2021	0
New Mexico	2022	0
	2023	2
	2021	0
New York	2022	0
	2023	2
	2021	0
North Carolina	2022	1
	2023	1

State	Year	Number of Transfers
	2021	0
Ohio	2022	10
	2023	1
	2021	0
Oregon	2022	0
	2023	1
	2021	1
Pennsylvania	2022	2
	2023	0
	2021	6
Texas	2022	8
	2023	6
	2021	0
Utah	2022	5
	2023	0
	2021	0
Virginia	2022	0
	2023	1
	2021	0
Washington	2022	2
	2023	1
	2021	26
Total	2022	35
	2023	42

Table No. 3
Status of Franchised Outlets
For Years 2021 to 2023

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2021	7	0	0	0	0	1	6
Alabama	2022	6	0	0	0	0	1	5
	2023	5	1	0	0	0	0	6
Alaska	2021	2	0	0	0	0	0	2

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	7	1	0	0	0	0	8
Arizona	2022	8	0	0	0	0	0	8
	2023	8	0	0	0	0	0	8
	2021	3	0	0	0	0	0	3
Arkansas	2022	3	0	0	0	0	1	2
	2023	2	0	0	0	0	0	2
	2021	55	4	0	0	0	1	58
California	2022	58	2	0	0	0	1	59
	2023	59	0	0	0	0	1	58
	2021	3	0	0	0	0	0	3
Colorado	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
	2021	4	0	0	0	0	0	4
Connecticut	2022	4	0	0	0	0	1	3
	2023	3	0	0	0	0	0	3
	2021	2	1	0	0	0	0	3
Delaware	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
	2021	34	2	0	0	0	4	32
Florida	2022	32	3	0	0	0	2	33
	2023	33	2	0	0	0	0	35
	2021	7	0	0	0	0	0	7
Georgia	2022	7	0	0	0	0	0	7
	2023	7	1	0	0	0	1	7
	2021	2	0	0	0	0	0	2
Hawaii	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year	
	2021	1	0	0	0	0	0	1	
Idaho	2022	1	0	0	0	0	0	1	
	2023	1	1	0	0	0	0	2	
	2021	20	1	0	0	0	0	21	
Illinois	2022	21	0	0	0	0	2	19	
	2023	19	0	0	0	0	2	17	
	2021	16	0	0	0	0	1	15	
Indiana	2022	15	0	0	0	0	0	15	
	2023	15	0	0	0	0	0	15	
	2021	4	0	0	0	0	0	4	
Iowa	2022	4	0	0	0	0	0	4	
	2023	4	1	0	0	0	0	5	
	2021	6	0	0	0	0	0	6	
Kansas	2022	6	0	0	0	0	0	6	
	2023	6	0	0	0	0	0	6	
	2021	4	0	0	0	0	1	3	
Kentucky	2022	3	0	0	0	0	0	3	
	2023	3	0	0	0	0	0	3	
	2021	13	0	0	0	0	2	11	
Louisiana	2022	11	0	0	0	0	2	9	
<b>-</b>	2023	9	0	0	0	0	0	9	
	2021	0	1	0	0	0	0	1	
Maryland	2022	1	0	0	0	0	0	1	
	2023	1	0	0	0	0	0	1	
	2021	5	0	0	0	0	1	4	
Massachusetts	2022	4	0	0	0	0	0	4	
<b> </b>	2023	4	0	0	0	0	0	4	
	2021	20	0	0	0	0	0	20	
Michigan	2022	20	0	0	0	0	0	20	
<b> </b>	2023	20	0	0	0	0	1	19	
) (°	2021	11	0	0	0	0	0	11	
Minnesota	2022	11	2	0	0	0	0	13	

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2023	13	2	0	0	0	1	14
	2021	1	0	0	0	0	0	1
Mississippi	2022	1	1	0	0	0	0	2
	2023	2	1	0	0	0	0	3
	2021	9	2	0	0	0	1	10
Missouri	2022	10	0	0	0	0	1	9
	2023	9	0	0	0	0	0	9
	2021	3	0	0	0	0	0	3
Montana	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
	2021	3	0	0	0	0	0	3
Nebraska	2022	3	1	0	0	0	0	4
	2023	4	0	0	0	0	0	4
	2021	2	0	0	0	0	0	2
Nevada	2022	2	2	0	0	0	0	4
	2023	4	0	0	0	0	0	4
	2021	3	0	0	0	0	0	3
New Hampshire	2022	3	0	0	0	0	0	3
титрытс	2023	3	0	0	0	0	0	3
	2021	9	2	0	0	0	0	11
New Jersey	2022	11	0	0	0	0	1	10
	2023	10	1	0	0	0	0	11
	2021	3	0	0	0	0	0	3
New Mexico	2022	3	1	0	0	0	0	4
	2023	17	1	0	0	0	2	16
	2021	16	0	0	0	0	1	15
New York	2022	15	1	0	0	0	2	14
	2023	14	1	0	0	0	2	13
North	2021	25	0	0	0	0	0	25
Carolina	2022	25	2	0	0	0	0	27
	2023	27	1	0	0	0	0	28

State	Year	Outlets at Start of Year	Outlets Opened	pened		Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2021	2	0	0	0	0	0	2
North Dakota	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
North Dakota 2021  2021 2023 2021 Ohio 2022		17	0	0	0	0	0	17
Ohio	2022	17	0	0	0	0	0	17
	2023	17	0	0	0	0	0	17
	2021	5	0	0	0	0	0	5
Oklahoma	2022	5	0	0	0	0	1	4
	2023	4	0	0	0	0	0	4
	2021 5	0	0	0	0	0	5	
Oregon	2022	5	0	0	0	0	1	4
	2023	4	1	0	0	0	0	5
	2021	10	1	0	0	0	0	11
Pennsylvania	2022	11	0	0	0	0	1	10
	2023	10	0	0	0	0	0	10
	2021	2	0	0	0	0	0	2
Rhode Island	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	7	1	0	0	0	0	8
South	2022	8	0	0	0	0	0	8
Carolina	2023	8	3	0	0	0	0	11
	2021	2	0	0	0	0	0	2
South Dakota	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	7	0	0	0	0	0	7
Tennessee	2022	7	0	0	0	0	0	7
	2023	7	0	0	0	0	1	6
	2021	58	6	0	0	0	0	64
Texas	2022	64	3	0	0	0	2	65
	2023	65	2	0	0	0	1	66
	2021	7	0	0	0	0	0	7
Utah	2022	7	0	0	0	0	1	6

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2023	6	0	0	0	0	0	6
	2021	1	0	0	0	0	0	1
Vermont	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	14	1	0	0	0	2	13
Virginia	2022	13	0	0	0	0	1	12
	2023	12	0	0	0	0	1	11
	2021	24	0	0	0	0	0	24
Washington	2022	24	0	0	0	0	3	21
	2023	21	0	0	0	0	0	21
	2021	0	0	0	0	0	0	0
West Virginia	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
	2021	4	0	0	0	0	0	4
Wisconsin	2022	4	0	0	0	0	0	4
	2023	4	0	0	0	0	0	4
	2021	1	0	0	0	0	0	1
Wyoming	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	1	0
	2021	466	23	0	0	0	15	474
Total	2022	474	18	0	0	0	24	468
	2023	468	18	0	0	0	12	474

<sup>\*</sup> If multiple events occurred affecting an outlet, this table shows the event that occurred last in time.

Table No. 4
Status of Company-Owned Outlets
For Years 2021 to 2023

State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired from Franchisees	Outlets Closed	Outlets Sold to Franchisees	Outlets at End of the Year
	2021	5	0	0	0	0	5
Maryland	2022	5	0	0	0	0	5
	2023	5	0	0	0	0	5

State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired from Franchisees	Outlets Closed	Outlets Sold to Franchisees	Outlets at End of the Year
	2021	1	0	0	0	0	1
New York	2022	1	0	0	1	0	0
	2023	0	0	0	0	0	0
	2021	6	0	0	0	0	6
Total	2022	6	0	0	1	0	5
	2023	5	0	0	0	0	5

<u>Table No. 5</u> Projected Openings as of December 31, 2023

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlets In the Next Fiscal Year	Projected New Company-Owned Outlets in the Next Fiscal Year
California	7	6	0
Delaware	1	1	0
Florida	3	2	0
Georgia	1	1	0
Idaho	2	0	0
Iowa	3	1	0
Maine	1	1	0
New Jersey	4	3	0
New York	1	1	0
North Carolina	1	1	0
Ohio	1	1	0
Pennsylvania	1	1	0
Tennessee	1	1	0
Texas	9	7	0
West Virginia	1	0	0
Total	37	26	0

Exhibit H lists the name, address, telephone number and email address of all current Sylvan franchisees, including those who have signed franchise agreements but are not yet open, for each of their outlets as of February 15, 2024. Exhibit H also lists the name, city, state and current business telephone number (or if unknown, the last known home telephone number) of every franchisee who had an outlet terminated, canceled, not renewed, or otherwise voluntarily or involuntarily ceased to do business during the most recently completed fiscal year or who has not communicated with Sylvan within 10 weeks of December

31, 2023. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

In some instances, current and former franchisees sign provisions restricting their ability to speak openly about their experiences, financings, and other matters with Sylvan. You may wish to speak with current and former franchisees, but be aware that not all such franchisees will be able to communicate fully with you.

Current franchisees may receive a referral bonus for referring a potential franchisee candidate to Sylvan; such bonuses, if awarded at all, are granted regardless of whether the referred potential candidate purchases a franchise and becomes a Sylvan franchisee. In addition, current franchisees may participate in conference calls with prospective franchisee candidates to discuss the opportunity of purchasing a Sylvan franchise; such conference calls may be set up by Sylvan, though representatives from Sylvan do not participate in such conference calls. If a current franchisee participates in such a discussion, he or she may be compensated nominally for his or her time regardless of whether any of the prospective franchisee candidates with whom he or she speaks purchases a Sylvan franchise. The opportunity to participate in such conference calls would be made available to the entire Sylvan System if at all. Franchisees who receive financial incentives to refer franchise prospects to Sylvan may be required to register as franchise brokers under the laws of Washington State.

Sylvan endorses the following trademark-specific franchisee organization associated with the franchise being offered you:

The Sylvan Franchise Owners Association, Inc. <a href="https://www.sylvanfoa.com">www.sylvanfoa.com</a> (517) 897-9368

#### **ITEM 21: FINANCIAL STATEMENTS**

Exhibit G to this disclosure document contains the audited consolidated balance sheets of Sylvan Learning, LLC as of December 31 of each of 2021, 2022, and 2023, and the related consolidated statements of income, stockholder's equity, and cash flows for each of the three years in the period ended December 31, 2023.

#### **ITEM 22: CONTRACTS**

Attached to this disclosure document are the following contracts:

Exhibit A	License Agreement for Sylvan and SylvanSync, Attachments, and State-Specific Amendments
Exhibit B	Sylvan Edge Package License Agreement
Exhibit C	ACE IT! Tutoring System License Agreement
Exhibit D	Multi-Territory Incentive Plan Amendment
Exhibit I	Sample Form of General Release

#### **ITEM 23: RECEIPTS**

The last two pages of this disclosure document, Exhibit L, are detachable duplicate Receipts. Please sign and date both copies of the Receipt. Keep one signed copy of the Receipt for your file and return to us the other signed copy of the Receipt. The Receipt page also contains the names, addresses and telephone numbers of our franchise sellers or brokers.

[THE DISCLOSURE DOCUMENT ENDS HERE.]

## STATE SPECIFIC APPENDIX TO FRANCHISE DISCLOSURE DOCUMENT

#### NO WAIVER OR DISCLAIMER OF RELIANCE IN CERTAIN STATES

The following provision applies only to franchisees and franchises that are subject to the state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin:

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

# INFORMATION REQUIRED BY THE STATE OF CALIFORNIA

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.

Neither SYLVAN nor any person listed in Item 2 of the Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a et seq., suspending or expelling such persons from membership in such association or exchange.

The License Agreement (the term SYLVAN uses for its franchise agreement) provides for termination upon bankruptcy. This provision may not be enforceable under Federal Bankruptcy Law (11 U.S.C.A. Sec. 101 et seq.).

The License Agreement contains a covenant not to compete which extends beyond the termination of the franchise. This provision may not be enforceable under California law (see the California Rider to the License Agreement).

The License Agreement provides that the laws of Texas shall govern the License Agreement. This provision may not be enforceable under California law.

Our website has not been reviewed or approved by the California Department of Financial Protection and Innovation. Any complaints concerning the content of this website may be directed to the California Department of Financial Protection and Innovation at <a href="https://www.dfpi.ca.gov">www.dfpi.ca.gov</a>.

California Business and Professions Code sections 20000 through 20043 (the Franchise Relations Act) provide rights to the franchisee concerning termination, transfer, or non-renewal of a franchise. If the franchise agreement contains a provision that is inconsistent with the law, the law will control.

In recognition of the requirements of the California Franchise Investment Law, the disclosure document for Sylvan for use in the State of California shall be amended as follows:

1. Item 6 and Item 10, Additional Disclosure:

The highest interest rate allowed by law in California is 10% annually.

2. Exhibit J Additional Disclosure. The following provision shall be added to the end of Exhibit J, Franchise Disclosure Questionnaire:

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

#### INFORMATION REQUIRED BY THE STATE OF HAWAII

SYLVAN HAS FILED THESE FRANCHISES UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. SYLVAN'S FILING DOES NOT CONSTITUTE THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS' APPROVAL, RECOMMENDATION OR ENDORSEMENT OR A FINDING BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS THAT THE INFORMATION PROVIDED IN THE DISCLOSURE DOCUMENT IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN HAWAII WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE, OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE, OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE DISCLOSURE DOCUMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS DISCLOSURE DOCUMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. YOU SHOULD REFER TO THE CONTRACT OR AGREEMENT FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH SYLVAN AND YOU.

#### INFORMATION REQUIRED BY THE STATE OF ILLINOIS

Notwithstanding anything to the contrary set forth in the Franchise Disclosure Document, the following provisions shall supersede and apply to all franchises offered and sold in the State of Illinois.

1. Item 17 of the Franchise Disclosure Document is amended by the addition of the following language at the beginning thereof:

# "Notice Required By Law

THE TERMS AND CONDITIONS UNDER WHICH YOUR FRANCHISE CAN BE TERMINATED AND YOUR RIGHTS UPON NON-RENEWAL MAY BE AFFECTED BY ILLINOIS LAW, 815 ILCS 705-19-705/20."

- 2. The provisions of the Illinois Franchise Disclosure Act of 1987 ("the Act") shall supersede any provisions of the License Agreement or Texas law which are in conflict with the Act.
- 3. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a license agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a license agreement may provide for arbitration to take place outside of Illinois.
- 4. Your rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act of 1987.
- 5. The provisions of Section 27 of the Act supersede the provisions of Section 29 of the License Agreement that set a limitation within one (1) year following the effective date of expiration or termination, to the extent that claims are brought under Section 26 of the Act.
- 6. In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

# INFORMATION REQUIRED BY THE STATE OF MARYLAND

- 1. The Maryland Rider to your License Agreement amends Section 9.3.9 of your License Agreement. Section 9.3.9 concerns our requirement that, in the event you transfer your licensed business to another person, you and that person execute a general release in our favor. If Maryland law applies to your License Agreement and the relationship between you and SYLVAN, then a release of this kind does not operate to cause you to waive any right you may have under the Maryland Franchise Registration and Disclosure Law.
- 2. Item 17.c ("Requirements for franchisee to renew or extend") and Item 17.m ("Conditions for franchisor approval of transfer") of the Franchise Disclosure Document are amended to provide that "The requirement that you provide a general release of all claims against us in order to renew or transfer your franchise shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law."
- 3. Item 17.g ("Cause defined curable defaults") of the Franchise Disclosure Document is amended to provide that "Termination upon filing of a bankruptcy petition against you or any shareholder may not be enforceable under federal bankruptcy law."
- 4. Item 17.v ("Choice of forum") in the Franchise Disclosure Document is amended to provide that "Franchisee may sue in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law."
- 5. Nothing in the Franchise Disclosure Document or in the License Agreement, or the Receipt of Franchise Related Documents is intended to nor shall act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Act.
- 6. Any claim arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

#### INFORMATION REQUIRED BY THE STATE OF MINNESOTA

Notwithstanding anything to the contrary set forth in the Franchise Disclosure Document and/or Franchise Agreement, as applicable, the following provisions shall supersede and apply to all franchises offered and sold in the State of Minnesota:

1. The Minnesota cover page is amended to add the following statements:

"THESE FRANCHISES HAVE BEEN REGISTERED UNDER THE MINNESOTA FRANCHISE ACT. REGISTRATION DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE COMMISSIONER OF COMMERCE OF MINNESOTA OR A FINDING BY THE COMMISSIONER THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE MINNESOTA FRANCHISE ACT MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WHICH IS SUBJECT TO REGISTRATION WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, AT LEAST 7 DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST 7 DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION, BY THE FRANCHISEE, WHICHEVER OCCURS FIRST, A COPY OF THIS PUBLIC OFFERING STATEMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE FRANCHISE. THIS PUBLIC OFFERING STATEMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE LICENSE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR AN UNDERSTANDING OF ALL RIGHTS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE."

- 2. Minnesota Rule 2860.4400D (the "Rule") provides that it is unfair or inequitable to require a franchisee to assent to a release, assignment, novation or waiver that would relieve any person from liability under Minnesota Statutes. To the extent of a conflict between any provision of the Franchise Agreement and the Rule, the Rule shall prevail.
- 3. Minnesota law provides franchisees with certain termination and non-renewal rights. It requires, except in certain specified cases, that a franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the franchise agreement (Minn. Stat. Sec. 80C.14, Subd. 3,4 and 5). Please see Section 1 of the Minnesota Rider to the License Agreement.
- 4. The Minnesota Rider to the License Agreement provides that, to the extent Minnesota law applies, SYLVAN will indemnify you for any damages for which you are held liable due to your use of our "Sylvan System" mark, if you have used the mark properly and notified us within 10 days of your knowledge of a claim against you. SYLVAN must have sole control over any litigation concerning our trademarks. This explanation supplements the information in Item 13 of the Disclosure Document.
- As provided in Section 4 of the Minnesota Rider to the License Agreement, Section 13 of the agreement regarding remedies, which provides that SYLVAN may collect liquidated damages in certain circumstances, is deleted in its entirety in accordance with Minnesota Rule 2860.4400J.

Therefore, the corresponding portions of Item 6 titled "Liquidated Damages for Breach" and paragraph (i) of Item 17 in this Disclosure Document do not apply.

- 6. As provided by Section 5 of the Minnesota Rider to the License Agreement, to the extent Minnesota law applies, Section 20 of the agreement, which states that you must sue in Texas, does not affect your rights under the Minnesota Franchise Law (Minnesota Statutes 1992, Chapter 80C), including any rights to bring suit against SYLVAN in Minnesota.
- 7. No release language set forth in the Franchise Agreement shall relieve Franchisor or any other person, directly or indirectly, from liability imposed by the laws concerning franchising of the State of Minnesota.
- 8. Minn. Rule 2860.4400J prohibits waiver of a jury trial. Accordingly, Item 17 of the Franchise Disclosure Document and Section 28 of the License Agreement are amended as follows:
  - "Nothing contained herein shall limited Franchisee's right to submit matters to the jurisdiction of the courts of Minnesota to the full extent required by Minn. Rule 2860.4400J."
- 9. Item 17 of the Franchise Disclosure Document and Section 21 of the License Agreement are amended as follows:
  - "Nothing contained herein shall limited Franchisee's right to submit matters to the jurisdiction of the courts of Minnesota to the full extent required by Minn. Rule 2860.4407J."
- 10. These states have statutes which limit the franchisor's ability to restrict your activity after the License Agreement has ended: California Business and Professions Code Section 16,600, Florida Statutes Section 542.33, Michigan Compiled Laws Section 445.771 *et seq.*, Montana Codes Section 30-14-201, North Dakota Century Code Section 9-08-06, Oklahoma Statutes Section 15-217-19, Washington Code Section 19.86.030. Other states have court decisions limiting the franchisor's ability to restrict your activity after the License Agreement has ended.
- 11. A provision in the License Agreement which terminates the franchise upon bankruptcy of the franchise may not be enforceable under Title 11, United States Code Section 101.
- 12. Franchisor will protect the Franchisee's right granted hereby to use the Marks or will indemnify Franchisee from any loss, costs or expenses arising out of any claim, suit, or demand regarding the use of the Marks.
- 13. Section 29 of the License Agreement is amended by adding the following:
  - "Any claims pursuant to Minn. Stat. Sec. 80C.17 may be commenced within the time period provided in Minn. Stat. Sec. 80C.17, subd. 5."

14. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including, fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed with the franchise.

# **INFORMATION REQUIRED BY THE STATE OF NEW YORK**

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CAN NOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is to be added at the end of Item 3:

With the exception of what is stated above, the following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal, or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature, or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10-year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation, or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

3. The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for a franchisee to renew or extend," and Item 17(m), entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687(4) and 687(5) be satisfied.

- 4. The following language replaces the "Summary" section of Item 17(d), titled "**Termination by franchisee**": You may terminate the agreement on any grounds available by law.
- 5. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum," and Item 17(w), titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or the franchisee by Article 33 of the General Business Law of the State of New York

- 6. Franchise Questionnaires and Acknowledgements--No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 7. Receipts--Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 et seq.), which describes the time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchisee before a sale may be made. New York law requires a franchisor to provide the Franchise Disclosure Document at the earlier of the first personal meeting, ten (10) business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

# INFORMATION REQUIRED BY THE STATE OF NORTH DAKOTA

Sections 12.5.1 and 12.5.2 of the License Agreement requires that:

- (i) during the term of the License Agreement, you not engage as an owner, operator or in any managerial capacity in any educational business other than as a Licensee of SYLVAN; and
- (ii) following termination or expiration of your License Agreement (including non-renewal or transfer of your License to another person), for a period of two years you not engage as an owner, operator or any managerial capacity in any educational business offering individualized diagnostic tests or academic or prescriptive educational programs which are designed to be personally taught, supervised or administered by trained instructors within your former Territory or at any location less than five miles from your former Territory's boundary.

To the extent the preceding provisions conflict with provisions of North Dakota law concerning restrictive covenants, these restrictive covenants are not enforceable (see Section 1 of the North Dakota Rider to the License Agreement).

Because liquidated damages provisions may be unenforceable in North Dakota and Section 13 of the License Agreement includes a liquidated damages provision, Section 2 of the North Dakota Rider to the License Agreement substitutes a new Section 13 to the License Agreement that does not contain a liquidated damages provision.

Section 3 of the North Dakota Rider to your License Agreement provides that Section 20 of your License Agreement is supplemented by two provisions. The first states that to the extent the requirement that any suit you bring against SYLVAN be brought in courts in Texas violates applicable North Dakota law (see N.D.C.C., 51-19-09), this requirement is void. The second provision states that each of you and SYLVAN appoints the Securities Commissioner, State of North Dakota, State Capitol, Bismarck, North Dakota 58505, to be each party's true and lawful agent to receive service of any lawful process and any civil litigation or proceeding arising under the License Agreement.

Because the North Dakota Franchise Investment Law provides that provisions in a franchise agreement purporting to operate as waivers to compliance with such law are void, Section 4 of the North Dakota Rider to the Franchise Agreement provides for adding a new Section 26.14 to the License Agreement that states the no-waiver requirements of the Franchise Investment Law.

# INFORMATION REQUIRED BY THE STATE OF RHODE ISLAND

Section 20 of the License Agreement provides that (i) Texas law governs the agreement and (ii) if SYLVAN brings suit against you, we must do so in the courts of the state where your Territory is located, and that if you bring suit against SYLVAN, you must do so in Tarrant County, Texas, the site of SYLVAN's headquarters. The Rhode Island Franchise Act states that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act." To conform to the requirements of Rhode Island law, the Rhode Island Rider to the License Agreement provides that to the extent Rhode Island law applies and is in conflict with Section 20 of the License Agreement, Rhode Island law prevails.

# INFORMATION REQUIRED BY THE STATE OF SOUTH DAKOTA

Section 11 of the License Agreement concerns termination of the License Agreement and the notice required to terminate (which is discussed in Item 17). It provides for 10 days' notice for monetary defaults. To conform to the requirements of South Dakota law, Section 1 of the South Dakota Rider to the License Agreement provides that SYLVAN must give you at least 30 days' notice and an opportunity to cure in the event of your non- or late payment of any amounts due to SYLVAN under the License Agreement.

Section 12 of the License Agreement imposes certain restrictions on your freedom to compete with SYLVAN during the term of your License Agreement or after its termination or expiration. Section 2 of the South Dakota Rider the License Agreement provides that to the extent that provisions of Section 12 of the License Agreement conflict with South Dakota law, South Dakota law prevails. This language is added because:

"Covenants not to compete upon termination or expiration of a franchise agreement are generally unenforceable in the State of South Dakota, except in certain instances as provided by law."

Section 13 of the License Agreement entitles SYLVAN to liquidated damages in the event of your breach of certain provisions of the Agreement. Because South Dakota law may make liquidated damages provisions void and unenforceable, Section 6 of the South Dakota Rider to the License Agreement substitutes a new Section 13 stating simply that SYLVAN shall be able to seek whatever damages are available under the law.

Section 20 of the License Agreement (i) provides for Texas governing law and (ii) requires that if SYLVAN brings suit against you, to do so in the courts of the state in which your Territory is located, and if you bring suit against us, to do so in courts in the State of Texas, the state where our principal place of business is located. Section 4 of the South Dakota Rider to the License Agreement amends Section 20 of the License Agreement to add the following, as required by South Dakota law:

"Notwithstanding the foregoing, (i) the law regarding license registration, employment, covenants not to compete, and other matters of local concern will be governed by the laws of the State of South Dakota; but as to contractual and other matters, this Agreement and all provisions of this instrument will be and remain subject to the application, construction, enforcement, and interpretation under the governing law of the State of Texas; and (ii) any provision which designates jurisdiction or venue or requires the licensee to agree to jurisdiction or venue in a forum outside of South Dakota is void with respect to any cause of action which is otherwise enforceable in South Dakota."

# INFORMATION REQUIRED BY THE STATE OF VIRGINIA

To the extent the Virginia Retail Franchising Act, Va. Code §§13.1-557 – 13.1-574 applies, the terms of this Addendum apply.

#### Item 17, Additional Disclosures:

Any provision in any of the contracts that you sign with us which provides for termination of the franchise upon the bankruptcy of the franchisee may not be enforceable under federal bankruptcy law (11 U.S.C. 101 et. seq.).

"According to Section 13.1 - 564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the franchise agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable."

# INFORMATION REQUIRED BY THE STATE OF WASHINGTON

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration involving a franchise purchased in Washington, the arbitration site shall be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration, or as determined by the arbitrator. In addition, if litigation is not precluded by the franchise agreement, a franchise may bring an action or proceeding arising out of or in connection with the sales of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee shall not include rights under the Washington Franchise Investment Protection Act except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, rights or remedies under the Act such as a right to a jury trial may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

# EXHIBIT A TO THE FRANCHISE DISCLOSURE DOCUMENT

# SYLVAN LEARNING, LLC LICENSE AGREEMENT

# SYLVAN LEARNING® AND SYLVANSYNC® LICENSE AGREEMENT

# **SUMMARY PAGE**

LICENSE AGREEMENT NO.:	
EFFECTIVE DATE:	
LICENSEE(S):	
ADDRESS FOR NOTICES:	
TELEPHONE NUMBER:	
E-MAIL ADDRESS:	
TERRITORY NAME:	
LICENSOR:	Sylvan Learning, LLC, a Delaware limited liability company
ADDRESS FOR NOTICE:	2350 Airport Freeway, Suite 505, Bedford, TX 76022
INITIAL LICENSE FEE:	\$36,900, including (applicable only if checked and respective separate agreement is executed):
	☐ EDGE Package License
	ACE IT! Tutoring System License
MONTHLY ROYALTY FEE:	16% of monthly gross revenues, with a total Minimum Annual Royalty of not less than \$32,000, subject to Section 6.2
TERRITORY ADVERTISING MINIMUM:	8% of monthly gross revenues, capped at a maximum of \$40,000 per territory per calendar year, subject to Section 7.8
RENEWAL FEE:	\$6,000, subject to Section 10.2

i

# SYLVAN LEARNING® AND SYLVANSYNC® LICENSE AGREEMENT

#### TABLE OF CONTENTS

1.	Grant of Licenses; The Territory; Site Location	1
2.	Confidential Operations Manual	
3.	Protection of Sylvan's Proprietary Rights.	
4.	Obligations of Sylvan; Supervision, Assistance and Services.	
5.	Training.	
6.	Fees.	
7.	Licensee's Obligation to Comply with Standards of Operation.	14
8.	Technology.	
9.	Transfer and Assignment.	23
10.	Term and Renewal	
11.	Termination.	27
12.	Relationship After Expiration or Termination.	30
13.	Injunctive Relief and Liquidated Damages	32
14.	Hold Harmless, Indemnification and Insurance	
15.	Parties as Independent Contractors.	34
16.	Notices	35
17.	Attorneys' Fees	35
18.	Construction	
19.	Organization of the Licensee	36
20.	Time is of the Essence.	37
21.	Jurisdiction and Venue.	37
22.	Waiver of Damages.	37
23.	Dispute Resolution by Binding Arbitration.	
24.	Acceptance by Licensee.	
25.	Waiver.	39
26.	Modification.	39
27.	Class Action Waiver.	39
28.	Waiver of Jury Trial	39
29.	Limitation on Claims.	
30.	Attachments	
31.	Acknowledgements	40

# STATE-SPECIFIC RIDERS

#### **ATTACHMENTS:**

Attachment A	A – 1	erri	tory
--------------	-------	------	------

Attachment B - Licensee's Owners and Key Personnel

Attachment C – Undertaking and Guaranty

Attachment D - Confidentiality and Non-Competition Agreement

Attachment E - Acknowledgment of Request to Contribute to Advertising Cooperative

Attachment F – Contact Center Franchisee Participation Agreement

Attachment G - SylvanSync Educational Software License and Terms of Use Agreement

Attachment H – Telephone Numbers Assignment Agreement

Attachment I – ACH Authorization Agreement

Attachment J – Dashboard Access Agreement

Attachment K – Bright Horizons Participation Agreement

Attachment L – mySylvan Marketplace+ Franchisee Participation Pilot Agreement

# SYLVAN LEARNING® AND SYLVANSYNC® LICENSE AGREEMENT

This LICENSE AGREEMENT ("Agreement") is made and entered into on the Effective Date reflected in the Summary Page by and between Sylvan Learning, LLC, a Delaware limited liability company, ("Sylvan," "Franchisor," or "Licensor") and the Licensee identified on the Summary Page ("Licensee," "Franchisee," or "you"). For purposes of the Agreement and its attachments and riders, "franchise agreement" and "license agreement" are used interchangeably and refer to this Agreement.

#### Background

- A. Sylvan or its predecessors have developed certain proprietary programs, systems, teaching and management techniques, individualized diagnostic tests and academic and prescriptive educational courses or programs which are designed to be taught, supervised, or administered by trained instructors (the "Sylvan System") for use in operating learning centers ("Sylvan Centers"). The Sylvan System is licensed to others to conduct business at such Sylvan Centers pursuant to the Sylvan System.
- B. Additionally, Sylvan, pursuant to rights granted to it by third parties, has developed and plans to further develop certain proprietary technology, digital assets, systems, business methods, techniques and methodologies designed to be used in connection with portable proprietary computer systems and the Internet to provide a digital learning environment incorporating assessment results, instruction and the management of student educational programs, referred to as the "SylvanSync System." The SylvanSync System is licensed to others to conduct business at locations within a defined geographical area pursuant to the SylvanSync System.
- C. The businesses utilizing the Sylvan System and the SylvanSync System which Sylvan and its Licensees operate (the "Sylvan Network") are now identified by certain respective trade names, trademarks, service marks, logos, symbols and other indicia of origin (collectively, the "Licensed Marks") identified in the Confidential Operations Manual (hereinafter defined), including but not limited to, "Sylvan," "Sylvan Learning Center," and "Sylvan Learning" in the Sylvan System, and "SylvanSync" in the SylvanSync System, and such other trade names, trademarks, service marks, associated logos and symbols as are now designated by Sylvan (and as may hereafter be designated by Sylvan in writing).
- D. Licensee desires to have Sylvan license the Sylvan System and the SylvanSync System to Licensee with the right to conduct business using the Sylvan System and the SylvanSync System under the respective Licensed Marks at the locations and within the Territory designated in this Agreement. The business that Licensee conducts in connection with these systems is referred to in this Agreement as the "Licensed Business."

NOW, THEREFORE, in consideration of the above premises, and of the additional mutual covenants and valuable consideration recited below, the parties agree as follows:

#### 1. <u>Grant of Licenses; The Territory; Site Location.</u>

- 1.1 <u>Grant of Sylvan System License</u>. Subject to the terms and conditions of this Agreement, Sylvan grants to Licensee the right and license, and Licensee accepts and undertakes the obligation, for the term established by Paragraph 10.1 of this Agreement: (a) to operate one or more Sylvan Centers (referred to hereinafter in this Agreement as "the Center," whether one or more) and to use the Sylvan System at such Centers, including Sylvan's proprietary programs, systems, and techniques and certain copyrighted materials, as the Sylvan System may be modified, changed, improved or further developed from time to time by Sylvan; and (b) to do so only at such site or sites mutually agreed upon as provided in Paragraphs 1.5 and 1.7.
- 1.2 <u>Grant of SylvanSync System License</u>. Subject to the terms and conditions of this Agreement, Sylvan grants to Licensee the right and license, and Licensee accepts and undertakes the obligation, for the term established by Paragraph 10.1 of this Agreement, to use the SylvanSync System, including Sylvan's proprietary programs, systems, and techniques and certain copyrighted materials, as the SylvanSync System may be modified, changed, improved or further developed from time to time by Sylvan to conduct business (a) within each Center, and, if Licensee desires, (b) at such other site or sites, including without limitation one or more Satellites as defined in this Agreement, approved by Sylvan as provided in Paragraphs 1.5 and 1.7, serving retail-pay customers and, upon separate approval by Sylvan, institutional-pay customers.

- 1.3 <u>Territory</u>. Sylvan determines the size and boundaries of licensed territories depending upon its evaluation of such factors as the number of school-aged children who reside within a territory; the geographic size, population, and socio-economic factors of any territory; and such other financial and market factors as Sylvan may deem pertinent. The geographic area granted to Licensee ("Territory") is described in Attachment A, which is attached hereto and made a part hereof by reference. For the avoidance of doubt, the Territory does not include the metaverse, and Sylvan reserves, maintains, and controls all rights with respect to said metaverse.
- 1.4 <u>Territorial Rights.</u> During the term of this Agreement, and provided that Licensee is in compliance with the terms and conditions of this Agreement, Sylvan will not itself, or through any affiliate or subsidiary, operate or establish, nor license any third party to operate or establish, another Sylvan Center using the Sylvan System at any location within the Territory, except as otherwise provided in this Agreement. Sylvan retains all other rights, and may, among other things, on any terms and conditions Sylvan deems advisable, and without granting Licensee any rights therein:
- 1.4.1 own and operate (outright, through contract, joint-ventures or otherwise) Sylvan Centers at any location(s) outside of the Territory;
- 1.4.2 grant licenses for the operation of Sylvan Centers at any location(s) outside of the Territory as Sylvan, in its sole and exclusive discretion, deems appropriate;
- 1.4.3 conduct and operate businesses, and offer and sell services, programs and products within the Territory which do not comprise a part of the Sylvan System and, in connection therewith, to exploit Sylvan's Licensed Marks, name, reputation and know-how;
- 1.4.4 use the Licensed Marks and other aspects of the Sylvan System in conjunction with activities not contemplated by Sylvan as being conducted solely through or at Sylvan Centers;
- 1.4.5 offer, distribute, use and/or sell components of the Sylvan System through courses or programs offered and conducted by methods or through facilities other than Centers, within the Territory or outside the Territory, including, but not limited to: (i) the Internet; (ii) CD-ROM, books, audio or video tapes and/or electronic media; (iii) sales for home use by students; or (iv) other concepts currently in existence or as may be developed in the future, for example, through advancements in technology;
- 1.4.6 sell, provide, and deliver, directly or through affiliates or third parties, Sylvan System programs pursuant to contracts with State, Federal or Local governmental entities and school districts, and corporate contracts which require the delivery or performance of Sylvan System services within the Territory ("in-Territory Services"). All operations by Sylvan, an affiliate or a third party in providing in-Territory Services pursuant to this Paragraph shall be conducted in compliance with the same standards and requirements established by Sylvan for its licensees generally; and
- 1.4.7 sell and operate directly and to license and franchise others to sell and use within the Territory under different names or marks and/or under the Licensed Marks services, programs and products which primarily emphasize vocational training and skills in the adult workplace including by way of illustration and not by way of limitation, various computer based training and testing software programs and instructional materials and services.
- 1.5 <u>Initial Center; Initial Center Site.</u> Within forty-five (45) days of the Effective Date of this Agreement, Licensee shall identify and submit to Sylvan for approval at least one proposed site for the location of the first Center to be opened in the Territory. Time is of the essence. Licensee shall submit fully completed site approval forms to Sylvan, as specified by Sylvan in the Confidential Operations Manuals. Sylvan shall review the proposed site approval forms, and may, but is not obligated to, conduct a physical on-site review of the proposed site. If Sylvan approves the site, Attachment A shall be amended to identify the site address for the Primary Center. Sylvan shall, if necessary or requested by Licensee, provide reasonable assistance in finding other site locations. Licensee shall pay for Sylvan's out-of-pocket expenses for travel, meals and lodging incurred in providing such assistance. Sylvan may withhold approval of a site that fails to meet its site location guidelines, which Sylvan may revise from time to time, provided that such approval shall not be withheld unreasonably. Should the site(s) proposed by Licensee not be approved by Sylvan, then Sylvan shall have the right to terminate this Agreement. Licensee shall not relocate any Center or portion thereof except within the Territory and then only with Sylvan's prior written approval of the proposed new site. Sylvan may withhold approval of a relocation site that fails to meet its then-current location guidelines, which Sylvan may revise from time to time.

- Unless an extension is granted in writing, Licensee shall commence full and continuous operation of at least one Center in the Territory operating the Sylvan System and the SylvanSync System (the "Primary Center") at a site approved by Sylvan within one hundred eighty (180) days from the Effective Date of this Agreement (the "Required Open Date"). Time is of the essence. Sylvan may terminate this License Agreement if Licensee fails to meet this obligation, in which event none of the Initial License Fee shall be refunded. The Primary Center is a "Center" for the purposes of this Agreement, and all terms and conditions for a Center apply, including without limitation the Minimum Required Hours. At the Primary Center, Licensee must offer all products, programs, and services in the Sylvan System and the SylvanSync System specified by Sylvan in the Confidential Operations Manual or otherwise in writing as mandatory. Notwithstanding, Licensee may not open the Primary Center if it has not met all of the pre-opening requirements in the Confidential Operations Manual, including but not limited to:
  - (a) Execution and delivery to Sylvan of the Telephone Number Assignment Agreement attached hereto as Attachment H;
  - (b) Execution and delivery to Sylvan of the most current ACH Agreement attached hereto as Attachment I for the Licensee entity operating under this License Agreement;
  - (c) Execution and delivery to Sylvan of the Dashboard Access Agreement attached hereto as Attachment J;
  - (d) Not being in material default under this License Agreement or any other agreements with Sylvan; and
  - (e) Being current on all obligations due to Sylvan, including payment of the initial license fee, royalty, and any other fees then due.
- 1.7 <u>Additional Locations by Licensee within Territory</u>. Contingent upon Sylvan's prior approval, Licensee may open additional locations within the Territory subject to the terms and conditions of this Paragraph 1.7, and each additional location shall be identified as one of:
  - (a) an additional Center which operates the Sylvan System and the SylvanSync System (an "Additional Center"). All terms and conditions for the Primary Center, including without limitation the Minimum Required Hours, shall apply to such Additional Centers; or
  - (b) an additional Center in which Licensee may offer any portion or part of the Sylvan System and/or the SylvanSync System, and/or offer other programs and services approved by Sylvan (a "Flex Center"). Except for the requirement to offer all mandatory Sylvan and SylvanSync products, programs, and services, which does not apply to any Flex Center, all other terms and conditions for a Center, including without limitation the Minimum Required Hours, shall apply to such Flex Centers; or
  - out of Center in the Territory at a location approved by Sylvan ("Satellite"), in which Licensee may offer any portion or part of the Sylvan System and/or the SylvanSync System, and/or offer other programs and services approved by Sylvan. For example, and not as a limitation, a Satellite may be a space in a school or a community center. Satellites are not required to meet all requirements for Centers but must meet such requirements as Sylvan sets forth in the Confidential Operations Manual. Subject to Sylvan's prior written approval, and if Licensee meets the criteria, terms, and conditions that Sylvan may establish from time to time in the Confidential Operations Manual or otherwise in writing, Licensee may also deliver SylvanSync programs through a "Virtual Satellite" in the Territory whereby the SylvanSync sessions are delivered through live, online platforms. Licensee and such Virtual Satellite must meet all requirements that Sylvan sets forth in the Confidential Operations Manual.
- 1.7.1 Licensee must notify Sylvan in writing of Licensee's intention to open another Center or Satellite and shall submit to Sylvan for approval other information as is then required by Sylvan of licensees opening additional centers or satellites in their territories, including but not limited to business plans, pro formas addressing new and existing Center and Satellite operations, and financial information.
- 1.7.2 Licensee must obtain Sylvan's approval of such new Center or Satellite in writing, which Sylvan may withhold in its sole discretion, applying Sylvan's then-current standards and procedures for approval, including, but not limited to, receipt of satisfactory required documentation, and evaluation of Licensee's prior

performance under this Agreement, and the proposed site and layout, decor and equipment as are then currently required for initial centers or additional locations, as determined by Sylvan.

- 1.7.3 For each additional Center, but not for any Satellite, Licensee must pay Sylvan a fee of One Thousand Dollars (\$1,000) ("Additional Center Fee") to compensate Sylvan for its expenses incurred in site review and additional center plan approval. Said sum is payable upon approval of the new Center in writing by Sylvan.
  - 1.7.4 Licensee shall have no right to sublicense under this License Agreement.
- 1.8 <u>The Sylvan System; The SylvanSync System; Products, Programs, and Services Offered and Not to Be Offered; Licensee's Acknowledgment.</u>
- 1.8.1 <u>The Sylvan System.</u> Sylvan may, in its sole discretion, change, modify, delete from, supplement, improve or further develop the Sylvan System and the services, programs, products, and other components comprising the Sylvan System from time to time, through the Confidential Operations Manual, as provided in Paragraph 2.1.4 or otherwise upon written notice to Licensee. Licensee agrees to comply with all Sylvan's requirements in such regard, including, without limitation, offering and selling products, programs, and services that Sylvan specifies.
- 1.8.2 The SylvanSync System. Sylvan may, in its sole discretion, change, modify, delete from, supplement, improve or further develop the SylvanSync System and the services, programs, products, and other components comprising the SylvanSync System from time to time, through the Confidential Operations Manual, as provided in Paragraph 2.1.4 or otherwise upon written notice to Licensee. Licensee agrees to comply with all Sylvan's requirements in such regard, including, without limitation, offering and selling products, programs, and services that Sylvan specifies.
- 1.8.3 <u>Products, Programs, and Services Offered.</u> At the Primary Center and at all additional Centers under Paragraph 1.7 (a), Licensee must offer all products, programs, and services that comprise the Sylvan System and the SylvanSync System as specified by Sylvan in the Confidential Operations Manual or otherwise in writing. Additional Centers under Paragraph 1.7 (b) and Satellites have different product, program, and services requirements as set forth in Paragraph 1.7.
- 1.8.4 <u>Products, Programs, and Services</u>. If Sylvan includes new products, programs, or services as part of the Sylvan System or as part of the SylvanSync System, and/or offers to Licensee the right and opportunity to offer new products, programs, and services at or from the Licensed Business, Sylvan may implement new and/or different standards of operations for such new products, programs, and services, and/or charge fees for such products, programs, and services that are different from those fees charged under this Agreement. In every Center and Satellite, Licensee must refrain from offering or selling products, programs, or services that are not then-approved by Sylvan and otherwise as directed by Sylvan.
- 1.8.5 <u>Licensee's Acknowledgment Concerning Separate Systems</u>. Sylvan began offering the Sylvan System license and the SylvanSync System license in a joint license agreement in April 2016, although such systems and license grants are distinct from one another. Previously, Sylvan offered the two systems and license grants under two separate documents: (i) a Sylvan System license, and (ii) an optional SylvanSync System product license. Licensee acknowledges and agrees that the Sylvan System and the SylvanSync System are separate systems, and that Sylvan includes both systems in this Agreement for efficiency only.
- 1.9 <u>Splitting the Territory</u>. If Licensee requests and Sylvan approves such request, the Territory may be divided into two separate territories (each a "Split Territory") in order for Licensee to develop a second Primary Center. For the newly created Split Territory, Sylvan will disclose to Licensee then-current Sylvan franchise disclosure document, and Licensee will execute our then-current form of Sylvan's license/franchise agreement and pay to Sylvan \$5,000 per Territory split (the "Territory Split Fee"). For example, if the Territory is split into three separate territories, Licensee shall pay \$10,000 as its Territory Split Fee. Concurrent with such execution of the second (or subsequent) license/franchise agreement, the Attachment A to this Agreement shall be amended to replace the current Territory with the smaller remaining Split Territory. Sylvan's approval to subdivide the current Territory is subject to conditions, which may include successful completion of training by a new manager (if we require you to have a second manager), Licensee's compliance under all license or franchise agreements and other agreements with Sylvan and its affiliates, and the subdivided territory(ies) independently meeting Sylvan's then-current standards for a Territory and Primary Center.

# 2. <u>Confidential Operations Manual.</u>

- 2.1 <u>Confidential Operations and Instruction Manual</u>. Sylvan shall lend to Licensee, at no additional charge, certain Confidential Operations and Instruction Manual (hereinafter referred to as the "Confidential Operations Manual," whether one or more). The Confidential Operations Manual contains both mandatory and required standards, methods, rules, and processes, and recommended policies and procedures.
- 2.1.1 Sylvan has determined that certain programs, and certain other policies, systems, techniques, and materials are essential to the successful operation of the Licensed Business, in accordance with the standards of the Sylvan System and the SylvanSync System. In order to protect the reputation and goodwill of the businesses operating under the Sylvan System and the under the SylvanSync System and to maintain standards of operation under the Licensed Marks, Licensee agrees to operate the Licensed Business in accordance with the Confidential Operations Manual, including such amendments thereto, as Sylvan may publish from time to time, all of which Licensee acknowledges belong solely to Sylvan and shall be on loan from Sylvan during the term of this Agreement. Should Licensee lose one or more of the Confidential Operations Manual or supplements, Sylvan may impose a charge for replacement manuals or supplements issued to Licensee. The standards, specifications, or requirements established by Sylvan for Licensee, unless otherwise indicated, shall be set forth in this Agreement or may, from time to time, be set forth by Sylvan in the Confidential Operations Manual.
- 2.1.2 The Confidential Operations Manual may consist of multiple volumes of printed text, video and/or audio tapes and files, computer disks, and other electronically stored data, and various and periodic or episodic operational and/or management bulletins, in any format, and Licensee acknowledges and agrees that Sylvan may provide a portion or all of the Confidential Operations Manual (including updates and amendments) and other instructional information and materials in, or via, electronic media, including, without limitation, through the Internet.
- 2.1.3 Licensee shall at all times use its best efforts to keep the information in the Confidential Operations Manual, the Confidential Operations Manual themselves, and any other manuals, materials, goods, and information created or used by Sylvan and designated for confidential use within the Sylvan System or the SylvanSync System confidential and shall limit access to employees of Licensee on a need-to-know basis.
- Licensee understands and acknowledges that each of the Sylvan System and the SylvanSync System serve markets and operates under market conditions that are continuously changing and evolving, and that new and different standards, procedures, and techniques will be developed and recommended by Sylvan in response to those changes, or other conditions in the Sylvan System or the SylvanSync System, to maintain and/or enhance the quality of Sylvan System and/or SylvanSync System services provided by Licensee in the Licensed Business, and the reputation and goodwill of the Sylvan System, the SylvanSync System, and the Sylvan Network. Accordingly, Licensee understands and acknowledges that Sylvan may, from time to time, revise the contents of the Confidential Operations Manual to implement new or different requirements for the Sylvan System and/or the SylvanSync System, and Licensee expressly agrees to comply with all such changed requirements which are by their terms mandatory. Sylvan may exercise certain rights, to the fullest extent permitted by then-applicable law, with respect to pricing of access to the products and services offered within the Sylvan System or the SylvanSync System, including, but not limited to, establishing policies with respect to the maximum and minimum retail prices which Licensee may charge customers of the Licensed Business, recommended pricing and minimum advertised pricing for some or all of the products or services sold at the Licensed Business, which Sylvan may compel Licensee to observe and honor. Sylvan further reserves the right to establish price promotions or package promotions which may directly or indirectly impact Licensee's retail prices, and in which Sylvan may compel Licensee to participate. Franchisor may engage in any such activity periodically or throughout the term and may engage in such activity in some geographic areas but not others, or with regard to certain subsets of licensees but not others.
- 2.1.5 Licensee shall at all times ensure that its copy of the Confidential Operations Manual is kept current and up to date. In the event of any dispute as to the contents of the Confidential Operations Manual, the terms and dates of the master copy of same maintained by Sylvan at its principal place of business shall be controlling.

# 3. Protection of Sylvan's Proprietary Rights.

3.1 <u>Confidential Information and Trade Secrets</u>. Licensee understands and agrees that he/she will come into possession of certain of Sylvan's trade secrets concerning the manner in which it conducts business including, but not necessarily limited to, the following: methods of doing business or business processes; strategic business plans; customer lists and information; marketing and promotional campaigns; software; and Sylvan's materials clearly marked or labeled as trade secrets. Licensee agrees that the foregoing information, which may or may not be

considered "trade secrets" under prevailing judicial interpretations or statutes, is private, valuable, and constitutes trade secrets belonging to Sylvan and its affiliates. Licensee agrees that Sylvan derives independent economic value from the foregoing information not being generally known to, and not being readily ascertainable through proper means by, another person. Licensee acknowledges Sylvan's proprietary rights in the Sylvan System and the SylvanSync System programs, systems, techniques, and manuals, which may constitute trade secrets, are confidential information which will be disclosed to Licensee under this Agreement, and further acknowledges that Licensee does not acquire any right or interest therein beyond the rights expressly granted to it under this Agreement. By execution of this Agreement, Licensee also acknowledges that the know-how and operating format related to Sylvan's proprietary methods and materials is also proprietary to Sylvan and will not be disclosed by Licensee except in connection with Licensee's required activities as authorized by this Agreement. Licensee agrees to maintain adequate security in the control, use, and handling of proprietary materials and/or trade secrets Sylvan supplies to Licensee in accordance with the practices described in the Confidential Operations Manual or as stated otherwise in writing from time to time. Unless expressly waived by Sylvan in writing, Licensee shall require all persons who have or could have access to Sylvan proprietary materials and/or trade secrets to sign written confidentiality agreements pertaining to the security of Sylvan proprietary materials and/or trade secrets, which agreement shall be in a form reasonably acceptable to, and approved in advance by, Sylvan. Licensee shall immediately notify Sylvan of any unauthorized use of Sylvan's proprietary information and/or trade secrets, or any infringement of Sylvan's copyrights. Sylvan shall have sole discretion to take such action or inaction as it deems appropriate. Notwithstanding any other provision of this Agreement, there may be certain, limited circumstances where applicable law allows for the disclosure of certain trade secrets, as specified in the Confidential Operations Manual.

#### 3.2 Licensed Marks.

- 3.2.1 Licensee expressly acknowledges Sylvan's rights in and to the Licensed Marks and agrees not to represent in any manner that Licensee has acquired any ownership rights in the Licensed Marks. Licensee agrees not to use any of the Licensed Marks or any marks, names or indicia which are or may be confusingly similar in its own corporate or business name except as authorized in this Agreement. Licensee further acknowledges and agrees that any and all goodwill associated with each of the Sylvan System and the SylvanSync System and identified by the Licensed Marks shall inure directly and exclusively to the benefit of Sylvan and that, upon the expiration or termination of this Agreement for any reason, no monetary amount shall be attributable to any goodwill associated with Licensee's use of the Licensed Marks.
- 3.2.2 Licensee understands and agrees that any use of the Licensed Marks other than as expressly authorized by this Agreement, without Sylvan's prior written consent, is an infringement of Sylvan's rights in the Licensed Marks and that the right to use the Licensed Marks granted herein does not extend beyond the termination or expiration of this Agreement. Licensee expressly covenants that, during the term of this Agreement and thereafter, Licensee shall not, directly or indirectly, commit any act of infringement or contest or aid others in contesting the validity of Sylvan's right to use the Licensed Marks or take any other action in derogation thereof.
- 3.2.3 Licensee acknowledges an obligation to monitor its own and other parties' use of the Licensed Marks and agrees to do so. Licensee shall promptly notify Sylvan of any claim, demand, or cause of action that Sylvan may have based upon or arising from any unauthorized attempt by any person or legal entity to use the Licensed Marks, any colorable variation thereof, or any other mark, name, or indicia in which Sylvan has or claims a proprietary interest. Licensee shall assist Sylvan, upon request and at Sylvan's expense, in taking such action, if any, as Sylvan may deem appropriate to halt such activities, but shall take no action nor incur any expenses on Sylvan's behalf without Sylvan's prior written approval. If Sylvan undertakes the defense or prosecution of any litigation relating to the Licensed Marks, Licensee agrees to cooperate with and assist Sylvan as Sylvan may reasonably request in the defense or prosecution of any such action.
- 3.2.4 Licensee further agrees and covenants to use the Licensed Marks solely in the manner prescribed by Sylvan; to observe all laws with respect to the registration of trade names and assumed or fictitious names, to include in any application therefore a statement that Licensee's use of the Licensed Marks is limited by the terms of this Agreement, and to provide Sylvan with a copy of any such application and other registration document(s); and to observe such requirements with respect to trademark and service mark registrations and copyright notices as Sylvan may, from time to time, require, including, without limitation, affixing "SM", "TM", or "®" adjacent to the Licensed Marks. Licensee agrees that it shall not commit any act causing an incurable tarnishment of the Licensed Marks or the reputation and goodwill associated with the Sylvan System or the SylvanSync System.

- 3.2.5 Sylvan reserves the right, in its sole discretion, to designate and require that Licensee use one or more new, modified, or replacement Licensed Marks. If as a result of such requirement, Licensee should be required to alter any signage or outdoor advertising, Licensee shall have one hundred twenty (120) days from receipt of Sylvan's requirement to comply with same, at Licensee's sole cost and expense, except that if Licensee shall be required more than once during the initial term or during any renewal term of this Agreement to alter any signage or outdoor advertising as a result of such requirement.
- 3.2.6 In connection with Sylvan's or a permitted licensee's or third party's conduct of any business or enterprise in the Territory and under the Licensed Marks, that does not comprise a part of the Sylvan System or of the SylvanSync System, Sylvan agrees that it shall not commit, or permit any of its licensees to commit, any act that will cause an incurable tarnishment of the Licensed Marks or the reputation and goodwill associated with the Sylvan System or of the SylvanSync System.
- 4. Obligations of Sylvan; Supervision, Assistance and Services.
- 4.1 <u>Before Opening</u>. The obligations and services to be performed by Sylvan prior to the opening of the Licensed Business and at no charge to Licensee are:
- 4.1.1 <u>Site Selection</u>. Sylvan will assist Licensee in the selection of a location for the Center and, following the guidelines and standards contained in the Confidential Operations Manual, make recommendations and suggestions regarding the layout and décor of the Center to ensure compliance with Sylvan standards for the Sylvan System and for the SylvanSync System. Sylvan must approve all floor plans and signage of your Center and, if any, your Satellite.
- 4.1.2 <u>Confidential Operations Manual</u>. The Confidential Operations Manual will be loaned to Licensee.
- 4.1.3 Training. Sylvan will provide an initial training program as described in the Confidential Operations Manual for up to three members of Licensee's staff, inclusive of Licensee or, if Licensee is a corporation, partnership, limited liability company, or limited liability partnership, one of Licensee's principals who is designated to supervise the operation of the Licensed Business, has been previously approved by Sylvan, owns a beneficial equity interest in the Licensee entity, and has executed the Undertaking and Guaranty appended to this Agreement as Attachment C (the "Managing Principal"), and such other employee(s) of Licensee whom Sylvan shall have the right to approve. Sylvan reserves the right to charge a non-refundable initial training fee (the "Initial Training Fee") of up to One Thousand Dollars (\$1,000) to cover Sylvan's administrative expenses. Licensee or Managing Principal must successfully complete the initial training program by the deadline specified by Sylvan and comply with any other requirements that Sylvan may designate from time to time. Sylvan may at its discretion elect not to charge the Initial Training Fee unless the Managing Principal does not complete the initial training program by the deadline specified by Sylvan; under such circumstances. Sylvan may at its discretion elect not to charge the Initial Training Fee. However, if the Managing Principal does not complete the initial training program by the deadline specified by Sylvan, then under such circumstances, the Initial Training Fee will be charged and Licensee must submit payment upon invoice. Licensee or Managing Principal shall attend and successfully complete, to Sylvan's satisfaction, the initial training program offered by Sylvan within one hundred eighty (180) days of the Effective Date. Time is of the essence. If Licensee or Managing Principal does not successfully complete (to Sylvan's satisfaction) the initial training program within one hundred eighty (180) days of the Effective Date (provided that the training is made available during such time), Sylvan may terminate this Agreement, in which case Sylvan shall not refund any portion of the Initial License Fee or the Initial Training Fee. Sylvan's training obligation hereunder automatically expires one hundred eighty (180) days after the Effective Date, after which Sylvan may charge fees in its discretion for providing such training. All training is intended to guide Licensee and, through Licensee, certain of Licensee's employees in the delivery and sale of products and services in accordance with the standards of the Sylvan System and the SylvanSync System. Licensee shall be fully and solely responsible for all employment actions and decisions related to its employees during training, and during their employment in the Licensed Business.
- 4.1.4 <u>Training Materials and Forms</u>. Sylvan will furnish Licensee with certain materials deemed by Sylvan to be appropriate for instruction of Licensee's personnel in the conduct of the licensed instructional programs. Sylvan reserves the right to charge separate fees for (a) additional training materials provided to Licensee for Licensee's employees; (b) specialized training materials for specific programs, courses, or materials; (c) specialized training conducted utilizing different methodologies or delivery systems, including, without limitation,

electronically delivered training programs or materials; and/or (d) additional or supplemental training, charged on a per-person basis.

- 4.1.5 Additional Training. If Sylvan determines, in its sole discretion, that Licensee is in need of additional supervision or supplemental training, Sylvan may require that Licensee receive such training from Sylvan, in which case Licensee agrees that it shall pay Sylvan's then-current per diem additional training charges and out-of-pocket training expenses, which shall be as set forth in the Confidential Operations Manuals or otherwise in writing (as of the Effective Date, such fees are \$500 per day). If Licensee requests that Sylvan provide additional supervision or supplemental training or that any training programs offered or required by Sylvan, then Licensee further agrees that it shall pay Sylvan's then-current per diem additional training charges and out-of-pocket training expenses, set forth in the Confidential Operations Manuals or otherwise in writing.
- 4.2 <u>During Term at No Charge</u>. The obligations of Sylvan during the term of this Agreement, for which Licensee shall not be subject to any charge, are:
- 4.2.1 <u>Action Plan</u>. Sylvan will assist Licensee in developing an initial action plan for Licensee to follow and implement in operating the Licensed Business, which includes a marketing strategy, an estimate of expenses, and sales goals. Neither Sylvan nor Licensee shall be bound to implement the action plan, nor shall any action plan constitute a representation by Sylvan with respect to actual, average, projected, or forecasted volume, sales, profits, or earnings of Licensee if Licensee follows and implements such plan.
- 4.2.2 <u>Toll-Free Number</u>. Sylvan will maintain a toll-free number for Licensee's use in contacting employees of Sylvan at Sylvan's headquarters. Sylvan reserves the right to charge a fee for maintenance of a toll-fee number for Licensee, pursuant to Section 6.11 below.
- 4.2.3 <u>Advertising Materials</u>. Sylvan will provide to Licensee, from time to time, certain advertising, marketing, and/or promotional materials, in forms and media determined by Sylvan in its sole discretion, approved and updated at least every two (2) years by Sylvan.
- Quality Assurance Review. Sylvan will maintain a procedure and checklist for the review of all aspects of Licensee's operation of the Licensed Business ("Quality Assurance Review") whereby Sylvan can determine the extent to which Licensee's operations are or are not in compliance with the standards, methods, and procedures Sylvan establishes for its licensees. Approximately one hundred twenty (120) days after Licensee first opens the Licensed Business, Sylvan will conduct a Quality Assurance Review of Licensee's Center. Such Quality Assurance Review may be conducted on-site and in person by a Sylvan representative, or through an off-site or remote review of documents, materials, and other information as determined by Sylvan. At Licensee's request, Sylvan will have a Sylvan representative conduct a Quality Assurance Review of the Licensed Business once in every two (2) calendar year period if Sylvan has not itself initiated such action. Licensee may request that such Quality Assurance Review be conducted on-site and in person by a Sylvan representative, and Sylvan shall have a Sylvan representative conduct such Quality Assurance Review, provided that Licensee is in compliance with all financial and reporting obligations to Sylvan and is not subject to default or termination under this Agreement. If Licensee is not in such compliance, and/or is subject to default or termination under this Agreement, Licensee still may request that such Quality Assurance Review be conducted on-site and in person by a Sylvan representative, and Sylvan shall have a Sylvan representative conduct such Quality Assurance Review, provided that Licensee shall reimburse Sylvan for its and its representative's travel, lodging, meal, and other actual costs incurred in connection with providing such onsite Quality Assurance Review in person. At Licensee's request, Sylvan will provide Licensee with the materials and forms used in the Quality Assurance Review process for Licensee's use in reviewing Licensee's operations.
- 4.2.5 Ongoing Assistance. If a Sylvan representative has not already conducted two visits to the Licensed Business for a Quality Assurance Review or otherwise, then Sylvan will at Licensee's request cause a Sylvan representative to provide assistance and offer suggestions and recommendations within his or her knowledge toward the betterment of the Licensed Business, such assistance to be provided either via up to two (2) one (1) day on-site visits each calendar year or via other customary means of providing advice, at Sylvan's sole discretion. Visits to the Licensed Business under this paragraph shall be scheduled by Sylvan according to the availability and prior commitments of Sylvan representatives.
- 4.2.6 <u>Latest Developments</u>. Sylvan shall endeavor to keep abreast of the most recent and advanced knowledge concerning instructional equipment and new reading and mathematics or other instructional techniques. Sylvan will, based on its good faith determination of the applicability of such knowledge to licensee operations, make certain such knowledge is available to Licensee and to licensees generally.

8

- 4.2.7 <u>Sylvan Hardware and Sylvan Software for Use in the SylvanSync System.</u> Sylvan shall loan or otherwise provide to Licensee certain Sylvan Hardware and Sylvan Software (as those capitalized terms are defined in Paragraph 8.1.7 of this Agreement) that Licensee must use as part of Licensee's operation of the SylvanSync System. The Sylvan Hardware and Sylvan Software to be loaned or provided at any given time shall be identified by Sylvan in the Confidential Operations Manual and are subject to change by Sylvan at any time and from time to time.
- 4.2.8 <u>Mandatory Assessments</u>. Sylvan shall make available to Licensee at no charge each assessment that Sylvan deems mandatory in any mandatory Sylvan System or SylvanSync System program, and Licensee may use such assessment at no charge in all Sylvan System and SylvanSync System programs, whether optional or mandatory, in which Sylvan approves use of such assessment.
- 4.3 <u>During Term and May be Subject to Charge</u>. The services which Sylvan must provide during the term of this Agreement, and for which Sylvan may charge Licensee, are:
- 4.3.1 <u>National Conference</u>. From time to time, but at least bi-annually, Sylvan will hold a national conference of not more than five (5) days duration at a location to be selected by Sylvan. Sylvan shall determine the topics and agenda for such conference to serve the purpose, among other things, of updating licensees on new developments affecting licensees, and exchanging information between licensees and Sylvan personnel regarding Sylvan System and SylvanSync System operations and programs. Sylvan may charge a fee for each attendee for attendance at a national conference, which is payable upon invoice and before Licensee and its employees attends such national conference, and nonrefundable upon payment. All travel, room and board, and registration expenses for Licensee and Licensee's personnel attending a national conference shall be borne by Licensee.
- 4.4 <u>Other: Regional Training</u>. From time to time, Sylvan may conduct training sessions at a location other than its headquarters. Attendance at such training may be made mandatory for Licensee and Licensee's Sylvancertified personnel. Sylvan may charge a fee for attendance at regional training sessions. All travel, room and board, and registration expenses for Licensee and Licensee's personnel attending the training shall be borne by Licensee.

#### 5. Training.

- 5.1 Training Programs and Requirements. The success of every student enrolled at a Sylvan location depends upon the quality and integrity of the program in which that student receives instruction, and in turn the quality and integrity of the program depends upon the proper initial and continuous training of Licensee's personnel involved in the delivery of the Sylvan System and SylvanSync System programs. To assist Licensee in the operation of its Licensed Business and the proper delivery of the Sylvan System and the SylvanSync System and the student instructional programs under each system, Sylvan agrees to provide "Training Programs" as set forth in the Confidential Operations Manual. Licensee agrees that it shall comply with Sylvan's standards and requirements for certifications Licensee (or Licensee's Managing Principal) and Licensee's personnel must obtain, and for training Licensee (or Licensee's Managing Principal) and Licensee's personnel, including without limitation Licensee's obligations under Paragraph 4.1.3 of this Agreement, and including any training subscription fees or other payments required by Sylvan from time to time, as designated in the Confidential Operations Manual and any changes, additions or modifications thereto. Prior to any training, Licensee and all other persons to be trained must sign, whenever requested, a confidentiality, non-disclosure, and covenant not to compete agreement on the form then used by Sylvan. The training that Sylvan provides to Licensee and to Licensee's employees does not constitute employment, direction, or supervision of Licensee or Licensee's employees by Sylvan. Sylvan does not hire, fire, manage, supervise, schedule, discipline, or set compensation for Licensee or Licensee's employees, and all decisions and actions related to Licensee's employees are the sole responsibility of Licensee.
- 5.2 <u>Expenses</u>. Travel and living expenses for Licensee and Licensee's staff while attending Sylvan's Training Programs are at Licensee's expense. Sylvan may charge a fee for attendance at Regional Training Programs and National Conference. Sylvan may require Licensee to pay a fee to attend the initial training program, as described in Section 4.1.3.

#### 6. <u>Fees.</u>

- 6.1 <u>Initial License Fee</u>. Licensee agrees to pay to Sylvan an Initial License Fee of Thirty-Six Thousand, Nine Hundred Dollars (\$36,900) in one lump sum, unless otherwise specified on the Summary Page. Licensee acknowledges and agrees the Initial License Fee is fully earned by Licensor when paid and is not refundable.
  - 6.2 <u>Monthly Royalty Fee; Minimum Annual Royalty.</u>

- 6.2.1 Monthly Royalty Fee. At all times after the commencement of operations by Licensee, Licensee shall pay Sylvan a monthly royalty for the preceding month equal to sixteen percent (16%) of gross revenues (hereinafter defined). The term "gross revenues," as used in this Agreement, shall mean the aggregate gross amount of all revenues from whatever source derived (based upon the cash method of accounting) which arise from or are derived by Licensee or by any other person from business conducted or which originated in, on, from, or through the Licensed Business or from the sale of any products or services associated with the use of the Licensed Marks, whether such business is conducted in compliance with or in violation of the terms of this Agreement, excluding only sales or other tax receipts (the collection of which is required by law), and any documented refunds, chargebacks, credits, and allowances made to customers in good faith by Licensee. Gross revenues shall not include any amounts paid to Licensee by Sylvan pursuant to any subcontract that may be entered into between Licensee and Sylvan under Paragraph 1.4.6 above.
- Minimum Annual Royalty. To induce Sylvan to enter into this Agreement with Licensee, Licensee agrees and acknowledges that Licensee shall generate sufficient gross revenues so that the royalty fees that Licensee pays to Sylvan under Paragraph 6.2.1 for each calendar year, net of all subsequent reductions to and refunds from such gross revenues, shall be at least equal to the Minimum Annual Royalty, as defined in this Paragraph, beginning with the calendar year that follows the Required Open Date. The "Minimum Annual Royalty" is Thirty-Two Thousand Dollars (\$32,000) per calendar year; provided, however, that if the Required Open Date falls in the second half of a calendar year, then the Minimum Annual Royalty for and only for the immediately following calendar year shall be Twenty Thousand Dollars (\$20,000). If Licensee fails to open a Primary Center by the original Required Open Date, as set forth in Paragraph 1.6 above, and Sylvan extends such Required Open Date as an accommodation to Licensee, Sylvan reserves the right to enforce the Minimum Annual Royalty requirement as if Licensee had opened such Primary Center on time, according to the original requirement. Licensee agrees and acknowledges that the Minimum Annual Royalty is not a projection, forecast, or guarantee of Licensee's actual gross revenues in the Territory, and that Sylvan expressly disclaims making any representation concerning Licensee's actual gross revenues and performance in the Territory. The Minimum Annual Royalty shall apply in the initial term and each renewal term of this Agreement. Upon the first and second times in a given term of this Agreement that Licensee fails to generate sufficient gross revenues for the royalty fees due under Paragraph 6.2.1 to meet or exceed the Minimum Annual Royalty, such failure shall constitute a curable material default of this Agreement; such default shall be deemed cured by payment to Sylvan of the difference (the "Shortfall") at the same time as and in addition to the monthly royalty payment due in the January immediately following such calendar year. In the event of any subsequent failure (each a "Subsequent Failure"), after the first two failures in a term, by Licensee to generate sufficient gross revenues for the royalty fees due under Paragraph 6.2.1 to meet or exceed the Minimum Annual Royalty in a calendar year during a given term of this Agreement, regardless of whether the earlier failures were cured in accordance with this Agreement or waived by Sylvan, such Subsequent Failure shall be deemed an incurable material default of this Agreement and, subject to the appeal provisions of Paragraph 6.2.3, Sylvan may terminate this Agreement at any time from the monthly royalty payment due date in the January immediately following the calendar year until June 30 of such calendar year (the "Termination Period"). If Sylvan does not terminate this Agreement by written notice to Licensee during the Termination Period, then Sylvan shall be deemed to have waived its rights to terminate this Agreement for such Subsequent Failure, and such Subsequent Failure shall not be deemed an incurable material default for the purposes of Paragraph 10.2.3 and Paragraph 11.3.9. Sylvan's notice of termination to Licensee shall stipulate an appeal period ("Appeal Period") of not less than twenty (20) days before the termination takes effect. During the Appeal Period, Licensee may appeal the termination pursuant to the process set forth in Paragraph 6.2.3; otherwise, the termination shall take effect upon the expiration of the Appeal Period. Payment of and/or Sylvan's acceptance of payment of a given Shortfall shall not constitute a cure or stay of termination for any Subsequent Failure, except in the event that Sylvan approves an appeal pursuant to Paragraph 6.2.3
- 6.2.2.1 Notwithstanding Paragraph 6.2.2, the Minimum Annual Royalty shall be waived for the Transferor and the Transferee for the calendar year of the Effective Date of this Agreement if Licensee enters into this Agreement as the Transferee in an approved Transfer.
- 6.2.3 <u>Process to Appeal Termination for Subsequent Failure</u>. Licensee may appeal Sylvan's termination for a Subsequent Failure by appealing in writing to Sylvan within the Appeal Period as defined in Paragraph 6.2.2 and promising to pay the Shortfall, and all late fees, interest, and penalties in connection therewith, within fifteen (15) days following Sylvan's written ruling on Licensee's appeal (the "Cure Date"). Beginning on the date Sylvan receives Licensee's written appeal (provided same is received within the Appeal Period), termination pursuant to Paragraph 6.2.2 and Paragraph 11.3.9 shall be stayed until Sylvan sends to Licensee Sylvan's written

ruling on the appeal. Sylvan shall have the right to deny or accept Licensee's appeal, at Sylvan's sole discretion, and is not required to accept any appeal. During the Appeal Period, Licensee's obligation to pay the Minimum Annual Royalty under appeal is suspended, but all of Licensee's other obligations, financial and otherwise, remain in effect, including without limitation the requirement to pay ongoing royalties and other amounts. If Sylvan denies the appeal in writing, the termination automatically shall become effective ten (10) days after the date of Sylvan's written denial to Licensee. If Sylvan accepts the appeal in writing, termination shall be stayed until the Cure Date or such later date as Sylvan stipulates. If, Sylvan having accepted the appeal in writing, Licensee subsequently pays to Sylvan the Shortfall by the Cure Date, then the termination automatically shall be rescinded, and the underlying Subsequent Failure shall not be deemed an incurable material default for the purposes of Paragraph 10.2.3 and Paragraph 11.3.9; otherwise, the termination shall become effective automatically on the day after the Cure Date without further notice or opportunity to cure, and Licensee shall not be obligated to pay the Shortfall. The provisions in this Paragraph apply only to termination for Subsequent Failure as defined in Paragraph 6.2.2 and do not apply to any other event of default or termination.

- Monthly Royalty Payment Date. The payments under Paragraph 6.2 shall be due and payable to Sylvan at Sylvan's office or at such location as Sylvan shall establish in the Confidential Operations Manual, or otherwise in writing, by the 15th day of the calendar month for revenues received by Licensee in the immediately preceding month. If payment is made by mail, the payment must be postmarked not later than the 15th day of the calendar month to be considered paid when due. Sylvan reserves the right to require in the Confidential Operations Manual, or otherwise in writing, that Licensee make payments directly to Sylvan, its affiliates, its bank, or such financial institution account that Sylvan specifies, by electronic funds transfer, pre-authorized auto-draft arrangement, or such other method as Sylvan from time to time may specify (collectively, "electronic transfer of funds") on the 15th day of the calendar month. If the 15th day of the calendar month falls on a U.S. national holiday or Saturday or Sunday, the electronic transfer of funds shall be made on the next business day on which U.S. banks are open. Licensee agrees to comply with such requirement and shall furnish to Sylvan or its designee such information and authorizations as may be necessary to permit Sylvan or its designee to make withdrawals by electronic funds transfer or auto-draft arrangement. Sylvan shall promptly return any funds to Licensee that were improperly or erroneously deposited into a Sylvan account via an electronic transfer of funds. Within 15 days after expiration or termination of this Agreement, Licensee shall pay Sylvan the royalty for the period immediately preceding expiration or termination. Licensee's obligation to pay sums due under Paragraph 6.2 for revenues received during the term of this Agreement shall survive termination of this Agreement.
- 6.3.1 <u>Late Fees and Interest Charges</u>. For all amounts due to Sylvan pursuant to this Agreement, Licensee agrees to pay a late charge of ten percent (10%) upon any sum not paid when due, plus interest of one and one-half percent (1½%) per month or portion thereof that the amount due remains unpaid. The obligation to pay a late charge is in addition to the potential liability for termination under Paragraph 11. Acceptance by Sylvan of any payments or partial payments due to it under this Agreement shall not be deemed a waiver by Sylvan of any preceding or succeeding breach by Licensee of any terms, provisions, covenants, or conditions of this Agreement (including Sylvan's right to require Licensee to pay a late fee and interest charges), or other amounts due.
- Gross Revenue Report. Licensee's monthly payment shall be accompanied by fully completed and accurate (a) gross revenue and (b) income and expense reports on forms as required by Sylvan, setting forth the gross revenues received during the previous month, the number of students currently enrolled by Center or program category, any advertising expenditures paid, and any further information Sylvan may request. Licensee's reports shall also include such information as Sylvan requests on services provided on a scholarship or reduced- or no-fee basis. Such forms and information required therein may be changed by Sylvan from time to time. Licensee shall keep adequate records and retain appropriate data and receipts to verify the information provided in such reports and shall provide such backup data as Sylvan may request from time to time. Licensee shall certify the accuracy of such data on each such report. Annually, not later than April 30th for the prior year ended December 31, Licensee shall provide Sylvan with a profit and loss statement and a balance sheet for Licensee's operation, in such form and method of transmission as requested by Sylvan or prescribed in the Confidential Operations Manual. Such profit and loss statement and balance sheet may at Sylvan's specific request be a compilation prepared by a Certified Public Accountant. Also, annually, Licensee shall provide Sylvan with federal and state income tax and sales tax returns within one hundred twenty (120) days of the end of Licensee's tax year. Sylvan may require that Licensee submit monthly reports electronically, no later than five days (excluding holidays) following the end of each calendar month for the immediately preceding calendar month. Licensee shall have until the 10<sup>th</sup> day of the calendar month to submit an amended report for the purpose of correcting errors. Sylvan may require submission of reports more frequently,

including submission of daily automated reports, as established in the Confidential Operations Manual, or otherwise in writing, provided, however, that Sylvan agrees that it shall not begin to require daily reporting until such time as Sylvan's technology and accounting systems allow for automatic submission of same.

- 6.5 <u>Establishment of Tuition Charges</u>. Licensee shall establish the tuition and other charges to Licensee's students and shall make such revenue information available to Sylvan on request, as necessary for the purposes of Sylvan's audit procedures and other purposes.
- 6.6 <u>Records, Inspection and Auditing.</u> Licensee shall use a Sylvan approved bookkeeping and accounting system. Licensee shall accurately record gross revenues in the Computer System on the day they are received, or in such other manner and frequency that Sylvan prescribes. Licensee is responsible for hiring its own accountant. Licensee shall sign such documents as Sylvan may at any time request which will authorize Sylvan to endorse for deposit in Sylvan's account checks payable to Licensee's order in the event Sylvan elects to manage the Licensed Business pursuant to Paragraph 9.5.

Licensee shall allow Sylvan or its representatives to inspect all premises of the Licensed Business, the books of account and federal and state income tax and sales tax returns of Licensee pertaining to the Licensed Business during reasonable times of the business day with or without notice, and, in the event Sylvan or its representatives conduct such inspection remotely, Licensee agrees to make available to Sylvan or its representatives for inspection all such materials, or certified true, correct and complete copies thereof, via electronic or physical transmission in the form and on or before the delivery time designated by Sylvan. Materials that Licensee must maintain in the Computer System or other electronic format pursuant to this Agreement shall be made available immediately upon request. For materials that Licensee is not required to maintain electronically pursuant to this Agreement, and that Licensee certifies are not available in electronic format, Sylvan may designate a delivery time no fewer than two (2) calendar days following Sylvan's request. Sylvan shall use reasonable efforts not to disrupt the operations of the Licensed Business during on-site inspections. Licensee shall maintain on the Center premises, or at a mutually agreed upon location, for a period of at least three (3) years, all sales slips, cancelled checks and receipts, and all books of account. If Sylvan's audit discloses a liability for any fees due to Sylvan, Licensee shall pay the amount of the deficiency, including late charges due under Paragraph 6.3.1, within thirty (30) days of Sylvan's notice to Licensee of the deficiency. Should the audit reflect a balance due Licensee, Sylvan agrees to pay said balance within thirty (30) days of completion of the audit report. If the deficiency (excluding late charges) is three percent (3%) or more of the fees Licensee was required to pay hereunder for the relevant period, Licensee shall pay Sylvan its costs incurred in performing the audit in addition to the amount of the deficiency. Nothing in this paragraph shall limit Sylvan's right to terminate Licensee under Section 11 for Licensee's failure to pay when due royalties owing Sylvan.

Sylvan agrees that it will maintain the confidentiality of all financial information it shall obtain about Licensee's operations, and shall not disclose such financial information to any third party who is not bound to maintain the confidentiality of such information; provided however, that (i) Sylvan may use such information in preparing earnings claims, financial performance representations, or similar franchise disclosures, so long as the information about Licensee's operations is not identified or indicated as pertaining to Licensee, and (ii) Sylvan may prepare a composite list of financial performances by its licensees for dissemination among the licensees, identifying key operating metrics and financial information of the Licensed Business, and (iii) Sylvan may disclose such information in response to a government or judicial subpoena, provided, however, that Sylvan shall use reasonable efforts to notify Licensee of any such subpoena so that Licensee may seek a stay of enforcement of the subpoena.

In addition to the reports described in Paragraph 6.5 and Paragraph 6.6 above, Licensee shall provide to Sylvan such additional or substitute reports, at such times and with such data and information as Sylvan may request from time to time, and in the form and manner as specified by Sylvan in the Confidential Operations Manual, which reports may be electronic reports transmitted via the Computer System as defined in Paragraph 8.1 of this Agreement, or other methods specified by Sylvan.

Sylvan may also require you to participate in brand-wide management and reporting systems, which Licensee must contribute requested data and otherwise participate in. Licensee must also execute Attachment J, the Dashboard Access Agreement, which gives Licensee access to Sylvan's current reporting system. Licensee may be required to participate in other systems in the future, which it must participate in and incorporate into its reporting procedures at its own cost and expense.

6.7 <u>Licensee to Pay Taxes</u>. Licensee shall pay all taxes, regardless of their nature, assessed against it when due and before delinquent except when being contested in good faith by appropriate proceedings. Licensee

agrees to pay, in addition to the monthly royalty fee, an amount equal to any sales, gross receipts, or similar tax imposed on Sylvan and calculated solely on payments required to be made by Licensee to Sylvan under this Agreement, unless such tax is an optional alternative to an income tax otherwise payable by Sylvan. Licensee shall pay such taxes on Sylvan's behalf except when being contested in good faith by appropriate proceedings. In no event shall Licensee be required to pay any income tax otherwise payable by Sylvan.

- 6.8 Payment of Accounts. Sylvan may, from time to time, and at its sole option and discretion, allow Licensee to purchase goods, materials, equipment, forms, printed matter, and programs on an open or revolving account, or through special installment payment plans. Any limitation on the amount Licensee may charge to an account or payment plan will be determined by Sylvan, taking into consideration the credit worthiness, financial capability, and business history of Licensee. All items charged to Licensee's account(s) shall be supported by invoice. Licensee shall promptly pay Sylvan for all charges made to Licensee's account(s). Invoiced charges are due and payable thirty (30) days after date of the invoice. If not paid by then, the unpaid balance will be considered past due, and balances not paid when due shall be subject to late charges and interest as set forth in Paragraph 6.3.1 of this Agreement. No payment by Licensee, nor any endorsement or statement on any check or writing accompanying any check or payment, may be deemed an accord and satisfaction unless agreed by Sylvan in a separate writing. Sylvan may accept any such check or payment without prejudice to Sylvan's right to recover any balance remaining or pursue any remedy available to Sylvan. Sylvan may require payment by electronic transfer of funds as provided in Paragraph 6.3.
- CPP) agreements entered into by Sylvan on behalf of the Sylvan Network. Licensee acknowledges that Sylvan alone shall negotiate and establish discounts and procedures with CPP clients that will be uniformly adhered to by all CPP-participating licensees, and Licensee will offer and provide the Sylvan-established discounts from Licensee-established tuition rates as indicated in the Confidential Operations Manual. Sylvan will represent to CPP clients that the Licensed Business is a participant in the program, and will in turn authorize the CPP client to represent to members of the group defined in the CPP agreement (as examples and not as limitations, the defined group may be employees, members, and/or customers of the CPP client, consumers responding to a particular promotion, etc.) that the Licensed Business is a participant in the program. Licensee shall not be required to honor or provide any other discounts to customers who enroll in the Licensed Business pursuant to a CPP program but may do so at Licensee's discretion. Licensee may refuse to enroll CPP customers if available space or class time is limited due to full capacity enrollment. Licensee shall periodically report the number of students, newly enrolled and existing, in the Licensed Business pursuant to CPP programs.
- Refurbishments), Paragraph 7.11 (Educational Programs), Paragraph 6.2.2 (Minimum Annual Royalty), Paragraph 7.8.1 (the cap on Monthly Territory Advertising Minimums), and Paragraph 6.11 (Technology) may be adjusted annually ("Cost of Living Adjustment") by Sylvan as set forth herein based upon changes in the Revised Consumer Price Index for Urban Wage Earners and Clerical Workers: All items (1982-1984 = 100) ("Consumer Price Index") published by the Bureau of Labor Statistics, United States Department of Labor. The base index ("Base Index") shall be the Consumer Price Index for January 1 of the calendar year of the Effective Date. Sylvan may adjust any, some, all, or none of the four (4) flat dollar amounts identified in this Paragraph so that the adjusted flat dollar amount for a given year shall be equal to the product of (a) a fraction, the numerator of which is the Consumer Price Index for January of the year of such adjustment, and the denominator of which is the Base Index, multiplied by (b) the flat dollar amount set forth in such Paragraph. The Cost of Living Adjustment shall apply to the flat dollar amounts in Paragraph 7.9.4 (Upkeep and Refurbishments), Paragraph 7.11 (Educational Programs), and Paragraph 6.11 (Technology) from the Effective Date, and to the flat dollar amounts in Paragraph 6.2.2 (Minimum Annual Royalty) and Paragraph 7.8.1 (the cap on Monthly Territory Advertising Minimums) beginning in the sixth calendar year of the term of this Agreement.
- 6.11 Technology Related Fees. Sylvan reserves the right to incorporate new technologies and charge related fees, as indicated in the Confidential Operations Manual. Licensee is required to utilize the Hub Technology Platform (or equivalent point of sale and merchant processing system, if Sylvan chooses a substitute platform), for which Licensee shall pay to Sylvan a then-current associated monthly fee indicated in the Confidential Operations Manual (as of the Effective Date, this fee is \$114 per Center). Sylvan reserves the right to charge a fee for maintenance of a toll-fee number for Licensee, pursuant to Section 4.2.2. above, as indicated in the Confidential Operations Manual (as of the Effective Date, this fee is \$.20/minute). Further, All fees hereunder are subject to Paragraph 6.10 above, and may be modified from time to time due to changes in costs of third-party service providers.

6.12 Payment of Fees. Licensee must participate in Sylvan's then-current electronic funds transfer program authorizing Sylvan to utilize a pre-authorized bank draft system, and sign the ACH Authorization form attached hereto as Attachment I. All royalty and other amounts owed under this Agreement, including interest charges, are payable monthly and must be received by Sylvan or credited to Sylvan's account by pre-authorized bank debit before 5:00 p.m. on the date such payment is due, as specified in the Confidential Operations Manual (the "Due Date"). On each Due Date, Sylvan will transfer from Licensee's commercial bank operating account ("Account") the fees due pursuant to this Article 6 and elsewhere in this Agreement based on the gross revenue reported to Sylvan by Licensee or as determined by Sylvan by the records contained in the cash registers/computer terminals of the Licensed Business. Declining or revoking Licensee's participation (directly or indirectly) in Sylvan's then-current electronic funds transfer program is a material breach of this Agreement for which Sylvan may terminate Licensee's agreement.

For the sake of clarity, Licensee must include in gross revenue all revenue you receive in connection with the operation of the Licensed Business, including without limitation payments that you may receive from third parties that "host" programs and disbursements Licensee receive from any third party sales platform, in each case whether authorized or unauthorized (provided, Sylvan's acceptance of fees paid by Franchisee in connection with unauthorized programs or third party service providers will not constitute a waiver of any right or remedy of Sylvan under this Agreement or applicable law). If Licensee has not reported to Sylvan gross revenue for any reporting period, Sylvan will transfer from the Account an amount calculated in accordance with its estimate of the Licensed Business' gross revenue during the reporting period, which estimate may be based on, among other things, historical financial performance of the Licensed Business or current and historical performance of other franchisees. If, at any time, Sylvan determines that Licensee have underreported gross revenue or underpaid the Royalty Fee or other amounts due to Sylvan under this Agreement, or any other agreement, Sylvan shall initiate an immediate transfer from the Account in the appropriate amount in accordance with the foregoing procedure, including interest as provided in this Agreement. Any overpayment will be credited against future royalty and other payments due under this Agreement.

In connection with the payment by electronic funds transfer, Licensee shall: (1) comply with procedures specified by Sylvan in the Confidential Operations Manual or otherwise in writing; (2) perform those acts and sign and deliver those documents as may be necessary to accomplish payment by electronic funds transfer as described in this <u>Paragraph 6.12</u>; (3) give Sylvan an authorization in the form designated by Sylvan to initiate debit entries and credit correction entries to the Account for payments of the royalty and other amounts payable under this Agreement, including any interest charges; and (4) make sufficient funds available in the Account for withdrawal by electronic funds transfer no later than the Due Date for payment thereof.

Notwithstanding the provisions of this <u>Paragraph 6.12</u>, Sylvan reserves the right to modify, at its option, the method by which Licensee pay the royalty and other amounts owed under this Agreement, including interest charges, upon receipt of written notice by Sylvan. Licensee's failure to have sufficient funds in the Account shall constitute a default of this Agreement pursuant to <u>Article 11</u>. Licensee shall not be entitled to set off, deduct, or otherwise withhold any royalty, interest charges, or other monies payable to Sylvan under this Agreement on grounds of any alleged nonperformance by Sylvan of any of its obligations or for any other reason.

#### 7. <u>Licensee's Obligation to Comply with Standards of Operation.</u>

7.1 Compliance with Standards. Licensee agrees that the Licensed Business shall be managed and administered in a competent manner, in full compliance with this Agreement and the Confidential Operations Manual to preserve, maintain, and enhance the goodwill associated with the Sylvan System, the SylvanSync System, and the Licensed Marks. Licensee shall use only the Sylvan System and SylvanSync System educational training and other programs that Sylvan licenses to it under this Agreement, or courses which Sylvan specifically approves in writing. Licensee agrees to comply with Sylvan's standards, specifications, and other requirements which Sylvan may make in writing from time to time concerning the décor, design, and layout of the Licensed Business. Sylvan may prescribe specific standards for all signs identifying the Licensed Business locations, and Licensee shall place and maintain at least one sign at each location meeting Sylvan's standards for such type of location, subject to local ordinances. Licensee agrees to offer for sale at the Center only the programs, products, and services that Sylvan designates or approves from time to time, which Sylvan may modify at its option.

Licensee shall purchase from Sylvan all required diagnostic tests, education programs, programs, and student record forms, parent information booklets, and explanatory and promotional brochures developed and copyrighted by Sylvan. These materials are unique and are available from no other source. Licensee must at all times maintain an inventory of such materials sufficient to service customers and students, and to maintain the quality standards of the Sylvan System and of the SylvanSync System.

Sylvan warrants that proprietary products purchased by Licensee from Sylvan hereunder meet Sylvan's specifications for such products. Sylvan neither makes, nor does it authorize Licensee to make any other warranties, express or implied, with respect to proprietary products delivered under the License Agreement, and Sylvan expressly excludes and disclaims all implied warranties of merchantability and fitness for a particular purpose with respect to proprietary products delivered under the License Agreement.

Licensee shall also utilize Sylvan's designated suppliers or distributors to purchase certain commercially available equipment, software, point-of-sale systems, merchant processing systems, insurance carriers or brokers, marketing services, and other products, services, and materials required by Sylvan in the Confidential Operations Manual or otherwise in writing for use in a licensed business operating the Sylvan System or the SylvanSync System. Sylvan reserves the right to modify, add, and discontinue use of such suppliers or distributors at any time at its sole discretion. Licensee agrees to promptly comply with all such requirements within Sylvan's designated timeframe, and at Licensee's sole expense. Sylvan shall designate the approved sources, suppliers, and distributors of such equipment and materials in the Confidential Operations Manual. Licensee is free to choose in accordance with the approved sources list the wholesale or retail distributor from which it purchases such equipment and materials. The equipment, materials, and furniture used by Licensee must be consistent with the type, quantity, quality, and variety as Sylvan requires in the Confidential Operations Manual or otherwise in writing. Additional sources may, in Sylvan's discretion, be approved in the future. Licensee may request that Sylvan consider a proposed additional source by following the procedure stated in the Confidential Operations Manual.

Licensee shall also purchase certain commercially available furniture products which meet the specifications of Sylvan. Sylvan will provide Licensee with specifications as set forth in the Confidential Operations Manual. If the furniture does not meet Sylvan's specifications, Licensee may not use such furniture in the Licensed Business, and a Center will not be considered open for purposes of Paragraph 1.6 unless and until such conforming furniture is installed.

Licensee shall not purchase equipment, furniture items or other materials for which Sylvan has designated specifications from any supplier which does not meet Sylvan's specifications.

Sylvan intends that Sylvan company-owned Centers comply with the foregoing standards.

# 7.2 <u>Center Staff Requirements; Accreditation.</u>

- 7.2.1 <u>Staff Requirements.</u> To maintain the high image and quality standards of Sylvan's proprietary programs and materials, Licensee agrees to provide sufficient and competent management (including, without limitation, operational, financial, and marketing management), staff, and Sylvan-certified personnel at the Licensed Business to deliver high-quality instruction and service across all programs. Sylvan does not permit absentee management by Licensee. Licensee or, if Licensee is a corporation, partnership, limited liability company, or limited liability partnership, Licensee's Managing Principal, shall be fully involved in the operation of the Licensed Business, providing hands-on leadership, oversight, and support to employees. The operation, growth, and success of the Licensed Business must be the Managing Principal's primary professional responsibility; the Managing Principal may serve in whichever full-time role benefits the business most. Licensee or Managing Principal must be available to engage and communicate with members of the Sylvan Corporate staff regarding the Licensed Business during the operational hours required by Sylvan. Licensee agrees that it shall comply with Sylvan's standards and requirements for staffing the Licensed Business, as designated in the Confidential Operations Manual, and any changes, additions, or modifications thereto.
- 7.2.2 <u>Accreditation</u>. Licensee agrees that if accreditation for supplemental education programs or services offered by Licensee is available from an accrediting body or organization, Licensee will at Sylvan's request comply with the standards necessary to attain and maintain such accreditation.
- 7.3 Advertising To Promote Sylvan System Services and SylvanSync System Services. Except for media forms whose circulation cannot be limited to Licensee's Territory, Licensee shall not, under any circumstances, place advertisements promoting and publicizing Sylvan System or SylvanSync System programs in another Licensee's territory or in the territory of a Sylvan Company-owned Center without Sylvan's prior written permission. For media forms whose circulation cannot be limited, Licensee shall conform to Sylvan's guidelines set forth in the Confidential Operations Manual or otherwise in writing.
- 7.3.1 Sylvan will make available to Licensee from time to time certain advertising materials which have been approved by Sylvan as being suitable for advertising and promotional purposes. Licensee shall not

make any changes to such pre-approved advertising materials without the advance written consent of Sylvan. Sylvan may subsequently withdraw its approval of any previously approved advertising materials in which event Licensee shall cease using such advertising within a reasonable period of time, which shall in no event exceed ninety (90) days.

- 7.3.2 Licensee shall not publish any advertising which has not been previously approved or provided by Sylvan. If Licensee wishes to publish any advertising which has not been previously approved by Sylvan, Licensee shall submit a copy of each new advertisement to Sylvan prior to the first publication of that advertisement, and Licensee shall not publish any advertising without the express written approval of Sylvan. Any advertisement submitted to Sylvan for approval shall be deemed approved if not expressly disapproved by Sylvan within twenty-one (21) calendar days after receipt by Sylvan. Failure to obtain prior approval constitutes a breach of this Agreement. Once an advertisement has been approved or deemed approved by Sylvan, Licensee may update Licensee's address, telephone number(s), fax number(s), web address(es), and operating hours on the advertisement from time to time without submitting same to Sylvan for approval, provided no material changes are made to the advertisement. Licensee shall not use the name of any public figure or celebrity in its advertising without Sylvan's prior written approval. Licensee's obligations hereunder also apply to advertising and marketing activities in which Licensee engages through any local or regional cooperative as further described in Paragraph 7.8.2.
- 7.3.3 In transmitting and causing any other party to transmit advertisements or solicitations by e-mail or other electronic media, Licensee shall comply with and shall cause other parties to comply with Sylvan's standards and policies and the Confidential Operations Manual.
- Answering Service and Response Times. In order to assure the quality of services Licensee offers under this Agreement, all telephone calls to the Licensed Business must be answered by a live person during the period from 08:00 a.m. to 08:00 p.m. local time in the Territory during at least six (6) days of each calendar week. The live person shall be either a member of Licensee's personnel or work through an answering service or contact center. In addition, Licensee shall ensure that all web and email inquiries to the Licensed Business are responded to in an appropriately personalized and courteous manner within one (1) business day but in no event longer than two (2) calendar days and via the communications channel(s) that Sylvan reasonably designates in the Confidential Operations Manual from time to time, which may include, as examples and not as limitations, email or voice/telephone. Licensee shall maintain at Licensee's expense a telephone answering function that is active during all times when the Licensed Business is closed. The required message for the answering function shall be designated in the Confidential Operations Manual or otherwise in writing. Sylvan has established a contact center program whereby Sylvan either elects to provide certain contact center services and/or designates a third-party vendor to provide certain contact center services to Licensee (the "Contact Center Program"). Commencing on the date the Center opens for operation and continuing until the one (1)-year anniversary thereof, Licensee is required to participate in the Contact Center Program and to pay the associated fees that Sylvan designates from time to time. Licensee must also enter into the form of agreement designated by Sylvan in connection with Licensee's participation in the Contact Center Program, and Sylvan's current form of "Contact Center Franchisee Participation Agreement" is attached to this License Agreement as Attachment F. Sylvan reserves the right to modify the Contact Center Program fees, and modify the Contact Center Program agreement, from time to time. All fees that Sylvan designates in connection with the Contact Center Program shall be paid by Licensee to Sylvan or a third-party, at Sylvan's option, at such time and in such manner as Sylvan designates.

#### 7.5 Sylvan Advertising.

- 7.5.1 <u>Licensee Advertising Advisory Committee</u>. Sylvan shall establish and administer a Licensee Advertising Advisory Committee (the "Committee") that shall consult with Sylvan concerning the selection and placement of Sylvan Advertising as defined in Paragraph 7.5.2. The Committee shall be composed of three (3) licensees appointed by Sylvan. Each licensee member shall serve for a one (1) year term corresponding to the calendar year. Any licensee member may be successively reappointed, and any vacancy shall be filled by prompt appointment by Sylvan. Each licensee member must be in good standing under all of its license agreements with Sylvan during such licensee member's term on the Committee; otherwise, Sylvan may remove such licensee member from the Committee. Sylvan shall meet with the Committee as appropriate, but Sylvan shall not be obligated to participate in more than four (4) such meetings per calendar year. All meetings shall be held by telephone or other communications channel designated by Sylvan.
- 7.5.2 <u>Advertising Allocation; Advertising, Marketing and Public Relations</u>. Subject to the other provisions of this Agreement, Sylvan has the right to direct some amount (the "Advertising Allocation") of Licensee's royalty toward funding advertising and marketing programs and activities intended to increase the public's awareness

of the Sylvan Network, the Sylvan System, and the SylvanSync System generally ("Sylvan Advertising"). Sylvan undertakes no obligation to ensure that any individual licensee benefits directly or on a pro rata basis from Sylvan Advertising. The amount and frequency of the Advertising Allocation shall be determined at Sylvan's sole discretion and are subject to change at any time. Sylvan may, in its sole discretion, modify, decrease, suspend, and/or increase expenditures and/or operations of Sylvan Advertising from time to time. Sylvan has sole discretion to determine what advertising, marketing, and public relations programs and activities Sylvan undertakes within Sylvan Advertising, and has sole discretion over the creative concepts, materials, and endorsements used in such Sylvan Advertising and the geographic, market, and media placement and allocation of advertising and marketing materials in connection with Sylvan Advertising. Licensee agrees to participate in all Sylvan Advertising activities and programs that Sylvan designates, provided that Licensee is not required to make expenditures outside of the Advertising Allocation and/or Licensee's local advertising expenditure requirements as set forth in and limited by this Agreement. Sylvan shall report to licensees annually concerning advertising, marketing, and public relations activities conducted and proposed for Sylvan Advertising and also concerning amounts expended on same. Sylvan has no obligation to audit the amounts collected in connection with the Advertising Allocation, however, it may choose to make unaudited financial statements regarding the Advertising Allocation available upon request.

- 7.5.3 Website. Sylvan shall arrange for maintenance of a Website for the benefit of Sylvan and the Sylvan brand and Sylvan Network generally as funded from Advertising Allocations. Provided that Licensee is not in default under this Agreement, Sylvan shall provide one or more references or webpage(s) to Licensee's Center, as Sylvan may periodically designate, within Sylvan's Website. Licensee shall ensure the accuracy and timeliness of all information in all such references and webpage(s) concerning Licensee's Center. (The term "Website" as used in this Agreement means one or more related documents, designs, pages, or other communications that can be accessed through electronic means, including but not limited to the Internet, World Wide Web, social networking sites (including but not limited to Facebook, X (formerly known as Twitter), LinkedIn, Google Wave, YouTube, etc.), blogs, vlogs, and other applications, etc.). Licensee agrees not to establish or use any Website or authorize any other party to establish a Website that relates in any manner to the Licensed Business or refers to the Licensed Marks except with Sylvan's prior written approval, which Sylvan may withhold in its sole discretion.
- 7.5.4 <u>Legacy Forms and Alternative Funding Formula</u>. Certain legacy license agreements provide for an "alternative funding formula" ("AFF") that sets the level of those licensees' spend from time to time on advertising of the type described in Paragraph 7.5.2. If an AFF is proposed to licensees in the United States pursuant to such legacy license agreements, Licensee shall be deemed to have agreed to such AFF with respect to this Agreement, and Licensee's Advertising Allocation to Sylvan Advertising shall be in the amount set forth in the AFF for the term of the AFF, if the AFF is approved by licensees owning at least two-thirds (2/3) of outstanding license agreements in the United States. No third party is a beneficiary to the provisions in this Paragraph 7.5.4. In all other cases, Sylvan will determine the Advertising Allocation amount as described in Paragraph 7.5.2.
- 7.6 <u>Licensee to Obtain Permits and Licenses</u>. Prior to commencing business operations, Licensee shall obtain all local and state permits, licenses, and certifications necessary to operate the Licensed Business. Licensee shall also comply at all times with all educational, safety, building, or other local, state, and federal statutes, ordinances, regulations, codes, licensing requirements, or standards applicable to the Licensed Business. Sylvan shall in no manner be liable for advising Licensee of such matters or determining Licensee's compliance therewith and Licensee will defend, indemnify, and hold Sylvan harmless from any claims or loss relating to same.
- System and SylvanSync System services at and from the Licensed Business and will use its primary and best efforts to cultivate, develop, and expand the market for same within the Territory consistent with Sylvan's evaluation of the potential of the Territory taking into consideration such factors as the size of population in and market potential of the Territory, competition in the marketplace, the prior performance of Licensee or other licensees in other geographic areas, and such other financial and market factors as Sylvan may deem pertinent. Licensee agrees that the Licensed Business shall be open and fully staffed during periods specified in the Confidential Operations Manual or otherwise by Sylvan in writing (subject to the limitations in the next sentence) to the extent that such hours do not violate the provisions of Licensee's premises lease. Each Center must be open and fully staffed for instruction for at least the "Minimum Instructional Hours," defined as not fewer than three (3) consecutive hours per day, five (5) days per week, of which one day must be a weekend day, and in no event shall Sylvan require any Center to be open in excess of the Minimum Instructional Hours during the term of this Agreement. Licensee acknowledges and agrees that the Minimum Instructional Hours are an absolute minimum and that Sylvan makes no guarantee that the Minimum Instructional Hours will result in any minimum performance, revenue, or outcome for the Licensed Business.

V.2024.051024

- 7.8 <u>Monthly Territory Advertising Minimum and New Primary Center Advertising Minimum; Local and Co-operative Advertising Expenditures.</u>
- 7.8.1 In order to promote its Licensed Business and the Sylvan System and SylvanSync System generally, Licensee agrees that it shall expend each month, on approved local advertising using the type of advertising specified in Paragraph 7.3 or on other marketing and advertising materials and activities approved by Sylvan, a "Monthly Territory Advertising Minimum" equal to eight percent (8%) of Licensee's prior month's gross revenues. Licensee shall expend not less than Ten Thousand Dollars (\$10,000) on advertising and marketing approved by Sylvan to promote the Primary Center during the three (3) month period beginning in the month before the Primary Center opens, and such expenditures shall be recognized toward the Monthly Territory Advertising Minimum for the month in which they are expended and in the subsequent month(s) until the total expenditure has been so recognized; this Primary Center expenditure obligation does not apply to relocations. Licensee shall keep appropriate records and receipts for all advertising or marketing expenditures for at least three years, and shall provide substantiating data for such expenditures upon request by Sylvan. Notwithstanding the foregoing, if and at such time as Licensee has expended Forty Thousand Dollars (\$40,000) per Territory in a given calendar year on local advertising under this Paragraph approved by Sylvan, then Licensee's Monthly Territory Advertising Minimum obligations automatically shall be waived for the remainder of such calendar year.
- 7.8.2 Upon Sylvan's request, Licensee shall be required to contribute up to one hundred percent (100%) of the amounts provided in Paragraph 7.8.1 to a local or regional advertising cooperative comprising a geographic region and such licensee operated licensed businesses and Sylvan operated licensed businesses therein as Sylvan may designate in its sole judgment. For absence of doubt, the parties agree that a cooperative may consist of a single licensed business or more than one licensed business. At Sylvan's request, a representative of Licensee shall participate in the administration of such advertising cooperative. All advertising and marketing decisions of any such cooperative, including without limitation, content and media placement, shall be subject to Sylvan's prior approval and shall comply with the terms and conditions of this Agreement, including without limitation Paragraph 7.3. Licensee's payment obligations to such cooperative shall be offset by Licensee's verified New Center Advertising expenditures described in Paragraph 7.8.1 of this Agreement.
- 7.9 Quality Assurance. In order to enhance the public image and reputation of the Sylvan System and the SylvanSync System and maintain and ensure the quality and consistency of services provided thereunder, Sylvan may, without notice, perform a Quality Assurance Review either remotely or by sending a Sylvan representative or other designated representative to the Licensed Business or any part thereof, and Licensee shall allow Sylvan's and Sylvan's representative to observe Licensee's operation and methods. Licensee will fully cooperate with such representative and will make available any data concerning the Licensed Business and its operation as reasonably requested by such representative. For each Quality Assurance Review, Sylvan's representative will produce and deliver to Licensee a Quality Assurance Review document which will describe the results of the Review and include a Correction Plan setting forth any material deficiencies observed and recommendations relating thereto. Licensee will, within the time limits set forth in such Correction Plan, comply with and fully implement any corrective action designated therein as mandatory or required, and will consider in good faith any other recommendations or suggestions. The representative will endeavor to minimize the disruption of the educational services being provided by Licensee while the representative is present in the Licensed Business.
- 7.9.1 Licensee shall maintain the Licensed Business locations, the equipment in the Licensed Business (including, without limitation, computers, computer and other software, telephones, Internet, telecommunications or other communications equipment, utilities, interior signage, fixtures, furniture, furnishings, décor, and interior), and the signs and exterior of the Licensed Business locations in good condition and repair continually during the term of this Agreement. Licensee agrees to adhere to Sylvan's standards, specifications and procedures for signs, fixtures, furnishings, carpeting, interior paint, equipment, color schemes and trade dress for locations in the Licensed Business as described in the Confidential Operations Manual or otherwise by Sylvan in writing from time to time.
- 7.9.2 Notwithstanding Paragraph 7.9.1 above, in the event that Sylvan finds Licensee in default for failure to comply with the provisions of Paragraph 7.9.1, Licensee may seek suspension of such default if Licensee demonstrates that the Sylvan licensed businesses in general owned by Sylvan and its affiliates fail at that time to meet the provisions of Paragraph 7.9.1. In interpreting this Paragraph 7.9.2, the Sylvan licensed businesses owned by Sylvan

V.2024.051024

and its affiliates shall be considered generally so that Licensee may not seek suspension of its obligations under Paragraph 7.9.1 by comparing the Licensed Business to any single Sylvan licensed business or group of Sylvan licensed businesses not constituting all of the Sylvan licensed businesses in general owned by Sylvan and its affiliates.

- 7.9.3 Sylvan shall have the sole right to add to, modify, or delete from the standards, specifications, policies, and procedures for signs, furnishings, and equipment for the Licensed Business to conform to the trade dress, color schemes, and presentations of the marks in a manner consistent with the then-current image of Sylvan licensed businesses. If Sylvan modifies, adds to, or deletes from its standards, specifications, and procedures, Licensee agrees to update, refurbish, and improve the Licensed Business as Sylvan may require as soon as reasonably possible, or discontinue the use of equipment, furnishings, or equipment that Sylvan specifies.
- 7.9.4 Maintenance, refurbishment, modification, and upgrading of the Licensed Business shall be at Licensee's expense; provided, however that Sylvan shall not require Licensee to expend sums on such modifications, upgrades, additions, or deletions in excess of the following amounts:
- (a) Five thousand dollars (\$5,000) per Center, not to exceed Ten Thousand dollars (\$10,000) per Territory, annually for the Computer System, excluding licensing fees;
- (b) In addition to its maintenance obligations set forth in Paragraph 7.1 and 7.9.1, Five Thousand dollars (\$5,000) per Center, not to exceed Ten Thousand dollars (\$10,000) per Territory, annually for refurbishing and upgrading fixtures, furnishings, décor, and trade dress, excluding costs and expenses required under Paragraph 7.9.4(a) and (c);
- (c) Once during the term of this Agreement, but not before the third (3<sup>rd</sup>) anniversary of the Effective Date hereof, and once during each renewal term, Seventy-Five Thousand dollars (\$75,000) total for the Territory, which may be divided among any number of Centers in the Territory or allocated to one Center, in Sylvan's discretion, toward bringing the Center(s) into compliance with the then-current standards for center design, layout floor plan, furnishings, signage, and trade dress ("Center Remodel"). Expenses for one relocation in each term shall count toward the expenditure cap under (c) in that term.

The expenditures required in (a) and (b) above shall not exceed thirty-six thousand dollars (\$36,000) minus expenditures on mandatory Sylvan System programs, enhancements, and updates, up to the maximum set forth in Paragraph 7.11 of this Agreement, in any three (3) year period. Each year's required expenditure cap under this Paragraph 7.9.4 shall be unique to that year and shall not be affected by expenses incurred in any other year. Licensee's voluntary (that is, not in response to a direction by Sylvan) expenditures in (a), (b), and (c) above shall count toward the maximums herein to the extent that such purchases further and do not diminish the Center's overall compliance with Sylvan's standards. No expenditure shall entitle Licensee to offset or fail to meet any other obligation pursuant to this Agreement, including without limitation obligations to meet operational requirements.

- 7.10 <u>Attendance at Conferences</u>. The Licensee shall attend national and regional conferences, and Licensee's Sylvan-certified personnel shall attend regional conferences designated as mandatory or required in the Confidential Operations Manual or otherwise by Sylvan. If Licensee is a partnership, corporation, or other business entity, the requirements of this paragraph will be met if the national conference is attended by Licensee's Managing Principal. Licensee must bear all expenses of attending such conferences, including but not limited to registration and/or attendance fees, travel, lodging, meals, and entertainment. All current registration and/or attendance fees paid to Sylvan shall be indicated in the Confidential Operations Manual or other written directive.
- 7.11 Compliance with Educational Standards; Mandatory Programs and Training. Sylvan has established certain defined standards in teaching methods and procedures for the delivery of its educational and instructional programs and will offer help and assistance to the Licensee in attaining and maintaining those standards. Licensee shall comply with all educational standards designated as required or mandatory by Sylvan in its Confidential Operations Manual or otherwise in writing. From time to time, Sylvan may develop or have developed for it new or enhanced programs or modules designated for use in the Sylvan System or the SylvanSync System and shall make such programs or modules available to Licensee as either mandatory or optional items. In the event a Sylvan System item is designated as mandatory, Sylvan may not require implementation of that item in the Licensed Business earlier than one hundred eighty (180) days after the item is generally made available to Sylvan's licensees. In the event that a SylvanSync System item is designated as mandatory, Sylvan may not require implementation of same earlier than thirty (30) days after the item is generally made available to Sylvan's licensees. Mandatory Sylvan System and SylvanSync System programs and enhancements or updates to same shall be provided to Licensee at Sylvan's cost. Licensee shall be required to purchase, implement, and make available to its customers each such mandatory program,

enhancement, or update to mandatory program no later than the required implementation date for each as determined by Sylvan in accordance with the provisions of this paragraph. Optional Sylvan System and SylvanSync System programs and updates to same will be made available to Licensee for such charges as Sylvan shall establish. Licensee acknowledges and agrees that Sylvan may require Licensee, Managing Principal, and/or Licensee's personnel to successfully complete designated training and/or obtain designated certifications before releasing programs, updates, or modules to Licensee's failure to complete such requirements or cause personnel to complete such requirements (provided that the training is made available to Licensee) shall not constitute a defense of Licensee's failure to obtain and implement any mandatory program or module on or before the required implementation date. For any calendar year in which Licensee is required to expend more than Five Thousand Dollars (\$5,000) per Center, not to exceed Ten Thousand Dollars (\$10,000) per Territory, (which maximum shall be adjusted each calendar year based upon the cost of living adjustment in Paragraph 6.10 of this Agreement) altogether on mandatory Sylvan System and/or SylvanSync System programs required to be implemented beginning in that year, Licensee shall be permitted to extend payment of the amount in excess of such maximum to the following calendar year(s) for so long as Licensee's obligations in any given year exceed the maximum. Expenditures on mandatory programs, enhancements or updates purchased and implemented after their required implementation date has already passed shall not count toward the maximums in this Paragraph 7.11 and in Paragraph 7.9.4. As set forth in Paragraph 4.2.8, Sylvan shall make available to Licensee at no charge each assessment that Sylvan deems mandatory in any mandatory Sylvan System or SylvanSync System program, and Licensee may use such assessment at no charge in all Sylvan System and SylvanSync System programs, whether optional or mandatory, in which Sylvan approves use of such assessment. Optional assessments in the Sylvan System and SylvanSync System will be made available to Licensee for such charges as Sylvan shall establish.

#### 7.12 <u>Customer Information</u>.

- 7.12.1 All information that Licensee collects, receives, or otherwise possesses relating to past, current, future, or prospective customers is referred to in this Agreement as "Customer Information." Sylvan may periodically establish policies respecting the collection, maintenance, storage, and use of the Customer Information by Sylvan franchisees, including Licensee, and Licensee agrees to comply with such policies. Licensee shall input and maintain in the Computer System all Customer Information which is capable of being so stored.
- 7.12.2 Licensee may not use the Customer Information for any purpose other than in the normal conduct of the Licensed Business. Licensee may not sell, loan, give, or otherwise disclose any Customer Information, including customer lists, to any party without Sylvan's prior written permission except in the normal conduct of the Licensed Business. Licensee must use commercially reasonable efforts to keep the Customer Information secure at all times.
- 7.12.3 Sylvan may have independent access to the Customer Information via connection to the Computer System. Nevertheless, upon Sylvan's request, Licensee shall promptly provide Sylvan with a copy of all Customer Information in electronic or other format Sylvan designates.
- 7.12.4 Upon expiration or termination of this Agreement (except solely for termination as a result of an uncured default by Sylvan hereunder), Licensee shall send to Sylvan a true, accurate, and complete copy of all Customer Information in electronic or other format designated by Sylvan. Licensee acknowledges that, upon expiration or termination of this Agreement (except solely for termination of this Agreement as a result of an uncured default by Sylvan hereunder), Licensee shall no longer have any right to use, and Licensee shall not use, the Customer Information for any purpose whatsoever.
- 7.12.5 Both during and after the term of this Agreement, Sylvan may use, and authorize others to use, the Customer Information only for purposes that, in Sylvan's exclusive business judgment, will benefit the Sylvan Network or the Sylvan brand. Except for the purposes set forth in the foregoing sentence, Sylvan shall not share Customer Information with any third party, including without limitation persons or entities that, even though affiliated with Sylvan, do not operate under, or perform functions in support of, the Sylvan Network or the Sylvan brand. Before using or authorizing others to use the Customer Information for any purpose other than data analysis, research, or maintenance, Sylvan shall advise the Licensee Advertising Advisory Committee as it exists at such time, or a successor to the Licensee Advertising Advisory Committee, of how and when the Customer Information is proposed to be used or to be authorized to be used. Sylvan shall not share with any other Sylvan franchisee any segment of Customer Information that personally identifies customers of the Licensed Business, except with Licensee's written consent, and shall not require Licensee to do so except in the normal course of facilitating the instruction of a transferring student.

- 7.12.6 Neither Sylvan nor Licensee shall use the Customer Information in any manner that would violate any applicable laws governing the privacy rights of any customer, and their use of Customer Information is subject to the provisions of Paragraph 8.3 concerning privacy. In the event either Sylvan or Licensee provides any Customer Information to a third party, the furnishing party shall require the receiving third party to also comply with all applicable laws governing the privacy rights of any customer.
- 7.13 Membership in FOA during the Initial Period. The Sylvan Learning Center Franchise Owners Association ("FOA") is an organization of Sylvan licensees that is administered by The Sylvan Franchise Owners Association, Inc., a Maryland corporation. Licensee shall become a member of the FOA on the Effective Date, and shall maintain Licensee's FOA membership during the "Initial Period," which is the remainder of the calendar year of the Effective Date plus the subsequent calendar year. Licensee's membership in the FOA after the Initial Period is optional. The FOA collects annual membership dues from its members. FOA annual membership dues are based on the number of Centers a licensee is licensed to operate. Licensee is responsible for Licensee's FOA membership costs, dues, fees, and activities. Licensee acknowledges and agrees that Sylvan does not own or control any part of the FOA or of The Sylvan Franchise Owners Association, Inc. In the event that fewer than twenty-five percent (25%) of active Sylvan licensee's obligations in this Paragraph shall not apply during the subsequent calendar year.
- 7.14 Purpose of Standards, Training, and Assistance. Sylvan and its representatives provide standards, training, and assistance to Licensee and Licensee's employees for the sole purpose of helping Licensee to properly utilize the Sylvan System and the SylvanSync System in conducting the Licensed Business. It is understood and agreed that standards, training, and assistance from Sylvan do not constitute or create any employer/employee relationship or joint employer relationship with Licensee or Licensee's employees. Licensee is solely responsible for all employment matters in connection with the Licensee's business, including without limitation all hiring, termination, supervision, compensation, and discipline matters.

# 8. <u>Technology.</u>

- 8.1 <u>Computer System.</u> Sylvan has the right to specify certain information and communications technologies, equipment, systems, and formats, including, but not limited, to electronic personal computers, networks and network configurations, software ("Required Software"), hardware, data storage, retrieval, and transmission systems, and internet access mode and speed, and minimum operating specifications for same, that Licensee must use in connection with its operation of the Licensed Business and communications with Sylvan and others. Collectively, all of the items of hardware, software, firmware, systems, and other elements referred to above in this Paragraph 8.1 are referred to in this Agreement as the "Computer System." Licensee agrees to comply with Sylvan's requirements with respect to the Computer System. In this regard:
- 8.1.1 Sylvan has the sole right to determine and designate which elements and/or components of the Computer System are mandatory, and which are optional.
- 8.1.2 Licensee agrees to install and use the Computer System and Required Software in the manner that Sylvan requires.
- 8.1.3 Subject to the cost protections provided in Paragraph 7.9.4(a), Licensee agrees to implement and periodically upgrade, update, and make other changes to the Computer System and Required Software as Sylvan may request in writing (collectively, "Computer Upgrades"). Sylvan may, but is not obligated to, implement such required Computer Upgrades for or on behalf of Licensee, and Licensee shall reimburse Sylvan for Sylvan's costs and expenses, under the following circumstances: (i) in the event that Licensee does not meet any obligation concerning Computer Upgrades within the time period specified by Sylvan; and (ii) for Computer Upgrades that can be remotely performed by Sylvan.
- 8.1.4 Licensee agrees to comply with Sylvan's written specifications (whether in the Confidential Operations Manual or otherwise) with respect to the Computer System and the Required Software, and with respect to Computer Upgrades, at Licensee's own expense subject to the maximums set forth in Paragraphs 7.11 and 7.9.4 of this Agreement.
- 8.1.5 Licensee agrees to afford Sylvan unimpeded access to Licensee's Computer System and Required Software in the manner, form, and at the times set forth in this Agreement and the Confidential Operations Manual and otherwise upon Sylvan's request.

- 8.1.6 Licensee shall establish and maintain Internet access and an electronic mail address and shall keep Sylvan informed of such address and all changes thereto during the term of this Agreement. Licensee shall use such electronic addresses or addresses and associated mailboxes that Sylvan specifies from time to time.
- 8.1.7 To the extent that any hardware ("Sylvan Hardware") or software ("Sylvan Software") for use within the Computer System is the property of Sylvan and is loaned or licensed by Sylvan to Licensee:
- (a) Licensee shall maintain the Sylvan Hardware in good condition, normal wear and tear excepted. Licensee shall be responsible for loss of, or damage to, the Sylvan Hardware due to fire, theft, or other catastrophe, and agrees to keep the Sylvan Hardware insured against such catastrophic loss. Upon Sylvan's request, Licensee shall provide Sylvan with a certificate evidencing such insurance coverage;
- (b) Sylvan shall provide certain maintenance and support services for the Sylvan Hardware and for designated software loaded onto the Sylvan Hardware. The definition, scope, nature, and extent of the support services will be set forth in a written or electronic service level agreement, which agreement may be modified and updated by Sylvan from time to time. Sylvan agrees to be bound by its obligations under such service level agreement.
- (c) <u>No Warranties</u>. LICENSEE UNDERSTANDS AND ACKNOWLEDGES THAT THE SYLVAN HARDWARE AND SYLVAN SOFTWARE IS PROVIDED TO LICENSEE "AS IS." SYLVAN DISCLAIMS ANY AND ALL WARRANTIES AND GUARANTEES RELATED TO THE SYLVAN HARDWARE OR SYLVAN SOFTWARE, INCLUDING WITHOUT LIMITATION THE WARRANTY OF MERCHANTABILITY AND THE WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE.
  - (d) Such Sylvan Hardware and Sylvan Software shall remain the property of Sylvan.
- 8.2 <u>Data and Reporting.</u> Sylvan has the right to periodically specify in writing, in the Confidential Operations Manual or otherwise, the information that Licensee must collect and maintain on the Computer System, including, without limitation, the Customer Information as defined in Section 7.12, and Licensee agrees to provide Sylvan with the reports that Sylvan may reasonably request from the data so collected and maintained. Licensee agrees to: (a) upload to Sylvan daily (or at such other intervals that Sylvan may reasonably require) all information and materials that Sylvan may require in connection with Licensee's operation of the Licensed Business; and (b) permit Sylvan access to download information from Licensee.
- 8.3 <u>Privacy Laws; Laws Pertaining to E-Mails</u>. Licensee and Sylvan agree to abide by all applicable laws pertaining to the privacy of consumer, employee, and transactional information ("Privacy Laws").
- 8.3.1 Licensee agrees to comply with Sylvan's standards and policies pertaining to Privacy Laws. If there is a conflict between Sylvan's standards and policies pertaining to Privacy Laws and actual applicable law, Licensee shall: (i) comply with the requirements of applicable law; (ii) immediately give Sylvan written notice of said conflict, identifying the applicable law; and (iii) promptly and fully cooperate with Sylvan and Sylvan's counsel in determining the most effective way, if possible, to meet Sylvan's standards and policies pertaining to Privacy Laws within the bounds of applicable law.
- 8.3.2 Licensee agrees not to publish, disseminate, implement, revise, or rescind a data privacy policy without Sylvan's prior written consent as to said policy.
- 8.3.3 In addition to any other provision of this Agreement, each party shall be responsible for its respective compliance with any laws pertaining to sending e-mails, including but not limited to the Controlling the Assault of Non-Solicited Pornography and Proprietary Marketing Act of 2003 (known as the "CAN-SPAM Act of 2003").
- 8.4 <u>License Agreements Concerning Websites</u>. If Sylvan ever does approve in writing a request for Licensee to use a separate Website (but by including these terms, Sylvan is not deemed to have promised or represented that it will approve a separate Website for Licensee), then Sylvan has the right to require that Licensee meet any or all of the following requirements:
- 8.4.1 Licensee agrees that any separate Website that it owns or that is maintained for its benefit shall be deemed "advertising" under this Agreement and will be subject to, among other things, the terms and conditions of this Agreement, including, without limitation, Paragraph 7.3 above, and Sylvan's prior written approval.

- 8.4.2 Licensee agrees that, before establishing any Website, Licensee shall submit to Sylvan, for Sylvan's prior written approval, a sample of the proposed Website domain name, format, visible content (including, without limitation, proposed screen shots), and non-visible content (including, without limitation, meta data and meta tags) in the form and manner Sylvan may reasonably require.
- 8.4.3 Licensee agrees not to use or modify any such Website without Sylvan's prior written approval as to such proposed use or modification.
- 8.4.4 Licensee agrees to comply with Sylvan's written standards and specifications for Websites, as set forth in the Confidential Operations Manual and otherwise, in addition to any other applicable requirements. Such standards and specifications may concern, as an example and not as a limitation, Licensee's use of social media.
- 8.5 <u>Use of Licensed Marks</u>. Licensee agrees not to use or permit the use or display of Licensed Marks as part of any Internet domain name or website, or any other electronic identifier (including but not limited to e-mail addresses, account names in a social media site, and the like) of Licensee or the Licensed Business in any forum or medium, except with Sylvan's prior written approval and in compliance with Sylvan's policies concerning same as set forth from time to time.
- 8.6 <u>Changes to Technology</u>. Because changes to technology are dynamic and not predictable within the term of this Agreement, and in order to provide for inevitable but unpredictable changes to technological needs and opportunities, Licensee agrees: (a) that Sylvan will have the right to establish, in writing, new standards to address new technologies, whether published in the Confidential Operations Manual or otherwise in writing, and that Sylvan has the right to implement those changes in technology into the Computer System and to add or modify fees accordingly, subject to the maximum required expenditures identified in this Agreement; and (b) to implement upgrades and updates to the Computer System and all other technology aspects of the Licensed Business at such times, and in the manner, specified by Sylvan.

# 9. <u>Transfer and Assignment.</u>

- 9.1 <u>Transfer or Assignment by Sylvan</u>. This Agreement and all rights and duties of Licensor hereunder may be freely assigned or transferred by Sylvan at its sole discretion to any person or legal entity which agrees to assume Sylvan's obligations hereunder, and shall be binding upon and inure to the benefit of Sylvan's successors and assigns including, without limitation, any entity which acquires all or a portion of the capital stock of Sylvan or any entity resulting from or participating in a merger, consolidation, or reorganization in which Sylvan is involved, and to which Sylvan's rights and duties hereunder are assigned or transferred.
- 9.2 <u>Licensee's Obligations Are Personal; Approval of Transferees.</u> Licensee understands and acknowledges that the rights and duties created by this Agreement are personal to Licensee, and that Sylvan has granted the Licenses in reliance on many factors, including, without limitation, the individual or collective character, skill, aptitude, and business and financial capacity of Licensee and Licensee's principals named on the signature page hereof. Accordingly, neither Licensee nor any person named on the signature page hereof shall, without Sylvan's prior written consent, which Sylvan may condition on such procedures as Sylvan may require, directly or indirectly sell, assign, transfer, convey, give away, pledge, mortgage, or otherwise encumber any interest in the Licenses, this Agreement or any portion or aspect thereof, or any equity or voting interest in Licensee (each a "Transfer"). Any purported Transfer occurring by operation of law or otherwise, including any assignment by a trustee in bankruptcy, without Sylvan's prior written consent shall be a material default of this Agreement. Any purchase and sale agreement entered into by Licensee for any Transfer must provide for and require that the Center must continue to operate without interruption during the Transfer.
- 9.3 Approval for Transfer of Majority Interest. Prior to Licensee effecting any Transfer hereunder, the effect of which would be to reduce the cumulative percentage interest of all persons named on the signature page to less than fifty-one percent (51%), Licensee must first have i) complied with all provisions and met all requirements of this Section 9, this Paragraph 9.3, and Paragraphs 9.3.1 through 9.3.10 hereto, and ii) obtained Sylvan's conditional approval of the proposed Transfer subject to compliance with this Section 9. Provided that Licensee and the proposed transferee comply with the provisions of this Section 9, Sylvan shall not unreasonably withhold its consent to a Transfer by Licensee of any interest in this License, this Agreement, or any equity or voting interest in Licensee (including, without limitation, any direct or indirect interest, such as by management contract, lease, proxy for voting of stock, or otherwise).

- 9.3.1 The proposed transferee shall demonstrate to Sylvan's sole satisfaction that it meets all of Sylvan's requirements for becoming a licensee, including, without limitation, that it meets Sylvan's managerial, financial, and business standards then in effect; meets the financial requirements Sylvan shall deem appropriate in light of the terms and conditions of the proposed transfer; possesses a good moral character, business reputation, and satisfactory credit rating; will comply with all instruction and training requirements of Sylvan; and has the aptitude and ability to operate the Licensed Business (as may be evidenced by prior related business experience or otherwise).
- 9.3.2 As of the effective date of the proposed Transfer, all obligations of Licensee under this Agreement and under any other agreements between Licensee and Sylvan, its subsidiaries and affiliates are fully satisfied. Licensee shall have paid all outstanding debts and obligations to Sylvan, its subsidiaries, and affiliates, and have brought current all accounts or balances Licensee may have with any advertising cooperative of which Licensee is a member.
- 9.3.3 As of the effective date of the proposed Transfer, all obligations of the proposed transferee to Sylvan, its subsidiaries, and affiliates must be fully satisfied.
- 9.3.4 Sylvan shall provide the prospective transferee with Sylvan's then-current form of disclosure document required by the Federal Trade Commission's Trade Regulation Rule on Franchising and/or other applicable state franchise registration/disclosure laws, and receipt for same shall be delivered to Sylvan, provided, however, Sylvan shall not be liable for any representations other than those contained in such disclosure document. Sylvan shall provide a copy of said disclosure document to Licensee upon request; provided, however, that in complying with Licensee's request, Sylvan shall not be required to register any disclosure document not then duly registered and approved by the authority having jurisdiction over said process. The transferee may be required to execute a disclosure letter concerning the Transfer and a waiver and release by the transferee of Sylvan for any amount paid to or representation made by Licensee.
- 9.3.5 If Sylvan is then offering license agreements to new licensees, the proposed transferee must execute such documents as are then required by Sylvan including Sylvan's then-current form of license or franchise agreement (or equivalent) which may contain terms and conditions different from those in this Agreement. If Sylvan is not then offering License Agreements to new licensees, Licensee must execute Sylvan's most recent form of License Agreement offered to new licensees, which may contain terms, conditions, and fees substantially different from those in this Agreement. Further, upon Transfer, when the transferee executes Sylvan's then-current form of Agreement, the transferee shall obtain the remaining term under this Agreement and not a new 10-year term.
- 9.3.6 Licensee shall pay to Sylvan a transfer fee in the amount of 50% of the then-current initial license fee for the proposed transfer if the proposed transferee is not currently a qualified Sylvan licensee, or 25% of the then-current initial license fee for a proposed transfer to a licensee with a Sylvan License Agreement in effect at the time of the Transfer; provided, however, to qualify for such reduced transfer fee, such qualified Sylvan proposed licensee must have (1) completed the initial training requirements, and (2) operated a separate Sylvan Learning Center for at least six (6) months prior to the date of the Transfer. The transferee agrees to complete, to Sylvan's satisfaction, the training required by Sylvan, and agrees to pay the fee that Sylvan designates for such training (the "Supplemental Training and Administrative Fee").
- 9.3.7 Sylvan may perform a Quality Assurance Review with respect to the Licensed Business to be transferred and will make the results thereof available to Licensee and the proposed transferee, provided, however, that if Sylvan has not performed a Quality Assurance Review within one (1) year of the proposed transfer date, Sylvan shall perform such review. Thereafter, Licensee shall prepare and deliver to Sylvan, prior to the transfer, a written statement specifying how any deficiencies set forth in the Review will be corrected, and whether the expense thereof will be borne by Licensee or the proposed transferee; provided, however, that Sylvan assumes no liability or responsibility to Licensee or the proposed transferee with respect thereto.
- 9.3.8 Licensee agrees in writing to transfer to the proposed transferee all books, programs, materials, and other items reasonably necessary to operate the Licensed Business, as set forth in the Quality Assurance Review referenced in Paragraph 9.3.7 above.
- 9.3.9 Licensee shall have executed a general release under seal where required, in the thencurrent form satisfactory to Sylvan, of any and all claims against Sylvan, its current and former affiliates, and their respective past and present owners, officers, directors, agents, representatives, heirs, administrators, successors, and assigns, including, without limitation, claims arising under federal, state, and local laws, rules, and ordinances arising out of, or connected with, the performance of this Agreement.

- 9.3.10 Sylvan's consent to a transfer of any interest in the Licenses granted herein shall not constitute a waiver of any claims it may have against the transferring party, nor shall it be deemed a waiver of Sylvan's right to demand exact compliance with any of the terms of this Agreement by the transferee.
- 9.3.11 If Licensee elects to participate in Sylvan's resale program in connection with the transfer of the Licensed Business pursuant to this Section 9.3., Licensee must comply with Franchisor's then-current resale program requirements, which may include the execution of Franchisor's then-current resale program agreement and payment of the then-current resale program administration fee and other fees.
- 9.3.12 At Franchisor's sole discretion, Franchisor may require the Territory be split into two or more smaller protected areas (each a "Fractional Area"), each of which shall meet the then-current standards for a Territory set forth in Item 12 of the franchise disclosure document that existed as of the Effective Date of this Agreement. The Fractional Area that contains the Primary Center shall serve as the new Territory (or similarly described territory) in the transferee's franchise agreement and shall be the only territory transferred under the purchase agreement between the Licensee and transferee. Sylvan, at its sole discretion, may offer the remaining Fractional Areas to transferee or Licensee under a separate then-current license agreements, if Licensee or transferee meet Sylvan's then-current standards for multi-unit development; if transferee or Licensee undertakes development of the remaining Fractional Areas, they shall pay the Territory Split Fee for each Fractional Area in lieu of an initial license fee. If the Licensee's Territory is split into a Fractional Area, then i) the Licensee shall not be required by pay a Transfer Fee with respect to the transfer that resulted in the Fractional Area and ii) Sylvan will provide the purchasing franchisee a right of first refusal to purchase the newly created Fractional Areas not subject of their license agreement. The term of such right of first refusal will commence upon closing of the Transfer and shall expire ninety days later.
- 9.4 <u>Record of Ownership.</u> Without regard to whether Sylvan's consent to the transfer is or may be required, Licensee shall submit to Sylvan prior to any proposed transfer of any equity or voting interest in Licensee, and at any other time upon request, a list of all holders of direct or indirect equity and voting interests of record reflecting their respective present and/or proposed direct or indirect interests in Licensee, in such form as Sylvan may require.
- Rights of Licensee's Heirs or Personal Representative upon the Death, Absence, or Incapacity of Licensee. This Agreement shall inure to and bind the respective heirs, executors, administrators, or successors of Licensee. In the event of death or incapacity of the Licensee or the death or absence from active participation in the business of any individual upon whose ability or experience Sylvan relied in granting the licenses of Paragraphs 1.1 and 1.2, Sylvan shall have the right, upon written notice to Licensee (or heirs), to assist the Licensee in providing for an orderly transition in finding a satisfactory substitute to assume control of the Licensed Business. Mutual agreement shall be reached in the selection of the individual to assume control of the Licensed Business. Failure of Licensee to provide an agreeable substitute shall be considered a breach hereof and cause for termination, unless cured as provided herein. During any period in which the Licensed Business is not being operated as required herein by a person upon whose ability or experience Sylvan relied in granting the licensee, Sylvan shall have the right to enter upon the Licensee's premises to manage the Licensed Business on an interim basis in return for a management fee to be paid to Sylvan as invoiced by Sylvan. The heirs or personal representatives of Licensee have the same rights and obligations as Licensee if they desire to sell or transfer the License Agreement.
- Right of First Refusal. Before any Transfer described in Paragraph 9.2 or 9.3 above, other than a Transfer to an existing holder of a minority interest in Licensee which results in such transferee holding a minority interest in Licensee, and other than a transfer to an heir of the first degree of the holder of a majority interest in Licensee, to be exercised once only, Licensee must advise Sylvan in writing of the proposed transfer, including the name and background experience of the proposed transferee and all terms of the proposal set forth in a binding agreement. Sylvan shall have the right of first refusal, to be exercised within thirty (30) days of Sylvan's receipt of Licensee's written notice, to acquire such interest on the identical bona fide terms proposed by the transferee, or, at Sylvan's option, for the cash equivalent thereof. Licensee may not agree to a closing date for the transfer that is less than forty-five (45) days after Sylvan's receipt of Licensee's notice. If Sylvan exercises its right of first refusal, Licensee agrees to assign to Sylvan all of the rights Licensee granted to the transferee under the acquisition documents, and Sylvan agrees to assume transferee's obligations thereunder. In the event Sylvan does not elect to exercise its right of first refusal as provided, the other conditions of a sale, assignment or transfer stated herein must nevertheless be met, and Licensee may conclude the proposed transfer, but only if such transaction is closed within sixty (60) days of Sylvan's election not to purchase, and only if the transaction closes on the same terms and conditions as was advised

to Sylvan. In the event the transaction is not so closed, Licensee must again comply with all of the terms of this right of first refusal before transferring to the proposed transferee or any other transferee.

9.7 <u>Pledge of the License Agreement</u>. The restrictions of Paragraph 9.2 to the contrary notwithstanding, Licensee may, from time to time, pledge its rights under this Agreement as collateral for the benefit of a bank or other commercial lender, but only with the prior written consent of Sylvan, which consent may be conditioned upon the requirement that the security agreement or other document evidencing the pledge or granting of a security interest in the License shall contain the following language:

"The sale, transfer or assignment of the security interest created hereunder in and to Sylvan License Agreement No.\_\_\_\_\_\_\_, dated \_\_\_\_\_\_\_, (the "franchise") together with the rights created hereunder or by law authorizing the Secured Party to repossess or foreclose, and subsequently retain or sell or dispose of said franchise, shall be subject to all those restrictions contained in Section 9 and Section 11 of said License Agreement.

The referenced provisions of the License Agreement are incorporated herein by reference to the same extent as if fully set forth herein at length, such that any sale, transfer or assignment of the security interest created hereunder, or any sale of the franchise, the licensed rights or interest of Licensee under said License Agreement pursuant to or following foreclosure or repossession under this security agreement, shall be void if made without the express written consent of the Licensor under the franchise, pursuant to said License Agreement provisions."

Any purported assignment of the interest of the secured party, or foreclosure of the security interest, by operation of law or otherwise, without Sylvan's written consent shall not be binding upon Sylvan and shall be a material default of this Agreement. Unless otherwise agreed to in writing, Sylvan shall not be obligated to provide any notice required to be provided to Licensee under this Agreement to any secured party or pledgee claiming any interest in this Agreement, under this paragraph or otherwise.

#### 10. Term and Renewal.

- 10.1 <u>Expiration Date</u>. The initial term of this Agreement shall begin on the Effective Date indicated on the Summary Page and shall expire at midnight on the date ten (10) years thereafter. Notwithstanding the foregoing, nothing contained in this paragraph will limit Sylvan's termination rights set forth in Paragraph 11 of this Agreement
- 10.2 <u>Renewal of License</u>. Subject to this Paragraph 10.2, Licensee may serially renew the licenses granted pursuant to this Agreement for two additional, consecutive five (5)-year renewal terms. Without waiving its rights not to renew as provided in this paragraph, Sylvan will notify Licensee of the upcoming expiration date, and the procedures for securing approval of renewal, at least two hundred seventy (270) days prior to the expiration date. Renewal will be granted provided that Licensee meets the following conditions:
- 10.2.1 Licensee is in full compliance with all conditions and terms of this Agreement at the time of renewal.
- 10.2.2 Licensee has, during the last three years of the term of this Agreement, never failed to timely cure any default as to which Licensee was given a Notice of Default.
- 10.2.3 Licensee has never, during the term of this Agreement, committed a material breach of this Agreement which cannot be cured and as to which Licensee was given a Notice of Default.
- 10.2.4 Licensee gives Sylvan notice of its intention to renew in the last year of the ten (10) year term of the Agreement, but not less than one hundred eighty (180) days prior to the scheduled final day of the term. If Licensee fails to give such notice or to so agree, this Agreement shall expire on the expiration date of the term without further notice to Licensee.
- 10.2.5 Licensee has complied with Sylvan's requirements for additional performances or obligations of Licensee other than those set forth herein such as correction of deficiencies, modernization or updating, which Sylvan believes in its good faith judgment shall be required and which may require significant additional capital expenditures by Licensee.

- 10.2.6 Licensee has paid Sylvan a renewal fee of Six Thousand Dollars (\$6,000). If Sylvan is then offering license agreements to new licensees, then Licensee must execute Sylvan's then-current form of license agreement, which may contain terms, conditions, and fees substantially different from those in this Agreement. If Sylvan is not then offering License Agreements to new licensees, Licensee must execute Sylvan's most recent form of License Agreement offered to new licensees, which may contain terms, conditions, and fees substantially different from those in this Agreement.
- 10.2.7 Licensee shall have executed a general release under seal where required, in the thencurrent form satisfactory to Sylvan, of any and all claims against Sylvan, its current and former affiliates, and their respective past and present owners, officers, directors, agents, representatives, heirs, administrators, successors, and assigns, including, without limitation, claims arising under federal, state, and local laws, rules, and ordinances arising out of, or connected with, the performance of this Agreement.

This Agreement does not grant any automatic rights to a renewal term and Sylvan is not obligated to offer Licensee a renewal term if the requirements of this Paragraph 10.2. are not strictly and timely met.

- Month-To-Month License. If Licensee fails to give notice of its election to renew the Licenses as provided in Paragraph 10.2 and Sylvan nonetheless permits Licensee to continue to operate the Licensed Business under the Licensed Marks and/or Sylvan System and/or the SylvanSync System, all terms of this Agreement shall be deemed to be extended on a month-to-month basis. In that event either party may terminate this Agreement effective the last day of any calendar month, upon thirty (30) days' notice to the other party.
- 10.4 <u>Termination before Expiration</u>. Notwithstanding Paragraphs 10.1 and 10.2, this Agreement may be terminated as provided in Section 11 hereof.

#### 11. Termination.

- 11.1 <u>Default and Termination Notices</u>. If Licensee is in breach of any material provision of this Agreement, including, but not limited to Paragraphs 11.2.1 to 11.2.21 of this Agreement, Sylvan may give a Notice of Default and Opportunity to Cure which shall be not less than ten (10) days with respect to curable monetary defaults and thirty (30) days with respect to curable non-monetary defaults, after the date of the Notice. In no event, however, shall the number of days of notice be less than that required by any applicable statute or regulation of the state in which the Territory lies. If Licensee is determined by Sylvan to have cured the breach before the date stated in the Notice, such Notice shall be of no further effect. If the breach has not been cured before the cure date stated in the Notice, then Sylvan may give a Notice of Termination stating its intention to terminate on a certain date. With respect to breaches stated herein or otherwise determined to be incurable, Sylvan need not provide a period for cure.
- 11.2 <u>Material Breaches That May Be Cured</u>. Sylvan may terminate this Agreement, by delivery of written notice of default, upon the occurrence of any of the following events of default and Licensee's failure to take appropriate corrective action during the applicable cure period. Provided, however, that although Sylvan intends to allow Licensee to cure the defaults listed hereunder, if in Sylvan's good faith judgment, reasonably exercised, such default cannot reasonably be cured (such as, but not limited to damages to Sylvan from misrepresentations or warranties referenced in Paragraph 11.2.2), then such breaches shall be deemed incurable under Paragraph 11.3 hereof, and same shall be grounds for termination, effective on the earlier of delivery or three days after mailing by Sylvan of a written declaration of termination to Licensee.
- 11.2.1 To offer programs in the Licensed Business that are not approved by Sylvan or not part of the Sylvan System or the SylvanSync System.
- 11.2.2 Expressly or impliedly to misrepresent, warrant or guarantee the results of the licensed programs, except as specifically authorized by Sylvan.
- 11.2.3 To fail to pay when due any invoice of Sylvan or of any third party for goods or services Licensee has received; or any sales or other taxes when due and before delinquent, except when being contested in good faith by appropriate proceedings; or any rental or other amount due under a lease covering the premises of the Licensed Business.
- 11.2.4 To fail to provide at the Licensed Business adequately trained personnel as required by this Agreement, or to fail to cause the Licensee or its Managing Principal to attend and successfully complete Sylvan's initial training program.

- 11.2.5 To fail to resolve disputes among the Owners of Licensee respecting conduct of the Licensed Business.
- 11.2.6 Notorious behavior by a principal, director, officer, agent, or employee of Licensee which would tend to bring the Licensed Business or Sylvan's techniques of instruction into disrepute. If the activity is the type described in this subparagraph, Sylvan may require Licensee, in order to cure the breach adequately, to sever its relationship with the person whose behavior is offensive.
- 11.2.7 Absentee management of the Licensed Business. As used in this Paragraph, "absentee management" means that the Licensee or Managing Principal is not fully engaged in the operation of the Licensed Business, is not providing hands on leadership, oversight, and support to staff members on a full-time basis, or is not available to engage and communicate with members of the Sylvan Corporate staff regarding the Licensed Business during normal working hours.
- 11.2.8 To fail to open and commence full and continuous operation of the Center within one hundred eighty (180) days from the Effective Date of this Agreement or other mutually agreed required opening date that Sylvan approves in writing.
- 11.2.9 To fail to provide when due any Gross Revenue Report, any profit and loss statement, or any tax return as required under this Agreement or to fail to pay when due any royalty payment or any other fee or payment due hereunder, including, as an example and not as a limitation, any fee due to an advertising cooperative, but expressly excluding any failure to meet the Minimum Annual Royalty, which failure is a separate breach as described in Paragraph 11.2.20. In the event that Licensee fails four (4) times within any twelve (12) month period to pay any one or more of such sums or to provide any gross revenue report within ten days after the date same are due, and has each time been notified by Sylvan of said failure, Licensee may be deemed by Sylvan to have committed an incurable breach of this Agreement.
- 11.2.10 To fail to comply with mandatory or required standards and policies set forth in the Confidential Operations Manual and any changes, additions or modifications thereto, or to fail to implement a required or mandatory program, module or other materials, or to adhere to mandatory or required educational policies or standards, or to implement new mandatory programs, modules or materials as required by Sylvan.
- 11.2.11 To operate the Licensed Business without obtaining and maintaining all necessary permits and licenses, or to fail to comply with applicable laws.
- 11.2.12 To fail to permit Sylvan to inspect the premises, or observe the operation, or conduct an audit of the Licensed Business.
- 11.2.13 To allow unlawful activities to occur or to sell, utilize or exchange any unauthorized or illegal material at the Licensed Business.
  - 11.2.14 To fail to maintain insurance as required by this Agreement.
- 11.2.15 To fail to provide a suitable individual to assume control of the Licensed Business in the event of death or incapacity of the Licensee or the death or absence from active participation in the Licensed Business of any individual upon whose ability or expertise Sylvan relied in granting the Licenses.
- 11.2.16 To fail to obtain Sylvan's written approval of the site, design, and layout for each Center and Satellite that is opened within the Territory.
- 11.2.17 To fail to implement or fully comply within the time prescribed, any items designated as mandatory or required on a Correction Plan delivered to Licensee pursuant to Paragraph 7.9 hereof.
- 11.2.18 To place an advertisement or cause an advertisement to be circulated in another Licensee's exclusive territory or in the territory of a Sylvan owned licensed business without the prior written approval of Sylvan, except pursuant to Paragraph 7.3 hereof, or to violate Sylvan's policies, as established from time to time, regarding advertising.
- 11.2.19 To fail to obtain Sylvan's prior approval, if Sylvan has asserted its approval right in accordance with Paragraph 7.8.2, for content and media placement by the cooperative in which Licensee is a member, and to which Licensee has not objected in writing.

- 11.2.20 Each of the first two failures during the initial term or during a renewal term of this Agreement to generate sufficient gross revenues in a calendar year such that the royalty fees that Licensee pays to Sylvan under Paragraph 6.2.1 meet or exceed the Minimum Annual Royalty.
- 11.2.21 To fail to adhere to the requirements under Paragraph 6.12, including failing to provide or maintain a valid ACH Authorization form with Sylvan;
- 11.2.22 To fail to adhere to any other material provision of this Agreement, any other agreements between Licensee and Sylvan, its subsidiaries or affiliates, or any authorized specification, standard, policy or operating procedure prescribed by Sylvan as mandatory or required.
- 11.3 <u>Material Breaches That Cannot Be Cured.</u> Sylvan may terminate this Agreement, effective on the sooner of delivery or three days after mailing a written declaration of termination to the Licensee if any one or more of the following material, non-curable breaches of this Agreement occur:
- 11.3.1 An assignment or sale without Sylvan's prior written consent which is prohibited under Section 9 of this Agreement.
- 11.3.2 Any incurable breach of Licensee's confidentiality obligations under Paragraph 3 of this Agreement.
  - 11.3.3 Any act causing a tarnishment of Sylvan's reputation.
- 11.3.4 Assuming that Licensee operates but one Center in the Territory, the abandonment of the Center, including any closing of the Center for a period of ten (10) or more consecutive days, without the written approval of Sylvan. Sylvan's approval shall be given in those instances when the Center has been closed as the result of an act of God, fire or weather-related catastrophe, condemnation, or illness of Licensee, but only on the condition Licensee shall submit to Sylvan a plan for re-opening the Center within ninety (90) days of its closing. Sylvan's consent shall not be given in the event Licensee's closing of the Center is related to or pursuant to the events described in Paragraphs 11.3.5 to 11.3.7 below, or due to any adverse financial or economic condition suffered by Licensee.
  - 11.3.5 The insolvency, bankruptcy, or reorganization of the Licensee under the bankruptcy laws.
  - 11.3.6 The making by Licensee of an assignment for the benefit of creditors.
- 11.3.7 The appointment of a receiver or trustee of all or any part of the assets of Licensee or the Licensed Business.
- 11.3.8 Any act by Licensee or any person listed on the signature page hereof that a) results in a charge of, conviction of or pleading of nolo contendere or no contest to: (i) a misdemeanor relevant to the operation of the Licensed Business, (ii) a misdemeanor that involves moral turpitude, or (iii) a misdemeanor that involves a person under the age of 19; b) results in any charge of, conviction of or pleading of nolo contendere to a felony; c) results in any civil or regulatory action that involves a person under the age of 19; or d) in Sylvan's sole judgment impairs the goodwill of the Licensed Marks. In the event of a charge filed pursuant to subparagraph a) or b) above without plea or conviction, Sylvan may require Licensee during a period of twelve (12) months to sell the Licensed Business at no less than fair market value. Fair market value shall be determined by the previous twelve-month average licensed business sales prices for the Sylvan system. If Licensee fails to sell the Licensed Business in such twelve (12) month period, Sylvan may terminate this Agreement.
- 11.3.9 A third or any subsequent failure during an initial term or during a renewal term of this Agreement to generate sufficient gross revenues in a calendar year such that the royalty fees that Licensee pays to Sylvan under Paragraph 6.2.1 meet or exceed the Minimum Annual Royalty, whether or not the earlier such breaches were cured or waived, each such breach, if not waived, having been the subject of a separate notice by Sylvan to Licensee, but subject to the process set forth in Paragraph 6.2.3.
- 11.3.10 The breach of any of the terms of this Agreement four (4) or more times within any twelve (12) month period, whether or not such breach was cured, each breach having been the subject of a separate notice by Sylvan to Licensee.
- 11.3.11 Failure of the Licensee to attend at least two (2) mandatory national conferences out of any three (3) consecutive national conferences.

- 11.3.12 Failure of the Licensee's Director or Director of Education to attend at least two (2) mandatory regional conferences out of any three (3) consecutive regional conferences.
- 11.3.13 Licensee, or anyone acting on Licensee's behalf, maintains books or records which in Sylvan's judgment are intentionally false, submits reports to Sylvan which in Sylvan's judgment are intentionally false, or submits revenue reports that understate gross revenues or are otherwise substantially incorrect, provided that the foregoing shall not apply in the first instance if Licensee discovers, discloses to Sylvan, corrects all false reports, and promptly, though in no event later than thirty (30) days after the discovery, remits all amounts due to Sylvan, pursuant to such corrected reports and this Agreement, before receiving notification thereof from Sylvan.
- 11.4 <u>Remedies</u>. The remedies hereby reserved are independent, cumulative, and concurrent, and the exercise of any one shall not preclude the simultaneous or subsequent exercise of the same, or any other remedy. Sylvan's right of termination is in addition to any other rights or remedies it may have under Paragraph 11.1, 11.2 or 11.3. Sylvan's failure to exercise any remedy available to it shall not be deemed a waiver of its right to exercise any other right or remedy under this Agreement.
- 11.5 <u>Licensee's Notice of Breach by Sylvan</u>. It is the intention of Sylvan to comply with all obligations, performances and duties required of it under this Agreement. In the event that Licensee shall claim that Sylvan has failed to meet any obligation under this Agreement, Licensee shall provide Sylvan with a notice of such claim. Licensee's notice shall describe Sylvan's alleged deficiency and shall provide Sylvan with an opportunity to cure same within at least thirty (30) days from the date of receipt of the notice by Sylvan.

## 12. Relationship After Expiration or Termination.

- 12.1 <u>Obligations of Licensee</u>. Upon the expiration or termination of this Agreement, Licensee agrees that it shall immediately take the following actions:
- 12.1.1 Licensee shall cease to do business under any name incorporating or similar to SYLVAN LEARNING CENTER or SYLVANSYNC and shall cease to use any and all of the Sylvan System, the SylvanSync System, the Licensed Marks, proprietary programs, systems, techniques, or materials.
- 12.1.2 Licensee shall promptly return to Sylvan all proprietary, confidential or trade secret materials in its possession, custody, or control, including without limitation, the Confidential Operations Manual, all computer software and database material, customer lists, customer records and customer files. Licensee hereby covenants and agrees that it will not destroy, damage, hide or take any steps so as to prevent Sylvan from obtaining any such information, nor will Licensee retain any printouts, disks or tapes containing any of the programs or data stored in its computer system. Licensee shall retain no copy or record of any of the foregoing, except Licensee's copy of this Agreement, any correspondence between the parties and any other documents which Licensee reasonably needs for compliance with any provision of law. Licensee expressly affirms and agrees that the foregoing software, materials, lists, records, and files shall be deemed to be the property of Sylvan, and that Sylvan shall have the right to enter upon Licensee's premises and to remove such materials.
- 12.1.3 Licensee shall immediately remove and destroy within thirty (30) days after such termination all signs, designs and insignia in any way indicating or suggesting that Licensee's business establishment is related to or connected with Sylvan, its subsidiaries or affiliates, or any of Sylvan's licensees.
- 12.1.4 Licensee acknowledges that there will be substantial confusion in the mind of the public if, after the expiration or termination of the License Agreement, Licensee continues to use any listed or unlisted telephone numbers, email addresses, social media accounts or (if any) Internet domain names or the like (each a "Communications Account" and collectively the "Communications Accounts") used in the Licensed Business and/or listed in any directory under the name "Sylvan" or some other name confusingly similar thereto. Therefore, Licensee agrees that promptly after the expiration or termination of this Agreement for any reason whatsoever, Licensee will forthwith cease and desist using such Communications Accounts and, upon written demand of Sylvan, direct the provider(s) servicing the Licensed Business to transfer the Communications Accounts and their listings to Sylvan, or to such other person and at such location as Sylvan may direct. If Licensee does not promptly so direct the provider(s), Licensee irrevocably appoints Sylvan as its attorney-in-fact to so direct the provider(s) to make such transfers under the terms of this Agreement.
- 12.1.5 Except as provided in Paragraph 12.2 below, Licensee shall immediately refrain from engaging in any and all contacts with customers, whether with respect to providing service to customers, or for any other purpose whatsoever.

- 12.2 <u>Settlement of Outstanding Contracts with Students</u>. Licensee agrees that it is solely responsible to settle all claims of students enrolled in courses taught by Licensee whose courses of instruction are not completed on the effective date of expiration or termination of this Agreement. Upon termination, Licensee may not in any event accept any new students in the Licensed Business. However, if in Sylvan's sole judgment, the breach that led to termination is not such that Licensee should be immediately prevented from any further instruction of students, Sylvan may, in writing, allow Licensee to continue to complete the instruction for some or all of Licensee's existing students. Sylvan may in its sole discretion waive the application of Paragraph 12.1 for a period not to exceed thirty (30) days, to the extent necessary to allow Licensee to complete the students' course of instruction.
- 12.3 <u>Cessation of Sales Activities</u>. If Licensee is in breach of any material provision of this Agreement which is incurable or which has not been cured after Sylvan has given Licensee a Notice of Default and Opportunity to Cure, Sylvan shall have no obligation whatsoever to sell any products, equipment, or services to Licensee on cash or credit. Licensee shall not thereafter have any right to offer or sell Sylvan System or SylvanSync System products or services, and Licensee shall not, as a result thereof, have a defense at law or equity based upon impossibility of Licensee's performance, nor will Licensee as a result have any claim against Sylvan.
- 12.4 <u>Payment of All Obligations Due</u>. The expiration or termination of this Agreement shall not relieve Licensee from its obligation to pay Sylvan and other creditors when due any and all obligations and indebtedness incurred, or license fee accrued, as of the effective date of termination, and Licensee's obligations under Paragraph 3 and Paragraph 6 shall remain in full force and effect.
- Licensee and Principals Not to Compete. Licensee and its Owners, shareholders, partners, and/or members (collectively, "shareholders"), if any, acknowledge that they are to receive information which Sylvan has developed over time at great expense, including but not limited to methods of site selection; marketing methods; diagnostic techniques; prescription writing methods; and service methods and skills. They further acknowledge that this information is not generally known in the industry and is beyond their own present skills and experience, and that to develop it themselves would be expensive, time-consuming and difficult. They further acknowledge that Sylvan's information provides a competitive advantage and will be valuable to them in the development of their business, and that gaining access to it is the primary reason they are entering into this Agreement. Accordingly, Licensee and its shareholders, if any, agree that Sylvan's information as described above constitutes trade secrets belonging to Sylvan. In consideration of Sylvan's confidential disclosure of these trade secrets, Licensee, for itself, and if a corporation or a partnership, for its partners, directors, officers, and stockholders as well, covenants, individually:

# 12.5.1 During the term of this Agreement:

- (a) To use its best efforts in operating the Licensed Business and in recommending, promoting, and encouraging patronage of all Sylvan Network businesses; and
- (b) Not to engage as an owner, operator, or in any managerial capacity in any educational business, whether or not such business is not-for-profit, including, without limitation, any preschool, daycare, tutoring, test preparation, study skills, college advisory, technology-based instruction, blended instruction or other similar goods or services, other than as a licensee of the Sylvan System and the SylvanSync System, whether directly or indirectly (including through a spouse, parent, or child), unless such business is associated with an affiliate of Sylvan under a separate franchise agreement.
- 12.5.2 In the event this Agreement is terminated, expires, or is not renewed, or if Licensee assigns or transfers its interest herein to any person or business organization, then in such event Licensee covenants, for a period of two (2) years after such termination, expiration, nonrenewal, transfer or assignment, not to engage as an owner, operator, or in any managerial capacity, whether directly or indirectly (including through a spouse, parent, or child), in any educational business offering individualized diagnostic tests or academic or prescriptive educational programs which are designed to be taught, supervised, or administered by trained instructors, or in any educational business offering programs or services competitive with programs or services available at any Center or Satellite during the term of this Agreement and at the time of termination or expiration of this Agreement, and which programs or services Licensee offered in the Licensed Business during the two (2) year period prior to the termination or expiration of this Agreement, within the Territory or at any location less than fifty (50) miles from the Territory boundary or less than twenty-five (25) miles from any other corporate-affiliate-owned or licensee-owned Sylvan Learning Center business, other than as an authorized licensee of another Sylvan System and/or SylvanSync licensed business.

- 12.5.3 This Paragraph 12.5 shall not prevent any person affiliated with Licensee from accepting employment in any Center, Satellite or school which is licensed by state governmental authorities to provide a curriculum required by law, as an instructor in an established course of instruction previously offered at that Center, Satellite, or school. However, this exception shall not relieve Licensee of its obligation under Paragraph 3. The parties agree that each of the foregoing covenants shall be construed as independent of any other covenant or provision of this Agreement. Should any part of the restrictions contained in this Paragraph 12.5 be found to be unenforceable by virtue of its scope in terms of area, business activity prohibited or length of time, and should such part be capable of being made enforceable by reduction of any or all thereof, Licensee and Sylvan agree that the same shall be enforced to the fullest extent permissible under the law. Sylvan may, unilaterally, at any time, in its sole discretion, revise any of the covenants in this Paragraph 12.5 so as to reduce the obligations of Licensee hereunder. The running of any period of time specified in this Paragraph 12.5 shall be tolled and suspended for any period of time in which the Licensee is found by a court of competent jurisdiction to have been in violation of any such restrictive covenant. Licensee expressly agrees that the existence of any claim it may have against Sylvan, whether or not arising from this Agreement, shall not constitute a defense to the enforcement by Sylvan of the covenants in this Paragraph 12.5.
- 12.5.4 If any part of these restrictions contained in this Paragraph 12.5 is found to be unreasonable in time or distance, each month of time or mile of distance may be deemed a separate unit so that the time or distance may be reduced by appropriate order of the court or arbiter to that deemed reasonable. If, at any time during the two-year period following the expiration, termination, or approved transfer of this Agreement or the date any Owner ceases to be an Owner under this Agreement, Licensee or any of its Owners fails to comply with its obligations under this Article 12, that period of non-compliance will not be credited toward satisfaction of the two-year period. Licensee understands and acknowledges that Sylvan shall have the right, in its sole discretion, to reduce the scope of any covenant set forth in this Paragraph 12.5, or any portion thereof, without Licensee's consent, effective immediately upon receipt by Licensee of written notice thereof; and Licensee agrees that it shall comply forthwith with any covenant as so modified, which shall be fully enforceable notwithstanding the provisions of Articles 14, 21, and 23 hereof.
- 12.6 <u>Territory upon Termination</u>. Upon termination of this Agreement for any reason, Sylvan may immediately license the Territory or any portion thereof to another person or may itself operate a licensed business within the Territory.

#### 13. Injunctive Relief and Liquidated Damages.

Licensee acknowledges that Sylvan's remedy at law for Licensee's breach of Paragraph 3, Paragraph 12.1, or Paragraph 12.5 would be inadequate and that, accordingly, in such events, Sylvan shall be entitled to immediate injunctive relief. In addition to, and not in derogation of, the right to obtain an injunction, Sylvan shall be entitled to liquidated damages until such injunction issues. Due to the difficulties of computing actual damages from any such breach, damages shall be fixed in an amount equal to twenty-four (14) multiplied by the average monthly royalty Licensee had paid Sylvan over the thirty-six (36) month period (or such shorter period if Licensee held this Agreement for a lesser period) preceding termination or expiration of this Agreement for each month or part thereof that Licensee is in breach of Paragraph 3, Paragraph 12.1, or Paragraph 12.5. The parties agree that the stated sum is not a penalty, but reasonably represents the actual damages Sylvan will suffer in the event of a breach of any one of the enumerated paragraphs.

Licensee further acknowledges that Sylvan will suffer lost future profits in the event that Licensee ceases operation of the Licensed Business prior to the expiration or renewal of the term of this Agreement. Due to the difficulties of computing actual damages to Sylvan based on such early closure, damages shall be fixed in an amount equal to the average monthly royalty Licensee had paid Sylvan over the thirty-six (36) month period (or such shorter period if Licensee held this Agreement for a lesser period) preceding Licensee's closure of the Licensed Business, multiplied by twenty-four (24), or if there are fewer than twenty-four (24) months remaining until the expiration of the term of this Agreement, multiplied by the number of months remaining of the term. The parties agree that the stated sum is not a penalty, but reasonably represents Sylvan's actual damages under the circumstances.

#### 14. Hold Harmless, Indemnification and Insurance.

14.1 <u>Hold Harmless and Indemnification</u>. LICENSEE SHALL DEFEND AT ITS OWN COST AND INDEMNIFY AND HOLD HARMLESS TO THE FULLEST EXTENT PERMITTED BY LAW, SYLVAN AND ITS AFFILIATES, AND THEIR RESPECTIVE PARENTS, SUBSIDIARIES, DIRECTORS, OFFICERS, EMPLOYEES, AGENTS, SHAREHOLDERS, DESIGNEES, AND REPRESENTATIVES

(COLLECTIVELY, THE "FRANCHISOR INDEMNITEES") FROM ALL LOSSES AND EXPENSES INCURRED IN CONNECTION WITH ANY ACTION, SUIT, PROCEEDING, CLAIM, CAUSE OF ACTION, DEMAND, INVESTIGATION, OR FORMAL OR INFORMAL INQUIRY (REGARDLESS OF WHETHER ANY OF THE FOREGOING IS REDUCED TO JUDGMENT), OR ANY SETTLEMENT OF THE FOREGOING, WHICH ACTUALLY OR ALLEGEDLY, DIRECTLY OR INDIRECTLY, ARISES OUT OF, IS BASED UPON, IS A RESULT OF, OR IS IN ANY WAY RELATED TO ANY OF THE FOLLOWING: (1) ANY ACTUAL OR ALLEGED INFRINGEMENT OR ANY OTHER ACTUAL OR ALLEGED VIOLATION OF ANY PATENT, TRADEMARK, COPYRIGHT, OR OTHER PROPRIETARY RIGHT OWNED OR CONTROLLED BY THIRD PARTIES BY LICENSEE OR THE LICENSED BUSINESS OR ANY OF ITS OR ITS RESPECTIVE OWNERS, OFFICERS, DIRECTORS, PERSONNEL. EMPLOYEES. SERVANTS. MANAGEMENT AGENTS, CONTRACTORS. SUBCONTRACTORS, PARTNERS, PROPRIETORS, AFFILIATES OR REPRESENTATIVES, OR ANY THIRD PARTY ACTING ON BEHALF OF OR AT THE DIRECTION OF SUCH PERSONS OR ENTITIES, WHETHER IN CONNECTION WITH THE LICENSED BUSINESS OR OTHERWISE (COLLECTIVELY, THE "LICENSEE INDEMNITORS"); (2) ANY ACTUAL OR ALLEGED VIOLATION OR BREACH OF ANY CONTRACT, FEDERAL, STATE, OR LOCAL LAW, REGULATION, RULING, STANDARD, OR DIRECTIVE OF ANY INDUSTRY STANDARD BY LICENSEE OR ANY OF THE OTHER LICENSEE INDEMNITORS; (3) ANY ACTUAL OR ALLEGED LIBEL, SLANDER, OR ANY OTHER FORM OF DEFAMATION BY LICENSEE OR ANY OF THE OTHER LICENSEE INDEMNITORS; (4) ANY ACTUAL OR ALLEGED VIOLATION OR BREACH OF ANY WARRANTY, REPRESENTATION, AGREEMENT, OR OBLIGATION IN THIS AGREEMENT BY LICENSEE OR ANY OF THE OTHER LICENSEE INDEMNITORS: (5) ANY AND ALL ACTS, ERRORS, OR OMISSIONS ENGAGED IN BY LICENSEE OR ANY OF THE OTHER LICENSEE INDEMNITORS, ARISING OUT OF OR RELATED TO THE DESIGN, CONSTRUCTION, CONVERSION, BUILD OUT, OUTFITTING, REMODELING, RENOVATION, UPGRADING, OR OPERATION OF THE LICENSED BUSINESS, WHETHER ANY OF THE FOREGOING WAS APPROVED BY SYLVAN, INCLUDING, BUT NOT LIMITED TO, ANY PERSONAL INJURY, DEATH, OR PROPERTY DAMAGE SUFFERED OR CAUSED BY ANY CUSTOMER, VISITOR, OPERATOR, EMPLOYEE, OR GUEST OF THE LICENSED BUSINESS; (6) ANY PERSONAL INJURY, DEATH, OR PROPERTY DAMAGE SUFFERED OR CAUSED BY LICENSEE OR ANY OF THE OTHER LICENSEE INDEMNITORS; (7) ALL LIABILITIES ARISING FROM OR RELATED TO ITS MARKETING, ADVERTISING, PROMOTION, OFFER, SALE, OR DELIVERY OF PRODUCTS OR SERVICES AS CONTEMPLATED BY THIS AGREEMENT; (8) ANY AND ALL LATENT OR OTHER DEFECTS IN THE LICENSED BUSINESS, WHETHER OR NOT DISCOVERABLE BY SYLVAN OR LICENSEE; (9) THE INACCURACY OR LACK OF AUTHENTICITY OF ANY INFORMATION DISCLOSED TO ANY CUSTOMER OF THE LICENSED BUSINESS; (10) CRIMES COMMITTED ON OR NEAR ANY OF THE PREMISES OR FACILITIES OR VEHICLES USED BY ITS LICENSED BUSINESS; (11) ANY SERVICES OR PRODUCTS PROVIDED BY ANY AFFILIATED OR NONAFFILIATED PARTICIPATING ENTITY; (12) ANY ACTION BY ANY CUSTOMER, VISITOR, OPERATOR, EMPLOYEE, OR GUEST OF THE LICENSED BUSINESS OR ANY OTHER FACILITY OF ITS LICENSED BUSINESS; (13) ANY AND ALL ACTS, ERRORS, OR OMISSIONS ENGAGED IN BY LICENSEE OR ANY OF THE OTHER LICENSEE INDEMNITORS, ARISING OUT OF OR RELATED TO ANY DISCRIMINATION, HARASSMENT, DISABILITY, HOUR AND WAGE CLAIMS, OR OTHER EMPLOYMENT PRACTICES IN ANY WAY RELATED TO THE OPERATION OF THE LICENSED BUSINESS, WHETHER ANY OF THE FOREGOING WAS APPROVED BY SYLVAN, (14) ANY AND ALL CLAIMS RELATED TO ITS NONCOMPLIANCE OR ALLEGED NONCOMPLIANCE WITH ANY LAW, ORDINANCE, RULE OR REGULATION, INCLUDING ANY ALLEGATION THAT WE ARE A JOINT EMPLOYER OR OTHERWISE RESPONSIBLE FOR ITS ACTS OR OMISSIONS RELATING TO ITS EMPLOYEES, AND (15) ANY DAMAGE TO THE PROPERTY OF LICENSEE OR SYLVAN, ITS AND OUR RESPECTIVE AGENTS, OR EMPLOYEES, OR ANY THIRD PERSON, FIRM, OR CORPORATION, WHETHER OR NOT SUCH LOSSES, CLAIMS, COSTS, EXPENSES, DAMAGES, OR LIABILITIES WERE ACTUALLY OR ALLEGEDLY CAUSED WHOLLY OR IN PART THROUGH THE ACTIVE OR PASSIVE NEGLIGENCE OF SYLVAN OR ANY OF ITS AGENTS OR EMPLOYEES, OR RESULTED FROM ANY STRICT LIABILITY IMPOSED ON SYLVAN OR ANY OF ITS AGENTS OR EMPLOYEES.

THE INDEMNIFICATION REQUIRED UNDER THIS PARAGRAPH 14.1 SHALL APPLY TO ALL CLAIMS, INCLUDING THOSE THAT ARISE, OR ARE ALLEGED TO ARISE, AS A RESULT OF

SYLVAN'S OWN NEGLIGENCE OR GROSS NEGLIGENCE, IF ANY, REGARDLESS OF WHETHER SYLVAN'S NEGLIGENCE OR GROSS NEGLIGENCE IS ALLEGED TO BE THE SOLE, CONTRIBUTING, OR CONCURRING CAUSE OF SUCH ALLEGED DAMAGES THAT MIGHT BE ASSERTED.

For purposes of this Agreement, the term "Losses and Expenses" means, without limitation, all claims, losses, liabilities, costs, and expenses including compensatory, exemplary, incidental, consequential, statutory, or punitive damages or liabilities; fines, penalties, charges, expenses, lost profits, attorneys' fees, expert fees, costs of investigation, court costs, settlement amounts, judgments, compensation for damages to Sylvan's reputation and goodwill, costs of or resulting from delays, and financing; travel, food, lodging, and other expenses necessitated by Sylvan's need or desire to appear before, or witness the proceedings of, courts or tribunals (including arbitration tribunals), or governmental or quasi-governmental entities, including those incurred by Sylvan's attorneys or experts to attend any of the same; costs of advertising material and media/time/space, and costs of changing, substituting, or replacing the same; and any and all expenses of recall, refunds, compensation, public notices, and all other amounts incurred by Sylvan in connection with the matters described above. All such Losses and Expenses incurred by Sylvan will be chargeable to and payable by Licensee pursuant to this Paragraph 14.1, regardless of any actions, activities, or defenses undertaken by Sylvan or the subsequent success or failure of such actions, activities, or defenses.

Licensee shall give Sylvan written notice of any event of which Licensee is aware for which indemnification is required within three days of its actual or constructive knowledge of such event. At its expense and risk, Sylvan may elect to assume (but under no circumstance is obligated to undertake) the defense or settlement thereof, provided that Sylvan will seek Licensee's advice and counsel. Any assumption by Sylvan shall not modify Licensee's indemnification obligation. Sylvan may, in its sole and absolute discretion, take such actions as it deems necessary and appropriate to investigate, defend, or settle any event, or take other remedial or corrective actions with respect thereof as may be, in Sylvan's sole and absolute discretion, necessary for the protection of the Franchisor Indemnities or the System. Under no circumstances will Sylvan or the Franchisor Indemnities be required to seek recovery from third parties or to otherwise mitigate their losses to maintain a claim against Licensee; in no event will a failure to pursue recovery from third parties or to mitigate loss reduce the amounts recoverable by Sylvan or the Franchisor Indemnities from Licensee. The indemnification obligations provided by this Paragraph 14.1 will survive the expiration or termination of this Agreement.

Insurance. Licensee shall, at its expense and no later than upon commencement of the business contemplated by this Agreement, produce and maintain in full force and effect throughout the term of this Agreement the types of insurance enumerated in the Confidential Operations Manual which shall be in such amounts as may from time to time be required by Sylvan, including but not limited to worker's compensation insurance, all risk protection with replacement cost coverage on equipment, furnishings, fixtures, and building, and comprehensive public liability insurance. Licensee may, but is not required to, also secure teacher's professional errors and omission coverage if available. Sylvan may at any time amend the amount of insurance Licensee is required to carry. Sylvan shall be named as an insured under the comprehensive public liability insurance policies, and otherwise as its interest may appear. Licensee shall keep and maintain any and all insurance required hereunder, and under any lease or any loan agreement, in full force and effect at all times during the term of this Agreement. Licensee shall make timely delivery of certificates of all required insurance to Sylvan, each of which shall contain a statement by the insurer that the policy will not be cancelled or materially altered without at least thirty (30) days prior written notice to Sylvan, and other evidence of coverage that Sylvan may require from time to time. The procurement and maintenance of such insurance shall not relieve Licensee of any liability to Sylvan under any indemnity requirement of this Agreement. In the event that Licensee fails to carry and maintain any and all insurance as required hereunder, Licensee authorizes Sylvan to secure such insurance for Licensee and to charge Licensee for such premiums.

# 15. <u>Parties as Independent Contractors.</u>

The parties acknowledge and agree that this Agreement does not create a fiduciary relationship between them, that Licensee shall operate the Licensed Business as an independent contractor, Sylvan and Licensee do not intend to be partners, associates, joint venturer, employee, employer, agents, or joint employers in any way, Sylvan shall not be construed to be jointly liable for any of Licensee's acts or omissions under any circumstances, and that nothing in this Agreement shall be construed to create a partnership, joint venture, agency, employment, fiduciary relationship, master-servant relationship, or legal relationship of any kind. Sylvan shall have no relationship with Licensee's employees and Licensee have no relationship with Sylvan's employees.

None of Licensee's employees will be considered employees of Sylvan or its affiliates. Neither Licensee nor any of Licensee's employees whose compensation Licensee pay may in any way, directly or indirectly, expressly or by implication, be construed to be an employee of Sylvan or its affiliates for any purpose, including with respect to any mandated or other insurance coverage, tax or contributions, or requirements pertaining to withholdings, levied or fixed by any city, state, or federal governmental agency. Neither Sylvan nor its affiliates will have the power to hire or fire Licensee's employees. Sylvan has no right or duty to supervise, manage, control or direct Licensee's employees in the course of their employment for Licensee. Licensee expressly agree, and will never contend otherwise, that Sylvan's authority under this Agreement to certify certain of Licensee's employees for qualification to perform certain functions for Licensee's Licensed Business does not directly or indirectly vest in Sylvan or its affiliates the power to hire, fire, or control any such employee. Licensee further acknowledge and agree, and will never contend otherwise, that Licensee alone will exercise day-to-day control over all operations, activities, and elements of Licensee's Licensed Business and that under no circumstance shall Sylvan or its affiliates do so or be deemed to do so. Licensee further acknowledge and agree, and will never contend otherwise, that the various requirements, restrictions, prohibitions, specifications, and procedures of Sylvan System which Licensee is required to comply with under this Agreement, whether set forth in the Confidential Operations Manual or otherwise, do not directly or indirectly constitute, suggest, infer, or imply that Sylvan or its affiliates controls any aspect or element of the day-to-day operations of Licensee's Licensed Business, which Licensee alone control, but constitute only standards to which Licensee must adhere when exercising Licensee's control of the day-to-day operations of Licensee's Licensed Business. Licensee is solely responsible for all terms and conditions of employment of Licensee's employees.

Except as otherwise expressly authorized by this Agreement, neither party hereto will make any express or implied agreements, warranties, guarantees, or representations or incur any debt in the name of or on behalf of the other party, or represent that the relationship between Sylvan and Licensee is other than that of franchisor and franchisee. Sylvan does not assume any liability, and will not be deemed liable, for any agreements, representations, or warranties made by Licensee which are not expressly authorized under this Agreement, nor will Sylvan be obligated for any damages to any person or property which directly or indirectly arise from or relate to the operation of the Licensed Business.

During the term of this Agreement, Licensee shall identify itself as the owner of the Licensed Business operating under a franchise granted by Sylvan, and shall apply for all permits, certificates of occupancy, and business licenses in Licensee's own name. Additionally, Licensee's individual name (if Licensee is an individual) or Licensee's corporate name (if Licensee is a Business Entity) must appear prominently on all invoices, order forms, receipts, business stationery, and contracts. Licensee shall not use the Licensed Marks to incur or secure any obligation or indebtedness on behalf of Sylvan. In all dealings with third parties including, without limitation, employees, suppliers, and customers, Licensee shall disclose in an appropriate manner acceptable to Sylvan that it is an independent entity licensed by Sylvan. Licensee shall prominently indicate on all letterheads, business forms, and the like that it is a Licensee of Sylvan by using language substantially stating Licensee is "A Licensee of Sylvan Learning, LLC."

#### 16. Notices.

Any notice required or permitted by this Agreement shall be in writing, addressed to a party at the address given above, or to such other address as the party by notice informs the other, shall be mailed by registered or certified mail if not hand delivered or transmitted via facsimile or electronically, and shall be effective on the sooner of delivery or three (3) days after mailing. Licensee authorizes the transmission of e-mail by Sylvan and Sylvan's employees, vendors, and affiliates (on matters pertaining to the business contemplated hereunder) to Licensee and Licensee's employees and designees during the term of this Agreement, and agrees that such persons shall not opt-out, or otherwise ask to no longer receive e-mails, during the term of this Agreement or, concerning Licensee's employees and designees, during the term of their employment with or designation by Licensee, respectively.

# 17. <u>Attorneys' Fees.</u>

In the event either Sylvan or Licensee institutes a suit or action to enforce any term or provision of this Agreement, the most prevailing party in the suit or action, or on appeal, shall be entitled to recover from the losing party attorneys' fees, costs, and expenses (and interest on such fees, costs, and expenses) to be set by the trial or appellate court in addition to costs or disbursements provided by law.

### 18. Construction; Severability.

All captions and headings are for reference purposes only and shall not be deemed to be a part of this Agreement. If any part of this Agreement shall, for any reason whatsoever, be declared invalid or unenforceable, the

affected portion shall nevertheless be enforced to the fullest extent allowed by law, and such declaration shall not affect the validity of the remaining provisions, which shall remain in full force and effect. The parties hereby declare their intention that they would have executed the remaining portion of this Agreement without including herein any portions which may, for any reason, be declared invalid in the future; provided, however, that if Sylvan determines that such finding of invalidity or illegality adversely affects the basic consideration of this Agreement, Sulvan, at its option, may terminate this Agreement..

# 19. Organization of the Licensee

- Representations. If the Agreement was signed under the Licensee's individual capacity and Licensee then utilizes a Business Entity to operate under this Agreement, then such individuals are required to execute the Sylvan's form of the assignment and assumption agreement assigning this Agreement to the Business Entity within 30 days of the Effective Date. If Licensee is a Business Entity, Licensee makes the following representations and warranties: (1) the Business Entity is duly organized and validly existing under the laws of the state of its formation; (2) it is qualified to do business in the state or states in which the Licensed Business is located; (3) execution of this Agreement and the development and operation of the Licensed Business is permitted by its governing documents; and (4) unless waived in writing by Sylvan, its charter documents and its governing documents shall at all times provide that the activities of the Business Entity are limited exclusively to the development and operation of Sylvan businesses. If Licensee is an individual, or a partnership comprised solely of individuals, Licensee make the following additional representations and warranties: (1) each individual has executed this Agreement; (2) each individual shall be jointly and severally bound by, and personally liable for the timely and complete performance and a breach of, each and every provision of this Agreement; and (3) notwithstanding any transfer for convenience of ownership pursuant to Article 9 of this Agreement, each individual shall continue to be jointly and severally bound by, and personally liable for the timely and complete performance and a breach of, each and every provision of this Agreement. "Business Entity" means a corporation, limited liability company, limited partnership, or other entity created pursuant to statutory authority.
- Governing Documents. If Licensee is a corporation, copies of its Articles of Incorporation, bylaws, other governing documents, and any amendments, including the resolution of the Board of Directors authorizing entry into and performance of this Agreement, and all shareholder agreements, including buy/sell agreements, must be furnished to Sylvan. If Licensee is a limited liability company, copies of its Articles of Organization, operating agreement, other governing documents, and any amendments, including the resolution of the managers authorizing entry into and performance of this Agreement, and all agreements, including buy/sell agreements, among the members must be furnished to Sylvan. If Licensee is a general or limited partnership, copies of its written partnership agreement, other governing documents and any amendments, as well as all agreements, including buy/sell agreements, among the partners must be furnished to Sylvan, in addition to evidence of consent or approval of the entry into and performance of this Agreement by the requisite number or percentage of partners, if that approval or consent is required by its written partnership agreement or applicable law. When any of these governing documents are modified or changed, Licensee must promptly provide copies of the modifying documents to Sylvan.
- Ownership Interests. If Licensee is a Business Entity, Licensee represents that all of its equity interests are owned as set forth on Attachment B to this Agreement. In addition, if Licensee is a corporation, Licensee shall maintain a current list of all Owners, including owners of record and all beneficial owners of any class of voting securities of the corporation (and the number of shares owned by each). If Licensee is a limited liability company, Licensee shall maintain a current list of all members (and the percentage membership interest of each member). If Licensee is a partnership, Licensee shall maintain a current list of all owners of an interest in the partnership (and the percentage ownership interest of each general and limited partner). Licensee shall comply with Article 9 of this Agreement prior to any change in ownership interests and shall execute any necessary addenda to Attachment C as changes occur to ensure the information contained in Attachment B is true, accurate, and complete at all times. Licensee must also provide a copy of the Business Entity's EIN and execute a new ACH Authorization Agreement (Attachment I), if the EIN changes. "Owner(s)" means you if you are an individual, or each individual or entity holding more than a ten percent equity interest in you if you are a Business Entity (regardless of voting rights), and the licensee individual(s) or entity(ies) that enter into the License Agreement if you are a Business Entity. It includes all officers, directors, and shareholders of a corporation, all managers and members of a limited liability company, all general and limited partners of a limited partnership, and the grantor and the trustee of the trust. If any Owner is a Business Entity, then the term "Owner" also includes the Owners of that Business Entity.

- Restrictive Legend. If Licensee is a corporation, Licensee shall maintain stop-transfer instructions against the transfer on Licensee's records of any voting securities, and each stock certificate of the corporation shall have conspicuously endorsed upon its face the following statement: "Any assignment or transfer of this stock is subject to the restrictions imposed on assignment by the Sylvan® License Agreement(s) to which the corporation is a party." If Licensee is a limited liability company, each membership or management certificate or other evidence of interest in the limited liability company shall have conspicuously endorsed upon its face the following statement: "Any assignment or transfer of an interest in this limited liability company is subject to the restrictions imposed on assignment by the SylvanM® License Agreement(s) to which the limited liability company is a party." If Licensee is a partnership, Licensee's written partnership agreement shall provide that ownership of an interest in the partnership is held subject to, and that further assignment or transfer is subject to, all restrictions imposed on assignment by this Agreement.
- 19.5 <u>Guarantees</u>. If Licensee is a Business Entity, each Owner (and if Licensee is a limited partnership, each of Licensee's general partner's Owners) shall execute the Undertaking and Guaranty attached hereto as Attachment C and the Confidentiality and Non-Competition Agreement attached hereto as Attachment D.

#### 20. Time is of the Essence.

Time is of the essence of this Agreement.

#### 21. Jurisdiction and Venue.

- 21.1. Governing Law. Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act), the Federal Arbitration Act, the Copyright Act, or the Patent Act, this Agreement (and all matters arising out of or relating to this Agreement) are governed by, and shall be construed in accordance with, the laws of the State of Texas, without giving effect to the conflict of laws provisions thereof to the extent such principles or rules would require or permit the application of the laws of any jurisdiction other than those of the State of Texas. By agreeing to the application of Texas law, the parties do not intend to make this Agreement or their relationship subject to any franchise, distributorship, business opportunity, or similar statute, rule, or regulation of the State of Texas to which this Agreement or the parties' relationship otherwise would not be subject. As of the Effective Date, Sylvan has a place of business in the State of Texas, and Texas otherwise bears a reasonable relationship to this Agreement, the parties' relationship established by this Agreement, and the parties. Licensee and Sylvan acknowledge and agree that the choice of applicable state law set forth in this Section provides each of the parties with the mutual benefit of uniform interpretation of this Agreement and the parties' relationship created by this Agreement. Licensee and Sylvan further acknowledge the receipt and sufficiency of mutual consideration for such benefit, and that each party's agreement regarding applicable state law has been negotiated in good faith and is part of the benefit of the bargain reflected in this Agreement.
- 21.2. <u>Consent to Jurisdiction</u>. Subject to the arbitration obligations in Article 23 below, any judicial action must be brought in a court of competent jurisdiction in the state, and in (or closest to) the county, where Franchisor's principal place of business is then located. Each of the parties irrevocably submits to the jurisdiction of such courts and waives any objection to such jurisdiction or venue. Notwithstanding the foregoing, Sylvan may bring an action for a temporary restraining order or for temporary or preliminary injunctive relief, or to enforce an arbitration award or judicial decision, in any federal or state court in the county in which Licensee resides or the Licensed Business is located.

# 22. <u>Waiver of Damages.</u>

SYLVAN AND LICENSEE HEREBY WAIVE TO THE FULLEST EXTENT PERMITTED BY LAW ANY RIGHT TO OR CLAIM OF ANY PUNITIVE, SPECIAL, CONSEQUENTIAL, OR EXEMPLARY DAMAGES AGAINST THE OTHER, AND AGREE THAT IN THE EVENT OF A DISPUTE BETWEEN THEM EACH SHALL BE LIMITED TO THE RECOVERY OF ANY ACTUAL DAMAGES SUSTAINED BY IT.

# 23. <u>Dispute Resolution by Binding Arbitration.</u>

23.1.1 Any dispute or claim between (A) Licensee and/or any Owner and (B) Sylvan or any Franchisor Indemnitee, including but not limited to any dispute or claim arising out of or relating in any way to

23.1.1.2 this Agreement or any other agreement between Licensee and/or any Owner and Sylvan or any Franchisor Indemnitee,

- 23.1.1.3 the offer and sale of the franchise opportunity,
- 23.1.1.4 any representations made prior to the execution of this Agreement,
- 23.1.1.5 the validity, enforceability, or scope of this Agreement and this arbitration

agreement, and

### 23.1.1.6 the relationship of the parties

must be submitted to binding arbitration before the American Arbitration Association ("AAA") pursuant to its Commercial Arbitration Rules in effect at the time the arbitration demand is filed. The AAA rules are available online at www.adr.org. The only disputes or claims that shall not be subject to arbitration shall be those that relate to the protection or enforcement of Sylvan's or Franchisor Indemnitees' rights in and to Intellectual Property (including, but not limited to, the Licensed Marks). The number of arbitrators shall be one. This arbitration agreement and the arbitration shall be subject to and governed by the Federal Arbitration Act, and not any state arbitration law.

- 23.1.2 Licensee, the Owners, Sylvan, and the Franchisor Indemnitees agree that arbitration will be conducted on an individual, not a class-wide or representative basis, that only Licensee, the Owners, Sylvan, and the Franchisor Indemnitees may be the parties to any arbitration proceeding described in this Paragraph, and that no such arbitration proceeding may be consolidated or joined with another arbitration proceeding involving Licensee, the Owners, Sylvan, the Franchisor Indemnitees or any other person or entity. The arbitrator shall have no power to preside over or consider any form representative, joint, consolidated, collective or class proceeding. Despite the foregoing or anything to the contrary in this Paragraph, if any court or arbitrator determines that all or any part of this Paragraph is unenforceable with respect to a dispute that otherwise would be subject to arbitration, then Sylvan and Licensee agree that this arbitration clause will not apply to that dispute, and such dispute will be resolved in a judicial proceeding in accordance with Paragraph 23.H.
- 23.1.3 Sylvan and Licensee will be bound by any limitation under this Agreement or applicable law, whichever expires first, on the timeframe in which claims must be brought. Sylvan and Licensee further agree that, in connection with any arbitration proceeding, each must submit or file any claim constituting a compulsory counterclaim (as defined by the then-current Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding as the claim to which it relates. Any such claim not submitted or filed in the proceeding will be barred. The arbitrator does not have the right to consider any settlement discussions or offers either Sylvan and Licensee made.
- 23.1.4 Unless prohibited by law, the arbitration shall occur in Tarrant County, Texas. The arbitrator will have no authority to select a different hearing locale other than as described in the prior sentence.
- 23.1.5 Except as may be required by law, neither Licensee, its Owners, nor an arbitrator may disclose the existence, content, or results of any arbitration under this Paragraph without the prior written consent of all parties.
  - 23.1.6 The arbitrator must follow, and may not disregard, the applicable law.
- 23.1.7 The arbitrator has the right to award any relief he or she deems proper in the circumstances, including money damages (with interest on unpaid amounts from the date due), specific performance, injunctive relief, and attorneys' fees and costs (in accordance with Paragraph 17.), provided that: (i) the arbitrator has no authority to declare any Licensed Mark generic or otherwise invalid; and (ii) Sylvan and Licensee waive to the fullest extent the law permits any right to or claim for any punitive, exemplary, treble, and other forms of multiple damages against the other. The arbitrator's award and decision will be conclusive and bind all parties covered by this Paragraph, and judgment upon the award may be entered in a court specified or permitted in Paragraph 21 above.
- 23.1.8 The decision of the arbitrator will be final and binding on all parties to the dispute. A judgment may be entered upon the arbitration award in any federal or state court having jurisdiction.
- 23.1.9 Despite the existence of the arbitration clause, the parties shall have the right to seek temporary restraining orders, preliminary injunctions, and similar equitable relief from a court pending arbitration of the merits of the claims. Any injunctive relief may be given without the necessity of Sylvan posting bond or other security and any such bond or other security is hereby waived.
- 23.1.10 To the fullest extent permitted by law, the parties agree that any actions permitted to be brought under this Agreement by either party in any court may only be brought in a federal or state court serving the

judicial district in which Sylvan's principal headquarters is located at the time litigation is commenced. Licensee hereby irrevocably submits to the jurisdiction of the federal and state courts serving the judicial district in which Sylvan's principal headquarters is located at the time litigation is commenced and waive any objection Licensee may have to the jurisdiction of or venue in such courts.

#### 24. Acceptance by Licensee.

This Agreement shall be binding upon Licensee at the time it is signed by Licensee, and delivered to Sylvan at its principal place of business. This Agreement shall not be binding upon Sylvan until it is accepted in writing by a principal officer of Sylvan.

#### 25. No Waiver.

Sylvan reserves the right and privilege, in its sole discretion, to vary standards for any licensee based upon the peculiarities of a particular site or circumstance, demographics, or any other conditions which Sylvan deems to be of importance to the successful operation of a Sylvan licensed business. Licensee shall have no recourse against Sylvan on account of any licensee and shall not be entitled to require Sylvan to grant Licensee a like or similar variation hereunder. No failure of Sylvan to exercise any power reserved to it by this Agreement, or to insist upon strict compliance by Licensee with any obligation or condition hereunder, and no custom or practice of the parties at variance with the terms hereof, shall constitute a waiver of Sylvan's right to demand exact compliance with any of the terms herein. Sylvan's waiver of any particular default by Licensee shall not affect or impair Sylvan's rights with respect to any subsequent default of the same, similar, or different nature, nor shall any delay, forbearance, or omission of Sylvan to exercise any power or right arising out of any breach or default by Licensee of any of the terms, provisions, or covenants hereof affect or impair Sylvan's right to exercise the same, nor shall such constitute a waiver by Sylvan of any right hereunder or the right to declare any subsequent breach or default and to terminate this Agreement prior to the expiration of its term. Subsequent acceptance by Sylvan of any payments due to it hereunder shall not be deemed to be a waiver by Sylvan of any preceding breach by Licensee of any terms, covenants, or conditions of this Agreement.

# 26. <u>Modification; Entire Agreement.</u>

Except as otherwise provided herein, neither Sylvan nor Licensee may modify this Agreement except by an instrument in writing between Sylvan and Licensee. This Agreement, all attachments to this Agreement, and all ancillary agreements executed contemporaneously with this Agreement constitute the final and fully integrated agreement between the parties regarding the subject matter of this Agreement and supersede any and all prior negotiations, understandings, representations and agreements. Nothing in this agreement or any related agreement is intended to disclaim the representation made in the disclosure document provided to Licensee and Sylvan.

#### 27. Class Action Waiver.

Any action brought by Licensee or Sylvan shall be conducted on an individual basis, and not as part of a consolidated, common, or class action, and Licensee and Licensee's owners waive any and all rights to proceed on a consolidated, common, or class basis. <u>LICENSEE HEREBY AGREES NOT TO SEEK JOINDER OF ANY OF CLAIMS WITH THOSE OF ANY OTHER PARTY.</u>

#### 28. Waiver of Jury Trial.

TO THE FULLEST EXTENT PERMITTED BY LAW, LICENSEE, EACH OWNER, AND THE FRANCHISOR INDEMNITIES KNOWINGLY, WILLINGLY, AND VOLUNTARILY, WITH FULL AWARENESS OF THE LEGAL CONSEQUENCES, AFTER CONSULTING WITH COUNSEL (OR AFTER HAVING WAIVED THE OPPORTUNITY TO CONSULT WITH COUNSEL) AGREE TO WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY DISPUTE BETWEEN THEM. THE RIGHT TO A TRIAL BY JURY IS A RIGHT SUCH PARTIES WOULD OR MIGHT OTHERWISE HAVE HAD UNDER THE CONSTITUTIONS OF THE UNITED STATES OF AMERICA AND THE STATE IN WHICH THE LICENSED BUSINESS IS LOCATED.

# 29. <u>Limitation on Claims.</u>

Except for those Claims brought under the indemnification or insurance coverage provisions of this Agreement (including, but not limited to, those provisions contained in Paragraph 14), upon expiration or termination of this Agreement, neither party will commence any Claim related to or arising out of this Agreement, unless commenced within one (1) year following the effective date of expiration or termination. "Claims" means any

allegation, challenge, demand, cause of action, lawsuit, arbitration, dispute, controversy, investigation or administrative proceeding.

30. <u>Attachments.</u> The following Attachments are a part of this Agreement and are incorporated herein by reference:

Attachment A – Territory

Attachment B - Licensee's Owners and Key Personnel

Attachment C – Undertaking and Guaranty

Attachment D - Confidentiality and Non-Competition Agreement

Attachment E - Acknowledgment of Request to Contribute to Advertising Cooperative

Attachment F – Contact Center Franchisee Participation Agreement

Attachment G – SylvanSync Educational Software License and Terms of Use Agreement

Attachment H – Telephone Numbers Assignment Agreement

Attachment I – ACH Authorization Agreement

Attachment J – Dashboard Access Agreement

Attachment K - Bright Horizons Participation Agreement

Attachment L – mySylvan Marketplace+ Franchisee Participation Pilot Agreement

# 31. Acknowledgements.

#### 31.1 Patriot Act.

You represent and warrant that to your actual knowledge: (i) neither Licensee, nor its officers, directors, managers, members, partners or other individual who manages the affairs of Licensee, nor any Licensee affiliate or related party, or any funding source for the Licensed Business, is identified on the lists of Blocked Persons, Specially Designated Nationals, Specially Designated Terrorists, Specially Designated Global Terrorists, Foreign Terrorists Organizations, and Specially Designated Narcotics Traffickers at the United States Department of Treasury's Office of Foreign Assets Control (OFAC), or the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, commonly known as the "USA Patriot Act," as such lists may be amended from time to time (collectively, "Blocked Person(s)"); (ii) neither Licensee nor any Licensee affiliate or related party is directly or indirectly owned or controlled by the government of any country that is subject to an embargo imposed by the United States government; (iii) neither Licensee nor any Licensee affiliate or related party is acting on behalf of the government of, or is involved in business arrangements or other transactions with, any country that is subject to such an embargo; (iv) neither Licensee nor any Licensee affiliate or related party are on the United States Department of Commerce Denied Persons, Entity and Unverified Lists, or the United States Departments of State's Debarred List, as such lists may be amended from time to time (collectively, the "Lists"); (v) neither Licensee nor any Licensee affiliate or related party, during the term of this Agreement, will be on any of the Lists or identified as a Blocked Person; and (vi) during the term of this Agreement, neither Licensee nor any Licensee affiliate or related party will sell products, goods or services to, or otherwise enter into a business arrangement with, any person or entity on any of the Lists or identified as a Blocked Person. You agree to notify Sylvan in writing immediately upon the occurrence of any act or event that would render any of these representations incorrect.

[Please initial to acknowledge that you have read and understand this Paragraph 31.1]

The acknowledgments in clauses 31.2 through 31.8 below apply to all franchises and franchises except not to any franchises and franchises that are subject to the state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

### 31.2 Reasonable Restrictions.

You have carefully considered the nature and extent of the restrictions upon you set forth in this Agreement, including, without limitation, the covenants not to compete, the restrictions on assignment, and the rights, obligations, and remedies conferred upon you under this Agreement. You acknowledge that such restrictions, rights, obligations, and remedies: (1) are reasonable, including, but not limited to, their term and geographic scope; (2) are designed to preclude competition which would be unfair to Sylvan; (3) are fully required to protect Sylvan's legitimate business interests; and, (4) do not confer benefits upon Sylvan that are disproportionate to your detriment.

[Please initial to acknowledge that you have read and understand this Paragraph 31.2]

# 31.3 Receipt of Disclosure Document.

You hereby acknowledge that you received from Sylvan its current franchise disclosure document, together with a copy of all proposed agreements related to the sale of the Franchise, at least 14 calendar days prior to the execution of this Agreement or at least 14 days before you paid us any consideration in connection with the sale or proposed sale of the Franchise granted by this Agreement.

## [Please initial to acknowledge that you have read and understand this Paragraph 31.3]

#### 31.4 Receipt of Agreement.

You hereby acknowledge that you received from Sylvan this Agreement with all blanks filled in at least seven calendar days prior to the execution of this Agreement. You represent that you have read this Agreement in its entirety and that you have been given the opportunity to clarify any provisions that you did not understand and to consult with an attorney or other professional advisor. You further represent that you understand the terms, conditions, and obligations of this Agreement and agree to be bound thereby.

## [Please initial to acknowledge that you have read and understand this Paragraph 31.4]

#### 31.5 Independent Investigation.

You acknowledge and represent that you are entering into this Agreement, all attachments hereto, and all ancillary agreements executed contemporaneously with this Agreement, as a result of your own independent investigation of all aspects relating to the Licensed Business, and not as a result of any representations about Sylvan or your reliance on any such representations (if made) by its stakeholders, officers, directors, employees, agents, representatives, independent contractors, or franchisees which are contrary to the terms set forth in this Agreement or any franchise disclosure document required or permitted to be given to you pursuant to applicable law. You have been advised and given the opportunity to independently investigate, analyze, and construe the business opportunity being offered under this Agreement, the terms and provisions of this Agreement, and the prospects for the Licensed Business, using the services of legal counsel, accountants, or other advisers of your own choosing; you have either consulted with these advisors or have deliberately declined to do so. You further recognize that the business venture contemplated by this Agreement involves business risks and that its success will be largely dependent upon your skills and ability as an independent business person. This offering is not a security as that term is defined under applicable federal and state securities laws.

#### [Please initial to acknowledge that you have read and understand this Paragraph 31.5]

#### 31.6 No Representations; No Reliance.

You acknowledge and represent that, except for representations made in Sylvan's current franchise disclosure document, neither Sylvan nor its Affiliates, nor any of their respective stakeholders, officers, directors, employees, agents, representatives, independent contractors, has made any representations, warranties, or guarantees, express or implied, as to the potential revenues, profits, expenses, sales volume, earnings, income, or services of the business venture contemplated under this Agreement, and that you have not relied on any such representations (if made) in making your decision to purchase an Sylvan Learning® and SylvanSync® franchise. You further acknowledge and represent that neither Sylvan nor its representatives have made any statements inconsistent with the terms of this Agreement.

#### [Please initial to acknowledge that you have read and understand this Paragraph 31.6]

# 31.7 No Financial Performance Representations; No Reliance.

You specifically acknowledge that the only financial performance information furnished by Sylvan is set forth in Item 19 of its current franchise disclosure document; that no officer, director, employee, agent, representative or independent contractor of Sylvan is authorized to furnish you with any other financial performance information; that, if they nevertheless do, you will not rely on any such financial performance information given to you by any such individual; and, that if any such individual attempts to or actually does give you any such financial performance information in contravention of this provision, you will immediately communicate such activity to us. For the purpose of this Paragraph 24.G., "financial performance information" means information given, whether orally, in writing, or visually which states, suggests or infers a specific level or range of historic or prospective sales, expenses and/or profits of franchised or Sylvan-owned facilities.

# [Please initial to acknowledge that you have read and understand this Paragraph 31.7]

# 31.8 No Licensure Representations; No Reliance.

You acknowledge that neither Sylvan nor its Affiliates, nor any of their respective stakeholders, officers, directors, employees, agents, representatives, independent contractors, has made any representation or statement on which you have relied regarding your ability to procure any required license, permit, certificate or other governmental authorization that may be necessary or required for you to carry out the activities contemplated by this Agreement

# [Please initial to acknowledge that you have read and understand this Paragraph 31.8]

IN WITNESS WHEREOF, the parties hereto have set their respective hands, as of the Effective Date indicated on the Summary Page.

LICENSOR:	LICENSEE:	
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company,	
By:	By:	
John McAuliffe, Chief Executive Officer	, its Member	

#### CALIFORNIA RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

This Amendment pertains to franchises sold in the State of California that are subject to the California Franchise Investment Law (the "Act") and is for the purpose of complying with California statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act, as set forth in this Amendment, or otherwise. Notwithstanding anything which may be contained in the body of the License Agreement to the contrary, the Agreement is amended to include the following:

Franchisor and Franchisee hereby agree that the License Agreement dated \_\_\_\_\_\_, will be amended as follows:

- 1. Whenever the provisions of this Rider conflict with provisions contained elsewhere in this License Agreement, the provisions of this Rider shall prevail to the extent of such conflict.
- 2. Covenants not to compete which extend beyond the term of the License Agreement may not be enforceable under California law.
- 3. To the extent the California Franchise Investment Law applies, it shall prevail over any conflicting provisions of the License Agreement.
- 4. No disclaimer, questionnaire, clause or statement signed by a franchisee in connection with the commencement of the franchise relationship shall be construed or interpreted as waiving any claim of fraud in the inducement, whether common law or statutory, or as disclaiming reliance on or the right to rely upon any statement made or information provided by any franchisor, broker or other person acting on behalf of the franchisor that was a material inducement to a franchisee's investment. Any statements or representations signed by a franchisee purporting to understand any fact or its legal effect shall be deemed made only based upon the franchisee's understanding of the law and facts as of the time of the franchisee's investment decision. This provision supersedes any other or inconsistent term of any document executed in connection with the franchise.
- 5. The following statement shall be deemed to amend the License Agreement:

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1

LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By:	By:

# HAWAII RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

Law of the State of Hawaii ("the Act") and is for the p Signing this Amendment where the jurisdictional require	the State of Hawaii that are subject to the Franchise Investment surpose of complying with Hawaii statutes and regulations. It is ements of the Act are not met does not subject the parties to ich may be contained in the body of the License Agreement following:
The parties to the License Agreement dated Agreement will be amended as follows:	,, hereby agree that the License
In recognition of the requirements of the Franch Sylvan License Agreement (the "Agreement") agree as for	ise Investment Law of the State of Hawaii, the parties to the ollows:
1. The following statement shall be deem	ed to amend the License Agreement:
commencement of the franchise relationship s applicable state franchise law, including fraud in	at signed or agreed to by a franchisee in connection with the shall have the effect of (i) waiving any claims under any the inducement, or (ii) disclaiming reliance on any statement er person acting on behalf of the franchisor. This provision cuted in connection with the franchise.
LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By: John McAuliffe, Chief Executive Officer	By:

# ILLINOIS RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

Amen of the	osure Act (the "Act") and is for the purpose of adment where the jurisdictional requirements of t	in the State of Illinois that are subject to the Illinois Franchise complying with Illinois statutes and regulations. Signing this he Act are not met does not subject the parties to the provision ontained in the body of the License Agreement to the contrary
Licen	The parties to the License Agreement dated as Agreement will be amended as follows:	, hereby agree that the
1.	Illinois law governs the License Agreement.	
_		nois Franchise Disclosure Act, any provision in a franchise forum outside of the State of Illinois is void. However, a franchise atside of Illinois.
3. Disclo	Your rights upon Termination and Non-Rene osure Act.	ewal are set forth in sections 19 and 20 of the Illinois Franchise
		nois Franchise Disclosure Act, any condition, stipulation on my franchise to waive compliance with the Illinois Franchise
5.	The following statement shall be deemed to a	amend the License Agreement:
	commencement of the franchise relationship applicable state franchise law, including fraud	nent signed or agreed to by a franchisee in connection with the p shall have the effect of (i) waiving any claims under any in the inducement, or (ii) disclaiming reliance on any statemen other person acting on behalf of the franchisor. This provision xecuted in connection with the franchise.
LICE	ENSOR:	LICENSEE:
	AN LEARNING, LLC, aware limited liability company	[], a []
Rv.		Bv·

John McAuliffe, Chief Executive Officer

\_\_\_\_], its [\_\_\_\_

# MARYLAND RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

This Amendment pertains to franchises sold in the State of Maryland that are subject to the Maryland Franchise Registration and Disclosure Law (the "Act") and is for the purpose of complying with Maryland statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Notwithstanding anything which may be contained in the body of the License Agreement to the contrary, the Agreement is amended to include the following:

Franchisor and Franchisee hereby agree that the License Agreement dated \_\_\_\_\_, 20\_\_\_, will be amended as follows:

- 1. Whenever the provisions of this Rider conflict with provisions contained elsewhere in this License Agreement, the provisions of this Rider shall prevail to the extent of such conflict.
- 2. Section 9.3.9 of this License Agreement is hereby deleted in its entirety and the following is inserted in its place:
  - 9.3.9 The transferor shall have executed a general release under seal where required, in a form satisfactory to Sylvan, of any and all claims against Sylvan, its subsidiaries, affiliates and their officers and directors, including, without limitation, claims arising under federal, state, and local laws, rules, and ordinances arising out of, or connected with, the performance of this Agreement; provided, however, any such general release executed by the licensee upon transfer of the License Agreement shall not include any waiver of rights to which the licensee may be entitled under the Maryland Franchise Registration and Disclosure Law.
- 3. The following statement shall be deemed to amend the License Agreement:

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

- 4. Section 30 of this License Agreement is modified as described below.
  - a. Sections 30.2, 30.3, and 30.6 are hereby deleted in their entirety.
  - b. Sections 30.1 and 30.4 are hereby deleted in their entirety and replaced by the following new Sections 30.1 and 30.4:
  - 30.1 LICENSEE HAS CONDUCTED AN INDEPENDENT INVESTIGATION OF THE BUSINESS CONTEMPLATED BY THIS AGREEMENT AND UNDERSTANDS AND ACKNOWLEDGES THAT THE SUCCESS OF THE BUSINESS VENTURE IS DEPENDENT IN PART UPON THE BUSINESS ABILITIES AND PARTICIPATION OF LICENSEE AND ITS EFFORTS AS AN INDEPENDENT BUSINESS OPERATOR. SYLVAN EXPRESSLY DISCLAIMS THE MAKING OF, AND LICENSEE ACKNOWLEDGES THAT IT HAS NOT RECEIVED OR RELIED UPON, ANY WARRANTY OR GUARANTEE, EXPRESS OR IMPLIED, AS TO THE SUITABILITY OF THE SITE AS A SUCCESSFUL LOCATION FOR THE LICENSED BUSINESS.
  - 30.4 LICENSEE ACKNOWLEDGES THAT SYLVAN OR ITS AGENT HAS PROVIDED LICENSEE WITH A FRANCHISE DISCLOSURE DOCUMENT NOT LATER THAN THE EARLIER OF THE FIRST PERSONAL MEETING HELD TO DISCUSS THE SALE OF A LICENSE (IF LICENSEE IS IN IOWA OR NEW YORK), OR FOURTEEN (14) DAYS

# BEFORE THE EXECUTION OF THIS AGREEMENT OR BEFORE ANY PAYMENT OF ANY CONSIDERATION (TEN (10) BUSINESS DAYS IF LICENSEE IS IN NEW YORK).

5. With respect to Section 30 in the License Agreement, as written in the License Agreement, and not amended by this Rider, Franchisor may review the questions and acknowledgments with you during our pre-closing meeting, and may take notes of your verbal responses for our records.

LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By: John McAuliffe, Chief Executive Officer	By:

#### MINNESOTA RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

This Amendment pertains to franchises sold in the State of Minnesota that are subject to the Minnesota Franchise Act (Minn. Stat. Sec. 80C.1 et seq., the "Act") and is for the purpose of complying with Minnesota statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Notwithstanding anything which may be contained in the body of the License Agreement to the contrary, the Agreement is amended to include the following:

The parties to the License Agreement dated \_\_\_\_\_\_\_, \_\_\_\_\_\_, hereby agree that the License Agreement will be amended as follows:

- 1. Whenever the provisions of this Rider conflict with provisions contained elsewhere in this License Agreement, the provisions of this Rider shall prevail to the extent of such conflict.
- 2. Notwithstanding the provisions of Section 11 of this License Agreement, in the event of Licensee's breach of this License Agreement, Licensor may terminate or cancel the License if (i) Licensor gives Licensee written notice setting forth all the reasons for such termination or cancellation to the Licensee at least ninety (90) days in advance of such termination or cancellation, and (ii) Licensee fails to correct the reasons stated for termination or cancellation in the notice within sixty (60) days of receipt of the notice; except that the notice shall be effective immediately upon receipt where the alleged grounds are:
  - (i) Voluntary abandonment of the licensed business or license relationship by the Licensee;
  - (ii) The conviction of the Licensee in a court of competent jurisdiction of an offense directly related to the business conducted pursuant to the License; or
  - (iii) Failure to cure a default under the License Agreement which materially impairs the goodwill associated with the Licensor's trade name, trademark, service mark, logotype or other commercial symbol after the Licensee has received written notice to cure at least twenty-four (24) hours in advance of such termination.
- 3. Licensor may not terminate or cancel a License except for good cause, which shall be defined as failure by the Licensee substantially to comply with those reasonable requirements imposed by the License Agreement including, but not limited to, (1) the bankruptcy or insolvency of Licensee; (2) assignment for the benefit of creditors or similar disposition of the assets of the licensed business; (3) voluntary abandonment of the licensed business; (4) conviction or a plea of guilty or no contest to a charge of violating any law relating to the licensed business; or (5) any act by or conduct of Licensee which materially impairs the goodwill associated with the Licensor's trademark, trade name, service mark, logotype or other commercial symbol.
- 4. Section 3 of the License Agreement shall be amended by adding the following additional Subparagraph 3.3:

Sylvan reserves the right, in its sole discretion, to designate one or more new, modified or replacement Names or Marks for use by Licensee and to require the use by Licensee of any such new, modified or replacement Names or Marks in addition to or in lieu of any previously designated Names or Marks.

Sylvan will indemnify a Minnesota Licensee for damages for which such Licensee is held liable in any proceeding arising out of the use of the "Sylvan Learning Center" mark provided that the Licensee has used the mark properly and has notified Sylvan of any claim against the Licensee within ten days of Licensee's knowledge of such claim. Sylvan shall have sole control of any litigation involving the marks. Sylvan's indemnification obligation shall not apply to any licensee residing outside the State of Minnesota who purchases a license to be located outside of Minnesota.

5. Section 13 of the License Agreement is hereby deleted in its entirety and the following inserted in lieu thereof:

#### 13. <u>Remedies.</u>

Nothing herein contained shall bar Licensor's right to seek to obtain injunctive relief against threatened conduct that may cause it loss or damages, under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary injunctions.

- 6. The provisions of Section 20 of the License Agreement shall not in any way abrogate or reduce any rights of the licensee as provided for in Minnesota Statutes 1992, Chapter 80C, including the right to submit matters to the jurisdiction of the court of Minnesota.
- 7. Minnesota Rule 2860.4400D (the "Rule") provides that it is unfair or inequitable to require a franchisee to assent to a release, assignment, novation or waiver that would relieve any person from liability under Minnesota Statutes. To the extent of a conflict between the Rule and any provision of the Franchise Agreement, the Rule shall prevail.
  - 8. The following statement shall be deemed to amend the License Agreement:

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including, fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed with the franchise.

LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By:	By:

# NEW YORK RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

NEW TORK RIDER TO THE STEVANT	EEARMING, LEC LICENSE AGREEMENT
Franchise Act (New York State General Business Law, A of complying with New York statutes and regulations. Sig of the Act are not met does not subject the parties to the p	the State of New York that are subject to the New York article 33, Sec. 680 et seq., the "Act") and is for the purpose gning this Amendment where the jurisdictional requirements provisions of the Act. Notwithstanding anything which may contrary, the Agreement is amended to include the following:
The parties to the License Agreement dated Agreement will be amended as follows:	,, hereby agree that the License
1. Whenever the provisions of this Rider of Agreement, the provisions of this Rider shall prevail to the	conflict with provisions contained elsewhere in this License extent of such conflict.
rights enjoyed by the Licensee and any cause of action a	on 9.3.4 of the License Agreement, the Parties agree that all arising in its favor from the provisions of Article 33 of the regulations issued thereunder shall remain in force, it being of GBL Sections 687.4 and 687.5 be satisfied.
	hereby amended by adding the following additional sentence d not be considered a waiver of any rights conferred upon B.L."
inserted in its place: "The transferor shall have execute satisfactory to Sylvan, of any and all claims against Sylva including, without limitation, claims arising under federa or connected with, the performance of this Agreement; p and any causes of action arising in its favor from the pro-	ment is hereby deleted in its entirety and the following is a general release under seal where required, in a forman its subsidiaries, affiliates and their officers and directors, l, state, and local laws, rules, and ordinances arising out of rovided, however, that all rights enjoyed by the Franchisee ovisions of Article 33 of the General Business Law of Newmain in force; it being the intent of this proviso that the non-eral Business Law of New York State be satisfied."
terminate this License Agreement for any reason by giving	ne License Agreement is amended to read: "Licensee may g of a one hundred twenty (120) day written notice to Sylvan; e may possess to suspend or discontinue operations due to a
	t is hereby amended by adding the following additional faith effort to ascertain that any such successor or assignations to its licensees."
LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By: John McAuliffe, Chief Executive Officer	By:

# NORTH DAKOTA RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

This Amendment pertains to franchises sold in the State of North Dakota that are subject to the North Da	kota
Franchise Investment Law (the "Act") and is for the purpose of complying with North Dakota statutes and regulati	ons.
Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the partie	s to
the provisions of the Act. Notwithstanding anything which may be contained in the body of the License Agreen	nent
to the contrary, the Agreement is amended to include the follow	ing:
The License Agreement between ("Franchisee" or "You") ("Franchisor") dated as of, 20 (the "Agreement") shall be amer by the addition of the following language, which shall be considered an integral part of the Agreement "Amendment"):	ided

- 1. Whenever the provisions of this Rider conflict with provisions contained elsewhere in this License Agreement, the provisions of this Rider shall prevail to the extent of such conflict.
- 2. To the extent that Section 12 of this License Agreement conflicts with the provisions of North Dakota law regarding restrictive covenants (N.D.C.C., Sec. 9-08-06), such restrictive covenants may not be enforceable by Sylvan, in which event Sylvan shall otherwise retain its rights and remedies at common law.
  - 3. Section 13 is deleted and the following is substituted in lieu thereof:
  - 13. Remedies.

Licensee acknowledges that Sylvan's remedy at law for Licensee's breach of Sections 3, 12.1 or 12.5 would be inadequate and that, accordingly, in such events, Licensor shall be entitled to immediate injunctive relief. In addition to, and not in derogation of, the right to obtain an injunction, Sylvan shall be entitled to any other legally available damages or relief.

4. The License Agreement is hereby modified to add the first paragraph below to the end of the first paragraph of Section 20 and the second paragraph below as a new second paragraph of Section 20:

"Notwithstanding the foregoing, to the extent the requirement that venue lies with courts in Texas in the event that you bring suit against Sylvan violates applicable North Dakota law, such requirement shall be void.

The parties hereto agree that it is in their best interest to resolve disputes between them in an orderly fashion and in a consistent manner. Therefore, the parties irrevocably constitute and appoint the Securities Commissioner, State of North Dakota, State Capitol, Bismarck, North Dakota 58505, to be their true and lawful agent, to receive service of any lawful process in any civil litigation or proceeding arising under this Agreement, and service upon such agent shall have the same force and validity as if personal service had been obtained on the other party; provided that notice of service and a copy of any process served shall be sent by registered or certified mail, addressed to the other party at the address specified under Section 16 hereof."

- 5. Section 24 of the License Agreement is amended by adding a new Section 26.14 reading as follows:
- "Notwithstanding anything to the contrary in this Section 26, to the extent North Dakota law applies, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of the Franchise Investment Law of North Dakota or any rule or order thereunder is void."
- 6. The License Agreement requires the licensee to consent to a limitation of claims within one year. To the extent this requirement conflicts with North Dakota law, North Dakota law will apply.

V.2024.051024

LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By:	By:

# RHODE ISLAND RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

This Amendment pertains to franchises sold in the State of Rhode Island that are subject to the Rhode Island Franchise Investment Act (the "Act") and is for the purpose of complying with Rhode Island statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Notwithstanding anything which may be contained in the body of the License Agreement to the contrary, the Agreement is amended to include the following: The parties to the License Agreement dated \_\_\_\_\_\_, \_\_\_ hereby agree that the License Agreement will be amended as follows: Whenever the provisions of this Rider conflict with provisions contained elsewhere in this License Agreement, the provisions of this Rider shall prevail to the extent of such conflict. Section 20 of the License Agreement is amended by adding the following language at the end 2. thereof: "Notwithstanding the foregoing, to the extent Rhode Island law applies and so requires, nothing herein shall require Licensee to bring suit only in a forum outside of Rhode Island or operate to limit Licensee's rights under the Rhode Island Franchise Act." LICENSOR: LICENSEE: SYLVAN LEARNING, LLC, a Delaware limited liability company By: John McAuliffe, Chief Executive Officer

# SOUTH DAKOTA RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

Franchise Investme Signing this Amer the provisions of t	ent Law (the "Act") and is for the purpos adment where the jurisdictional requiren	e State of South Dakota that are subject to the South Dakota see of complying with South Dakota statutes and regulations, ments of the Act are not met does not subject the parties to the may be contained in the body of the License Agreement following:
The particular will be amended as		, hereby agree that the License Agreement
	Whenever the provisions of this Rider coovisions of this Rider shall prevail to the	onflict with provisions contained elsewhere in this License extent of such conflict.
at least 10 days' r Default providing	notice of termination for monetary defa Licensee at least thirty (30) days to cur	n 11 of the License Agreement providing that Sylvan give fulls by Licensee, Sylvan shall give Licensee a Notice of re such defaults. If Licensee cures a non-payment default trice of Default shall no longer be effective.
3. Paragraph 12.7:	Section 12 of the License Agreement i	s amended by the addition of the following language as
		extent the provisions of Section 12 concerning Dakota law, South Dakota law shall prevail."
4.	Section 13 is amended to read as follows	s:
	12.1 or 12.5 would be inadequate and the entitled to immediate injunctive relief. I	remedy at law for Licensee's breach of Section 3, nat, accordingly, in such events, Licensor shall be n addition to, and not in derogation of, the right to ntitled to any damages available under the law."
5. send of it:	Section 20 of the License Agreement is	amended by the addition of the following language at the
and other matters of and other matters, construction, enfor which designates j South Dakota is vo 6. I determined in antic	of local concern will be governed by the this Agreement and all provisions of this rement, and interpretation under the gourisdiction or venue or requires the liceroid with respect to any cause of action we Every contract in which the amount of eipation thereof is void to that extent except.	has registration, employment, covenants not to compete, laws of the State of South Dakota; but as to contractual is instrument will be and remain subject to the application, verning law of the State of Texas; and (ii) any provision hasee to agree to jurisdiction or venue in a forum outside of which is otherwise enforceable in South Dakota."  If damage or compensation for breach of an obligation is ept the parties may agree therein upon an amount presumed ractical or extremely difficult to fix actual damages. SDCL
LICENSOR:		LICENSEE:
SYLVAN LEARNING a Delaware limited	G, LLC, I liability company	[], a []
By: John McAuli	ffe, Chief Executive Officer	By:

# VIRGINIA RIDER TO THE SYLVAN LEARNING, LLC LICENSE AGREEMENT

Franchising Act Amendment whe of the Act. Notw	nendment pertains to franchises sold in (the "Act") and is for the purpose of contre the jurisdictional requirements of the withstanding anything which may be contamended to include the following:	nplying with Virginia s Act are not met does no	tatutes and regul t subject the part	ations. Signing this ies to the provisions
	ties to the License Agreement dated be amended as follows:		hereby agree	that the License
1. Agreement, the p	Whenever the provisions of this Rider or ovisions of this Rider shall prevail to the			here in this License
2. the Franchise Aş amended:	Notwithstanding anything to the contra greement contains provisions that are in			
	"According to Section 13.1-564 of the V to cancel a franchise without reasonabl the franchise agreement does not consti Virginia Retail Franchising Act or the I	le cause. If any ground itute "reasonable cause,	l for default or to a server or to a	ermination stated in ay be defined in the
3.	The following statement shall be deemed	ed to amend the License	e Agreement:	
	No statement, questionnaire, or acknow with the commencement of the franchis under any applicable state franchise la reliance on any statement made by any of the franchisor. This provision superse with the franchise.	e relationship shall hav aw, including fraud in franchisor, franchise se	e the effect of (i) the inducement, ller, or other per	waiving any claims or (ii) disclaiming son acting on behalf
	NESS WHEREOF, each of the undersconsents to be bound by all of its terms, a			
LICENSOR:		LICENSEE:		
SYLVAN LEARNII a Delaware limit	NG, LLC, ed liability company	[], a []		
By: John McAu	liffe, Chief Executive Officer	By:	]	

1

# WASHINGTON ADDENDUM TO THE LICENSE AGREEMENT, FRANCHISE DISCLOSURE QUESTIONNAIRE, AND RELATED AGREEMENTS

This Amendment pertains to franchises sold in the State of Washington that are subject to the Washington Franchise Investment Protection Act (the "Act") and is for the purpose of complying with Washington statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Notwithstanding anything which may be contained in the body of the License Agreement to the contrary, the Agreement is amended to include the following:

The parties to the License Agreement dated	 hereby	agree that	the License	Agreement
will be amended as follows:				

Notwithstanding anything to the contrary contained in the License Agreement, to the extent that the License Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:

RCW 19.100.180 may supersede the License Agreement in your relationship with the licensor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the License Agreement in your relationship with the licensor including the areas of termination and renewal of your franchise.

In any arbitration involving a franchise purchased in Washington, the arbitration site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration, or as determined by the arbitrator at the time of arbitration. In addition, if litigation is not precluded by the franchise agreement, a licensee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW shall prevail.

A release or waiver of rights executed by a licensee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the licensor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the License Agreement or elsewhere are void and unenforceable in Washington.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

IN WITNESS WHEREOF, each of the undersigned hereby acknowledges having read this Addendum, understands and consents to be bound by all of its terms, and agrees it shall become effective as of the Effective Date.

LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	[], a []
By:	By:

# ATTACHMENT A

# **TERRITORY**

Section 1.3 The boundaries of the Territory include the following portions of the zip codes and the map, where boundaries of the map control if there is a conflict with the zip codes: .

		[To Be Attached.]
Section 1.6	Primary Center address:	
Section 1.7	Additional Center address, if applied	cable:
	ITNESS WHEREOF, the undersigned we as of the Effective Date.	l, intending to be legally bound, have executed this Attachment
LICENSOR:		LICENSEE:
SYLVAN LEARN a Delaware lim	NING, LLC, ited liability company	a limited liability company
By:John M	McAuliffe, Chief Executive Officer	By:, its Member

# ATTACHMENT B

# LICENSEE'S OWNERS AND KEY PERSONNEL

A.	The following is a list of all shareholders, partners, members, or other investors owning a direct or indirect
	interest in the Licensee ("Owners"), and a description of the nature of their interest, each of whom shall
	execute the Undertaking and Guaranty substantially in the form set forth in Attachment C to the License
	Agreement:

NAME, ADDRESS, TELEPHONE NUMBER, AND EMAIL	OWNERSHIP INTEREST IN LICENSEE	NATURE OF INTEREST
	%	
	%	
	%	
	%	
	%	
	%	

B.	The following is a list of all of Licensee's Owners and key personnel, each of whom shall execute the
	Confidentiality Agreement and Non-Competition Agreement substantially in the form set forth in Attachment
	<u>D</u> to the License Agreement:

NAME, ADDRESS, TELEPHONE NUMBER, AND EMAIL	POSITION

	Email Address: .
D.	Licensee represents to Licensor that the persons identified in this <u>Attachment B</u> , <u>Sections A</u> and <u>B</u> re

D. Licensee represents to Licensor that the persons identified in this <u>Attachment B</u>, <u>Sections A</u> and <u>B</u> reflect a true and correct listing of the shareholders, partners, members, or other persons/companies owning a direct or indirect interest in the Licensee and a true and correct description of the nature of their interest.

1

a ,	limited liability company
Ву:	its Member

Licensee's Managing Principal is:

C.

LICENSEE:

#### ATTACHMENT C

#### UNDERTAKING AND GUARANTY

By virtue of executing a Sylvan License Agreement ("<u>License Agreement</u>") dated , ("<u>Licensee</u>") has acquired the right and franchise from Sylvan Learning, LLC ("<u>Licensor</u>") to establish and operate a SYLVAN and SYLVANSYNC licensed business ("<u>Licensed Business</u>") and the right to use in the operation of the Licensed Business the Licensed Marks and the Sylvan System, as they may be changed, improved, and further developed from time-to-time in Licensor's sole discretion. Terms not defined herein have the same meaning as defined in the License Agreement.

Pursuant to the terms and conditions of the License Agreement, each of the undersigned hereby acknowledges and agrees as follows:

- 1. I have read the terms and conditions of the License Agreement and acknowledge that the execution of this Undertaking and Guaranty and the undertakings of the Owners in the License Agreement are in partial consideration for, and a condition to, the granting of the rights under the License Agreement. I understand and acknowledge that Licensor would not have granted such rights without the execution of this Undertaking and Guaranty and the other undertakings of the Owners in the License Agreement.
- 2. I own a beneficial interest in the Licensee, and I am included within the term "Owner" as defined in the License Agreement.
- 3. I, individually and jointly and severally with the other Owners, hereby make all of the covenants, representations, warranties, and agreements of the Owners set forth in the License Agreement, and agree that I am obligated to and will perform thereunder, including, without limitation, the provisions regarding the use of confidential information in <a href="Article 4">Article 4</a>, compliance with the License Agreement and standards of operation in <a href="Article 7">Article 7</a>, the transfer provisions in <a href="Article 12">Article 9</a>, the restrictive covenants in <a href="Article 12">Article 12</a>, the indemnification obligations in <a href="Article 14">Article 14</a>, the choice of law and venue provisions in <a href="Article 21">Article 21</a>, and dispute resolution provisions in Article <a href="Article 23">Article 23</a>.
- 4. I, individually and jointly and severally with the other Owners, unconditionally and irrevocably guarantee to Licensor and its successors and assigns that all obligations of the Licensee under the License Agreement will be punctually paid and performed. Upon default by the Licensee or upon notice from Licensor, I will immediately make each payment and perform each obligation required of the Licensee under the License Agreement. Without affecting the obligations of any Owner under this or any other Undertaking and Guaranty, Licensor may, without notice to any Owner, waive, renew, extend, modify, amend, or release any indebtedness or obligation of the Licensee or settle, adjust, or compromise any claims that Licensor may have against the Licensee. I waive all demands and notices of every kind with respect to the enforcement of this Undertaking and Guaranty, including notices of presentment, demand for payment or performance by the Licensee, any default by the Licensee or any guarantor, and any release of any guarantor or other security for this Undertaking and Guaranty or the obligations of the Licensee. Licensor may pursue its rights against me without first exhausting its remedies against the Licensee and without joining any other guarantor and no delay on the part of Licensor in the exercise of any right or remedy will operate as a waiver of the right or remedy, and no single or partial exercise by Licensor of any right or remedy will preclude the further exercise of that or any other right or remedy. Upon Licensor's receipt of notice of the death of any Owner, the estate of the deceased will be bound by the foregoing Undertaking and Guaranty, but only for defaults and obligations under the License Agreement existing at the time of death, and in that event, the obligations of the Owners who survive such death will continue in full force and effect.
- 5. No modification, change, impairment, or suspension of any of Licensor's rights or remedies shall in any way affect any of my obligations under this Undertaking and Guaranty. If the Licensee has pledged other security or if one or more other persons have personally guaranteed performance of the Licensee's obligations, I agree that Licensor's release of such security will not affect my liability under this Undertaking and Guaranty.
- 6. I agree that any notices required to be delivered to me will be deemed delivered at the time delivered by hand; at the time delivered through e-mail; one Business Day after delivery by Express Mail or other recognized, reputable overnight courier; or three Business Days after placement in the United States Mail by Registered or Certified Mail, Return Receipt Requested, postage prepaid and addressed to the address identified on the signature line below. I may change this address only by delivering to Licensor written notice of the change.

- 7. I understand that Licensor's rights under this Undertaking and Guaranty shall be in addition to, and not in lieu of, any other rights or remedies available to Licensor under applicable law.
- 8. I agree to be bound individually to all of the provisions of the License Agreement including, without limitation, the litigation and dispute resolution provisions set forth in <u>Article 23</u> and I irrevocably submit to the jurisdiction of the state and federal courts serving the judicial district in which Licensor's principal headquarters are located at the time litigation is commenced. I hereby irrevocably submit to the exclusive jurisdiction of such courts and specifically waive any objection I may have to either the jurisdiction or exclusive venue of such courts.
- 9. I WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, INVOLVING LICENSOR, WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THE LICENSE AGREEMENT, THE PERFORMANCE OF ANY PARTY UNDER THE LICENSE AGREEMENT, AND/OR THE OFFER OR GRANT OF THE FRANCHISE.
- 10. This Agreement shall be construed under the laws of the State of Texas. The only way this Agreement can be changed is in writing signed by Licensor. Any capitalized terms contained in but not defined by this Guaranty and Personal Undertaking shall have the same meaning prescribed to that word in the License Agreement.
- 11. Should this Agreement be signed or endorsed by more than one person or entity, all of the obligations herein contained shall be considered the joint and several obligations of each signatory.

IN WITNESS WHEREOF, the undersigned, intending to be legally bound, has executed this Undertaking and Guaranty to be effective on the day and year first written above.

OWNER	

#### ATTACHMENT D

#### CONFIDENTIALITY AND NON-COMPETITION AGREEMENT

In consideration of my being an Owner of [	] ("Licensee") and by virtue of executing a Sylvan
Learning® and SylvanSync® License Agreement dated [	[ "License Agreement") and this Confidentiality
and Non-Competition Agreement (herein, "Agreement"), and	d other good and valuable consideration, the receipt and
sufficiency of which is acknowledged, I hereby acknowledge	e and agree as follows:

- Through the License Agreement, Licensee has acquired the right and license from SYLVAN LEARNING, LLC ("<u>Licensor</u>") to establish and operate a Sylvan Learning® and SylvanSync® license facility ("<u>Licensed Business</u>") and the right to use in the operation of the Licensed Business the Licensed Marks and the Sylvan System, as they may be changed, improved, and further developed from time-to-time in Licensor's sole discretion.
- 2. Licensor possesses certain proprietary and confidential information, knowledge, elements, and know-how which is utilized in the operation of the Sylvan System, including, without limitation, the Confidential Operations Manual, proprietary products and services, intellectual property, confidential information and other techniques and know-how which concerns Licensor's systems of operation, programs, services, products, customers, practices, materials, books, records, financial information, manuals, computer files, databases, or software, as further described in the License Agreement.
- 3. In addition to the Confidential Information identified in the License Agreement, any and all manuals, trade secrets, copyrighted materials, methods, information, knowledge, know-how, and techniques which Licensor specifically designates as confidential shall be deemed to be "Confidential Information" for purposes of this Agreement.
- 4. I acknowledge that, in my position with the Licensee, Licensor and Licensee have or will furnish me with valuable specialized training and will disclose Confidential Information to me in furnishing to me the training program and subsequent ongoing training and other general assistance during the term of this Agreement.
- 5. I will not acquire any interest in the Confidential Information, other than the right to utilize it in the operation of the Licensed Business during the term hereof, and I acknowledge that the use or duplication of the Confidential Information for any use outside the Sylvan System would constitute an unfair method of competition.
- 6. The Confidential Information is proprietary, involves trade secrets of Licensor, and is disclosed to me solely on the condition that I agree, and I do hereby agree, that I shall hold in strict confidence all Confidential Information and all other information designated by Licensor as confidential. Unless Licensor otherwise agrees in writing, I will not disclose and/or use the Confidential Information except in connection with the operation of the Licensed Business as an Owner of the Licensee, and then only in strict compliance with the Confidential Operations Manual and Sylvan System and only to such employees having a need to know; I will not directly or indirectly imitate, duplicate, or "reverse engineer" any Confidential Information or any other information designated by Licensor as confidential or aid any third party in such actions; and I will continue not to disclose and/or use any Confidential Information or any other information designated by Licensor as confidential even after I cease to be in that position unless I can demonstrate that such information has become generally known or easily accessible other than by the breach of an obligation of Licensee under the License Agreement.
- 7. Except as otherwise approved in writing by Licensor, I will not (either directly or indirectly, for myself or through, on behalf of, or in conjunction with any person, or legal entity) at any time while I am the Owner of, employed by, or associated with the Licensee, or at any time during the uninterrupted two (2)-year period (which will be tolled during any period of noncompliance) after I cease to be the Owner of, employed by, or associated with the Licensee (or the two (2)-year period after the expiration or earlier termination of the License Agreement, whichever occurs first):

- (a) Divert or attempt to divert any present or prospective customer of any Sylvan Learning® and SylvanSync® business to any Competitive Business, by direct or indirect inducement or otherwise, or do or perform, directly or indirectly, any other act that is harmful, injurious, or prejudicial to the goodwill associated with the Licensed Marks and the Sylvan System defined and described in the License Agreement; or
- (b) Own, maintain, advise, operate, engage in, be employed by, make loans to, invest in, provide any assistance to, or have any direct or indirect interest in (as owner or otherwise) or relationship or association with, any Competitive Business other than Sylvan Learning® and SylvanSync® businesses pursuant to a then-currently effective License Agreement with Licensor. While I am the Owner of, employed by, or associated with the Licensee, this restriction shall apply to any location within the United States, its territories or commonwealths, and any other country, province, state, or geographic area in which Licensor or its Affiliates have used, sought registration of, or registered the Licensed Marks or similar marks, or have operated or licensed others to operate a business under the Sylvan System or the Licensed Marks or similar marks. After I cease to be the Owner of, employed by, or associated with the Licensee (or after the expiration or earlier termination of the License Agreement, whichever occurs first), this restriction shall apply to any Competitive Business that either: (i) fifty (50) miles from the Territory boundary or less than twenty-five (25) miles from any other corporate-affiliate-owned or licensee-owned Sylvan Learning Center business, other than as an authorized licensee of another Sylvan System and/or SylvanSync licensed business in existence or under development at the time of such termination or transfer; or (ii) delivers services through the internet or mobile channels to customers within a 25-mile radius of the Sylvan Learning Center business.

I acknowledge that for purposes of this Agreement, "Competitive Business" means any business or enterprise that is the same or similar to Sylvan Learning® and SylvanSync® businesses, including without limitation any business or enterprise, franchised or non-franchised, that operates or grants franchises or licenses for the operation of any educational business offering individualized diagnostic tests or academic or prescriptive educational programs which are designed to be taught, supervised, or administered by trained instructors, or in any educational business offering programs or services competitive with programs or services available at any Sylvan Learning® and SylvanSync® business.

- 8. I agree that each of the foregoing covenants shall be construed as independent of any other covenant or provision of this Agreement. If all or any portion of a covenant in this Agreement is held unreasonable or unenforceable by a court or agency having valid jurisdiction in an un-appealed final decision to which Licensor is a party, I expressly agree to be bound by any lesser covenant subsumed within the terms of such covenant that imposes the maximum duty permitted by law, as if the resulting covenant were separately stated in and made a part of this Agreement.
- 9. I understand and acknowledge that Licensor shall have the right, in its sole discretion, to reduce the scope of any covenant set forth in this Agreement, or any portion thereof, without my consent, effective immediately upon receipt by me of written notice thereof, and I agree to comply forthwith with any covenant as so modified.
- 10. Licensor is a third-party beneficiary of this Agreement and may enforce it, solely and/or jointly with the Licensee. I am aware that my violation of this Agreement will cause Licensor and the Licensee irreparable harm; therefore, I acknowledge and agree that the Licensee and/or Licensor may apply for the issuance of an injunction preventing me from violating this Agreement, and I agree to pay the Licensee and Licensor all the costs it/they incur(s), including, without limitation, legal fees and expenses, if this Agreement is enforced against me. Due to the importance of this Agreement to the Licensee and Licensor, any claim I have against the Licensee or Licensor is a separate matter and does not entitle me to violate or justify any violation of this Agreement.
- 11. This Agreement shall be construed under the laws of the State of Texas. The only way this Agreement can be changed is in writing signed by both the Licensee and me. Any capitalized terms contained in but not defined by this Confidentiality and Non-Competition Agreement shall have the same meaning prescribed to that word in the License Agreement.

- 12. With respect to all claims, controversies and disputes, I irrevocably consent to personal jurisdiction and submit myself to the jurisdiction of the federal and state courts serving the judicial district in which Licensor's principal headquarters are located at the time litigation is commenced. I hereby irrevocably submit to the exclusive jurisdiction of such courts and specifically waive any objection I may have to either the jurisdiction or exclusive venue of such courts. Further, I acknowledge that this Agreement has been entered into in the State of Texas, and that I am to receive valuable information emanating from Licensor's headquarters in Bedford, Texas. In recognition of the information and its origin, I hereby irrevocably consent to the personal jurisdiction of the state and federal courts of Texas. Notwithstanding the foregoing, I acknowledge and agree that Licensor may bring and maintain an action against me in any court of competent jurisdiction for injunctive or other extraordinary relief against threatened conduct that will cause it loss or damages, under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary and permanent injunctions.
- 13. I WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, INVOLVING LICENSOR, WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THIS AGREEMENT, THE LICENSE AGREEMENT, THE PERFORMANCE OF ANY PARTY UNDER THE LICENSE AGREEMENT, AND/OR THE OFFER OR GRANT OF THE LICENSE.
- 14. Should this Agreement be signed or endorsed by more than one person or entity, all of the obligations herein contained shall be considered the joint and several obligations of each signatory.

	nding to be legally bound, have executed this Confidentiality
and Non-Competition Agreement to be effective on the da	y and year first written above.
ACKNOWLEDGED BY LICENSEE:	
a limited liability company	
a minica naomity company	
By:	
, its Member	

# ATTACHMENT E

Acknowledgment of Request to Contribute to Advertising Cooperative

Lice	asee:
Lice	ase Agreement and Center #:
1.	Recitals:
	a) A local advertising cooperative (the "Co-op") may exist in Licensee's Direct Marketing Area ("DMA"), having as its purpose, among others, the joint funding of advertising purchases to market the services of Sylvan Network licensed businesses in the DMA.
	b) Pursuant to Paragraph 7.8.2 of the License Agreement, upon Sylvan's request, Licensee shall contribute to such Co-op in accordance with the terms and conditions of Licensee's License Agreement.
2.	Request to Participate:
	Pursuant to Paragraph 7.8.2 of the License Agreement, Sylvan Learning, LLC hereby requests that Licenses contribute to such Co-op in accordance with the terms and conditions of Licensee's License Agreement.
3.	Contribution Alternatives:
	Licensee acknowledges that Sylvan has requested that it make a monthly contribution to the Co-op, pursuant to either one of the following funding alternatives:
	Alternative A: Monthly contribution on the same basis as a majority of other Co-op members;
	Alternative B: 8% of Licensee's Gross Revenues, such funding not to exceed the maximum of Forty Thousand Dollars (\$40,000) in each calendar year, as more extensively described in Paragraph 7.8.1 of the Agreement.
4.	Licensee acknowledges that it has read this form and understands that Sylvan has requested that Licensee contribute to the Co-op in Licensee's DMA. Licensee understands that it is not making a contribution election by executing this document.
LICI	ENSEE:
a	limited liability company
By:	, its Member

#### ATTACHMENT F

## <u>Contact Center</u> [New] Licensee Participation Agreement

This Contact Center Licensee Participation Agreement (the "Agreement") is entered into as of (the "Effective Date") by and between Sylvan Learning, LLC ("Sylvan"), a Delaware limited liability company, with a principal place of business at 2350 Airport Freeway, Suite 505, Bedford, TX 76022, and ("Licensee"), with a principal place of business at , Center #: <u>AgrmtNo.</u>, Payment Option (Model A or B): \_\_\_\_\_.

WHEREAS, Sylvan is a leading provider of remedial, enrichment and test preparation tutoring services to students in Kindergarten through Grade 12 through a network of independent franchisees; and

WHEREAS, Sylvan has entered into an agreement (the "Master Services Agreement") with WSOL, LLC ("Working Solutions"), under which Working Solutions shall provide contact center services ("Services") to participating Sylvan franchisees, including Licensee; and

WHEREAS, Licensee is an independent licensee of Sylvan authorized to provide Sylvan services and programs in one (1) or more licensed Territory or Territories, pursuant to one (1) or more Sylvan License Agreement(s) (the "License Agreement(s)"), as set forth in Attachment A, which is attached hereto and incorporated herein by reference;

WHEREAS, Licensee desires to have Working Solutions provide Services on its behalf in connection with certain Sylvan Learning Centers (each a "Participating Center" and, collectively, the "Participating Centers") within the Territory or Territories, such Participating Center(s) listed on Attachment A to this Agreement, pursuant to the terms and conditions contained herein.

NOW, THEREFORE, in consideration of the mutual promises contained herein and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties agree as follows:

- 1. <u>Services.</u> During the Term hereof, Sylvan shall cause Working Solutions to provide, on behalf of Licensee for the Territories and Participating Center(s) set forth in this Agreement, the Services specified in Attachment A which is attached hereto and incorporated herein by reference, and which may be revised from time to time as indicated in Sylvan's operations manual or other written notice. Licensee acknowledges that Working Solutions is responsible for all Services; Sylvan assumes no liability hereunder for any failure by Working Solutions to perform the Services.
- 2. <u>Payment; Billing.</u> Licensee shall select one of the two payment options set forth below each Participating Center by selecting the full-service (low variable fee) option ("Model A"), or the full service (low fixed fee) option ("Model B") as indicated in the header of this Agreement:

Fee Type	Model A - Low Variable Fee	Model B – Low Fixed Fee
Flat Fee	\$700 per Month	\$350 per Month
Variable Fee	\$10 per Inquiry	\$30 per Inquiry

Licensee shall have the option to modify its selection one (1) time for each Participating Center during the Initial Term hereof by providing thirty (30) days' prior written notice to Sylvan. Licensee may elect to add one (1) or more Center(s) to participate to receive Services pursuant to this Agreement after the Effective Date hereof by providing thirty (30) days' prior written notice to Sylvan (email notice shall be deemed sufficient). Sylvan shall invoice Licensee monthly for Services provided to each Participating Center under this Agreement during any portion of the preceding month; Sylvan shall not prorate the payments due under any circumstances. Payments shall be due in accordance with Licensee's obligations concerning payment of accounts as set forth in the License Agreement(s).

If Licensee closes one (1) or more of the Participating Centers during the Term hereof, Licensee shall provide at least ninety (90) days' prior written notice to Sylvan and Working Solutions prior to such closure(s), and Licensee shall not be invoiced for Services following the month that such Participating Center(s) closed. For avoidance of doubt, Licensee shall be responsible for all amounts billed through the time of closure.

If Licensee fails to provide such notice as required above, and Sylvan terminates Licensee's License Agreement(s) as a result of such closure(s), Licensee shall be charged an "Early Termination Fee" equal to the sum of the payments

1

SYLVAN LEARNING, LLC Franchise Disclosure Document | 2024 owed by Licensee under this Agreement for each such closed Participating Center during the two (2) months preceding the closure(s).

3. <u>Term.</u> Unless terminated earlier pursuant to the provisions of this Agreement, the term of this Agreement shall commence on the Effective Date and terminate after the expiration of twelve (12) months (the "Initial Term"). This Agreement may be renewed at the parties' discretion by mutual written agreement; provided, however, that this Agreement shall automatically renew in twelve-month increments if either party does not provide notice of its intention not to renew prior to the expiration of the Initial Term or any then-current renewal term. The Initial Term and any renewal terms of this Agreement shall collectively be referred to as the "Term" of this Agreement.

## 4. **Termination.**

- a) <u>Termination of License Agreement</u>. This Agreement shall automatically terminate in the event that Licensee's License Agreement(s) is or are terminated. If Licensee retains at least one (1) of the License Agreements following the termination of one (1) or more other License Agreement(s), then this Agreement shall remain in full force and effect as it pertains to Licensee's remaining Territory or Territories.
- b) <u>Termination of Master Services Agreement</u>. This Agreement shall automatically terminate in the event that the Master Services Agreement is terminated, and Sylvan does not transfer the rights and obligations of Working Solutions to an alternative third-party contact center services provider. Sylvan agrees to provide Licensee with as much notice as practicable in the event that the Master Services Agreement is terminated.
- c) <u>Termination for Convenience</u>. Licensee may not terminate this Agreement during the Initial Term hereof. Thereafter, during any renewal term, Licensee may terminate this Agreement for convenience by providing Sylvan with thirty (30) days' prior written notice.
- d) <u>Termination by Sylvan</u>. Sylvan shall have the right at any time during the Term of this Agreement, upon at least one (1) day's prior written notice to Licensee, to discontinue the Services provided by Working Solutions and to terminate this Agreement.
- 5. <u>Agave Pay Gateway Prerequisite</u>. Licensee must set up and enable an Agave Pay online payment gateway account for each Participating Center pursuant to a separate agreement between Licensee and Agave Pay in order for Working Solutions to provide Services to the Participating Center(s). Licensee must comply with all modifications to Sylvan's standards, specifications, and systems upon thirty (30) days' prior written notice.
- 6. Monthly Survey. Sylvan may require Licensee to take an online satisfaction survey (the 'Survey") each month regarding the Services received by Licensee under this Agreement for the previous month in connection with each Participating Center. The Survey may be accessible by a hyperlink delivered by Sylvan to Licensee via email, or via the Sylvan Advisor website, at Sylvan's sole discretion. Sylvan intends to use the results of the Survey to improve the quality of the Services delivered by Working Solutions (or any alternative third-party contact center services provider) to Licensee. Licensee must complete and submit the Survey to Sylvan within ten (10) days of the date that Sylvan opens the Survey to Licensee and notifies Licensee in writing that the Survey is ready. Licensee, or an employee of Licensee acting on Licensee's behalf, must submit to Sylvan a separate Survey for each Participating Center enrolled under this Agreement. Failure to timely complete and submit the Survey for each Participating Center shall constitute a material breach of this Agreement.
- 7. No Guarantee or Warranty. SYLVAN DOES NOT GUARANTY OR WARRANT THAT THE SERVICES PROVIDED BY WORKING SOLUTIONS WILL RESULT IN ANY MINIMUM VOLUME OF SALES OR ANY MINIMUM NUMBER OF CONVERSIONS FROM CALLS TO SALES, NOR THAT WORKING SOLUTIONS WILL BE ABLE TO RESOLVE ALL ISSUES PRESENTED BY FRANCHISEES. EXCEPT AS OTHERWISE PROVIDED IN THIS AGREEMENT, ALL SERVICES ARE BEING PROVIDED BY WORKING SOLUTIONS "AS IS." SYLVAN EXPRESSLY DISCLAIMS ALL REPRESENTATIONS, WARRANTIES, AND CONDITIONS, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OR CONDITIONS OF MERCHANTABILITY, OR FITNESS FOR A PARTICULAR PURPOSE.
- 8. <u>Exclusive Remedy</u>. Licensee's exclusive remedy against Sylvan relating to the performance of Services hereunder by Working Solutions shall be the right to terminate this Agreement in accordance with Paragraph 4(c) above.
- 9. <u>Default</u>. A breach by Licensee of any of its obligations set forth in this Agreement shall constitute a default by Licensee under the License Agreement(s).

- 10. <u>Licensee Cooperation</u>. Licensee acknowledges that Sylvan shall manage the relationship between Licensee and Working Solutions, or any alternative third-party contact center services provider. Notwithstanding the foregoing, Licensee agrees to cooperate with Working Solutions in the performance of this Agreement and agrees to provide Working Solutions with all information reasonably necessary for Working Solutions to provide Services on behalf of Licensee.
- 11. <u>Notice</u>. Any notice from Sylvan to Licensee, or from Licensee to Sylvan, shall be deemed duly served if delivered, mailed, or sent in accordance with the terms set forth in the License Agreement(s), or if sent by email, provided that the delivery and receipt of such email is confirmed by recipient or recipient's email system.
- 12. Assignment. This Agreement shall not be assigned by Licensee without the prior written consent of Sylvan. If Licensee transfers its Sylvan Learning business(es) to a third-party purchaser approved by Sylvan, such third-party purchaser shall assume Licensee's rights and obligations under this Agreement for the remainder of the Term hereof. This Agreement may be assigned by Sylvan to Sylvan's successor or to any subsidiary or affiliate of Sylvan and their respective successors. Licensee acknowledges that Sylvan may also transfer the rights and obligations of Working Solutions hereunder to a separate third-party contact center services provider in the event that Working Solutions fails to meet its obligations to Sylvan under the Master Services Agreement; provided, however, that such alternative third-party services provider shall be obligated to perform at a minimum the obligations set forth in Attachment A hereto, and to meet or exceed the service levels outlined in Attachment A hereto. Sylvan shall provide notice to Licensee of the migration to the alternative third-party contact center services provider as soon as is practicable.
- 13. <u>Integration</u>. This Agreement constitutes the entire agreement between the parties related to the subject matter hereof and supersedes all prior agreements and understandings. This Agreement may be amended or modified only in writing executed by both parties.
- 14. <u>Binding Agreement</u>. This Agreement shall inure to the benefit of, and be binding on Sylvan's and Licensee's respective heirs, executors, administrators, successors, and assigns.
- 15. <u>Litigation</u>. If Sylvan or Licensee brings an action in connection with, or to enforce its rights under, this Agreement, such litigation shall be brought in accordance with the venue provisions set forth in the License Agreement(s).
- 16. <u>Waiver</u>. No delay or omission on the part of any party hereto in exercising any right hereunder shall operate as a waiver of such right or any other right under this Agreement.
- 17. <u>Headings</u>. The Article and Section headings used herein are for reference and convenience only and shall not limit or control any term or provision of this Agreement or the interpretation or construction thereof.
- 18. <u>Applicable Law</u>. This Agreement shall be deemed to be entered into and shall be interpreted and construed in accordance with the internal laws of the State of Texas, without giving effect to its conflict of laws rules.
- 19. <u>Attachments</u>. All schedules, attachments, or attachments, if any, referred to in or attached to this Agreement are and shall be deemed to be an integral part of this Agreement as if fully set forth herein.
- 20. <u>Signatures</u>. A faxed, photocopied, scanned-in PDF format, or electronic signature shall have the same effect for all purposes as an ink-signed original.

IN WITNESS WHEREOF, the parties have hereunto set their hands on the date(s) set forth below.

LICENSEE:

Elem 1991.	ETCE: ISEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company
By: John McAuliffe, Chief Executive Officer	By:, its Member

LICENSOR.

#### ATTACHMENT A to CONTACT CENTER PARTICIPATION AGREEMENT

WORKING SOLUTIONS will provide Licensee with the following baseline services:

- Telephone Number Provision WORKING SOLUTIONS will provide local telephone number(s) to Licensee for each Participating Center. Licensee must use and advertise the telephone number(s) provided by WORKING SOLUTIONS as its inquiry telephone number on any marketing media controlled by the Licensee.
- Inbound Inquiry Handling WORKING SOLUTIONS will handle inbound calls and web-form inquiries
  from consumers.
- **Web-form Follow-up** WORKING SOLUTIONS will follow-up via telephone with consumers who submit inquiries via web-form.
- Outbound Follow-Up Calls WORKING SOLUTIONS will make consistent outbound follow-up calls to
  reach each consumer within a specified period of time from consumer's initial inquiry (the "Response
  Period"). If WORKING SOLUTIONS is unable to reach a consumer within the Response Period,
  WORKING SOLUTIONS will provide consumer's information to Licensee to allow Licensee to reach out
  to consumer directly. WORKING SOLUTIONS will not contact consumer following transfer of
  consumer's information to Licensee.
- **Disposition Reporting** WORKING SOLUTIONS will provide to Sylvan data containing the disposition of each inbound call, web-form, and email contact following the expiration of the applicable Response Period for each inquiry handled by WORKING SOLUTIONS. Sylvan will provide to Licensee, via SLCLink, a report concerning the disposition of contacts related to its Participating Center(s).
- Call Recordings WORKING SOLUTIONS will provide Sylvan with access to recordings of all calls between agents and consumers. Sylvan will provide Licensee with access to all call recordings related to its Participating Center(s) for thirty (30) days from the date of the call.

#### **Hours of Operation**

#### **Normal Hours:**

Dedicated Agent Services: Monday through Friday, 8:00 a.m. – 8:00 p.m., and Saturday 8:00 a.m. - 12:00 p.m. (local time of the applicable Territory)

## **Holiday Hours:**

WORKING SOLUTIONS may be closed during normal operating hours on the following Holidays: New Year's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Eve, Christmas Day, and New Year's Eve; if so, WORKING SOLUTIONS will provide a closed message announcement and voicemail box to be used in connection with all inbound calls.

#### **Facilities**

Services are to be provided from WORKING SOLUTIONS' facilities and other remote locations, as designated by WORKING SOLUTIONS.

Pursuant to the Master Services Agreement between WORKING SOLUTIONS and Sylvan, WORKING SOLUTIONS shall comply with the following service levels (to be measured monthly by Sylvan):

- Percentage of calls answered within 20 seconds > 80%
- Average abandonment rate of all Sylvan calls <5%
- Forced Busies or disconnects = 0
- Average time on-hold of all Sylvan calls = 20 seconds
- Web-form response time < 15 minute

#### ATTACHMENT G

SylvanSync Software License Agreement and Terms of Use



#### SYLVANSYNC EDUCATIONAL SOFTWARE

License and Terms of Use Agreement

Parties: Sylvan Learning, LLC, as "Licensor," and , as "Licensee"

IMPORTANT - READ CAREFULLY: This License and Terms of Use (this "License") sets forth the rights and obligations of a Sylvan licensee ("Franchisee" or "Licensee") and of each guarantor (the "Guarantors," collectively with Licensee, "You" or "Your") of Licensee's obligations pursuant to a License Agreement between the Licensee and Sylvan ("License Agreement"), on the one hand and Sylvan Learning, LLC ("Sylvan"), on the other hand, for the license of Sylvan's proprietary educational software for use in connection with delivering SylvanSync services ("SylvanSync Services"), installed on servers and/or on computer systems, and any associated user manuals and other explanatory documentation or materials, regardless of form or media, (all of the foregoing, including all versions thereof and modifications thereto, are collectively referred to as the "Software"), and the terms governing Your use of the Software.

- 1. LICENSE GRANT. Sylvan hereby grants You, as an active Sylvan and SylvanSync franchisee, limited rights (the "Grant") to use the Software solely for the operation of your Sylvan Learning business. The Software is licensed, not sold. Any third-party components embedded, included or otherwise provided with the Software that may be used when accessing or using the Software are also subject to the terms of this License.
- 2. LICENSE RESTRICTIONS. The License granted in Section 1 above is subject to the following restrictions. You may not: (a) copy all or any part of the Software; (b) permit use of the Software other than for its designated purpose, such prohibited uses to include, without limitation, use of the Software to sell or provide tutoring services except in the operation of SylvanSync Services in connection with your Sylvan Learning Center business in accordance with your Sylvan and SylvanSync franchise License Agreement (the "License Agreement"); (c) copy the Software, other than to make one copy of the Software for archival purposes; (d) sell, give, or otherwise distribute the Software to a third party; (e) separate the components of the Software for use individually or separately from the Software as a whole; (f) transfer, modify, reverse engineer, decompile, or disassemble the Software; (g) create derivative works or reduce the object code of the Software to human readable form; (h) sell, rent, lease, lend, distribute, market or create a lien or security interest against the Software; (i) transfer all or any part of Your rights under this License; or (j) remove any copyright, trademark or other proprietary or confidentiality notices, labels or markings. If Sylvan has provided the Software to You for evaluation and testing purposes, whether a pre-release or commercial version, You may only use the Software to determine its compatibility and/or feasibility of use for the evaluation period designated by Sylvan.

You may make one copy of the written documentation that accompanies the Software in support of authorized use of the Software. You may make one archival copy of the Software.

3. RESERVATION OF RIGHTS/OWNERSHIP CONCERNING THE SOFTWARE. ALL RIGHTS NOT EXPRESSLY GRANTED HEREIN ARE RESERVED BY SYLVAN. The Software is protected by all United States and international patent and copyright laws and treaties, as well as all other intellectual property laws and

treaties. All title, copyrights, patent and other proprietary rights, including trade secrets rights, in and to the Software and all accompanying materials, and any copies thereof in any media or format are owned by Sylvan, its designees or suppliers, as applicable. YOU MAY NOT COPY ANY PART OF THE SOFTWARE OR TRANSFER THE SOFTWARE TO ANOTHER COMPUTER OR NETWORK.

- 4. TERM AND TERMINATION. Unless earlier terminated in accordance with this License, the term of this License shall be coterminous with Your License Agreement, after which time this License terminates and You must stop using the Software. Without prejudice to any other rights it may have, Sylvan may terminate this License if You fail to comply with this License and the terms and conditions of any other agreement You may have with Sylvan. Sylvan may terminate this License if it determines that the Software no longer will be used in the operation of SylvanSync Services. Upon termination of this License, You may no longer access or use the Software regardless of whether or not it remains installed on Your computer. You must comply with all obligations with respect to Customer Information as set forth in Section 8 of this Agreement.
- **5. CONFIDENTIALITY/PRIVACY**. The Software and all materials and information relating to the Software is proprietary to, and constitute trade secrets of, Sylvan. You must maintain the secrecy of the Software.
- **6. LIMITED WARRANTY.** Sylvan warrants to You that, in the case of Software that is its own property, that Software will substantially conform to the applicable Sylvan specifications in effect at the date of installation. Sylvan's sole obligation under this warranty is limited to using commercially reasonable efforts to correct reported problems by supplying You with a corrected version of the Software. The Software may contain nonconformities, defects, errors or omissions.
- TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, THE FOREGOING SHALL BE YOUR SOLE AND EXCLUSIVE REMEDY AND SYLVAN'S TOTAL LIABILITY FOR A DEFECTIVE SOFTWARE. SYLVAN AND/OR ITS SUPPLIERS DISCLAIM ALL WARRANTIES, DUTIES AND CONDITIONS, WHETHER EXPRESS, IMPLIED OR STATUTORY, INCLUDING BUT NOT LIMITED TO WARRANTIES OF MERCHANTABILITY, FITNESS FOR PARTICULAR OR ANY PURPOSE, RELIABILITY, ACCURACY, LACK OF VIRUSES OR MALICIOUS CODE, NEGLIGENCE, TITLE AND NON-INFRINGEMENT AND ALL LIABILITY ARISING OUT OF OR RELATED TO THE PROVISION OF TECHNICAL SUPPORT OR OTHER SERVICES. SYLVAN AND ITS SUPPLIERS SHALL HAVE NO LIABILITY TO YOU FOR QUIET ENJOYMENT, POSSESSION OR INFRINGEMENT OF PROPRIETARY RIGHTS BY YOUR USE OF THE SOFTWARE OR ANY PORTION THEREOF. YOU ACKNOWLEDGE THAT THE INTERNET IS NOT SECURE AND YOUR USE OF THE SOFTWARE THROUGH THE INTERNET MAY CREATE SECURITY AND OTHER RISKS. YOU AGREE AND ACCEPT THE ENTIRE RISK OF ANY SECURITY BREACHES OF ANY KIND OR NATURE. WHETHER INTENTIONAL OR INADVERTENT, ARISING FROM YOUR ACTS OR OMISSIONS CONNECTED WITH YOUR USE OF THE SOFTWARE, INCLUDING ANY CLAIMS RELATED TO PRIVACY OR LOSS OF PRIVACY, DISCLOSURE OF PERSONAL INFORMATION, ADVERSE EFFECT ON YOUR NETWORK, SOFTWARE, HARDWARE OR DATABASE AND YOU WAIVE ANY CLAIMS YOU MAY HAVE OR ASSERT AGAINST SYLVAN WITH RESPECT THERETO. OTHER THAN AS SET FORTH HEREIN, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, IN NO EVENT SHALL SYLVAN OR ITS SUPPLIERS BE LIABLE TO YOU OR ANY OTHER THIRD PARTY FOR ANY DAMAGES WHATSOEVER, BE THEY DIRECT, SPECIAL, INCIDENTAL, INDIRECT, PUNITIVE, EXEMPLARY OR CONSEQUENTIAL (INCLUDING, WITHOUT LIMITATION, DAMAGES FOR LACK OF ACADEMIC IMPROVEMENT, FAILURE TO MEET ANY DUTY OF GOOD FAITH OR REASONABLE CARE, LOSS OF PROFITS, BUSINESS INTERRUPTION, PERSONAL INJURY OR PROPERTY DAMAGE, LOSS OF INFORMATION OF ANY SORT, OR ANY OTHER PECUNIARY LOSS) ARISING OUT OF THE USE OF OR INABILITY TO USE THE SOFTWARE AND/OR THE PROVISION OF OR FAILURE TO PROVIDE TECHNICAL SUPPORT SERVICES EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.
- **7.** CUSTOMER INFORMATION; USE IN NORMAL COURSE; RESTRICTIONS ON SYLVAN'S USE. For the purposes of this Agreement, "Customer Information" shall mean all information relating to past, current or prospective customers of Your Sylvan Learning Center which is stored on or contained in the Software, including without limitation, student demographic and academic data.

You may not use the Customer Information for any purpose other than in the normal conduct of operating SylvanSync Services in connection with a Sylvan Learning Center business in accordance with your License Agreement. By way of

example, You may not sell a mailing list containing Customer Information to a third-party marketing company. You may not sell, loan, give or otherwise disclose Customer Information, including customer lists, to any party without Sylvan's prior written permission. Sylvan may have access to the Customer Information via the Software. Nevertheless, upon Sylvan's request, You must provide Sylvan a copy of all Customer Information in electronic or other format as designated by Sylvan. Sylvan may periodically establish policies respecting the use of the Customer Information by Sylvan franchisees, including You.

Upon expiration or termination of this License or of Your License Agreement, You shall send to Sylvan a true, accurate and complete copy of all Customer Information in electronic or other format as reasonably designated by Sylvan. You hereby covenant and agree that you shall not destroy, damage, hide or take any steps so as to prevent Sylvan from receiving a true, accurate and complete copy of all Customer Information, nor shall You retain any printouts, disks or tapes containing any of the Customer Information except as set forth in this Section 7. Notwithstanding Your rights of use during the terms of this License, You expressly affirm and agree that, upon the expiration or termination of Your License Agreement, Your rights to Customer Information are controlled by the Your License Agreement, and You shall not use the Customer Information for any purpose except to the extent provided for pursuant to the License Agreement provisions concerning the period following expiration or termination of the License Agreement. Notwithstanding the foregoing, in the event that this License expires or terminates prior to the expiration or termination of the License Agreement, You shall have the right to continue to use the Customer Information in the normal conduct of a Sylvan Learning Center in accordance with the License Agreement for so long as You remain a Sylvan licensee pursuant to the License Agreement. Nothing in this License shall be construed to supersede or modify Your post-termination obligations under the License Agreement, including without limitation Your covenant not to compete and Your obligation not to contact former Customers as defined in the License Agreement.

Both during and after the term of this License, Sylvan may disclose, use, and authorize others to disclose and use, the Customer Information only for: (i) research, evaluation, and analysis in connection with the SylvanSync Services, and (ii) purposes that, in Sylvan's exclusive business judgment, will benefit the Sylvan system and the Sylvan brand. In no event shall Sylvan share Customer Information with any third party, including without limitation persons or entities that, even though affiliated with Sylvan, do not operate under or perform functions in support of the Sylvan system or the Sylvan brand, except for the purposes set forth in the foregoing sentences. Before using or authorizing others to use the Customer Information for any purpose other than data analysis, research or maintenance, Sylvan shall advise the franchisee members of the National Advertising Committee as it exists at such time, or a successor to the National Advertising Committee (collectively for the purposes hereof, the "NAC"), of how and when the Customer Information is proposed to be used or to be authorized to be used.

Sylvan shall not share with any other Sylvan franchisee any segment of Customer Information that personally identifies customers of Your Sylvan Learning Center, except with Your written consent.

Neither Sylvan nor You shall use the Customer Information in any manner that would violate any applicable laws governing the security and/or privacy rights of any customer, including but not limited to The Controlling the Assault of Non-Solicited Pornography And Marketing Act of 2003 ("CAN-SPAM" Act) (15 USCA § 7701 et seq.); The Telemarketing and Consumer Fraud Abuse Prevention Act. ("Do Not Call" Law) (15 USCA § 6101 et seq.); and, in Canada, The Personal Information Protection and Electronic Documents Act ("PIPEDA"). Both Sylvan and you must require any third party to whom it or you transfers Customer Information to also comply with all applicable laws governing the privacy rights of any customer.

With respect only to Customer Information, in the event of a conflict between the terms of this License and the terms of the License Agreement, the terms of this License shall govern.

8. MISCELLANEOUS. This License is personal to You, is specific to a single Sylvan territory, and is not transferable. You may not assign or otherwise transfer Your rights or delegate Your obligations under this License and these Terms of Use in whole or in part, and any attempt to do so is void. This License is governed by the laws of the State of Texas without giving effect to principles of conflict of laws and the applicable State or Federal court situated in the City of Bedford, State of Texas shall have exclusive jurisdiction. The provisions of this section notwithstanding, You agree that Sylvan may seek equitable relief in any court of competent jurisdiction to prevent or enjoin misappropriation, misuse, unauthorized disclosure of Software or infringement of any intellectual property rights in addition to any other remedies or damages to which Sylvan may be entitled. You agree to provide Sylvan with reasonable access to the Software to audit Your compliance with this License. The Software is subject to U.S. export control laws and regulations. Portions of the Software may be licensed from third party suppliers and, if so, Your use will be governed by any such third party's licensing terms in addition to this License. This License and any

Software enrollment terms constitute the complete agreement between You and Sylvan and supersede all prior or contemporaneous agreements or representations, written or oral, concerning its subject matter. Only the English version of this License is binding. Versions in other languages merely constitute a convenience translation for You and shall not be legally binding. Pre-printed terms and conditions contained in any purchase order or other ordering document shall have no force or effect. The provisions of Section 2 through and including Section 8 hereof shall survive expiration or termination of this License.

IN WITNESS WHEREOF, the undersigned, intending to be legally bound, have executed this  $\underline{\text{Attachment}}$   $\underline{\text{G}}$  to be effective as of the Effective Date.

LICENSOR:	LICENSEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company
By: John McAuliffe, Chief Executive Officer	By:, its Member

### ATTACHMENT H

#### TELEPHONE NUMBERS ASSIGNMENT AGREEMENT

This Telephone Numbers Assignment Agreement is made on , by and between ("<u>Assignor</u>") and Sylvan Learning, LLC or its designee ("<u>Assignee</u>").

## **BACKGROUND**

- A. The Assignee has developed and owns the proprietary system ("<u>System</u>") for the operation of a facility under the trademark and logo SYLVAN LEARNING, LLC ("<u>Licensed Business</u>");
- B. The Assignor has been granted a license to operate a Licensed Business pursuant to a License Agreement dated , in accordance with the System;
- C. To operate its Licensed Business, the Assignor shall be acquiring one or more telephone numbers, telephone listings and telephone directory advertisements; and
- D. As a condition to the execution of the License Agreement, the Assignee has required that the Assignor assign all of its right, title and interest in its telephone numbers, telephone listings and telephone directory advertisements to the Assignee in the event of a termination of the License Agreement.

#### **AGREEMENT**

In consideration of the foregoing, the mutual premises herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

- 1. <u>Assignment</u>. In the event of termination of the License Agreement, and in order to secure continuity and stability of the operation of the System, the Assignor hereby sells, assigns, transfers and conveys to the Assignee all of its rights, title and interest in and to certain telephone numbers, telephone listings and telephone directory advertisements pursuant to which Assignor shall operate its Licensed Business in accordance with the terms of the License Agreement; provided, however, such Assignment shall not be effective unless and until the License Agreement is terminated in accordance with the provisions thereof.
- 2. <u>Representation and Warranties of the Assignor</u>. The Assignor hereby represents, warrants and covenants to the Assignee that:
  - (a) As of the effective date of the Assignment, all of the Assignor's obligations and indebtedness for telephone, telephone listing services and telephone directory advertisement services shall be paid and current;
  - (b) As of the date hereof, the Assignor has full power and legal right to enter into, execute, deliver and perform this Agreement;
  - (c) This Agreement is a legal and binding obligation of the Assignor, enforceable in accordance with the terms hereof;
  - (d) The execution, delivery and performance of this Assignment does not conflict with, violate, breach or constitute a default under any contract, agreement or instrument to which the Assignor is a party or by which the Assignor is bound, and no consent of nor approval by any third party is required in connection herewith; and
  - (e) The Assignor has the specific power to assign and transfer its right, title and interest in its telephone numbers, telephone listings and telephone directory advertisements, and the Assignor has obtained all necessary consents to this Assignment.

3. <u>Miscellaneous</u>. The validity, construction and performance of this Assignment shall be governed by the laws of the State of Texas. All agreements, covenants, representations, and warranties made herein shall survive the execution hereof. All rights of the Assignee shall inure to its benefit and to the benefit of its successors and assigns.

In witness whereof, the undersigned, intending to be legally bound, have executed this Telephone Numbers Assignment Agreement to be effective on the day and year first written above.

ASSIGNEE:	ASSIGNOR:
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company
By:	By:
John McAuliffe, Chief Executive Officer	, its Member

#### ATTACHMENT I

## **ACH AUTHORIZATION AGREEMENT**

(VER. 08052021)

By executing below, the undersigned Licensee authorizes Sylvan Learning, LLC ("Licensor") to credit or debit the account identified below to pay all fees, charges, and any other amounts Licensee owes Licensor or its parents, affiliates, or subsidiaries pursuant to the applicable License Agreement, as amended, and any other agreements entered into between Licensor and Licensee, including, but not limited to, reimbursable or pass through expenses, the cost of any products or services Licensee purchases from Licensor, and, if necessary, to initiate adjustments for any transactions debited or credited in error. These debits and credits are related to the operation of the Licensed Business and the amount of each debit or credit will vary from month-to-month. This authorization will remain in full force and effect until Licensor has received written notification from Licensee of its termination in such time and in such manner as to afford Licensor a reasonable opportunity to act on it. Termination of this authorization may result in your License Agreement being terminated unless an alternate means of payment acceptable to Licensor is provided.

## **TERMS OF BILLING:**

Starting immediately and continuing thereafter until your License Agreement has expired or been terminated or alternate means of payment are approved by Licensor, Licensee authorizes Licensor to initiate either an electronic debit or credit or to create and process a demand draft against my bank account listed below on or about the 15<sup>th</sup> day of each month for those sums authorized herein.

Licensee's Bank Name:	
Bank ABA Number (Routing Number):	
Bank Account Number:	
Bank Account Type (Checking/Savings):	
Licensee Federal Tax ID Number:	
Territory Name:	
Licensee: (Insert legal name):	
Ву:	_
Printed Name:	
Title:	
Date:	

#### ATTACHMENT J

## DASHBOARD ACCESS AGREEMENT

This Dashboard Access Agreement ("<u>Agreement</u>") is entered into by Licensor and Licensee on the last date of execution below and amends the terms of the license agreement entered into by the parties ("<u>License Agreement</u>"). Capitalized terms not defined herein have the meaning ascribed in the License Agreement.

WHEREAS, Licensor created an online dashboard through Microsoft's Power BI to provide Sylvan Learning and SylvanSync licensees access to certain data, including, but not limited to, sales, operating expenses, membership sales and data, net promoter score, labor costs, and such other information as identified by Licensor ("Data"); and

WHEREAS, Licensee is indicating its desire to acquire a license for Power BI through Licensor, view the Data provided by Licensee and others, and share its Data on Power BI such that it is visible to other Sylvan Learning and SylvanSync licensees.

In light of the foregoing recitals and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows:

- 2. SHARING OF AND ACCESS TO DATA. Licensee acknowledges (a) Licensor may share Licensee's Data with other Sylvan Learning and SylvanSync licensees through the Power BI platform and such other platforms as identified by Licensor and (b) such Licensee's Data will be anonymous on the Power BI platform. Licensor makes no warranty or representation the Data will be representative of all Sylvan Learning and SylvanSync licensees. Further, Licensee acknowledges and agrees it will access and use the Data solely with its efforts to improve the operation of its Licensed Business pursuant to the License Agreement, and such Data is not provided in connection with the offer or sale of a franchise.
- 3. <u>CONFIDENTIALITY</u>. Licensee agrees all Data Licensor makes available to Licensee through Power BI is Confidential Information as defined in the License Agreement, and subject to confidentiality obligations and restrictive covenants set forth therein.
- 4. <u>MISCELLANEOUS TERMS</u>. This Agreement reflects the entire understanding of the parties regarding the subject matter hereof, may only be modified in writing, and supersedes any inconsistent or conflicting provisions of the License Agreement. The remaining terms of the License Agreement are unaffected by this Agreement and remain binding on the parties. The parties sign and deliver this Agreement to each other as shown below.

LICENSOR:	LICENSEE:	
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company	
By: John McAuliffe, Chief Executive Officer	By:, its Member	
SYLVAN LEARNING, LLC Franchise Disclosure Document   2024	1	License Agreement Attachment J

#### ATTACHMENT K

### **Bright Horizons Participation Agreement**

\_\_\_\_\_("I" or "me") is opting to receive enrollment referrals from Bright Horizons Family Solutions, LLC ("Bright Horizons"), an opportunity which is available to me through the relationship between Sylvan Learning, LLC ("Sylvan") and Bright Horizons, and I acknowledge and agree with all of the below terms and conditions of participation. This Bright Horizons Participation Agreement ("Agreement") amends the Sylvan License Agreement dated \_\_ for Territory #\_\_\_ ("Sylvan License Agreement").

- 1. <u>Services.</u> I will accept referrals from Bright Horizons and/or Sylvan under this program, and I will provide personalized in-person educational tutoring services in a variety of academic subjects (the "Services") to dependents of Bright Horizons Clients' employees ("Customers"). Group size shall be limited to no more than three (3) students to one (1) tutor.
- 2. <u>Staff Requirements</u>. Any staff member that is providing Services to a Customer student will meet the requirements for staff members set forth in Attachment A. I understand that Bright Horizons may request evidence of background screening and other required verifications from me as part of its quality review processes or otherwise in the event of an incident or complaint related to Services, and I agree to comply with any such request in a timely manner.
- 3. <u>Location</u>. I understand that all Services must be delivered at my Sylvan Learning Center or approved Satellite, and no services may be delivered virtually, at the homes of Customers, or at any other location.

# 4. <u>Compliance</u>.

- a. <u>Applicable Laws</u>. I agree to perform the Services consistent with the highest professional standards of quality and in accordance with all applicable laws, regulations, licenses, and certifications. I represent that my business (a) is in full compliance with all applicable regulatory requirements, (b) is in good standing with all regulatory authorities in the states or provinces in which it operates, and (c) possesses all licenses and permits required by law to provide the Services.
- b. <u>Data Security</u>. I shall fully comply with all applicable laws or regulations governing privacy, information security, and confidential information. Confidential Information includes, without limitation, proprietary information relating to Bright Horizons or Sylvan, whether written or oral, electronic or other form, and includes all personal information of Customers and enrolled students. I agree to bear all costs incurred to resolve a security breach of Confidential Information. I will notify Sylvan immediately in the case of any actual or suspected security breach, and I will cooperate fully with Sylvan and Bright Horizons to investigate and correct any such breach. I will ensure that all Confidential Information, including all personal and personally identifiable information, is preserved in a manner that prevents inappropriate disclosure, is retained only for as long as reasonably required, and is responsibly destroyed in a manner that prevents content recovery. Without limitation and notwithstanding any other provision of this Opt-In Agreement, I agree that I shall only use personal information provided by Bright Horizons for the purpose of communicating with Customers about the Services and for no other purpose whatsoever.
- c. <u>Bright Horizons Policies</u>. I agree to comply with the Bright Horizons Provider Program Policies attached as Attachment B. All information, forms, policies, and other materials made available by Bright Horizons or by Sylvan related to the services to be provided hereunder are subject to change. I shall review and familiarize myself with all information, forms, policies, and other materials provided to ensure compliance with all requirements relating to the services.

d. <u>Sylvan Policies</u>. I will perform the Services in a manner consistent with Sylvan brand standards, as outlined by my Sylvan License Agreement(s) and the Sylvan Confidential Operations Manual, and I ratify my commitment to comply with all requirements and obligations set forth in each such document.

#### 5. <u>Insurance</u>

- a. I agree to hold the following insurance policies which are required by Bright Horizons at the time that I commence Services, and I acknowledge that these coverage requirements are subject to change from time to time with notice from Sylvan or Bright Horizons:
  - i Comprehensive general liability insurance with limits of not less than \$1,000,000 combined single limit coverage for bodily injury or death and property damage;
  - ii Workers' compensation insurance as required by law; and
  - iii Physical and Sexual Abuse insurance with limits of not less than \$1,000,000 per occurrence and \$1,000,000 in the annual aggregate. Such insurance may be included as part of the Comprehensive General Liability policy or through a separate policy.
- b. I agree to add "Bright Horizons Family Solutions, LLC and its parent, subsidiary and affiliated companies" as an "Additional Insured" on the above-referenced insurance policies. The address for Bright Horizons is 2655 W. Midway Boulevard, Suite 330, Broomfield, CO 80020. I understand that Bright Horizons may request a copy of the applicable Certificate(s) of Insurance, and I agree to comply with any such request in a timely manner.

#### 6. <u>Reservation Process.</u>

- a. I understand that Customers will request a "Reservation" through the Bright Horizons benefits portal. Any Center that has opted into this program within a certain radius of the Customer's home address will be served to the Customer as an option for Services. I understand that Customers may be provided multiple Center options in addition to my Center, and Customers will be able to choose their preferred Center based on convenience and any factors they deem relevant.
- b. If the Customer selects my Center, then my Center will receive an email with the "Reservation Details", including the Customer and Student Name, the Reservation Number, the Reservation Creation Date, the Reservation Expiration Date, and the number of hours in the Reservation.
- c. When my Center receives a Reservation, the Center staff will input the Customer into Hub, create a task in Customer 360 to document the Reservation Number and Details, and contact the Customer to proceed with enrollment within twenty-four (24) hours of receipt of the Reservation. I will use reasonable efforts to accommodate the scheduling of Services in any academic subject that I currently support at a convenient time for the Customer.
- d. Once enrolled, if an Initial Assessment is conducted, it can be included as a billable instructional session.
- e. Only sessions that occur on or after the Reservation Creation Date and on or before the Reservation Expiration Date, up to the total value of the Reservation, are billable. The Reservation Expiration Date is determined by Bright Horizons and its Clients.
- f. I will not knowingly accept any initial requests for any Reservations from any Customer outside of the process outlined above.

#### 7. Invoicing Process.

- a. Sylvan will review all billable sessions for my Center, based on Reservation Details and Hub session attendance, on the 5<sup>th</sup> day of each month for the prior month.
- b. Sylvan will invoice Bright Horizons for all prior month's billable sessions in my Center on my behalf at the agreed upon rate of fifty-three United States dollars (\$53.00) per hour, and Bright Horizons will be required to pay the invoice within thirty (30) days of the invoice date.

- c. I understand that I must maintain accurate attendance records in Hub, as this will determine the amount that is invoiced to Bright Horizons. I agree to forfeit any amount that I may be due for Services if I did not properly record the session attendance in Hub by the end of the month in which the Services were delivered.
- d. When Sylvan receives the payment from Bright Horizons, Sylvan will remit the funds collected on my behalf to my Center. I understand that Sylvan will not be able to send me the funds until Sylvan has received such funds from Bright Horizons.
- e. I will report all revenue that I receive from Sylvan in connection with the Services on my Hub Revenue Report each month, and I will pay 16% royalty fees (or such other rate as my Sylvan License Agreement may specify) on the amount of revenue received from Sylvan in connection with the Services, and any other fees assessed on gross revenue under the Sylvan License Agreement. I acknowledge that if I am behind on any payments owed to Sylvan, then Sylvan shall have the right to withhold the required royalty fee from my Center's funds from Bright Horizons before it forwards such funds to my Center. Sylvan also reserves the right to withhold additional funds owed to my Center from Bright Horizons up to the full amount of the outstanding balances that my Center owes to Sylvan.
- f. I will not collect any payments of fees from any Customer unless the Customer's allotted hours for the year are exhausted; under such circumstances, if the Customer chooses to continue Services at my Center, then the Customer may continue as a private customer subject to my Center's regular instructional hourly rate.

#### 8. Cancelation.

- a. <u>Cancelation of Sessions</u>. Any changes or cancelations of scheduled tutoring sessions shall be communicated by the Customer to the Center. I agree to work directly with the Customer to reschedule any sessions for a convenient time.
- b. <u>Cancelation of the Reservation</u>. Any cancelation of the Reservation shall be communicated by the Customer through Bright Horizons. I agree to accept any cancelation requests that come from Bright Horizons or Sylvan.
- 9. Exclusivity. I agree not to offer, solicit to provide, or agree to provide employer-sponsored dependent care as part of any other dependent care provider network in any geographical location in which Bright Horizons provides such services, during the term of this Opt-In Agreement and for up to and one year following the termination or expiration hereof. I agree not to partner with any other party to offer, sell, or provide a combined employer sponsored back-up dependent care offering during the term and for up to one year following the termination or expiration hereof. To the extent that I currently participate in any other employer-sponsored dependent care network, I agree to cease participation in such network as soon as practicable and, in any event, before April 1, 2024.

#### 10. Additional Bright Horizons Opportunities.

a. College Coaching. By submitting this Opt-In Agreement, I may be permitted, on a pilot basis, to participate as a reseller of the College Coaching program offered by Bright Horizons to support rising and current high school senior students in developing their college applications. If so, I would be eligible to sell the College Coaching program to new and existing customers at my Sylvan Learning Center, and Bright Horizons would deliver the services under the College Coaching program, including, without limitation, by providing counseling sessions, webinars, and essay review. I would be responsible for collecting fees from customers who want to enroll in the College Coaching program and forwarding a portion of such fees to Bright Horizons or Sylvan, as directed. I agree that any revenue I may collect and retain in connection with the College Coaching program will be subject to 16% royalty fees due to Sylvan (or such other rate as my Sylvan License Agreement may specify). I understand that, given its pilot nature, details and processes related to the College Coaching program are subject to change. If I participate, I agree to comply with all policies and requirements of the College Coaching program, which may be provided separately in writing.

- b. Enhanced Family Supports. I agree to provide a 15% discount to current and future Bright Horizons Customers who are participating in the Bright Horizons Enhanced Family Supports program. This partnership offers families discounted services that are not subsidized by their employer, unlike the Services otherwise referenced in this Opt-In Agreement, which are subsidized by the employer. The discount referenced herein does not apply to the Services related to this Opt-In Agreement for in-person tutoring which is subsidized by the Bright Horizons' Clients.
- 11. <u>Term.</u> The term of this Opt-In Agreement shall commence on January 1, 2024 and continue for three (3) years. Thereafter, this Opt-In Agreement will renew for successive two (2) year renewal terms, unless I provide notice of non-renewal at least ninety (90) days in advance of any expiration date.
- 12. <u>Indemnification</u>. I agree to indemnify, defend and hold harmless Sylvan and Bright Horizons, and each of their affiliates, successors and assigns, employees, officers, directors, and agents from and against any and all liability, damages, settlements, losses, expenses, and costs (including reasonably attorneys' fees and expenses) arising out of or in connection with any third-party claims or causes of action, made or threatened, as a result of my breach of this Opt-In Agreement, or the negligence or willful misconduct of my staff or me in connection with delivery of the Services.
- 13. Governing Agreement; Release. I acknowledge that this Opt-In Agreement is governed by my Sylvan License Agreement(s) and that if I violate this Opt-In Agreement, my Sylvan License Agreement(s) will be subject to default and termination. Likewise, I acknowledge that if I am declared in default under my Sylvan License Agreement, Sylvan will have the right to terminate my participation with Bright Horizons. All of Sylvan's rights under my Sylvan License Agreement shall apply to this Opt-In Agreement. In consideration of the benefits being provided to me under this Opt-In Agreement, I agree to release Sylvan and Bright Horizons, and each of their affiliates, employees, officers, directors, and agents, from all claims, demands, damages, losses, and suits of every kind, known or unknown, foreseen or unforeseen, that arise out of or relate to my Sylvan License Agreement(s) or the Services to be provided hereunder.
- 14. <u>Third-Party Beneficiary</u>. I agree that Bright Horizons shall be considered a third-party beneficiary of this Opt-In Agreement and will be able to enforce fully the provisions of this Opt-In Agreement.

By signing this form, I confirm that I have reviewed all of the above terms and conditions related to my participation in Bright Horizons programs, and I agree to comply in all respects. This agreement is effective as of the Effective Date of the Licensee's Sylvan License Agreement.

#### LICENSEE:

a ,	limited liability company
By:	its Member

V.2024.051024

# Attachment A Staff Member Qualifications and Background Screening

**I. Staff Member Qualifications**. Each Member eligible to provide the Services must comply in full with the following requirements:

### Education/Experience

• Each Staff Member must be qualified and have the appropriate educational degrees and/or equivalent experience.

### References

• Each Staff Member must have a minimum of two (2) relevant and satisfactory employment references.

## Training

- At least one Staff Member on duty must be trained in First Aid and CPR.
- Each Staff Member must be trained in emergency protocols and procedures.

## Minimum Age

All Staff Members who are providing Services must be at least 18 years of age.

II. Background Screening. In addition to all background checks and employment verifications required by law and Provider's policies, each Licensee shall perform or cause to be performed a Background Check (within twelve months prior to being permitted to provide Services and at least once every thirty-six months), and Verification (prior to being permitted to provide Services) on each Staff Member. Each Staff Member must successfully pass a Background Check and Verification. Each Licensee agrees to use either a governmental law enforcement authority or similar regulatory agency or service, or a reputable professional background screening vendor who is accredited and in good standing with the National Association of Professional Background Screeners under its Background Screening Agency Accreditation Program, or a combination of such government authorities, and professional vendors, to perform all Background Checks on Staff Members. In the event of any actual or suspected incident or complaint or other concern involving a Staff Member, each Licensee shall perform additional background checks and verification reasonably requested by Bright Horizons including, without limitation, a 14-panel drug test.

[Continued on the following page.]

"<u>Background Check</u>" shall mean a background check using all known names/aliases consistent with the following requirements:

Category	Requirement
Social Security Number Trace	Social Security Number trace which identifies all
	addresses and aliases
Criminal Record Background Checks	Federal and state or county criminal history checks, plus national criminal record database check, for federal and state felony and misdemeanor criminal convictions. All information must be obtained from primary sources rather than database searches (except for the national criminal record database check). Any records identified in the national criminal record database check must be verified through a primary source.
Location and Time Period of	Criminal history checks shall include all locations where
Checks	employee has resided, been employed, or attended school in the past 7 years. International checks are required where employee has resided outside of the United States within the past 7 years.
OFAC Check	Search of U.S. Office of Foreign Assets Control Specially Designated National and export denial lists
Sex Offender Registries	Check national sex offender registry, and all state sex offender registries in states in which the employee has resided or worked

# "Verification" shall mean:

Category	Requirement
Personal Information	Name, date of birth and Social Security Number verified
<b>Licenses and Certifications</b>	Professional licenses and certifications verified
Legal Right to Work	Citizenship, most recent country of permanent residence and
	legal right to work verified through I-9 process
Education	Education verified in accordance with Provider's
	requirements
<b>Employment History</b>	Work experience verified
Reference Check	Two relevant and satisfactory reference checks

The following must be true: (A) each Staff Member providing the Services has successfully passed a Background Check and Verification, (B) no Staff Member providing the Services has a record of any prior conviction or other record of criminal activity inconsistent with the Staff Member's responsibilities and the safety and security of Dependents, and (C) no Staff Member or other employee shall provide Services or otherwise have access to any Dependent if that Staff Member or other employee: (1) has been convicted of a felony or misdemeanor (or the equivalent thereof under relevant non-U.S. law) other than for the following convictions: (a) a minor traffic violation (a moving traffic violation other than reckless driving,

hit and run, driving to endanger, vehicular homicide, driving while intoxicated or other criminal offense involving gross negligence, recklessness, intentional or willful misconduct while operating a motor vehicle), (b) a conviction that has been legally expunged, or (c) a conviction for a misdemeanor that occurred while the Staff Member was under the age of twenty-one years; or (2)is on a U.S. Government Specially Designated National or export denial list; or (3) is identified in a sexual offender registry; or (4) does not have the legal right to work in the United States.

#### Attachment B

## **In-Person Tutoring Program Provider Policies**

Incidents during In-Person Tutoring Program Services

- Parents should be promptly notified of any injuries/incidents that occur during the tutoring session by phone call, text, email, or in person at the end of the session.
- Provider shall promptly notify its insurers of any claim or possible claim and provide Bright Horizons with evidence of that notification upon request.

### **Quality Control and Safety**

Provider shall reserve the right to monitor tutoring sessions (live or through recordings, if permitted by a Dependent's parent/guardian) to ensure the quality of the Services and the health and safety of Dependents. Recordings shall be made available to Bright Horizons upon request. Provider shall immediately notify a Dependent's parents/guardians and Bright Horizons in the event of any inappropriate discussions between a Staff Member and Dependent that occur during the provision of Services or communications or attempted communications of any nature which occur outside of Provider's platform.

#### Ratios

Provider and its Licensees shall provide In-Person Tutoring in accordance with its standard ratios and shall ensure that at least two Staff Members meeting these requirements are present at the facility at all times during In-Person Tutoring.

### **Privacy**

Provider shall ensure Services comply with all applicable privacy laws, including, but not limited to the Children's Online Privacy Protection Act (COPPA), the California Consumer Privacy Act (CCPA) and any other federal, state or local privacy laws or regulations, or for Dependents located in Canada, any applicable Canadian national or provincial laws, including but not limited to the Personal Information Protection and Electronic Documents Act (PIPEDA). Provider shall cooperate with Bright Horizons with all requests for information to demonstrate compliance with privacy or other legal requirements.

## **Emergency Situations**

In the event a Staff Member observes an emergency situation, the Staff Member will immediately notify the Dependent's parents/guardians in person, by phone and text messaging, and/or through calling 911 if a parent/guardian is not immediately available and shall also immediately notify Bright Horizons.

#### Photos, Videos and Social Media

Except as directed by Bright Horizons and except in accordance with Provider's standard procedures for internal quality control, Provider will not use photos or video of Dependents for any purpose, including, but not limited to, use on social media platforms, Provider's website or promotional materials or any other website, unless a parent or guardian has granted consent to use of photos/videos by agreeing to Provider's standard photo and video release terms.

# **Required Provider Policies**

Provider shall ensure it has in place and complies with appropriate policies addressing the following:

- Session sign-in and sign-out process
- Positive behavior guidance and management

# ATTACHMENT L

# MYSYLVAN MARKETPLACE+SM FRANCHISEE PARTICIPATION PILOT AGREEMENT



# mySylvan Marketplace+SM Franchisee Participation Pilot Agreement

This mySylvan Marketplace+SM Franchisee Participation Pilot Agreement (this "Pilot Agreement") is entered into (the "Effective Date") by and between Sylvan Learning, LLC ("Sylvan") and ("Franchisee" or "Licensee") concerning certain operations within Sylvan Learning Territory# (the "Territory"), governed by that certain Sylvan Learning License Agreement# , dated , to which Franchisee and Sylvan are each a party (the "License Agreement").

WHEREAS, Sylvan, pursuant to rights granted to it by third parties, has developed and plans to develop further certain proprietary technology, digital assets, systems, business methods, techniques, and methodologies designed to be used in connection with portable proprietary computer systems and the Internet to provide an online digital marketplace accessible via web browser and mobile applications available on Android and iOS devices (the "Marketplace Platform") through which Sylvan facilitates scheduling of and payment for tutoring services offered by independent contractors to customers;

WHEREAS, Sylvan, pursuant to its rights, desires to expand the area in which it advertises and offers services through the Marketplace Platform;

**WHEREAS**, Franchisee desires to participate in such expansion on a pilot basis, subject to the terms and conditions of this Pilot Agreement;

**WHEREAS**, Sylvan has agreed to expand the Marketplace Platform to Franchisee's Territory on a pilot basis (the "Pilot"), subject to the terms and conditions of this Pilot Agreement;

**NOW, THEREFORE**, in consideration for the mutual promises contained herein, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties agree as follows:

- 1. <u>Incorporation</u>. The foregoing recitals shall be incorporated herein by reference.
- 2. General. The Pilot, through the Marketplace Platform, is designed to (a) connect consumers seeking one-on-one supplemental education services for students in grades K-12 (each a "Consumer") with qualified, professional tutors contracted by Sylvan who are able to meet the requested learning needs (each a "Tutor"); (b) facilitate scheduling of tutoring sessions directly between the Consumer and the selected Tutor (each a "Marketplace Tutoring Session") at a mutually agreed upon location acceptable to Sylvan, including through live, online platforms; (c) enable and process direct online payment by the Consumer to the Tutor; and (d) collect the administrative fees ("Tutor Fees") owed by Tutor to Sylvan in connection with each Marketplace Tutoring Session delivered by the Tutor. Tutors are recruited and screened by Sylvan; once approved, they enter into an independent contractor agreement (the "Tutor Agreement") with Sylvan, contingent upon the successful completion of a background investigation. Pursuant to the Tutor Agreement, each Tutor has the right to deliver tutoring services to students under the "Sylvan In-Home" or "mySylvan Marketplace+" brand names. Sylvan reserves the rights to rebrand the Marketplace Platform, as needed, and to grant or revoke the rights to use such brand names, as applicable. Tutors must set their own cancelation policy, hourly rate, and scheduling availability on their online tutor profiles within the Marketplace Platform. Generally, Marketplace Tutoring Sessions are offered at a one-to-one tutor-student ratio; however, some exceptions are made under certain circumstances. Tutors who are certified teachers may opt to become "Certified in the Sylvan Method" by completing all necessary online SylvanSync product training. Tutors who are Certified in the Sylvan Method may offer SylvanSync Programs (as defined in the License Agreement) to Consumers, and are denoted on the Marketplace Platform with a badge in their online tutor profile. All Tutors must use their own equipment, and Tutors who are Certified in the Sylvan Method must use their own compatible iPads to deliver SylvanSync Programs. Tutors and Consumers directly control their own scheduling of Marketplace Tutoring Sessions through the Marketplace Platform, and all payments are processed automatically upon the completion of a Marketplace Tutoring Session directly though the Marketplace Platform using the credit card saved to the Consumer's online account.
- 3. <u>Sylvan's Rights</u>. Pursuant to Paragraph 1.4 of the License Agreement, Sylvan is permitted (a) to operate businesses, and offer and sell services, programs, and products within the Territory which do not comprise a part of the Sylvan System and, in connection therewith, to exploit Sylvan's Licensed Marks (as defined in the License

Agreement), name, reputation, and know-how; (b) to use the Licensed Marks and other aspects of the Sylvan System in conjunction with activities not contemplated by Sylvan as being conducted solely through or at Sylvan Centers; and (c) to offer, distribute, use, and/or sell components of the Sylvan System through courses or programs offered and conducted by methods or through facilities other than Centers within the Territory or outside the Territory. Based on the foregoing, Franchisee acknowledges and agrees that Sylvan has the right to offer and deliver services through the Marketplace Platform (including, without limitation, advertising the Marketplace Platform, recruiting Tutors, and collecting Tutor Fees) within Franchisee's Territory without seeking or obtaining consent by Franchisee or compensating Franchisee in connection therewith. Notwithstanding the foregoing, Sylvan has elected, at its sole discretion, to allow certain franchisees, including Franchisee, to participate in and benefit from the expansion of the Marketplace Platform during the term of this Pilot Agreement.

- 4. Ownership of Intellectual Property. All intellectual property related to Marketplace Platform, including copyrights, trademarks, and patents (the "IP"), is owned by Sylvan and third parties with whom Sylvan has contracted and from whom Sylvan has licensed such IP. Franchisee acquires no rights in the IP. Sylvan's maintenance and renewal of the IP shall be at Sylvan's sole discretion. Franchisee hereby assigns to Sylvan all intellectual property rights associated with Franchisee's ideas, suggestions, and improvements relating to the Marketplace Platform.
- 5. <u>Delivery and Maintenance of the Marketplace Platform Software</u>. Sylvan will be responsible for the maintenance and support of the Marketplace Platform software, which shall remain the property of Sylvan. Sylvan will control the access to the Marketplace Platform by Tutors and Consumers within Franchisee's Territory. This Pilot Agreement expressly does not grant to Franchisee any administrative access to the Marketplace Platform, including the right to add or view Tutor or Consumer data, or to modify Marketplace Tutoring Sessions scheduled within the Marketplace Platform. All Tutor and Consumer data, along with all data related to Marketplace Tutoring Sessions, shall belong solely to Sylvan.
- 6. <u>Tutor Fees</u>. The Tutor Fees charged by Sylvan in the Territory shall be Twenty-Five Dollars (\$25.00) per Marketplace Tutoring Session. Sylvan also reserves the right to make additional Tutor Fees available to other franchisees participating in the Pilot based on demographics, marketability, and other factors, without making the same Tutor Fees available to Franchisee.
- 7. Session Fees. Sylvan shall provide a credit to Franchisee, during the term of this Pilot Agreement, in the amount of the following hourly session fees ("Session Fees") relating to the delivery of Marketplace Tutoring Sessions completed within Franchisee's Territory, as shown on Attachment A to the License Agreement and incorporated herein by reference. The Session Fees shall be paid from the Tutor Fees collected by Sylvan from Tutors, and shall be based upon the duration and location of the Marketplace Tutoring Session. A "Session" is defined generally as instruction delivered either online or in-person at the Consumer's home or other mutually agreeable appropriate location by one (1) Tutor to one (1) or more Consumers lasting in duration at least thirty (30) and not more than ninety (90) consecutive minutes. The credit shall be applied against any royalty or statement balances owed by Franchisee to Sylvan and shall be reflected in the monthly billing statement issued by Sylvan to Franchisee in the month following the month in which the Marketplace Tutoring Sessions were delivered in Franchisee's Territory. Franchisee shall be required to continue to pay all royalties and other payments due to Sylvan on a monthly basis pursuant to one or more license agreements between Sylvan and Franchisee, including the License Agreement.

#### **Marketplace Tutoring Session Fees:**

Location of	Hourly Tutor Fee	Session Fee Paid by	Amount
<u>Marketplace</u>	Collected by	Sylvan to	Retained by
Tutoring Session	Sylvan	Franchisee	Sylvan
In-Territory	\$25.00	\$15.00	\$10.00

If the amount of the Tutor Fee changes based on a promotion or other special rate offered by Sylvan, Sylvan shall pay to Franchisee a prorated Session Fee based on the rate offered (e.g., if the Tutor Fee is reduced by half, the Session Fee to be paid to Franchisee will be reduced by half). Franchisee understands that no Session Fee shall be paid by Sylvan to Franchisee for Marketplace Tutoring Sessions delivered outside of Franchisee's Territory.

- 8. Franchisee Obligations. Franchisee agrees to perform the following obligations.
  - a) Providing Feedback. Franchisee agrees to participate in focus groups and provide requested feedback relating to Franchisee's experiences with the Marketplace Platform and Marketplace Tutoring Sessions, including any feedback from Consumers in Franchisee's Territory of which Franchisee may be aware. At Sylvan's request, Franchisee shall participate in monthly calls with Sylvan and other participating franchisees to discuss the Pilot.
  - b) Local Marketing. Franchisee shall use its best efforts to market the Marketplace Platform to both Tutors and Consumers in the Territory. Franchisee shall not promote or market the Marketplace Platform outside of the Territory. Franchisee may discuss and demonstrate the Marketplace Platform with customers and prospective customers.
- 9. Sylvan Obligations. Sylvan agrees to perform the following obligations.
  - a) Tutor Recruitment. Sylvan shall use reasonable efforts to recruit qualified local professionals in the Territory and the surrounding area as independent contractors to serve as Tutors. Sylvan may use certain advertising services, including, without limitation, Indeed, Care.com, and referrals. Employees of Franchisee may apply to become Tutors, but Sylvan shall have no obligation to approve an employee of Franchisee as a Tutor. All Tutors shall be independent contractors of Sylvan; no Tutor shall have any direct relationship with Franchisee.
  - b) Candidate Screening. Sylvan shall screen all tutor applicants by reviewing resumes, conducting phone interviews, and checking references.
  - Background Investigation. Sylvan has contracted a third-party background investigation company to perform background investigations and education verification services on behalf of Sylvan for all qualified candidates with whom Sylvan intends to enter into a Tutor Agreement. The initial background investigation must be successfully completed prior to the Tutor becoming "live" (i.e., accessible and able to view) on the Marketplace Platform. Pursuant to the Tutor Agreement, the cost for the initial background investigation is generally paid by the Tutor upfront and reimbursed by Sylvan if and at such time as the Tutor has met certain conditions. Sylvan reserves the right to modify the terms of the Tutor Agreement without notice to Franchisee. It is the current process that Sylvan or its designated third-party background investigation company shall re-perform a background investigation on each Tutor on an annual basis. Franchisee acknowledges that the process and designated third-party company are subject to change without notice. Sylvan reserves the right to terminate any Tutor Agreement if the Tutor's annual background investigation discovers criminal conduct or otherwise inappropriate behavior.
  - d) Training and Onboarding. Sylvan shall perform all obligations relating to the product training and onboarding needs of all Tutors, including, without limitation, providing online training in the Marketplace Platform and SylvanSync, if applicable.
  - e) Customer Service. Sylvan shall manage the customer service needs of Tutors and Consumers participating in the Pilot.
  - f) Reporting. Sylvan shall provide monthly reports to Franchisee concerning the number of Marketplace Tutoring Sessions delivered within the Territory. Each report shall contain the length of each Marketplace Tutoring Session, the zip code of the Consumer, and the credit to be issued by Sylvan to Franchisee based on the delivery of such completed Marketplace Tutoring Sessions. Sylvan may, though it shall not be required to, provide more detailed information in monthly reports to Franchisee.
- 10. Sylvan License Agreement. Franchisee shall market the Marketplace Platform pursuant to the terms and conditions of this Pilot Agreement and in conjunction with Franchisee's operation of an approved Sylvan Learning Center pursuant to the License Agreement. Franchisee acknowledges and agrees that the Marketplace

Platform offering is separate from the Sylvan System and the SylvanSync System; however, this Pilot Agreement shall be governed by the terms and conditions of the License Agreement, unless otherwise stated herein. Any breach of this Pilot Agreement by Franchisee shall constitute a material breach of the License Agreement. Notwithstanding the foregoing, in the event of an express conflict between the License Agreement and this Pilot Agreement concerning only the subject matter hereof, this Pilot Agreement shall govern.

- 11. No Performance Representations. Sylvan expressly disclaims the making of, and Franchisee acknowledges that it has not received or relied upon, any statement, representation, warranty, or guarantee, express or implied, concerning Franchisee's potential performance with respect to the Pilot, including, but not limited to, potential sales volume, cost savings, profitability, success, or effect on Franchisee's Sylvan Learning business. No such statement, representation, warranty, or guarantee has been authorized by Sylvan. Franchisee has conducted an independent investigation of the Pilot and understands and acknowledges that it involves certain business risks which make the outcome of Franchisee's business venture largely dependent upon factors outside of Franchisee's control, as well as the business abilities and participation of Franchisee and its efforts as an independent business operator. Franchisee assumes all risks related to the Pilot, including the possibility that the Sylvan Learning Center(s) operated by Franchisee in the Territory may lose potential customers due to the introduction of the Marketplace Platform in the Territory and the surrounding area. Sylvan expressly disclaims all liability for any such losses. Franchisee acknowledges and agrees that Franchisee is accountable for Franchisee's actions, inactions, and omissions in connection with Franchisee's business.
- 12. <u>Franchise Offering Disclaimer</u>. Nothing in this Pilot Agreement shall be construed as an offer to sell a franchise. The offers contained herein are available only to existing Sylvan Learning franchisees. No franchises concerning the Marketplace Platform will be sold to any party until the offering has been duly registered and declared effective by the applicable jurisdiction and the required Franchise Disclosure Document has been delivered to the prospective franchisee before the sale in compliance with applicable law. Nothing in this Paragraph should be construed as a waiver of any applicable exemption provisions that may be available to Sylvan.
- 13. <u>Limitations of Pilot Participation</u>. The parties acknowledge and agree that the Marketplace Platform is available in the Territory on a pilot basis and that it will evolve over time as the parties gain a better understanding of its impact on the market. Nothing in this Pilot Agreement binds either party to continue participation in the Pilot following the expiration or termination of this Pilot Agreement; moreover, nothing in this Pilot Agreement shall create any binding precedent for terms and conditions of any future or subsequent agreement which may be offered in connection with the Marketplace Platform. Notwithstanding anything to the contrary in this Pilot Agreement, Franchisee expressly acknowledges and agrees that (a) the Pilot may be discontinued or modified at any time by Sylvan at its sole discretion; (b) Sylvan shall have the right, but not the obligation, to offer the Pilot model or a variant thereof as a license of franchise program (the "Future Program") and on such terms and conditions as Sylvan deems appropriate in its sole discretion; (c) Sylvan shall not be obligated to offer to Franchisee the right to participate in any Future Program; and (d) Franchisee shall not be obligated to participate in any Future Program. The rights secured by Franchisee pursuant to this Pilot Agreement are limited solely to those explicitly stated herein and Franchisee has no other rights, express or implied, during the term of this Pilot Agreement or after its termination by either party.
- 14. Confidentiality. Franchisee acknowledges that, during the term of this Pilot Agreement, Franchisee may receive from Sylvan certain confidential information relating to the Pilot or the Marketplace Platform, including, without limitation, proprietary information related to technology and marketing (the "Confidential Information"). The terms of this Pilot Agreement shall be considered Confidential Information and shall be subject to the confidentiality provisions of the License Agreement. In addition, Franchisee agrees that, until such time as the Confidential Information enters the public domain through no fault of Franchisee, Franchisee and Franchisee's employees will never, directly or indirectly, use, disseminate, or disclose to a third party the Confidential Information without the prior written consent of Sylvan. Franchisee further agrees to protect the Confidential Information from use or disclosure using no less than a reasonable degree of care. Franchisee's obligations under this Paragraph shall survive termination of this Pilot Agreement.

- 15. Term and Termination. The term of this Pilot Agreement shall commence on the Effective Date and continue until the earliest to occur of: (a) Franchisee and Sylvan enter into a subsequent agreement (if any) concerning the Marketplace Platform or similar offering; (b) Franchisee terminates this Pilot Agreement pursuant to this Paragraph; (c) Sylvan terminates this Pilot Agreement pursuant to this Paragraph; (d) the License Agreement terminates for any reason; or (e) December 31, 2025. Franchisee may not terminate this Pilot Agreement for any or no reason during the first one hundred twenty (120) days following the Effective Date. Following such initial 120-day period, Franchisee may terminate this Pilot Agreement for convenience upon sixty (60) days' advance written notice to Sylvan. Sylvan may terminate this Pilot Agreement for convenience upon sixty (60) days' prior written notice. Sylvan may terminate this Pilot Agreement immediately upon ten (10) days' prior written notice if (a) Franchisee commits any material breach of this Pilot Agreement or the License Agreement; or (b) Sylvan licenses to a third-party franchisee the right to open and commence operation of a Sylvan Learning Center in an area within close proximity to the Territory such that the area surrounding the Territory is no longer controlled solely by Franchisee or a set of franchisees all willing to participate in the Pilot. The term "day" as used in this Pilot Agreement shall mean a calendar day, rather than a business day.
- 16. No Warranties. FRANCHISEE UNDERSTANDS AND ACKNOWLEDGES THAT THE MARKETPLACE PLATFORM AND THE PILOT GENERALLY ARE BEING PROVIDED HEREUNDER ON AN "AS-IS" BASIS. SYLVAN DISCLAIMS ANY AND ALL WARRANTIES AND GUARANTEES RELATED TO EACH OF THEM INCLUDING, WITHOUT LIMITATION, THE WARRANTY OF MERCHANTABILITY AND THE WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE.
- 17. Month-to-Month Basis. Unless terminated earlier pursuant to Paragraph 15, Sylvan reserves the right to extend this Pilot Agreement on a month-to-month basis for up to twelve (12) months following December 31, 2025, or longer by written notice. During any such period that this Pilot Agreement is so extended, Franchisee may terminate this Pilot Agreement upon 30 days' prior written notice to Sylvan.
- 18. Waiver; Release. Sylvan reserves the right and privilege in its sole discretion as it may deem in the best interest of all concerned in any specific incident, to vary standards for any franchisee participating in the Pilot based upon peculiarities of a particular set of circumstances, demographics, or other conditions which Sylvan deems to be of importance to the successful operation the Pilot. Franchisee shall have no recourse against Sylvan on account of any variations and shall not be entitled to require Sylvan to grant Franchisee like or similar terms. No failure of Sylvan to exercise any power reserved to it by this Pilot Agreement, and no custom or practice of the parties at variance with the terms hereof, shall constitute a waiver of Sylvan's right to demand exact compliance with the terms herein. No waiver or approval by Sylvan of any particular breach or default by Franchisee, nor any delay, forbearance, or omission by Sylvan to act or give notice of default or to exercise any power or right arising by reason of such default hereunder, shall be considered a waiver or approval by Sylvan of any preceding or subsequent breach or default by Franchisee of any term, covenant, or condition of this Pilot Agreement. In consideration of the benefits provided to Franchisee hereunder, Franchisee agrees to fully and completely release, acquit, and forever discharge Sylvan and its affiliates, and each of their respective employees, officers, and directors, collectively, separately, and severally, of and from any and all demands, damages, or claims of every kind and nature whatsoever, foreseen and unforeseen, known or unknown, that arise out of or relate to the License Agreement or this Pilot Agreement.
- 19. Entire Agreement. This Pilot Agreement shall constitute the entire agreement between the parties related to the subject matter hereof, and shall supersede all prior oral or written statements, understandings, or negotiations concerning the Pilot and the Marketplace Platform. Except as otherwise provided herein, neither Sylvan nor Franchisee may modify this Pilot Agreement except by an instrument in writing between Sylvan and Franchisee.
- 20. <u>Power and Authority</u>. Each party to this Pilot Agreement represents and warrants that it is validly in existence; that its, his, or her representations herein are true and correct; and that it, he, or she has full power and authority to execute this Pilot Agreement and all documents contemplated herein.
- 21. <u>Choice of Law.</u> This Pilot Agreement shall in all respects be governed by and be construed under the internal laws of the State of Texas, without giving effect to its conflict of laws rules.

22. <u>Counterparts</u>. This Pilot Agreement may be executed in one or more counterparts, all of which taken together shall constitute one and the same instrument. An electronic signature shall be given the same effect as an original in-ink signature.

IN WITNESS WHEREOF, the parties to this Pilot Agreement have executed it as of the date first set forth above.

LICENSOR:	LICENSEE/FRANCHISEE:
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company
By:	By:, its Member
John McAuliffe, Chief Executive Officer	, its Member

# EXHIBIT B TO THE FRANCHISE DISCLOSURE DOCUMENT

## SYLVAN EDGE PACKAGE LICENSE AGREEMENT

#### Sylvan Edge Package License Agreement

This Sylvan Edge Package License Agreement (this "Agreement") is entered into as of (the "Effective Date") by and between Sylvan Learning, LLC ("Sylvan") and \_\_\_\_ ("Licensee"). This Agreement amends Sylvan License Agreement # dated (the "Sylvan License Agreement"), to which Sylvan and Licensee are parties and which governs Sylvan Learning territory # (the "Territory").

WHEREAS, Sylvan, itself and pursuant to rights granted to it by third parties, has developed certain portable robotics, computer programming, engineering, and mathematics learning programs, content, activities (including without limitation computer-based activities), sequences of activities, technology, specifications, training, techniques, methodologies, know-how, and trade dress and other indicia for same, to be delivered under certain of Sylvan's famous marks, which are distinct from the Sylvan System and the SylvanSync System as defined in the Sylvan License Agreement, and are designed for delivery to and use by students in small group settings (the "Package").

WHEREAS, Licensee operates the Licensed Business in the Territory, and desires that Sylvan approve and license Licensee to use the Package in connection with such Licensed Business.

WHEREAS, Sylvan agrees to approve and license Licensee to use the Package in connection with the Licensed Business in the Territory, for the Term (as defined in this Agreement) and subject to the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration for the mutual promises contained herein, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties agree as follows:

1. <u>Grant of License</u>. Sylvan hereby approves and grants to Licensee the right and license during the Term to use the Package in connection with the Licensed Business in the Territory, subject to the terms and conditions of this Agreement.

## 2. Reserved.

- 3. <u>Sylvan License Agreement Terms</u>. This Agreement is an addendum to the Sylvan License Agreement, and amends the Sylvan License Agreement to address the subject matter hereof. Except as otherwise set forth in this Agreement, Licensee's use of the Package is, and shall be treated and governed as, an optional program under the Sylvan License Agreement, including, without limitation, the Sylvan License Agreement terms and conditions concerning the Territory, payment dates, reporting, and covenants not to compete. Notwithstanding the foregoing, in the event of an express conflict between the Sylvan License Agreement and this Agreement concerning the subject matter hereof, this Agreement shall govern. Except as set forth in this Agreement, the Sylvan License Agreement remains unchanged and in full force and effect.
- 4. <u>Package Materials</u>. Sylvan's Confidential Operations Manual sets forth the materials in the Package, including, without limitation, the Package programs (the "Materials"); the costs that Licensee agrees to pay to acquire, use, or license such Materials; and the required and approved sources of such Materials. Sylvan's Confidential Operations Manual is subject to change by Sylvan from time to time in its sole discretion, in accordance with the terms of the Sylvan License Agreement concerning optional programs, and Licensee shall comply with such changes.
- 5. <u>Minimum Start-Up Purchase</u>. To induce Sylvan to enter into this Agreement, Licensee shall purchase at least the Minimum Start-Up Purchase, as defined in Sylvan's Confidential Operations Manual, not later than ninety (90) days following the Effective Date of this Agreement.
- 6. Sylvan Edge Royalty. At all times after the commencement of operations by Licensee under this Agreement, Licensee shall pay to Sylvan a monthly royalty for the preceding month equal to twelve percent (12%) of all revenues which derive from the Package ("Package Revenues"); and an additional amount equal to any sales, gross receipts, or similar tax imposed on Sylvan and calculated solely on payments required to be made by Licensee to Sylvan under this Agreement, unless such tax is an optional alternative to an income tax otherwise payable by Sylvan.

1

All royalties paid by Licensee under this Agreement shall be in addition to Licensee's monthly royalty obligations under the Sylvan License Agreement. In all other respects, the terms of the Sylvan License Agreement shall govern.

- 7. Sylvan Edge Local Marketing and Advertising. In order to promote the Package, Licensee agrees that it shall expend each month, on the type of advertising specified in the Sylvan License Agreement or on other marketing and advertising material and activities approved by Sylvan, not less than a "Monthly Territory Edge Advertising Minimum" equal to three percent (3%) of the preceding month's Package Revenues. Licensee shall keep appropriate records and receipts for all advertising or marketing expenditures for at least three years, and shall provide substantiating data for such expenditures upon request by Sylvan. The Monthly Territory Edge Advertising Minimum shall be in addition to the Monthly Territory Advertising Minimum in the Sylvan License Agreement; however, if and at such time as the aggregate of the amounts expended on monthly territory advertising under the Sylvan License Agreement and this Agreement reach Forty Thousand Dollars (\$40,000) per Territory in a given calendar year, then Licensee's Monthly Territory Edge Advertising Minimum obligations automatically shall be waived for the remainder of such calendar year. In all other respects, the terms of the Sylvan License Agreement shall govern.
- 8. <u>Training</u>. Licensee shall cause itself and its staff as designated by Sylvan to complete successfully all initial and subsequent training required by Sylvan before Licensee may offer and use the Package.
- 9. <u>Authorized Locations</u>. Licensee may offer and use the Package in connection with the Licensed Business at the following "Authorized Locations": (i) Licensee's approved Center(s) in the Territory; (ii) Licensee's approved Satellite(s) in the Territory, provided that Licensee provides at least ten (10) days' prior written notice to Sylvan that Licensee intends to offer and use the Package at Licensee's approved Satellite(s) in the Territory, with same to be deemed approved unless Sylvan provides written notice otherwise to Licensee within ten (10) days after receiving Licensee's notice; and (iii) other locations, such as but not limited to schools, within the Territory that are proposed by Licensee and approved by Sylvan. Sylvan has no obligation to approve any proposed location.
- 10. <u>Exclusivity</u>. During the Term, if Licensee is in material compliance with all of the terms of this Agreement and the Sylvan License Agreement, Sylvan shall not use, nor license any third party to use, the Package in the Territory. Sylvan reserves the right to use the Licensed Marks and certain aspects of the Package in conjunction with activities not contemplated by Sylvan as being conducted solely through Centers; and to offer, distribute, use and/or sell components of the Package through courses or programs offered and conducted by methods or other channels of distribution or through facilities other than Centers, within the Territory or outside the Territory, including but not limited to: (i) the Internet; (ii) CD-ROM, books, audio or video tapes; (iii) sales for home use by students; or (iv) other concepts currently in existence or as may be developed in the future, for example, through advancements in technology.
- Intellectual Property and Use of Third-Party Marks. All intellectual property related to the Package, including, without limitation, copyrights, trademarks, and patents (the "IP"), is owned by Sylvan and third parties with whom Sylvan has contracted and from whom Sylvan has licensed such IP. The IP includes, without limitation, certain marks, the rights to which are owned and may be owned by third parties in the future (the "Third-Party Marks"). Licensee agrees and covenants to use the Third-Party Marks solely in connection with the Package in accordance with this Agreement, and solely in the manner prescribed by Sylvan, and to observe such requirements with respect to trademark, service mark, and copyright notices as Sylvan may, from time to time, require, including, without limitation, affixing "SM," "TM," "®," and/or disclaimer language adjacent to the Third-Party Marks. Licensee agrees that it shall not commit any act causing a tarnishment of the Third-Party Marks or the reputation and goodwill associated with any third party. Licensee's commission of any act causing a tarnishment of the Third-Party Marks or the reputation and goodwill associated with any third party constitutes an incurable material default under this Agreement, and in such event, Sylvan may terminate this Agreement immediately upon written notice to Licensee in accordance with the termination notice provisions of the Sylvan License Agreement, without further notice or opportunity to cure. Licensee acquires no rights in the IP other than the right to use the IP in strict accordance with the terms and conditions of this Agreement. Licensee hereby assigns all intellectual property rights associated with Licensee's ideas, suggestions, and improvements relating to the Package to Sylvan. Licensee acknowledges and agrees that third parties reserve all rights to use or license others to use parts and/or components of the Package (as an example but not a limitation, LEGO® pieces and coding exercises) within the Territory. Notwithstanding the foregoing, Sylvan does not currently license or authorize any third-party owner to use the "Sylvan" and "SylvanSync" trademarks (collectively, the "Sylvan Marks") in connection with the Package in the Territory. Third parties directly

or indirectly providing any part of the Materials to Licensee for the Package may require Licensee to accept, enter into, and otherwise comply with certain other agreements, including, without limitation, end user license agreements and terms of service agreements. Such third-party licensors may have independent rights of enforcement against Licensee if Licensee violates the terms of any such agreements. In addition, such third-party licensors of Sylvan may have the right, as third-party beneficiaries of this Agreement, to bring claims against Licensee.

- 12. <u>Prior Agreements</u>. This Agreement shall supersede any prior agreements concerning the Package with respect to the subject matter hereof.
- Minimum Annual Edge Participation Threshold; Failure to Meet Minimum Annual Edge Participation Threshold. Licensee agrees and acknowledges that it is entering into this Agreement with the intent to generate revenue using the Package, and Licensee is obligated to use best efforts to promote the Package and to offer and sell the Package to customers. To induce Sylvan to enter into this Agreement with Licensee, Licensee agrees and acknowledges that Licensee shall generate sufficient Package Revenues so that the royalty fees that Licensee pays under this Agreement, net of all subsequent reductions to and refunds from such Package Revenues, shall be not less than Two Thousand, One Hundred Dollars (\$2,100) during the first full calendar year of the Term, and not less than Four Thousand, Two Hundred Dollars (\$4,200) during the second calendar year and all subsequent calendar years of the Term (the "Minimum Annual Edge Participation Threshold"). The Minimum Annual Edge Participation Threshold may be adjusted annually pursuant to Paragraph 6.10 (Cost of Living Adjustments) of the Sylvan License Agreement. The Minimum Annual Edge Participation Threshold shall be in addition to Licensee's Minimum Annual Royalty obligations under the Sylvan License Agreement. Licensee's first failure to meet the Minimum Annual Edge Participation Threshold in a given calendar year during the Term of this Agreement shall constitute a curable material default of this Agreement; such default shall be deemed cured by payment to Sylvan of the difference between the Minimum Annual Edge Participation Threshold and the earned Package Revenues for such calendar year (the "Shortfall"). Any subsequent failure (each a "Subsequent Failure") by Licensee to meet the Minimum Annual Edge Participation Threshold in a calendar year during the Term of this Agreement, regardless of whether the first failure or any other prior failure was cured in accordance with this Agreement and the Sylvan License Agreement or waived by Sylvan, shall be an incurable material default of this Agreement and, subject to the appeal provisions of Paragraph 14, Sylvan may terminate this Agreement at any time from the monthly Edge royalty payment due date in the January immediately following the calendar year until June 30 of such calendar year (the "Termination Period"). If Sylvan does not terminate this Agreement by written notice to Licensee during the Termination Period, then Sylvan shall be deemed to have waived its rights to terminate this Agreement for such Subsequent Failure, and such Subsequent Failure shall not be deemed an incurable material default for the purposes of Paragraph 21 of this Agreement. Sylvan's notice of termination to Licensee shall stipulate an appeal period ("Appeal Period") of not less than twenty (20) days before the termination takes effect. During the Appeal Period, Licensee may appeal the termination pursuant to the process set forth in Paragraph 14; otherwise, the termination shall take effect upon the expiration of the Appeal Period. Payment of and/or Sylvan's acceptance of payment of a given Shortfall shall not constitute a cure or stay of termination for any Subsequent Failure, except in the event that Sylvan approves an appeal pursuant to Paragraph 14. Licensee agrees and acknowledges that the Minimum Annual Edge Participation Threshold is not a projection, forecast, or guarantee of Licensee's actual gross revenues from the Package in the Territory, and that Sylvan expressly disclaims making any representation concerning Licensee's actual gross revenues from the Package and performance in the Territory.
- 14. Process to Appeal Termination for Subsequent Failure. Licensee may appeal Sylvan's termination for a Subsequent Failure by appealing in writing to Sylvan within the Appeal Period as defined in Paragraph 13 and promising to pay the Shortfall, and all late fees, interest, and penalties in connection therewith, within fifteen (15) days following Sylvan's written ruling on Licensee's appeal (the "Cure Date"). Beginning on the date Sylvan receives Licensee's written appeal (provided same is received within the Appeal Period), termination pursuant to Paragraph 13 and Paragraph 21 shall be stayed until Sylvan sends to Licensee Sylvan's written ruling on the appeal. Sylvan shall have the right to deny or accept Licensee's appeal, in Sylvan's sole discretion, and is not required to accept any appeal. During the Appeal Period, Licensee's obligation to pay the Minimum Annual Edge Participation Threshold under appeal is suspended, but all of Licensee's other obligations, financial and otherwise, remain in effect, including, without limitation, the requirement to pay ongoing royalties and other amounts. If Sylvan denies the appeal in writing, the termination automatically shall become effective ten (10) days after the date of Sylvan's written denial to Licensee. If Sylvan accepts the appeal in writing, termination shall be stayed until the Cure Date or such later date as Sylvan stipulates. If, Sylvan having accepted the appeal in writing, Licensee subsequently pays to Sylvan the Shortfall by the

Cure Date, then the termination automatically shall be rescinded, and the underlying Subsequent Failure shall not be deemed an incurable material default for the purposes of Paragraph 21; otherwise, the termination shall become effective automatically on the day after the Cure Date without further notice or opportunity to cure, and Licensee shall not be obligated to pay the Shortfall. The provisions in this Paragraph apply only to termination for Subsequent Failure as defined in Paragraph 13 and do not apply to any other event of default or termination.

- 15. Optional Package. Licensee acknowledges and agrees that the Package is optional under the Sylvan License Agreement and that Licensee may terminate this Agreement and all grants and rights hereunder without penalty in accordance with the terms and conditions in Paragraph 20.
- Optional and Mandatory Programs. Licensee shall comply with all educational standards designated as required or mandatory by Sylvan in its Confidential Operations Manual or otherwise in writing. Sylvan may, in its sole discretion, designate a program mandatory under this Agreement which was previously optional under the Confidential Operations Manual by providing notice to Licensee in writing. From time to time, Sylvan may develop or have developed for it new or enhanced programs or modules designated for use in connection with the Package, and shall make such programs or modules available to Licensee as either mandatory or optional. In the event that a Package program is designated as mandatory, Sylvan may not require implementation of such item earlier than ninety (90) days after the item is generally made available to all Sylvan Edge Package licensees. Mandatory Package programs and enhancements or updates to same shall be provided to Licensee at Sylvan's cost. Licensee shall be required to purchase, implement, and make available to its customers each such mandatory program, enhancement, or update to mandatory program no later than the required implementation date for each as determined by Sylvan in accordance with the provisions of this Paragraph. Optional Package programs and updates to same will be made available to Licensee for such charges as Sylvan shall establish. Licensee acknowledges and agrees that Sylvan may require Licensee to successfully complete designated training and/or obtain designated certifications before releasing programs, updates, or modules to Licensee. Licensee's failure to complete such requirements or cause personnel to complete such requirements (provided that the training is made available to Licensee) shall not constitute a defense of Licensee's failure to obtain and implement any mandatory program or module on or before the required implementation date. For any calendar year in which Licensee is required to expend more than Ten Thousand Dollars (\$10,000) (which maximum shall be adjusted each calendar year based upon the cost of living adjustment in the Sylvan License Agreement) altogether on mandatory Package, Sylvan System, and/or SylvanSync System programs required to be implemented beginning in that year, Licensee shall be permitted to extend payment of the amount in excess of such maximum to the following calendar year(s) for so long as Licensee's obligations in any given year exceed the maximum. Expenditures on mandatory programs, enhancements, or updates purchased and implemented after their required implementation date has already passed shall not count toward the maximums in this Paragraph.
- 17. <u>Reporting</u>. Licensee shall timely and accurately report academic, operational, and financial information of the type and in the form that Sylvan identifies from time to time.
- 18. <u>Standards and Quality Assurance</u>. Licensee shall use the Package in compliance with (i) this Agreement; (ii) all manuals that address the Package; and (iii) the Sylvan License Agreement, including, without limitation, the associated Confidential Operations Manual.
- 19. <u>Term.</u> The term of this Agreement shall commence on the Effective Date and shall run concurrently with the term of the Sylvan License Agreement (the "Term"), unless earlier terminated in accordance with this Agreement.
- 20. <u>Termination by Licensee</u>. Licensee may terminate this Agreement for any reason upon sixty (60) days' prior written notice to Sylvan. Licensee shall be obligated to fulfill all contractual obligations to customers, including completing all courses of instruction for which Licensee has contracted to provide and/or has received Package Revenues, prior to the expiration or termination of this Agreement. As otherwise stated in this Agreement, the post-termination obligations of Licensee under this Agreement are set forth in the Sylvan License Agreement.
- 21. <u>Termination by Sylvan; Automatic Termination</u>. Sylvan may not unilaterally terminate this Agreement except as set forth in this Paragraph. Unless earlier terminated in accordance with this Agreement, Sylvan may terminate this Agreement (a) in accordance with Paragraph 11 of this Agreement, addressing tarnishment of Third-Party Marks; (b) in accordance with Paragraph 13 of this Agreement, addressing failure to meet the Minimum

Annual Edge Participation Threshold; and/or (c) in the event that Licensee fails to comply with any other term of this Agreement and fails to cure such default within twenty (20) days following receipt of written notice of default from Sylvan. This Agreement shall automatically terminate upon termination of Licensee's Sylvan License Agreement for any reason.

- 22. <u>Breach of the Sylvan License Agreement</u>. Licensee's failure to comply with any material provision of this Agreement shall not, in and of itself, constitute a breach of the Sylvan License Agreement unless, in the course of such failure to comply with this Agreement, Licensee is in default of its obligations under the Sylvan License Agreement and fails to cure such default (if such default is curable), in which case the Sylvan License Agreement may be terminated according to its terms. As an example, and not a limitation, Licensee's failure to pay any amount when due in connection with this Agreement shall be a curable material default of the Sylvan License Agreement.
- 23. <u>No Warranties.</u> LICENSEE UNDERSTANDS AND ACKNOWLEDGES THAT THE PACKAGE IS BEING PROVIDED HEREUNDER "AS IS." SYLVAN DISCLAIMS ANY AND ALL WARRANTIES AND GUARANTEES RELATED TO THE PACKAGE, INCLUDING WITHOUT LIMITATION THE WARRANTY OF MERCHANTABILITY AND THE WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE.
- 24. <u>No Modifications</u>. Except as set forth herein, this Agreement may not be modified except with the express written consent of all parties hereto.
- 25. <u>Defined Terms</u>. Any capitalized terms not specifically defined herein shall have the meanings ascribed to them in the Sylvan License Agreement.

IN WITNESS WHEREOF, the parties to this Agreement have executed it as of the date first set forth above.

LICENSOR:	LICENSEE:	
SYLVAN LEARNING, LLC, a Delaware limited liability company	a limited liability company	
By: John McAuliffe, Chief Executive Officer	By:, its Member	

## ILLINOIS RIDER TO THE SYLVAN EDGE PACKAGE LICENSE AGREEMENT

To the extent the Illinois Franchise Disclosure Act, Ill. Comp. Stat.  $\S\S705/1 - 705/44$  applies, the terms of this Addendum apply.

- 1. Illinois law governs the Sylvan Edge Package License Agreement.
- 2. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
- 3. Your rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 5. The following statement shall be deemed to amend the License Agreement:

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1

a ,	limited liability company
Ву:	
	its Member

LICENSEE:

# EXHIBIT C TO THE FRANCHISE DISCLOSURE DOCUMENT

## ACE IT! TUTORING SYSTEM LICENSE AGREEMENT



#### ACE IT! Tutoring System License Agreement

This ACE IT! Tutoring System License Agreement (this "Agreement") is entered into as of (the "Effective Date") by and between Sylvan Learning, LLC ("Sylvan") and <u>LicenseeName</u> ("Licensee"). This Agreement amends Sylvan License Agreement # dated (the "Sylvan License Agreement"), to which Sylvan and Licensee are parties and which governs Sylvan Learning territory # (the "Territory").

WHEREAS, Sylvan has developed certain proprietary programs, systems, teaching, and management techniques to provide academic programs for students pursuant to contracts funded by third parties, such as school districts, other government entities, and not-for-profit organizations, ("Third-Party Contracts"), which are personally taught by trained instructors and which are distinct from the Sylvan System and the SylvanSync System (as defined in the Sylvan License Agreement) and which are approved for use by Sylvan to be delivered both in Sylvan Learning Centers and Satellites (as defined in the Sylvan License Agreement), as well as in un-branded locations, such as schools and community Centers, under the ACE IT! TUTORING® brand name (the "ACE IT! System");

WHEREAS, Sylvan licenses the ACE IT! System to persons and entities who have entered into Sylvan License Agreements to enable such persons and entities to conduct business using the ACE IT! System within their respective territories and in connection with the Licensed Business (the "ACE IT! Business");

WHEREAS, Licensee operates the Licensed Business in the Territory and desires that Sylvan approve and license Licensee to use the ACE IT! System in connection with the Licensed Business; and

WHEREAS, Sylvan agrees to approve and license Licensee to use the ACE IT! System in connection with the Licensed Business in the Territory, for the Term (as defined hereinafter) and subject to the terms and conditions set forth in this Agreement and the Sylvan License Agreement.

NOW, THEREFORE, in consideration for the mutual promises contained herein, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties agree as follows:

- Grant of License. Sylvan hereby approves and grants to Licensee the right and license during the Term to use the ACE IT! System, which includes, but is not limited to, its proprietary programs, systems, and techniques, and certain copyrighted materials, and Licensed Marks (as defined hereinafter), to conduct the ACE IT! Business in connection with retail services offered directly to consumers, as approved by Sylvan, and in connection with Third-Party Contracts with schools, other government entities, community Centers, and other third parties at approved locations in the Territory, subject to the terms and conditions of this Agreement. The term "Licensed Marks" shall be used in this Agreement to describe collectively certain trade names, trademarks, service marks, logos, symbols, and other indicia of origin identified in the Confidential ACE IT! Tutoring Operations and Education Manual(s), including, but not limited to, ACE IT! and ACE IT! TUTORING, and such other trade names, trademarks, service marks, associated logos and symbols as are now designated by Sylvan (and as may hereafter be designated by Sylvan in writing). Sylvan may, in its sole discretion, change, modify, delete from, supplement, improve or further develop the ACE IT! System and the services, programs, products and other components comprising the ACE IT! System from time to time, through the Confidential ACE IT! Tutoring Operations and Education Manual(s) and any other manuals issued by Sylvan for use in the ACE IT! System, or otherwise upon written notice to Licensee. Licensee agrees to comply with all of Sylvan's requirements in such regard, including, without limitation, offering and selling products, programs, and services that Sylvan specifies.
- 2. Reservation of Rights. Licensee expressly understands and agrees that Sylvan shall have the right, in its sole discretion and regardless of proximity to the Territory and the financial impact on Licensee's ACE IT! Business or its Sylvan License, to provide (outright, through contract, joint-ventures or otherwise) ACE IT! System programs at any locations outside of the Territory; to grant licenses for the provision of ACE IT! System programs at any locations outside of the Territory as Sylvan, in its sole and exclusive discretion, deems appropriate; to

1

conduct and operate businesses, and offer and sell services, programs and products within the Territory which do not comprise a part of the ACE IT! System and, in connection therewith, to exploit its Licensed Marks, name, reputation and know-how; to use the Licensed Marks and other aspects of the ACE IT! System in conjunction with activities not contemplated by Sylvan as being conducted solely in connection with Third-Party Contracts; and to offer, distribute, use and/or sell components of the ACE IT! System through courses or programs offered and conducted other than under Third-Party Contracts, within the Territory or outside the Territory; and to sell and distribute, and grant others the right to sell and distribute, at retail or wholesale, products associated with the ACE IT! System, within the Territory or outside the Territory, including, but not limited to: (i) the Internet; (ii) CD-ROM, books, audio or video tapes; (iii) sales for home use by students; or (iv) other concepts currently in existence or as may be developed in the future, for example, through advancements in technology. Sylvan agrees that, so long as this Agreement is in effect, it will not itself, or through any affiliate or subsidiary, utilize the ACE IT! System in the Territory for Third-Party Contracts.

#### 3. Reserved.

- 4. Sylvan License Agreement Governs. This Agreement is an addendum to the Sylvan License Agreement, and amends the Sylvan License Agreement to address the subject matter hereof. Except as otherwise set forth in this Agreement, Licensee's use of the ACE IT! System is, and shall be treated and governed as, an optional program under the Sylvan License Agreement, including, without limitation, the Sylvan License Agreement terms and conditions concerning the Territory, payment dates, record keeping, reporting, taxes, covenants not to compete, and other provisions. Notwithstanding the foregoing, in the event of an express conflict between the Sylvan License Agreement and this Agreement concerning the subject matter hereof, this Agreement shall govern. Except as set forth in this Agreement, the Sylvan License Agreement remains unchanged and in full force and effect.
- 5. <u>Standards and Quality Assurance</u>. Licensee shall use the ACE IT! System in compliance with (i) this Agreement; (ii) the Confidential ACE IT! Tutoring Operations and Education Manual(s); and (iii) the Sylvan License Agreement, including, without limitation, the associated Confidential Operations Manual.
- Confidential ACE IT! Tutoring Operations and Education Manual(s). Sylvan shall lend to Licensee, at no additional charge, certain Confidential Operations and Education Manual(s); and Licensee agrees to conduct its ACE IT! Business and to maintain standards of operation in accordance with the provisions set forth in such Confidential ACE IT! Tutoring Operations and Education Manual(s). Sylvan may impose a reasonable charge for replacement manuals or supplements issued to Licensee. The standards, specifications, and requirements established by Sylvan for Licensee, unless otherwise indicated, shall be set forth in this Agreement or may, from time to time, be set forth by Sylvan in the Confidential ACE IT! Tutoring Operations and Education Manual(s). Licensee understands and acknowledges that Sylvan may, from time to time, revise the contents of the Confidential ACE IT! Tutoring Operations and Education Manual(s) to implement new or different requirements for the ACE IT! System, and Licensee expressly agrees to comply with all such changed requirements which are by their terms mandatory. Licensee shall at all times ensure that its copy of the Confidential ACE IT! Tutoring Operations and Education Manual(s) is kept current and up to date. In the event of any dispute as to the contents of the Confidential ACE IT! Tutoring Operations and Education Manual(s), the terms and dates of the master copy of same maintained by Sylvan at its principal place of business shall be controlling. Licensee shall at all times use its best efforts to keep the information in the Confidential ACE IT! Tutoring Operations and Education Manual(s), the Confidential ACE IT! Tutoring Operations and Education Manual(s) themselves and any other manuals, materials, goods and information created or used by Sylvan and designated for confidential use within the ACE IT! System confidential and shall limit access to employees of Licensee on a need-to-know basis.

#### Training.

- 7.1 Sylvan agrees to provide an initial training program as described in the Confidential ACE IT! Tutoring Operations and Education Manual(s) (as hereinafter described) for up to two (2) members of Licensee's staff, including Licensee and one other staff person as designated by Sylvan.
- 7.2 Sylvan agrees to furnish Licensee certain materials deemed by Sylvan to be appropriate for instruction of Licensee's personnel in the conduct of the ACE IT! Tutoring programs.

2

- 7.3 Licensee agrees to cause itself and its staff, as required by Sylvan, to complete successfully any initial or subsequent training, as Sylvan may require as set forth in the Confidential ACE IT! Tutoring Operations and Education Manual(s), prior to offering any programs under this Agreement. Prior to any training, Licensee and all other persons to be trained must sign a confidentiality, non-disclosure, and covenant not to compete agreement on the form then used by Sylvan. The training that Sylvan provides to Licensee and to Licensee's employees does not constitute employment, direction, or supervision of Licensee or Licensee's employees by Sylvan. Sylvan does not hire, fire, manage, supervise, schedule, discipline, or set compensation for Licensee or Licensee's employees.
- 8. Authorized Locations. Licensee may offer and use the ACE IT! System in connection with the Licensed Business at the following "Authorized Locations": (i) schools and institutions within the Territory; (ii) schools and institutions outside of the Territory, provided a) that such area is not licensed by Sylvan to any other third party at such time; b) that Licensee submits a written request to offer and use the ACE IT! System at such location and that Sylvan grants approval for such offering and use in writing to Licensee; and c) that Licensee agrees that Licensee shall cease offering any programs under this Agreement in such location if and at such time as Sylvan licenses the territory which includes the location to a third party; (iii) Licensee's approved Center(s) in the Territory, provided that Licensee submits a written request to offer and use the ACE IT! System at such Center and that Sylvan grants approval for such offering and use in writing to Licensee; and (iv) Licensee's approved Satellite(s) in the Territory, provided that Licensee submits a written request to offer and use the ACE IT! System at such Satellite and that Sylvan grants approval for such offering and use in writing to Licensee. Sylvan shall have no obligation to approve any request to allow Licensee to offer the ACE IT! System in Centers or Satellites.
- 9. <u>Staff Requirements</u>. To maintain the high image and quality standards of Sylvan's proprietary programs and material, Licensee agrees to provide sufficient and competent management, staff, and ACE IT! certified personnel to deliver high quality instruction and service as designated in the Confidential ACE IT! Tutoring Operations and Education Manual(s) and any changes thereto for all programs. Licensee agrees that if accreditation for supplemental education programs or services offered by Licensee is available from an accrediting body or organization, Licensee shall attain and maintain such accreditation when designated in the Confidential ACE IT! Tutoring Operations and Education Manual(s) and any modifications thereto.
- 10. Permits, Licensing, and Approvals. Prior to commencing business operations, Licensee shall obtain all federal, local, and state permits, licenses, certifications, and federal, state, and/or local approvals (if required) necessary to provide ACE IT! System programs. Licensee shall also comply at all times with all educational, safety, building or other local, state and federal statutes, ordinances, regulations, codes, licensing requirements or standards applicable to education programs. Licensee shall comply with all local, state and federal requirements relating to confidential treatment of student identities. Sylvan shall in no manner be liable for advising Licensee of such matters or determining Licensee's compliance therewith and Licensee will defend, indemnify, and hold Sylvan harmless from any claims or loss relating to same. Licensee may not share permits, licenses, or approvals issued to another licensee of Sylvan or Sylvan itself. Licensee is responsible to ensure that the authorizing entity(ies) are aware of Licensee's status as an independently owned and operated business requiring separate and specific permits, licenses, and approvals.
- 11. Exclusivity. During the Term, if Licensee is in material compliance with all of the terms of this Agreement and the Sylvan License Agreement, Sylvan shall not use, nor license any third party to use, the ACE IT! System in the Territory.
- 12. <u>Establishment of Tuition Charges and Scholarship Students</u>. Licensee shall establish the tuition and other charges applicable under Licensee's Third-Party Contracts, and shall make such revenue information available to Sylvan upon request. Licensee agrees that, in any month, it may not provide more than 5% of services under the ACE IT! System on a scholarship, reduced-fee, or no-fee basis.
- 13. Protection of Sylvan's Proprietary Rights.
  - 13.1 Trade Secrets. Licensee acknowledges Sylvan's proprietary rights in the programs, systems, techniques, manuals, and trade secrets which will be disclosed to Licensee under this Agreement, and further acknowledges that Licensee does not acquire any right or interest therein beyond the rights expressly granted to it under this Agreement. By execution of this Agreement, Licensee also acknowledges that the know-how and operating format related to Sylvan's proprietary methods and materials is also proprietary to Sylvan and will not be disclosed by Licensee except in connection with Licensee's required activities as authorized by this Agreement. Licensee agrees to maintain adequate security in the control, use, and handling of proprietary materials Sylvan supplies to Licensee in accordance with the practices described in the Confidential ACE IT! Tutoring Operations

and Education Manual(s) or as stated otherwise in writing from time to time. Unless expressly waived by Sylvan in writing, Licensee shall require all persons who have or could have access to Sylvan proprietary materials, to sign written confidentiality agreements pertaining to the security of Sylvan proprietary materials, which agreement shall be in a form reasonably acceptable to Sylvan if requested by Sylvan. Licensee shall immediately notify Sylvan of any unauthorized use of Sylvan's trade secrets, or any infringement of Sylvan's copyrights. Sylvan shall have sole discretion to take such action or inaction as it deems appropriate in response.

13.2 Licensed Marks. Licensee expressly acknowledges Sylvan's rights in and to the Licensed Marks and agrees not to represent in any manner that Licensee has acquired any ownership rights in the Licensed Marks. Licensee agrees not to use any of the Licensed Marks or any marks, names, or indicia which are or may be confusingly similar in its own corporate or business name except as authorized in this Agreement. Licensee further acknowledges and agrees that any and all goodwill associated with the ACE IT! System and identified by the Licensed Marks shall inure directly and exclusively to the benefit of Sylvan and that, upon the expiration or termination of this Agreement for any reason, no monetary amounts shall be attributable to any goodwill associated with Licensee's use of the Licensed Marks. Licensee understands and agrees that any use of the Licensed Marks other than as expressly authorized by this Agreement, without Sylvan's prior written consent, is an infringement of Sylvan's rights in the Licensed Marks and that the right to use the Licensed Marks granted herein does not extend beyond the termination or expiration of this Agreement. Licensee expressly covenants that, during the term of this Agreement and thereafter, Licensee shall not, directly or indirectly, commit any act of infringement or contest or aid others in contesting the validity of Sylvan's right to use the Licensed Marks or take any other action in derogation thereof. Licensee acknowledges an obligation to monitor its own and other parties' use of the Licensed Marks and agrees to do so. Licensee shall promptly notify Sylvan of any claim, demand or cause of action known to Licensee that Sylvan may have based upon or arising from any unauthorized attempt by any person or legal entity to use the Licensed Marks, any colorable variation thereof, or any other mark, name or indicia in which Sylvan has or claims a proprietary interest. Licensee shall assist Sylvan, upon request and at Sylvan's expense, in taking such action, if any, as Sylvan may deem appropriate to halt such activities, but shall take no action nor incur any expenses on Sylvan's behalf without Sylvan's prior written approval; if Sylvan undertakes the defense or prosecution of any litigation relating to the Licensed Marks, Licensee agrees to cooperate with and assist Sylvan as Sylvan may reasonably request in the defense or prosecution of any such action, and Sylvan shall reimburse Licensee for Licensee's reasonable out-of-pocket expenses incurred in providing such assistance. Licensee further agrees and covenants to use the Licensed Marks solely in the manner prescribed by Sylvan; to observe all laws with respect to the registration of trade names and assumed or fictitious names, to include in any application therefor a statement that Licensee's use of the Licensed Marks is limited by the terms of this Agreement, and to provide Sylvan with a copy of any such application and other registration document(s); and to observe such requirements with respect to trademark and service mark registrations and copyright notices as Sylvan may, from time to time, require, including, without limitation, affixing "SM", "TM", or "®" adjacent to the Licensed Marks. Licensee agrees that it shall not commit any act causing an incurable tarnishment of the Licensed Marks or the reputation and goodwill associated with the ACE IT! System. Sylvan reserves the right, in its sole discretion, to designate and require that Licensee use one or more new, modified or replacement Licensed Marks. If as a result of such requirement, Licensee should be required to alter any signage or outdoor advertising, Licensee shall have 120 days from receipt of Sylvan's requirement to comply with same, at Licensee's sole cost and expense.

14. Intellectual Property and Use of Third-Party Marks. All intellectual property related to the ACE IT! System, including, without limitation, copyrights, trademarks, and patents (the "IP"), is owned by Sylvan and third parties with whom Sylvan has contracted and from whom Sylvan has licensed such IP. The IP may include, without limitation, certain marks, the rights to which may be owned by third parties in the future (the "Third Party Marks"). Licensee agrees and covenants to use the Third-Party Marks solely in connection with the ACE IT! System in accordance with this Agreement, and solely in the manner prescribed by Sylvan, and to observe such requirements with respect to trademark, service mark, and copyright notices as Sylvan may, from time to time, require, including, without limitation, affixing "SM," "TM," "®," and/or disclaimer language adjacent to the Third-Party Marks. Licensee agrees that it shall not commit any act causing a tarnishment of the Third-Party Marks or the reputation and goodwill associated with any third party. Licensee's commission of any act causing a tarnishment of the Third-Party Marks or the reputation and goodwill associated with any third party constitutes an incurable material default under this Agreement, and in such event, Sylvan may terminate this Agreement immediately upon written notice to Licensee in accordance with the termination notice provisions of the Sylvan License Agreement, without further notice or opportunity to cure. Licensee acquires no rights in the IP other than the right to use the IP in strict accordance with the terms and conditions of this Agreement. Licensee hereby

assigns all intellectual property rights associated with Licensee's ideas, suggestions, and improvements relating to the ACE IT! System to Sylvan.

#### 15. ACE IT! Royalty Payments and Requirements.

- 15.1 Monthly ACE IT! Royalty Payment. At all times after the commencement of operations by Licensee under this Agreement, Licensee shall pay to Sylvan a monthly royalty for the preceding month equal to twelve percent (12%) of gross revenues which derive from the ACE IT! System ("ACE IT! Revenues"); and an additional amount equal to any sales, gross receipts, or similar tax imposed on Sylvan and calculated solely on payments required to be made by Licensee to Sylvan under this Agreement, unless such tax is an optional alternative to an income tax otherwise payable by Sylvan. The annual amount paid by Licensee to Sylvan shall not be less than the Minimum Annual ACE IT! Royalty Threshold as set forth in Paragraph 15.2 of this Agreement. In all other respects, the terms of the Sylvan License Agreement shall govern.
- 15.2 Minimum Annual ACE IT! Royalty Threshold. Licensee agrees and acknowledges that it is entering into this Agreement with the intent to generate revenue using the ACE IT! System, and Licensee is obligated to use best efforts to promote the ACE IT! System and to offer and sell the ACE IT! System to customers. To induce Sylvan to enter into this Agreement with Licensee, Licensee agrees and acknowledges that the aggregate of the monthly royalty payments that Licensee pays to Sylvan for each full calendar year after Licensee commences operations under this Agreement shall not be less than One Thousand, Two Hundred Dollars (\$1,200) (the "Minimum Annual ACE IT! Royalty Threshold"), and that such amount may be adjusted annually pursuant to Paragraph 6.10 (Cost of Living Adjustments) of the Sylvan License Agreement. The Minimum Annual ACE IT! Royalty Threshold for the calendar year in which the Effective Date falls shall be prorated based on the number of full calendar months remaining between the four (4) month anniversary of the Effective Date and the end of that calendar year. The Minimum Annual ACE IT! Royalty Threshold shall be met by Licensee each calendar year regardless of the amount of gross revenue Licensee does receive or does not receive and/or the extent to which Licensee operates or does not operate the ACE IT! Business during such calendar year. The requirement to meet Minimum Annual ACE IT! Royalty Threshold shall continue through the initial term and each renewal term of this Agreement. The Minimum Annual ACE IT! Royalty Threshold shall be in addition to Licensee's Minimum Annual Royalty obligations under the Sylvan License Agreement; however, if and at such time as Licensee has entered into and is subject to the Sylvan Edge Package License Agreement in addition to this Agreement, the Minimum Annual Royalty payment required under the Sylvan License Agreement shall be reduced by Two Thousand, Four Hundred Dollars (\$2,400).
- 15.3 Failure to Meet Minimum Annual ACE IT! Royalty Threshold. Licensee's first failure to meet the Minimum Annual ACE IT! Royalty Threshold in a given calendar year during the Term of this Agreement shall constitute a curable material default of this Agreement; such default shall be deemed cured by payment to Sylvan of the Minimum Annual ACE IT! Royalty Threshold less any royalty payments paid to Sylvan by Licensee in the same calendar year in connection with the ACE IT! System (the "Shortfall"). Any subsequent failure (each a "Subsequent Failure") by Licensee to meet the Minimum Annual ACE IT!® Royalty Threshold in a calendar year during the Term of this Agreement, regardless of whether the first failure or any other prior failure was cured in accordance with this Agreement and the Sylvan License Agreement or waived by Sylvan, shall be an incurable material default of this Agreement and, subject to the appeal provisions of Paragraph 15.4, Sylvan may terminate this Agreement at any time from the monthly ACE IT! royalty payment due date in the January immediately following the calendar year until June 30 of such calendar year (the "Termination Period"). If Sylvan does not terminate this Agreement by written notice to Licensee during the Termination Period, then Sylvan shall be deemed to have waived its rights to terminate this Agreement for such Subsequent Failure, and such Subsequent Failure shall not be deemed an incurable material default for the purposes of Paragraph 20 of this Agreement. Sylvan's notice of termination to Licensee shall stipulate an appeal period ("Appeal Period") of not less than twenty (20) days before the termination takes effect. During the Appeal Period, Licensee may appeal the termination pursuant to the process set forth in Paragraph 15.4; otherwise, the termination shall take effect upon the expiration of the Appeal Period. Payment of and/or Sylvan's acceptance of payment of a given Shortfall shall not constitute a cure or stay of termination for any Subsequent Failure, except in the event that Sylvan approves an appeal pursuant to Paragraph 15.4.
- 15.4 Process to Appeal Termination for Subsequent Failure. Licensee may appeal Sylvan's termination for a Subsequent Failure by appealing in writing to Sylvan within the Appeal Period as defined in Paragraph 15.3 and promising to pay the Shortfall, and all late fees, interest, and penalties in connection therewith, within fifteen (15) days following Sylvan's written ruling on Licensee's appeal (the "Cure Date"). Beginning on the date Sylvan receives Licensee's written appeal (provided same is received within the Appeal Period), termination pursuant to Paragraph 15.3 and Paragraph 20 shall be stayed until Sylvan sends to Licensee Sylvan's

written ruling on the appeal. Sylvan shall have the right to deny or accept Licensee's appeal, in Sylvan's sole discretion, and is not required to accept any appeal. During the Appeal Period, Licensee's obligation to pay the Minimum Annual ACE IT! Royalty Threshold under appeal is suspended, but all of Licensee's other obligations, financial and otherwise, remain in effect, including, without limitation, the requirement to pay ongoing royalties and other amounts. If Sylvan denies the appeal in writing, the termination automatically shall become effective ten (10) days after the date of Sylvan's written denial to Licensee. If Sylvan accepts the appeal in writing, termination shall be stayed until the Cure Date or such later date as Sylvan stipulates. If, Sylvan having accepted the appeal in writing, Licensee subsequently pays to Sylvan the Shortfall by the Cure Date, then the termination automatically shall be rescinded, and the underlying Subsequent Failure shall not be deemed an incurable material default for the purposes of Paragraph 20; otherwise, the termination shall become effective automatically on the day after the Cure Date without further notice or opportunity to cure, and Licensee shall not be obligated to pay the Shortfall. The provisions in this Paragraph apply only to termination for Subsequent Failure as defined in Paragraph 15.3 and do not apply to any other event of default or termination.

- 16. Optional Program. Licensee acknowledges and agrees that the ACE IT! System is an optional program under the Sylvan License Agreement and that Licensee may terminate this Agreement and all grants and rights hereunder without penalty in accordance with the terms and conditions in Paragraph 19.
- 17. <u>Reporting</u>. Licensee shall timely and accurately report academic, operational, and financial information of the type and in the form that Sylvan identifies from time to time.
- 18. <u>Term.</u> The term of this Agreement shall commence on the Effective Date and shall run concurrently with the term of the Sylvan License Agreement (the "Term"), unless earlier terminated in accordance with this Agreement.
- 19. <u>Termination by Licensee</u>. Licensee may terminate this Agreement for any reason upon sixty (60) days' prior written notice to Sylvan. Licensee shall be obligated to fulfill all contractual obligations to customers, including completing all courses of instruction for which Licensee has contracted to provide and/or has received or will receive ACE IT! Revenues, prior to the expiration or termination of this Agreement. As otherwise stated in this Agreement, the post-termination obligations of Licensee under this Agreement are set forth in the Sylvan License Agreement.
- 20. <u>Termination by Sylvan; Automatic Termination</u>. Sylvan may not unilaterally terminate this Agreement except as set forth in this Paragraph. Unless earlier terminated in accordance with this Agreement, Sylvan may terminate this Agreement (a) in accordance with Paragraph 15.3 of this Agreement, addressing failure to meet the Minimum Annual ACE IT!® Royalty Threshold; and/or (b) in the event that Licensee fails to comply with any other term of this Agreement and fails to cure such default within twenty (20) days following receipt of written notice of default from Sylvan. This Agreement shall automatically terminate upon termination of Licensee's Sylvan License Agreement for any reason.
- 21. <u>Breach of the Sylvan License Agreement</u>. Licensee's failure to comply with any material provision of this Agreement shall not, in and of itself, constitute a breach of the Sylvan License Agreement unless, in the course of such failure to comply with this Agreement, Licensee is in default of its obligations under the Sylvan License Agreement and fails to cure such default (if such default is curable), in which case the Sylvan License Agreement may be terminated according to its terms. As an example, and not a limitation, Licensee's failure to pay any amount when due in connection with this Agreement shall be a curable material default of the Sylvan License Agreement.
- 22. No Warranties. LICENSEE UNDERSTANDS AND ACKNOWLEDGES THAT THE ACE IT! SYSTEM IS BEING PROVIDED HEREUNDER "AS IS." SYLVAN DISCLAIMS ANY AND ALL WARRANTIES AND GUARANTEES RELATED TO THE ACE IT! SYSTEM, INCLUDING WITHOUT LIMITATION THE WARRANTY OF MERCHANTABILITY AND THE WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE.
- 23. <u>No Modifications</u>. Except as set forth herein, this Agreement may not be modified except with the express written consent of all parties hereto.
- 24. <u>Defined Terms</u>. Any capitalized terms not specifically defined herein shall have the meanings ascribed to them in the Sylvan License Agreement.

IN WITNESS WHEREOF, the parties to this Agreement have executed it as of the date first set forth above.

LICENSOR:

LICENSEE:

SYLVAN LEARNING, LLC,
a Delaware limited liability company

a limited liability company

By:

By:

, its Member

John McAuliffe, Chief Executive Officer

#### ILLINOIS RIDER TO THE ACE IT! TUTORING SYSTEM LICENSE AGREEMENT

To the extent the Illinois Franchise Disclosure Act, Ill. Comp. Stat.  $\S\S705/1 - 705/44$  applies, the terms of this Addendum apply.

- 1. Illinois law governs the ACE IT! Tutoring System License Agreement.
- 2. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
- 3. Your rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
  - 5. The following statement shall be deemed to amend the License Agreement:

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1

, a	limited liability company
By:	
	. its Member

LICENSEE:

# EXHIBIT D TO THE FRANCHISE DISCLOSURE DOCUMENT

## MULTI-TERRITORY INCENTIVE PLAN AMENDMENT

1

#### MULTI-TERRITORY INCENTIVE PLAN AMENDMENT

This Multi-Territory Incentive Plan Amendment (this "Amendment") to the Sylvan License Agreer	nent dated
for a territory in ("License Agreement") is entered into as of	(the "Effective
Date") by and between Sylvan Learning, LLC ("Sylvan"), a Delaware limited liability company, with	a principal place
of business at 2350 Airport Freeway, Suite 505, Bedford, TX 76022, and ("License	ee").

WHEREAS, Licensee has expressed an interest in purchasing more than one new Sylvan territory as part of a "Multi-Territory Incentive Plan;" and

WHEREAS, Sylvan is willing to offer certain modified terms and conditions as part of the Multi-Territory Incentive Plan related to the amount and timing of the Initial License Fees due and the Required Open Dates for the Primary Centers in each new Sylvan territory; and

WHEREAS, the parties wish to amend the License Agreement as described in Section 1 below to reflect the Multi-Territory Incentive Plan described herein; and

NOW, THEREFORE, in consideration of the mutual promises contained herein and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties agree as follows:

- 1. <u>Incorporation; Each Territory operated under a Separate License or Franchise Agreement.</u> This Amendment forms an essential part of the License Agreement to which it is attached. This Amendment is executed concurrently with the License Agreement or shortly thereafter, and so, the first territory in Section 2 below shall be operated pursuant to the License Agreement. The license or franchise agreement for the second and third territory to be developed shall be on Sylvan's then-current license or franchise agreement, the terms of which may be materially different from the terms of the License Agreement. At the time Licensee is ready to develop the second and third territory, Licensee will be disclosed with the then-current Sylvan franchise disclosure document with the then-current form of license or franchise agreement. Each Sylvan center developed hereunder shall be at a specific location, which shall be designated in the respective license or franchise agreement that is within the attached territories.
- 2. <u>Required Open Date</u>. Paragraph 1.6 of the License Agreement is hereby amended such that the "Required Open Date" for the Primary Center is replaced with the applicable date set forth in the table below. Likewise, the equivalent cause regarding the required opening date in subsequent franchise or license agreements for the second and third territories are modified with the below respective required opening dates:

Sylvan License Agreement/Territory	Required Opening Date
First Territory	6 months from Effective Date of this Amendment
Second Territory	12 months from Effective Date of this Amendment
· ·	
Third Territory	18 months from Effective Date of this Amendment

3. <u>Development Fee</u>. Upon execution of this Amendment, Licensee shall pay to Sylvan a Development Fee in the amount set forth in the Development Fee Schedule below. The Development Fee is fully earned by Sylvan when paid and is not refundable, in whole or in part, under any circumstances. If the Licensee has paid the respective Development Fees in full, Licensee will not pay any additional initial franchise fees for any of the territories to be developed under this Amendment when Licensee executes subsequent franchise or license agreements, pursuant to Section 1 above.

Number of Territories	Total Development Fees
One Territory	\$36,900
Two Territories	\$56,900
Three Territories	\$76,900

1

#### 4. Material Defaults.

- 4.1. The following constitutes an additional curable material default under each of the License Agreements:
  - 4.1.1. Failure to commence full and continuous operation of any Primary Center by its Required Open Date.
- 4.2. Each of the following constitutes an additional incurable material default under each of the License Agreements:
  - 4.2.1. Failure to operate fully and continuously each Primary Center after its Required Open Date without the prior written approval of Sylvan.
  - 4.2.2. Failure to pay any Development Fee or portion thereof in full and on time according to the schedule above.
- 4.3. Each material default above shall be governed by the default and termination provisions set forth within the License Agreements.

IN WITNESS WHEREOF, the parties hereto have set their respective hands, as of the Effective Date.

LICE	NSOR:	LIC	ENSEE:
	N LEARNING, LLC, ware limited liability company	a ,	limited liability company
Ву: _		By:	
	John McAuliffe, Chief Executive Officer		, its Member

# EXHIBIT E TO THE FRANCHISE DISCLOSURE DOCUMENT

## OPERATIONS MANUAL TABLE OF CONTENTS

## © 2024 Sylvan Learning, LLC

#### **Contents**

introduction	
Introduction to Sylvan	
Company Background	
Types of Centers and Territories	
Mandatory and Recommended Operational and Educational Days/Hours	
The Sylvan Brand	20
The Power of a Brand	
Marketing Standards	20
Funding	
Exclusive Use of Sylvan-Approved Advertising	23
Customer-Facing Center-Created Forms	25
Territory Advertising Policy: Supporting Customer Choice	25
The Sylvan Learning Center Territory Advertising Policy	
Policy Center and Co-op Website Guidelines	27
Online Marketing Policy	27
Territory Public Relations Policy	27
Territory Media Relations Policy	28
Privacy, Data Collection, and Customer Contact	
National Media Relations	30
National Marketing Calendar	31
National Website	32
Appointment Requests	33
Receiving Web Form Data	33
Crisis Communication	33
Relationship Marketing	34
Events and Local Marketing	44
Why Grand Openings Are So Important	47
Key Steps to Hosting a Great Event	47
Promoting your Event	50
Marketing Channels	51
Summary: Local Marketing Importance	52
Voice of the Customer	55
Enroll: Breakthrough Selling	56
Enroll: Breakthrough Selling—the Sylvan Sales Approach	56
The Enroll Sales Approach	
Enroll: Breakthrough Selling Sales Process	
Overcoming Common Objections	59
The Customer Experience	61
Customer Focus.	61
Customer Response	61
Maintaining Quality Assurance	61
Instilling Quality Service	61
Sylvan Service Pledge	63
Answering Machine and/or Voicemail Greetings	63
Sylvan Customer Service Principles	64
The Customer Lifecycle	64
Education and Assessment	67
Assessment Practices	67
Information Release	67
STAR Assessment Delivery Area	67
Proctoring the Insight Assessments	68
Altering Standard Procedures	68

Operations Manual Table of Contents

SYLVAN LEARNING, LLC Franchise Disclosure Document | 2024

## © 2024 Sylvan Learning, LLC

Student Files/Folders	
Instruction	
Scheduling Teachers and Students	70
Scheduling Frequency for Students	70
Student-to-Teacher Ratio	
Tracking Attendance, Absences, and Make-Ups	
Preparing for Daily Instruction	
Preparing for a Student's First Hour of Instruction	73
Managing Student Notebooks	73
During Instruction	74
Instructional Management: Director Responsibilities	74
Instructional Floor Management	75
Instructional Management: Teacher Responsibilities	75
Learning Logs	
Transition Time	78
Interaction with Students and Teachers	78
Recording the Tokens Earned	78
The Sylvan Motivation Program	79
Buying at the Sylvan Store	
Monitoring/Supervising Students	
Managing Instructional Materials	
After Instruction	81
Paper-Based Instruction: Updating Notebooks	
Ongoing Conferences	
Care Calls	87
Progress Assessment	87
Teacher Conference/Contact	89
Exit Conference	89
Application for Approval of New Materials or Programs	91
Staffing	
Staff Management	93
Center Level Website Staff Bios	
Sylvan Staffing Requirements	
Dedicated Center Staff	
Sylvan Staff Identification	
Additional Requirements for SylvanSync Satellites	
Individual Employee Meetings	
Communication and Motivation	
Sample Job Descriptions	100
Establishing Your Role as a Franchisee	
Employer Recordkeeping	
Employee Handbooks	
Attracting Employment Candidates	
Initial Screening of Employment Candidates	
Interviewing Employment Candidates	
Sample Acceptance Letters	
Sample Rejection Letters	
Performance Appraisals	
Dealing with Employee Misconduct	
Employee Termination	
Employee Transfers from Center to Center	
Center Level Website Employment Form	132

## © 2024 Sylvan Learning, LLC

Society of Human Resources Management	
Troubleshooting Staffing	133
Training and Certification	134
Sylvan University	134
Initial Training and Sylvan 101	135
Director Training and Certification	136
Enroll: Breakthrough Selling Certification	137
GUIDE to Inquiry Excellence Certification	138
Math Center Certification	139
Teacher Training and Certification	139
Annual Conference	
Maintaining Quality Service	142
Addressing Customer Concerns	
Customer Complaints	
Complaints to the Better Business Bureau	
Measuring Customer Service	
Parent Surveys	
Inquiry Quality Check	
Tracking Key Business Indicators	
Quality Assurance Review (QAR)—Introduction	
Student Test Record Storage	
Franchise Business Consultants	
Setting Up Your Financial and Legal Structure	
Introduction	
Structuring Your Business	
Insurance	
Occupational Safety and Health Administration Standards	
Business Start-Up Planning	
Naming Your Business Entity; Assumed Business Name	158
Securing a Business License and/or Permit	
Company Tax Obligations	
Territory Development and New Center Growth	
Site Selection	
Site Requirements	
SylvanSync Satellite Site Criteria	
Center Layout and Design	
Introduction	
Center Floor Plans	
Paint Colors and Schemes	
Center Layout and Design Assistance	
Parent/Welcome Lobby	
Director's Office	
Director of Education's Office	
Administrator's Office	
Instructional Area	
Seminar and Testing Room	
Staff Work Room and Storage Area	
Vestibule (optional).	
Cleanliness	
Sylvan Policy on Animals in the Center	
Center Furniture	
Center Restroom Requirements	
1	

## © 2024 Sylvan Learning, LLC

Wall Graphics and Other Furniture	
Additional Furniture to Be Ordered	181
Signage	181
Sylvan Center Setup and Ordering Procedures	182
Office Furniture and Office Equipment	182
Franchise Reporting and Accounting	184
Corporate Email Standards	184
Credit Card Payment Setup	184
Statement Billing	184
Licensee Monthly Report	185
Annual Financial Statement	185
Accounting Overview	185
Accounting Software	186
Financial Statement Glossary of Terms	186
Break-Even Analysis	187
Calculating the Break-Even Point	187
Business Planning and Analysis	191
Potential Business Transfer	
Collection Procedures	191
Returned Checks	192
Refunds	193
Financial Planning	193
Developing a Financial Business Plan	193
Profit and Loss Projections	199
Twelve-Month Profit and Loss Projection	199
Three-Year Profit Projection (Optional)	199
Planning for Sylvan Seasonality	199
Planning for Technical Issues	200
Analysis	200
Document/File Retention	200
Sylvan Annual Plan	201
Key Personnel Profile	201
Staff Planning	201
SWOT Analysis	202
Planning Pro Forma Tool	202
Closing a Center	
Center Closing Procedures	
Forms/Files Currently Required (see SLCLink):	203
Disposal of Branded Materials	

# EXHIBIT F TO THE FRANCHISE DISCLOSURE DOCUMENT

## LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

## LIST OF STATE ADMINISTRATORS

STATE	STATE ADMINISTRATOR	STATE	STATE ADMINISTRATOR
CALIFORNIA	Dept. of Financial Protection and Innovation 320 W. 4 <sup>th</sup> St., Ste. 750 Los Angeles, CA 90013 213.576.7505 866.275.2677	NEW YORK	NYS Department of Law Investor Protection Bureau 28 Liberty St. 21st Fl New York, NY 10005 212-416-8222
HAWAII	Commissioner of Securities of the State of Hawaii Department of Commerce and Consumer Affairs Business Registration Division Securities Compliance Branch 335 Merchant St. Honolulu, HA 96813 808.586.2722	NORTH DAKOTA	North Dakota Securities Department 600 East Blvd. Ave. State Capitol, Fifth Floor, Dept. 414 Bismarck, ND 58505-0510 701.328.4712
ILLINOIS	Franchise Bureau Office of the Attorney General 500 S. Second St. Springfield, IL 62706 217.782.4465	RHODE ISLAND	Securities Division Dept. of Business Regulation 1511 Pontiac Ave. John O. Pastore Complex – Building 69-1 Cranston, RI 02920 401.462.9585
INDIANA	Securities Commissioner Indiana Securities Division 302 W. Washington St., Room E-111 Indianapolis, IN 46204 317.232.6681	SOUTH DAKOTA	Division of Insurance Securities Regulation 124 S. Euclid, Suite 104 Pierre, SD 57501 605.773.4823
MARYLAND	Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, MD 21202-2021 410.576.6360	VIRGINIA	State Corporation Commission Division of Securities and Retail Franchising 1300 E. Main St, 9 <sup>th</sup> Floor Richmond, VA 23219 804.371.9051
MICHIGAN	Michigan Dept. of Attorney General Consumer Protection Division Franchise Section 525 West Ottawa Street G. Mennen Williams Bldg, 1st Floor Lansing, MI 48909 517.373.1837	WASHINGTON	Dept. of Financial Institutions Securities Division 150 Israel Road SW Tumwater, WA 98501 360.902.8760
MINNESOTA	Minnesota Dept. of Commerce 85 7 <sup>th</sup> Place East, Ste 280 St. Paul, MN 55101-2198 651.539.1600	WISCONSIN	Securities and Franchise Registration Wisconsin Securities Commission 345 W. Washington St, 4 <sup>th</sup> Floor Madison, WI 53703 608.266.3364

1

## LIST OF AGENTS FOR SERVICE OF PROCESS

CALIFORNIA Commissioner of Financial Protection and Innovation Department of Financial Protection and Innovation 320 West Fourth Street, Suite 750 Los Angeles, California 90013-2344 (213) 576-7500 Toll Free: (866) 275-2677	MINNESOTA Commissioner of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 296-4026	VIRGINIA Clerk of the State Corporation Commission 1300 East Main Street, 1st Floor Richmond, Virginia 23219 (804) 371-9733
HAWAII Commissioner of Securities of the State of Hawaii Department of Commerce & Consumer Affairs Business Registration Division Securities Compliance Branch 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722	NEW YORK New York State Department of State 99 Washington Avenue, 6 <sup>th</sup> Floor Albany, New York 12231 (518) 473-2492	WASHINGTON Director of Department of Financial Institutions Securities Division – 3rd Floor 150 Israel Road, S.W. Tumwater, Washington 98501 (360) 902-8760
ILLINOIS Illinois Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465	NORTH DAKOTA North Dakota Securities Commissioner 600 Boulevard Avenue, State Capitol Fifth Floor Bismarck, North Dakota 58505-0510 (701) 328-4712	WISCONSIN Commissioner of Securities 4822 Madison Yards Way, North Tower Madison, WI 53705 (608) 261-9555
INDIANA Indiana Secretary of State 200 West Washington Street, Room 201 Indianapolis, IN 46204 (317) 232-6681	RHODE ISLAND Director of Department of Business Regulation Department of Business Regulation Securities Division Bldg. 69, First Floor John O. Pastore Center 1511 Pontiac Avenue Cranston, Rhode Island 02920 (401) 462-9527	
MARYLAND Maryland Securities Commissioner 200 St. Paul Place Baltimore, Maryland 21202-2020 (410) 576-6360	SOUTH DAKOTA Director of Division of Securities Division of Insurance – Securities Regulation 124 S. Euclid Avenue, Suite 104 Pierre, SD 57501-3185 (605) 773-3563	
MICHIGAN Dept. of Energy, Labor, & Economic Growth Corporations Division P.O. Box 30054 Lansing, Michigan 48909 7150 Harris Drive Lansing, Michigan 48909 (517) 373-7117	TEXAS: Registered Agents Inc. 5900 Balcones Drive, Suite 100 Austin, Texas 78731	

V.2024.051024

# EXHIBIT G TO THE FRANCHISE DISCLOSURE DOCUMENT

## FINANCIAL STATEMENTS

## FINANCIAL STATEMENTS

Sylvan Learning, LLC's audited consolidated balance sheets as of December 31 of each of 2021, 2022, and 2023, and the related consolidated statements of income, stockholder's equity, and cash flows for each of the three years in the period ended December 31, 2023.

## Sylvan Learning, LLC and Subsidiaries

Consolidated Financial Statements Together with Independent Auditors' Report

As of December 30, 2023, and December 31, 2022 and For the Years Ended December 30, 2023, December 31, 2022 and December 25, 2021



#### INDEPENDENT AUDITORS' REPORT

To the Board of Directors of Sylvan Learning, LLC and Subsidiaries:

## **Opinion**

We have audited the accompanying consolidated financial statements of Sylvan Learning, LLC and Subsidiaries, a Delaware limited liability company, which comprise the consolidated balance sheets as of December 30, 2023 and December 31, 2022, and the related consolidated statements of operations and member's equity (deficit), and cash flows for the period from August 21, 2023 through December 30, 2023 (Successor Company), the period from January 1, 2023 through August 20, 2023 (Predecessor Company 1), the year ended December 31, 2022 (Predecessor Company 1), the period from September 27, 2021 through December 25, 2021 (Predecessor Company 1) and for the period from January 1, 2021 through September 26, 2021 (Predecessor Company 2) and the related notes to the consolidated financial statements.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Sylvan Learning, LLC and Subsidiaries as of December 30, 2023 and December 31, 2022, and the results of its operations and its cash flows for the period from August 21, 2023 through December 30, 2023 (Successor Company), the period from January 1, 2023 through August 20, 2023 (Predecessor Company 1), the year ended December 31, 2022 (Predecessor Company 1), the period from September 27, 2021 through December 25, 2021 (Predecessor Company 1) and for the period from January 1, 2021 through September 26, 2021 (Predecessor Company 2) in accordance with accounting principles generally accepted in the United States of America.

## **Basis for Opinion**

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Our responsibilities under those standards are further described in the Auditors' Responsibilities for the Audit of the Consolidated Financial Statements section of our report. We are required to be independent of Sylvan Learning, LLC and Subsidiaries and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

#### Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about Sylvan Learning, LLC and Subsidiaries' ability to continue as a going concern within one year after the date that the consolidated financial statements are available to be issued.

## Auditors' Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditors' report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with generally accepted auditing standards will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with generally accepted auditing standards, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures
  that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the
  effectiveness of Sylvan Learning, LLC and Subsidiaries' internal control. Accordingly, no such
  opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about Sylvan Learning, LLC and Subsidiaries' ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

March 8, 2024

SC+H Attest Services. P.C.

## Sylvan Learning, LLC and Subsidiaries Consolidated Balance Sheets

(in thousands of dollars)	Sı	ıccessor	Pred	Prodocessor (1)		
As of	Decem	Predecessor (1)  December 31, 2022				
715 0)	Бесет	<i>bei</i> 50, 2025	Decem	ber 31, 2022		
Assets						
Current Assets						
Cash and cash equivalents	\$	4,927	\$	7,469		
Trade and notes receivables, net		3,561		3,456		
Other receivables		1,669		-		
Prepaid expenses and other current assets		465		457		
Total Current Assets		10,622		11,382		
Property and Equipment, net		251		584		
Intangible Assets						
Goodwill		15,548		12,361		
Tradenames		30,500		24,987		
Franchise rights, net		14,143		15,619		
Software and educational content, net		22,283		23,922		
Outside Land Birth of Handanda		205		1 105		
Operating Lease Right-of-Use Assets Other Assets and Notes Receivable, net		895 735		1,105 354		
Other Assets and Notes Receivable, net		/33		334		
Total Assets	\$	94,977	\$	90,314		
Liabilities and Member's Equity						
Current Liabilities						
Accounts payable and accrued expenses	\$	3,140	\$	2,478		
Accrued salaries and benefits		2,792		3,055		
Due to related party		438		483		
Current portion of deferred revenue		1,503		2,049		
Current operating lease liabilities		171		1,526		
Other current liabilities		69		83		
Total Current Liabilities		8,113		9,674		
Long-Term Liabilities						
Deferred revenue, net of current portion		2,487		2,508		
Non-current operating lease liabilities		757		54		
Total Long-Term Liabilities		3,244		2,562		
				<u> </u>		
Total Liabilities		11,357		12,236		
Commitment and Contingencies (Notes 8 and 11)						
Sylvan Learning, LLC and Subsidiaries Member's Equity		83,620		78,078		
Total Liabilities and Member's Equity	\$	94,977	\$	90,314		

The accompanying notes are an integral part of these financial statements.

# Sylvan Learning, LLC and Subsidiaries Consolidated Statements of Operations and Member's Equity (Deficit)

(in thousands of dollars)											
	Sı	iccessor	Predecessor (1)						Predecessor (2)		
	For the Period August 21, 2023 through December 30, 2023		For the Period January 1, 2023 through August 20, 2023		For the Year Ended December 31, 2022		For the Period September 27, 2021 through December 25, 2021		For the Perio January 1, 2021 througl September 26 2021		
Revenue											
Franchise revenue	\$	13,990	\$	25,564	\$	39,502	\$	9,154	\$	26,713	
Tuition and testing revenue	Ψ	788	Φ	1,817	Ψ	2,794	Ψ	512	Ψ	1,772	
Total Revenues		14,778		27,381		42,296		9,666		28,485	
Operating Expenses											
Labor and benefits		2,104		6,767		10,658		2,411		7,478	
Marketing and advertising		3,089		6,586		9,486		2,186		6,232	
Depreciation and amortization		2,173		5,327		7,974		1,822		5,634	
Other instructional and operating expenses		2,663		5,163		9,202		3,961		5,293	
Total Operating Expenses		10,029		23,843		37,320		10,380		24,637	
Operating Income (Loss)		4,749		3,538		4,976		(714)		3,848	
Other (Income) Expenses, net											
Interest (income) expense, net		(71)		(81)		-		2		2,629	
Other non-operating expense, net		5		7		17		2		15	
Foreign and state income taxes		94		176		286		68		190	
Total Other Expenses, net		28		102		303		72		2,834	
Net Income (Loss)		4,721		3,436		4,673		(786)		1,014	
Member's Equity (Deficit), Beginning of Year/Period		74,514		78,078		82,141		(6,624)		(7,638)	
Distributions		(4,256)		(7,000)		(8,686)		-		-	
Impact of Acquisition		8,641		_		(50)		89,551			
Member's Equity (Deficit), End of Year/Period	\$	83,620	\$	74,514	\$	78,078	\$	82,141	\$	(6,624)	

The accompanying notes are an integral part of these financial statements.

## Sylvan Learning, LLC and Subsidiaries Consolidated Statements of Cash Flows

	Su	ccessor	Predecessor (1)						Predecessor (2)		
	Au 2023 Dece	For the Period August 21, 2023 through December 30, 2023		he Period nuary 1, 3 through gust 20, 2023	Dece.	For the Year Ended December 31, 2022		the Period mber 27, through mber 25,	Jan 2021 Septe	ne Period uary 1, through mber 26,	
		2023		2023		.022		2021		.021	
Cash Flows from Operating Activities											
Net income (loss)	\$	4,721	\$	3,436	\$	4,673	\$	(786)	\$	1,014	
Adjustments to reconcile net income (loss) to net											
cash and cash equivalents provided by operating activities:				25.6				100		2.50	
Depreciation and amortization of property and equipment		143		276		421		102		258	
Amortization of intangible assets		2,030		5,051		7,553		1,720		5,376	
Amortization of operating lease right-of-use assets		28		707		1,454		(314)			
Interest paid, but incurred in prior year		-		-		-		-		(32	
Bad debt expense (recovery)		(76)		24		355		2		(126	
Loss on asset disposals		-		-		-		-		5	
Loss on derecognition of operating lease right-of-use assets		100		-		-		-			
Changes in operating assets and liabilities:											
Trade and notes receivables, net		(211)		158		(1,082)		435		904	
Other receivables		(1,669)		-		-		-			
Prepaid expenses and other current assets		157		(165)		137		(169)		(36	
Other assets and notes receivable, net		(207)		(174)		(68)		528		17	
Accounts payable and accrued expenses and due to related party		363		254		(2,525)		2,532		294	
Accrued salaries and benefits		688		(951)		350		730		347	
Deferred revenue		403		(970)		823		403		(15	
Operating lease liabilities and other liabilities		(17)		(1,207)		(1,075)		(223)		(134	
Net Cash and Cash Equivalents Provided by Operating Activities		6,453		6,439		11,016		4,960		7,872	
Cash Flows from Investing Activities  Cash paid for internally developed software and educational content		(1,656)		(2,436)		(3,377)		(749)		(2,578	
Capitalized interest on internally developed software and educational content		(1,030)		(2,430)		(3,377)		(/49)		(2,376	
Purchases of property and equipment		(56)		(30)		(51)		(8)		(43	
Purchases of property and equipment		(36)		(30)		(51)		(8)		(43	
Net Cash and Cash Equivalents Used in Investing Activities		(1,712)		(2,466)		(3,428)		(757)		(2,677	
Cash Flows from Financing Activities											
Repayments on debt				(= 000)		(0.606)		-		(2,700	
Distributions		(4,256)		(7,000)		(8,686)		-			
Payments on Paycheck Protection Program loan		-		-		-		-		(2,391	
Net change in capital lease obligations		-		-				-		(1	
Net Cash and Cash Equivalents Used in Financing Activities		(4,256)		(7,000)		(8,686)		-		(5,092	
Increase (Decrease) in Cash and Cash Equivalents		485		(3,027)		(1,098)		4,203		103	
Cash and Cash Equivalents, beginning of period		4,442		7,469		8,567		4,364		4,261	
Cash and Cash Equivalents, end of period	\$	4,927	\$	4,442	\$	7,469	\$	8,567	\$	4,364	
Supplemental Disclosure of Cash Flow Information:											
Cash paid for interest	s		s	7	\$	4	s	4	s	2,687	
•	D.	-	3	/	3	4	Þ	4	3	2,08	
Supplemental Disclosure of Non-Cash Operating Activities											
Acquisition of operating right-of-use assets and lease liabilities	\$	626	\$	359	\$	-	\$	-	\$		
Supplemental Disclosure of Non-Cash Investing and Financing Activities											
Impact of Acquisition	\$	8,641			\$	(50)	6	89,551	S		

The accompanying notes are an integral part of these financial statements.

# Sylvan Learning, LLC and Subsidiaries Notes to the Consolidated Financial Statements As of December 30, 2023 and December 31, 2022 and For the Years Ended December 30, 2023, December 31, 2022 and December 25, 2021

(In thousands of dollars)

#### 1. DESCRIPTION OF BUSINESS AND BASIS OF ACCOUNTING

## **Description of Business**

Sylvan Learning, LLC is a single-member limited liability company formed on June 15, 2015, in the state of Delaware. As of December 31, 2022, Educate Corporate Center Holdings, LLC (NLC) and Sylvan In-Home, LLC (SIH) are wholly owned subsidiaries of Sylvan Learning, LLC (collectively, Sylvan or the Company). Sylvan is wholly owned by Educate, Inc. (the Parent). Sylvan is a leading pre-K-12 education company delivering supplemental education services to students and their families.

The Sylvan Learning brand delivers personalized tutorial programs primarily to school-aged children through a network of 560 franchised and 5 corporate owned centers. During the year ended December 30, 2023, the Company sold 57 new centers to franchisees. For over 45 years, the Company has provided trusted, personalized instruction to millions of students improving their academic achievement and helping them experience the joy of learning. Sylvan Learning Centers are staffed by trained teachers who deliver high quality education programs that are personalized for each student based upon the results of extensive diagnostic testing.

NLC operates 5 corporate owned learning centers which are located in Maryland. Corporate owned centers are used as testing grounds for new content, new operational procedures and vendor relationships prior to release to the Sylvan franchise system.

SIH operates online marketplace technology which is available throughout the entire United States. The SIH marketplace is utilized by Sylvan and Sylvan franchisees to match tutors with students who need supplemental education services.

On September 27, 2021, the Franchise Group (Predecessor 1) acquired Educate, Inc. and its subsidiaries for a purchase price of \$83,882 from Guggenheim Partners (Predecessor 2). The Franchise Group is an owner and operator of franchised and franchisable businesses that continually looks to grow its portfolio of brands while utilizing its operating and capital allocation philosophies to generate strong cash flows.

During the period from September 27, 2021 through December 25, 2021, the Franchise Group determined a change in reporting entity from Educate to Sylvan was advantageous to simplify the audit approach as the debt that previously required the audit of Educate was paid off through the acquisition. In accordance with Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) 250-210, *Other Presentation Matters*, the change was applied retroactively to all periods presented. The change had no effect on the net income (loss) of prior periods.

On August 21, 2023, a buyer group including members of the Company's senior management in a financial partnership with a consortium of affiliate entities, including B. Riley Financial, Inc. and Irradiant Partners (collectively, Successor), completed a merger and acquisition of the Franchise Group. Upon completion of the merger, the Franchise Group became a private company and was no longer listed or traded on NASDAQ. There were no significant changes to business operations of the Franchise Group as a result of the merger.

(In thousands of dollars)

### 1. DESCRIPTION OF BUSINESS AND BASIS OF ACCOUNTING - cont'd.

### Change in Fiscal Year-End

On September 27, 2021, the Board of Directors of Sylvan approved a change in the Company's fiscal year-end from December 31 to the Saturday closest to December 31 of each year. The decision to change the fiscal year-end was done to align Sylvan's year end with the Franchise Group's fiscal year-end.

### **Basis of Accounting**

The accompanying financial statements have been prepared using the accrual basis of accounting in accordance with accounting principles generally accepted in the United States of America.

### **Principles of Consolidation and Variable Interest Entities**

In accordance with ASC 810, *Consolidation*, Sylvan consolidates investments where it has a controlling financial interest. The usual condition for a controlling financial interest is ownership of a majority of the voting interest and, therefore, as a general rule ownership, directly or indirectly, of over fifty percent of the outstanding voting shares is a condition for consolidation. For investments in Variable Interest Entities (VIEs), as defined by ASC 810, consolidation is required when an entity is determined to be the primary beneficiary of the variable interest entity. The accompanying consolidated financial statements include the operations of Sylvan Learning, LLC, its wholly owned subsidiaries, Educate Corporate Center Holdings, LLC and Sylvan In-Home, LLC, and investments in a VIE, SLC National advertising Fund, Inc. (SNAF). All significant intercompany account balances and transactions have been eliminated in consolidation.

SNAF was formed to operate, control and administer a pooled advertising fund for the benefit of Sylvan and its franchisees. All contributions and income earned by SNAF are restricted to expenditures for advertising, marketing and promotional programs, and no part of the members' equity may be distributed or inure to the benefit of any of its members. Sylvan and the Franchise Owner's Association (FOA) each have the right to designate two directors of SNAF or such greater number as may compose one half the number of directors then authorized.

Sylvan determined SNAF qualifies as a VIE because Sylvan has the power to direct their activities that most significantly affects their economic performance and the right to receive economic benefits from arrangements with SNAF that could potentially be significant. Prior to becoming wholly owned subsidiaries during the year ended December 31, 2022, NLC and SIH were determined to be VIEs of Sylvan and are included in the accompanying consolidated financial statements during the period from September 27, 2021 through December 25, 2021 and the period from January 1, 2021 through September 26, 2021.

(In thousands of dollars)

### 1. DESCRIPTION OF BUSINESS AND BASIS OF ACCOUNTING - cont'd.

### Principles of Consolidation and Variable Interest Entities - cont'd.

The following condensed balance sheets as of December 30, 2023 and December 31, 2022 and condensed statements of operations for the period from August 21, 2023 through December 30, 2023 and the period from January 1, 2023 through August 20, 2023 (collectively, the year ended December 30, 2023), the year ended December 31, 2022, and the period from September 27, 2021 through December 25, 2021 and the period from January 1, 2021 through September 26, 2021 (collectively, the year ended December 25, 2021) present the consolidated financial position and consolidated results of operations of Sylvan and the variable interest entities.

	<u>Successor</u>										
		Consolidated Balance Sheet as of December 30, 2023									
	Sylvan		SNAF		Eliı	minations	Consolidate				
Total current assets Property and equipment, net Intangible assets, net Other long-term assets	\$	11,737 251 82,474 1,630	\$	3,605	\$	(4,720) - - -	\$	10,622 251 82,474 1,630			
Total assets	\$	96,092	\$	3,605	\$	(4,720)	\$	94,977			
Total current liabilities Other long-term liabilities	\$	9,228 3,244	\$	3,605	\$	(4,720)	\$	8,113 3,244			
Total liabilities		12,472		3,605		(4,720)		11,357			
Total member's equity		83,620						83,620			
Total liabilities and member's equity	\$	96,092	\$	3,605	\$	(4,720)	\$	94,977			

	Predecessor (1)											
		Consolidated Balance Sheet as of December 31, 2022										
		Sylvan		SNAF	Elin	minations	Cor	ns olidate d				
Total current assets Property and equipment, net Intangible assets, net Other long-term assets	\$	12,407 584 76,889 1,459	\$	3,153	\$	(4,178) - - -	\$	11,382 584 76,889 1,459				
Total assets	\$	91,339	\$	3,153	\$	(4,178)	\$	90,314				
Total current liabilities Other long-term liabilities	\$	10,699 2,562	\$	3,153	\$	(4,178)	\$	9,674 2,562				
Total liabilities Total member's equity		13,261 78,078		3,153		(4,178)		12,236 78,078				
Total liabilities and member's equity	\$	91,339	\$	3,153	\$	(4,178)	\$	90,314				

(In thousands of dollars)

### 1. DESCRIPTION OF BUSINESS AND BASIS OF ACCOUNTING - cont'd.

Principles of Consolidation and Variable Interest Entities - cont'd.

	Successor	
Consolidated	Statement	of Operations

	Fo	For the period from August 21, 2023 through Decemb									
	-	Sylvan		SNAF	Eli	minations	Cor	ns o lidate d			
Total revenues	\$	12,990	\$	3,983	\$	(2,195)	\$	14,778			
Total costs and expenses	•	8,241	_	3,983	,	(2,195)	,	10,029			
Operating income		4,749		-				4,749			
Other expense		28		-		-		28			
Net income	\$	4,721	\$	-	\$	_	\$	4,721			

### <u>Predecessor (1)</u> Consolidated Statement of Operations

	For the period from January 1, 2023 through August 20, 2023								
	 Sylvan		SNAF	J	Eliminations	Consolidated			
Total revenues	\$ 23,962	\$	8,198	\$	(4,779)	\$	27,381		
Total costs and expenses	 20,424		8,198		(4,779)		23,843		
Operating income	3,538		-		-		3,538		
Other expense	102		-		-		102		
Net income	\$ 3,436	\$	-	\$	-	\$	3,436		

### Predecessor (1) Consolidated Statement of Operations Voor Ended December 31, 2022

			Ye	ar Ended De	cembe	r 31, 2022		
		Sylvan		SNAF	Eliı	minations	Con	s o lidate d
Total revenues	\$	36,529	\$	12,109	\$	(6,342)	\$	42,296
Total costs and expenses		31,553		12,109		(6,342)		37,320
Operating income	,	4,976		-		-		4,976
Other expense		303		-		-		303
Net income	\$	4,673	\$	-	\$	_	\$	4,673

### Predecessor (1) Consolidated Statement of Operations

	for the period from September 27, 2021 through December 25, 2021										
	 Sylvan		NLC		SIH		SNAF	Eliı	minations	Con	is olidate d
Total revenues Total costs and expenses	\$ 7,945 8,328	\$	431 587	\$	80 255	\$	2,974 2,974	\$	(1,764) (1,764)	\$	9,666 10,380
Operating loss	(383)		(156)		(175)		-		-		(714)
Other expense	 72		-		-		-				72
Net loss	\$ (455)	\$	(156)	\$	(175)	\$		\$		\$	(786)

(In thousands of dollars)

### 1. DESCRIPTION OF BUSINESS AND BASIS OF ACCOUNTING - cont'd.

Principles of Consolidation and Variable Interest Entities – cont'd.

	<u>Predecessor (2)</u> Consolidated Statement of Operations										
		For t	he period f	rom .	January 1, 2	021	through Sep	te mbe	er 26, 2021		
	Sylvan		NLC		SIH		SNAF	Elir	ninations	Cor	s o lidate d
Total revenues Total costs and expenses	\$ 22,955 18,258	\$	1,639 1,780	\$	133 841	\$	8,549 8,549	\$	(4,791) (4,791)	\$	28,485 24,637
Operating income (loss)	4,697		(141)		(708)		-		-		3,848
Other expense	2,834		-		-		-				2,834
Net income (loss)	\$ 1,863	\$	(141)	\$	(708)	\$	-	\$	_	\$	1,014

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

### Cash and Cash Equivalents

The Company considers all highly liquid investments with original maturities of three months or less at the time of purchase to be cash equivalents. Accounts are guaranteed by the Federal Deposit Insurance Corporation (FDIC) up to \$250 per depositor. The Company periodically maintains cash balances in excess of FDIC coverage. Management considers this to be a normal business risk.

### Trade and Notes Receivable and Allowance for Credit Losses

Accounts receivable result from transactions with customers and franchisees. Notes receivable are primarily due from franchisees for sales of corporate owned centers and working capital receivables. The Company reports accounts and notes receivable at their net realizable value. The Company maintains an allowance for credit losses for estimated losses resulting from the inability of its franchisees and debtors to make required payments. The Company calculates the allowance based on a specific analysis of past due balances and also considers current and expected economic, market and industry factors affecting the Company's customers, including their financial condition, the aging of account balances, historical credit loss experience and any other specific customer collection concerns. The Company also performs rigorous credit evaluation of those franchisees who are the counterparties to the notes receivable. Since the Company's trade receivables are largely similar, the Company evaluates its allowance for credit losses as one portfolio segment. The measurement and recognition of credit losses involves the use of judgment and represents management's estimate of expected lifetime credit losses based on historical experience and trends, current conditions and reasonable and supportable forecasts. At origination, the Company evaluates credit risk based on the aforementioned factors. On a continuing basis, data for each customer and franchisee is regularly reviewed based on past-due status to evaluate the adequacy of the allowance for credit losses; actual write-offs are charged against the allowance. Invoices are past due if not received by the due date which is typically 15-30 days. Receivable balances are written off when management determines that collection is unlikely. Actual collection experience has not differed significantly from the Company's estimates, due primarily to credit and collections practices. Accounts and notes receivable balances due from franchisees are secured by substantially all of the assets of the franchisee's business.

(In thousands of dollars)

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES - cont'd.

### Trade and Notes Receivable and Allowance for Credit Losses – cont'd.

Because the economic environment is subject to volatility, the ultimate amount of credit losses resulting from the trade and notes receivable may differ from the allowances established at December 30, 2023 and December 31, 2022 and such differences could be material. The allowance for credit losses totaled \$1,363 and \$1,437 as of December 30, 2023 and December 31, 2022, respectively.

Changes in the analysis of the allowance for credit losses are as follows:

	Suc	ecessor	Pred	lecessor (1)
	As of and for the		As o	f and for the
	period	August 21,	perio	d January 1,
	2023	through	202	23 through
	Dece	ember 30,	A	ugust 20,
		2023		2023
Allowance for credit losses, beginning of period	\$	1,473	\$	1,437
Additions: charges to credit loss expense		-		37
Deductions: write-offs		(34)		(1)
Deductions: recoveries		(76)		
Allowance for credit losses, end of period	\$	1,363	\$	1,473

### Fair Value of Financial Instruments

The amounts reported in the consolidated balance sheets for cash and cash equivalents, trade receivables, accounts payable and accrued expenses approximate fair value due to their short term nature. Management believes that the net book value of notes receivable after considering the allowance approximates the fair value of such notes as of December 30, 2023 and December 31, 2022.

### **Property and Equipment**

Property and equipment is stated at net book value. Expenditures for major additions and improvements are capitalized. When property and equipment are retired or otherwise disposed, the cost and accumulated depreciation and amortization are removed from the accounts and any resulting gain or loss is included in the statement of operations and member's equity (deficit) for the respective period. The cost of property and equipment is depreciated and amortized using the straight-line method over the following estimated useful lives:

Furniture and fixtures	2-7 years
Educational materials	2-5 years
Computer equipment and software	2-3 years

Leasehold improvements Lesser of useful life or remaining lease term

(In thousands of dollars)

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES - cont'd.

### Goodwill and Indefinite Lived Intangible Assets

The Company accounts for goodwill and intangible assets in accordance with ASC 350, *Intangibles – Goodwill and Other*. Goodwill and non-amortizing intangible assets are initially recorded at their fair values. These assets are not amortized but are evaluated as of the end of July of each fiscal year, and a more frequent evaluation is performed if an event occurs or circumstances change that would more likely than not reduce the assets fair values below their carrying values. Such events or circumstances could include, but are not limited to, significant negative industry or economic trends and unanticipated changes in the competitive environment.

For goodwill, the Company performs a qualitative and/or quantitative assessment to determine whether it is more likely than not that each reporting unit's fair value is less than its carrying value, including goodwill. If the Company determines that it is more likely than not that the fair value of the reporting unit is less than its carrying value, the Company then estimates the fair value. The Company uses either a market multiple method or a discounted cash flow method to estimate the fair value of its reporting units and recognizes goodwill impairment for any excess of the carrying amount of a reporting unit's goodwill over its estimated fair value.

For non-amortizing intangible assets, the Company evaluates its tradenames for impairment by comparing the fair value, based on an income approach using the relief-from-royalty method, to its carrying value. If the carrying value of the asset exceeds its estimated fair value, an impairment loss is recognized in an amount equal to that excess.

Prior to the acquisition in September of 2021, Sylvan followed Accounting Standards Update (ASU) No. 2014-02—*Intangibles—Goodwill and Other (Topic 350): Accounting for Goodwill.* The adopted guidance allows private companies to amortize goodwill over the useful life of the primary asset acquired in a business combination, not to exceed 10 years. The primary asset is the principal identifiable long-lived or intangible asset from which the acquired business derives its cash-flow generating capacity. Under the standard, goodwill should be tested for impairment only when a triggering event occurs that may reduce the fair value of an entity below its carrying amount. No triggering events occurred for the period from January 1, 2021 through September 26, 2021 that would have reduced the fair value of the entity below its carrying amount.

### **Amortizable Intangible Assets**

Amortizable intangible assets consist principally of franchise license rights, patents and software and educational content. Intangible assets with finite lives are amortized over their estimated useful lives. The franchise license rights are being amortized over their expected useful lives of 15 years while patents and software and educational content are typically amortized over 5 years.

(In thousands of dollars)

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES - cont'd.

### **Software and Educational Content**

Sylvan incurs direct costs to develop software and educational content for use by our franchisees. Costs associated with programs which are deemed recoverable through operations of the franchise system are deferred until the program is released for use. The Company amortizes the capitalized software and educational content using the straight-line method over five years.

### **Impairment of Long-Lived Assets**

The Company periodically evaluates the net realizable value of long-lived assets, principally furniture and equipment, goodwill and intangible assets, relying on a number of factors including operating results, business plans, economic projections and anticipated future cash flows. An impairment in the carrying value of an asset is recognized whenever future cash flows (undiscounted) from an asset or asset group are estimated to be less than its carrying value. The amount of the impairment recognized is the difference between the carrying value of the asset and its fair value. There were no impairment losses recorded during the years ended December 30, 2023, December 31, 2022, and December 25, 2021.

### **Right-of-Use Assets**

Effective September 27, 2021, the Company has adopted a policy of capitalizing right-of-use assets of at least one year held under lease liabilities as defined by ASC 842. These assets are made up of office space leases (Note 8). The leased assets are recorded at the present value of the lease liability. Right-of-use assets are reviewed annually for impairment in accordance with ASC 360. During the years ended December 30, 2023, December 31, 2022, and December 25, 2021 there were no right-of-use assets that were deemed to be impaired.

### **Revenue Recognition**

ASC 606, *Revenue from Contracts with Customers*, outlines a single, comprehensive model for entities to use in accounting for revenue arising from contracts with customers and superseded most revenue recognition guidance issued by the FASB, including industry specific guidance.

ASC 606 includes the following steps for revenue recognition:

- Identification of the contract, or contracts with a customer.
- Identification of the performance obligations in the contract.
- Determination of the transaction price.
- Allocation of the transaction price to the performance obligations in the contract.
- Recognition of the revenue when, or as, the Company satisfies a performance obligation.

The Company has determined that the franchise fee received from sales of new franchises and the selling of existing franchises do not relate to separate and distinct performance obligations from the franchise right and those upfront fees will therefore be recognized as revenue over the term of each respective franchise agreement, which is typically 10 years.

(In thousands of dollars)

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES - cont'd.

### Franchise and Franchise-related Revenue

Most franchised learning centers pay a 16% monthly royalty fee based on cash receipts, payable by the fifteenth day of the following month. New franchises pay an upfront fee, which is deferred and recognized over the initial license term of 10 years or the customer life, whichever is shorter. Other franchise-related revenues include Contact Center charges, SylvanSync fees, HUB fees, and Content revenues, which are annual or monthly services charged in addition to the royalty. Franchise and franchise-related revenues are recorded in the month earned if collectability is reasonably assured, and when performance obligations have been satisfied. Estimates of royalties earned but unreported by franchisees at the balance sheet date are recorded as revenue and accounts receivable and are adjusted to actual amounts when reported and paid by the franchisee.

### **Tutoring Revenue and Media Revenue**

Fees from tutoring, supplemental education services and advertising are recognized as revenue in the period the services have been provided or performance obligations have been satisfied. Fees collected in advance of providing services are recorded as deferred revenue.

A summary of disaggregated revenue is as follows:

	Su	ccessor	Pred	ecessor (1)	Pred	ecessor (1)	Prede	cessor (1)	Prede	cessor (2)	
	For	the Period					For t	he period	For t	he period	
	Augu	st 21, 2023	For	the Period			from	September	Janua	ry 1, 2021	
	t	through		ary 1, 2023	For the year ended		27, 20	21 through	through		
	Dec	ember 30,	through	h August 20,	December 31,		Dece	ember 25,	September 26,		
		2023		2023	2022		2021		2021		
Disaggregation of Revenue											
Franchise revenue	\$	10,764	\$	19,502	\$	29,840	\$	7,116	\$	20,387	
Media revenue		1,674		3,188		5,383		1,206		3,683	
Tutoring revenue		788		1,817		2,794		512		1,772	
Contact center revenue		1,006		1,904		2,570		564		1,362	
Other		546		970		1,709		268		1,281	
	\$	14,778	\$	27,381	\$	42,296	\$	9,666	\$	28,485	

### **Contract Balances**

Contract assets consist of deferred commissions related to the sale of new franchises or the resale of existing franchise locations. Contract assets are presented in other assets and notes receivable, net on the consolidated balance sheets and total \$709 and \$329 as of December 30, 2023, and December 31, 2022, respectively.

(In thousands of dollars)

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES - cont'd.

### Contract Balances - cont'd.

Contract liabilities consist of deferred franchise sales revenue from the sale of new franchises or the resale of existing franchise locations and tutoring revenue for tutoring services prepaid and not yet provided. The current portion of deferred revenue totaled \$1,503 and \$2,049 as of December 30, 2023, and December 31, 2022. Long-term contract liabilities are in long-term liabilities on the consolidated balance sheets and are \$2,487 and \$2,508 as of December 30, 2023, and December 31, 2022, respectively.

### **Marketing and Advertising**

Sylvan expenses marketing and advertising costs as incurred.

### **Income Taxes**

Sylvan has elected to be taxed as a limited liability corporation under sections of Federal and state income tax laws. Accordingly, the Company is taxed under the partnership provisions of the Internal Revenue Code (the Code). Under the partnership provisions of the Code, the members of the limited liability company include the Company's income or loss on their respective tax returns. Amounts recorded on the consolidated statements of operations and member's equity (deficit) related to minimum state payments and amounts withheld from royalty from international franchisees.

The Company recognizes the financial statement benefit of an income tax position only after determining that the relevant taxing authority would more likely than not sustain the position following an audit. For tax positions meeting the more likely than not threshold, the amount recognized in the financial statements is the largest benefit that has a greater than 50% likelihood of being realized upon ultimate settlement with the relevant taxing authority.

The Company's policy is to treat interest and penalties associated with taxes as income tax expense. No interest or penalties were recognized for the years ended December 30, 2023, December 31, 2022, and December 25, 2021.

The Company is subject to routine audits by taxing jurisdictions; however, there are currently no audits for any tax periods in progress.

### **Use of Estimates**

The preparation of financial statements in accordance with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results may differ from those estimates.

(In thousands of dollars)

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES - cont'd.

### **Recently Adopted Accounting Principles**

In January 2017, the FASB issued ASU No. 2017, *Intangibles - Goodwill and Other (Topic 350):* Simplifying the Test for Goodwill Impairment. This standard eliminates step two from the goodwill impairment test. Instead, an entity should compare the fair value of a reporting unit with its carrying amount and recognize an impairment charge for the amount by which the carrying amount exceeds the reporting unit's fair value, not to exceed the total amount of goodwill allocated to the reporting unit. The ASU is effective for the Company for the 2023 fiscal year. The Company adopted ASU 2017-04 during the year ended December 30, 2023 on a prospective basis. The adoption of ASU 2017-04 did not have a material effect on the Company's financial position, results of operations or financial statement disclosures.

In June 2016, the FASB issued ASU 2016-13, *Financial Instruments – Credit Losses* (*Topic 326*). The objective of the amendments in this update is to provide financial statement users with more decision-useful information about the expected credit losses on financial instruments and other commitments to extend credit held by a reporting entity at each reporting date. The new expected credit loss methodology, which is based on historical experience, current conditions and reasonable and supportable forecasts, replaced the incurred loss model for measuring and recognizing expected credit losses. The Company used the modified retrospective approach to implement ASU 2016-13 during the year ended December 30, 2023 which does not require adjustments to comparative periods or modified disclosures in those comparative periods. Based on historical trends, the financial condition of the Company's customers and management's expectations of economic and industry factors affecting the Company's customers, the adoption of ASU 2016-13 did not have a material impact on the Company's financial position or results of operations.

### 3. PURCHASE ACCOUNTING

On September 27, 2021, The Franchise Group acquired Educate, Inc.. The preliminary fair value of the consideration transferred at the acquisition date was \$83,882. Acquisition fees of \$1,792 were incurred and are recorded in other instructional and operating expenses on the accompanying consolidated statements of operations and member's equity (deficit) for the period ending December 25, 2021.

The purchase price was allocated to assets acquired and liabilities assumed based on the estimate of fair value on the acquisition date. The excess of the purchase price over the fair value of the tangible and intangible assets acquired totaled \$13,367 and was recorded as goodwill on the balance sheet.

(In thousands of dollars)

### 3. PURCHASE ACCOUNTING - cont'd.

The following table summarizes the estimated assets and liabilities assumed as of the date of acquisition:

	(Pred	decessor 1)
		As of
	Septen	nber 27, 2021
Cash and cash equivalents	\$	4,364
Accounts receivable		3,166
Prepaid expenses and other assets		425
Property and equipment		1,048
Goodwill		13,367
Trademarks		24,987
Franchise rights		18,265
Software and educational content		26,423
Other non-current assets		3,059
Total assets		95,104
Accounts payable and expenses		2,954
Accrued salaries and benefits		1,975
Deferred revenue		1,057
Lease liabilities		2,875
Other liabilities		2,361
Total liabilities		11,222
Consideration transferred	\$	83,882

The preliminary estimates of the fair value of identifiable assets acquired and liabilities assumed are subject to revisions, which may result in an adjustment to the preliminary values presented above.

During December 2021, a working capital adjustment of \$956 resulted in a reduction of consideration transferred and goodwill. As of December 25, 2021, total consideration transferred and goodwill totaled \$82,926 and \$12,411, respectively.

The Company completed the final purchase price allocation on March 26, 2022. A working capital adjustment of \$50 was made and resulted in a reduction of consideration transferred and a reduction of goodwill. As of December 31, 2022, total consideration transferred and goodwill totaled \$82,876 and \$12,361, respectively.

(In thousands of dollars)

### 3. PURCHASE ACCOUNTING - cont'd.

On August 21, 2023, a buyer group including members of the Company's senior management in a financial partnership with a consortium of affiliate entities, including B. Riley Financial, Inc. and Irradiant Partners completed a merger and acquisition of the Franchise Group. The preliminary fair value of the consideration transferred at the acquisition date was \$86,207. There were no acquisition fees incurred related to the transaction.

The purchase price was allocated to assets acquired and liabilities assumed based on the estimate of fair value on the acquisition date. The excess of the purchase price over the fair value of the tangible and intangible assets acquired totaled \$23,397 and was recorded as goodwill on the balance sheet.

The following table summarizes the estimated assets and liabilities assumed as of the date of acquisition:

	(Successor)	
		As of
	Augu	st 21, 2023
Cash and cash equivalents	\$	4,442
Accounts receivable		3,274
Prepaid expenses and other assets		622
Property and equipment		338
Goodwill		16,399
Trademarks		31,700
Franchise rights		15,700
Software and educational content		22,200
Other non-current assets		826
Total assets		95,501
Accounts payable and expenses		3,215
Accrued salaries and benefits		2,104
Deferred revenue		3,587
Lease liabilities		319
Other liabilities		69
Total liabilities		9,294
Consideration transferred	\$	86,207

The preliminary estimates of the fair value of identifiable assets acquired and liabilities assumed are subject to revisions, which may result in an adjustment to the preliminary values presented above.

During December 2023, an intangible asset and working capital valuation adjustment of \$3,151 resulted in a reduction of consideration transferred, goodwill, trademarks, and franchise rights. As of December 30, 2023, total consideration transferred, goodwill, trademarks, and franchise rights totaled \$83,056, \$15,548, \$30,500, and \$14,600, respectively.

(In thousands of dollars)

### 4. PROPERTY AND EQUIPMENT

Property and equipment consist of the following as of:

	Successor		Pred	ecessor (1)
	Decei	mber 30,	Dec	ember 31,
	2	023		2022
Furniture and fixtures	\$	106	\$	114
Computer equipment and software		188		320
Leasehold improvements		100		667
		394		1,101
Less: accumulated depreciation and amortization		(143)		(517)
Property and equipment, net	\$	251	\$	584

Depreciation is computed using the straight-line method over the estimated useful lives of the assets. Amortization of leasehold improvements is computed using the straight-line method over the lesser of the useful life of the asset or the remaining term of the lease. Depreciation and amortization expense for the period from August 21, 2023 through December 30, 2023, the period from January 1, 2023 through August 20, 2023, the year ended December 31, 2022, the period from September 27, 2021 through December 25, 2021, and the period from January 1, 2021 through September 26, 2021 totaled \$143, \$276, \$421, \$102, \$258, respectively.

### 5. GOODWILL

Changes in the carrying amount of goodwill are summarized as follows:

	Succ	essor	Pred	ecessor (1)	Prede	cessor (1)	Prede	ecessor (1)	Prede	cessor (2)				
	As of ar	nd for the	As of	and for the			As of	and for the	As of	and for the				
	period A	ugust 21,	period	period January 1,		of and for	period	September	period	January 1,				
	2023 through		202	2023 through the year endir		ear ending	27, 20	23 through	2021	through				
	Decen	nber 30,	Αι	ugust 20,	December 31,		December 31,		December 31,		Dec	ember 25,	Septe	ember 26,
	20	)23		2023	2022		2022			2021		2021		
Goodwill	\$	_	\$	12,361	\$	12,411	\$	-	\$	2,361				
Plus (Less): purchase accounting		15,548		(12,361)		(50)		12,411		(1,770)				
Less: amortization		-		-		-		-		(591)				
Goodwill, net	\$	15,548	\$	-	\$	12,361	\$	12,411	\$	-				

During the period January 1, 2021 through September 26, 2021, Sylvan incurred \$591 of goodwill amortization expense, based on a ten year amortization of goodwill. There was no goodwill impairment for the years ended December 30, 2023, December 31, 2022, and December 25, 2021.

(In thousands of dollars)

### 6. OTHER INTANGIBLE ASSETS

A summary of other intangible assets other than goodwill at December 30, 2023 and December 31, 2022 is as follows:

		Su	ccessor	Prede	ecessor (1)
	Useful Life	Dec	ember 30,	Dec	ember 31,
	In years		2023		2022
Indefinite-lived intangible assets not subject to amortization:					
Trademarks	Indefinite	\$	30,500	\$	24,987
			30,500		24,987
Intangibles subject to amortization:					
Software and educational content	5		23,856		30,548
Franchise Rights	15		14,600		18,265
			38,456		48,813
Less: accumulated amortization			(2,030)		(9,272)
Intangibles, net	·	\$	66,926	\$	64,528

Amortization expense for other intangible assets for the period from August 21, 2023 through December 30, 2023, the period from January 1, 2023 through August 20, 2023, the year ended December 31, 2022, the period from September 27, 2021 through December 25, 2021, and the period from January 1, 2021 through September 26, 2021 totaled \$2,030, \$5,051, \$7,553, \$1,720, and \$5,376, respectively. Of this amortization, amortization expense for software and educational content which is included in the preceding sentence, for the period from August 21, 2023 through December 30, 2023, the period from January 1, 2023 through August 20, 2023, the year ended December 31, 2022, the period from September 27, 2021 through December 25, 2021, and the period from January 1, 2021 through September 26, 2021 totaled \$1,573, \$3,947, \$5,305, \$1,321, \$3,900, respectively. As a result of the Successor acquisition, intangible assets and associated useful lives were revalued as of August 21, 2023. Based on the revalued useful lives of assets, the weighted average amortization period for software and patent intangible assets subject to amortization was five years as of December 30, 2023 and December 31, 2022. The weighted average amortization period for franchise rights subject to amortization was approximately 15 years as of December 30, 2023 and December 31, 2022. The weighted average of total intangible assets subject to amortization was 9 years as of December 30, 2023 and December 31, 2022.

(In thousands of dollars)

### 6. OTHER INTANGIBLE ASSETS - cont'd.

As of December 30, 2023, estimated future amortization expense of intangible assets subject to amortization is as follows:

2024	\$ 5,736
2025	5,736
2026	5,736
2027	5,736
2028	4,164
Thereafter	9,318
	\$ 36,426

### 7. DEBT

### **Acquisition by the Franchise Group**

On September 27, 2021, Sylvan was acquired by the Franchise Group in a cash free, debt free deal. The 2015 Term Loan, was paid off through the acquisition during the period from January 1, 2021 through September 26, 2021. As of December 30, 2023 and December 31, 2022, Sylvan has no debt outstanding.

### **Paycheck Protection Program Loan**

On April 13, 2020, the Company received \$2,374 from the Paycheck Protection Program (PPP). This program was created by the federal government to assist small businesses who were affected by the COVID-19 pandemic.

The PPP loan had a two-year maturity with a 1% interest rate. Interest accrued on the loan during the first 16 months, however, no payments were due.

The Company elected to pay the PPP loan back before maturity and in full, with interest, in March 2021. Interest expense incurred during the period from January 1, 2021 through September 26, 2021 totaled \$5.

### 8. LEASES

Effective September 27, 2021, the Company adopted ASU 2016-02, *Leases* (ASC 842), which modifies the guidance on lease accounting. Sylvan conducts a significant portion of its operations from leased facilities which consist of retail store and office locations. The lease terms of substantially all these leases are five years or less. Operating leases with an initial term of 12 months or less are not recorded on the consolidated balance sheets, and Sylvan recognizes rent expense for these leases on a straight-line basis over the lease term. For leases with an initial term in excess of 12 months, lease right-of-use assets and lease liabilities are recognized based on the present value of the future lease payments over the committed lease term at the lease commencement date.

(In thousands of dollars)

### 8. LEASES - cont'd.

Sylvan's leases do not provide an implicit rate; therefore, Sylvan uses its incremental borrowing rate and the information available at the lease commencement date in determining the present value of future lease payments. The incremental borrowing rate is the rate of interest that Sylvan would have to pay to borrow, on a collateralized basis over a similar term, an amount equal to the lease payments in a similar economic environment. Many of the leases contain options to renew, generally with one renewal period ranging from one to three years.

During December 2022, Sylvan exercised the early termination option outlined in the headquarters lease. The total exit payment was \$801 and was expensed on a straight-line basis over the remaining lease term through November 2023.

As a result of the Successor acquisition, the Company's operating leases were subject to market revaluation as of August 21, 2023. Operating leases with remaining terms of 12 months or less as of the valuation date were derecognized. The company recognized acquisition adjustments related to derecognition and above market rent valuation adjustments for operating right-of-use assets and lease liabilities totaling \$462 and \$627, respectively, during the period from August 21, 2023 through December 30, 2023.

Sylvan has lease agreements with lease and non-lease components, which the Company elects to combine as one lease component for all classes of underlying assets. Non-lease components include variable costs based on actual costs incurred by the lessor related to the payment of real estate taxes, common area maintenance, and insurance. These variable payments are expensed as incurred as variable lease costs.

Rent expense is recorded in other instructional and operational expenses on the consolidated statements of operations and member's equity (deficit) and amounted to the following:

	Suc	Successor Predecessor (1) Predecessor (1)		Predecessor (1)		Prede	cessor (2)			
							For th	ne period		
	For th	ne period					from S	eptember	For th	ne period
	from A	August 21,	For the	period	For	the year	27,	2021	from J	anuary 1,
	2023	through	from Ja	nuary 1,	6	ended	thi	rough	2021	through
	Dece	mber 30,	2023 t	hrough	Dece	ember 31,	Dece	mber 25,	Septe	mber 26,
	2	2023	August :	20, 2023		2022	2	2021	2	2021
Operating lease cost	\$	361	\$	782	\$	1,408	\$	341	\$	615
Variable operating lease cost		82		104		101		18		37
Total operating lease cost	\$	443	\$	886	\$	1,509	\$	359	\$	652

(In thousands of dollars)

### 8. LEASES - cont'd.

Future minimum lease payments, net of sublease payments, at December 30, 2023, by year and in the aggregate, under all noncancelable operating leases are as follows:

2024	\$ 171
2025	206
2026	216
2027	159
2028	150
Thereafter	26
Total	\$ 928

The weighted average remaining lease years and weighted average discount rates are as follows:

	Suc	cessor	Prede	cessor (1)
	Decei	December 30,		ember 31,
	2	023		2022
Weight average remaining lease terms (years)	\$	4	\$	1
Weighted average discount rates		10.48%		4.50%

### 9. FRANCHISEE NOTES RECEIVABLES

Sylvan provides loans to certain franchisees for working capital purposes and territory expansions. In prior years, NLC sold corporate owned centers and recorded financing for these sales with original terms ranging from 3 to 60 months, bearing 0% interest at rates as of December 30, 2023 and December 31, 2022.

Notes receivable, which is reported in trade receivables and other assets on the consolidated balance sheets, consists of the following at December 30, 2023 and December 31, 2022:

	Successor December 30,		Predecessor (1) December 31,	
	2	2023		2022
Sales of centers	\$	-	\$	133
Working capital		591		132
New territory loan		-		17
Unamortized discount and allowance for credit losses		(591)		(282)
		-		_
Less: current portion of notes receivable, net		-		
Total long-term notes receivable	\$	_	\$	

(In thousands of dollars)

### 10. EMPLOYEE RETENTION CREDIT

Under the provisions of the Coronavirus Aid, Relive, and Economic Security Act, the Company determined it was eligible for a refundable employee retention credit (ERC) subject to certain criteria. During the period ended December 30, 2023, the Company filed for ERCs totaling \$1,669 recognized as a reduction to labor and benefits expenses on the accompanying consolidated statements of operations and member's equity (deficit) during the period from August 21, 2023 through December 30, 2023. As of December 30, 2023, outstanding ERCs which have been filed, but not received totaled \$1,669 and are recorded in other receivables on the accompanying consolidated balance sheets.

### 11. COMMITMENTS AND CONTINGENCIES

The Company is subject to legal actions arising in the ordinary course of its business. The Company has adequate legal defenses, insurance coverage, and/or indemnifications from former subsidiaries with respect to such actions and does not believe any settlement would materially affect the Company's financial position.

The Company entered into a minimum purchase commitment of \$1,300 with a vendor who delivers educational content to certain franchisees. The amount of time to reach the minimum amount of purchases ended on December 31, 2020. The Company did not reach the minimum purchase amount and accrued \$459 for the minimum payment due as of December 31, 2020. As of December 31, 2022 \$119 was still outstanding and included in accounts payable and accrued expenses on the consolidated balance sheets. During the year ended December 30, 2023, the remaining minimum was fulfilled and no accrual balance is outstanding as of December 30, 2023.

### 12. EMPLOYEE BENEFIT PLAN

Sylvan sponsors defined contribution retirement plans under Section 401(k) of the Internal Revenue Code. The provisions of these plans allow for voluntary employee contributions of up to 20% of salary, subject to certain annual limitations. All employees are eligible after meeting certain minimum service requirements.

The Company may, at its discretion, make matching contributions, which are allocated to eligible participants. The Company did not make any discretionary matching contributions during the year ended December 30, 2023 and December 31, 2022. The Company made discretionary contributions of \$16, and \$62 during the period from September 27, 2021 through December 25, 2021 and the period from January 1, 2021 through September 26, 2021, respectively.

(In thousands of dollars)

### 13. RELATED PARTY TRANSACTIONS

Related party charges (reimbursements) are recorded in the consolidated statements of operations and member's equity (deficit) as follows:

	Succ	essor	Prede	cessor (1)	Predece	essor (1)		cessor (1)	Prede	cessor (2)				
							For tl	he period						
	For the	e period	For tl	ne period			f	rom	For t	he period				
	from A	ugust 21,	from J	anuary 1,	For the year		Septe	mber 27,	from.	January 1,				
	2023 t	hrough	2023	through	en	ded	2021	through	2021	through				
	Decem	iber 30,	Aug	gust 20,	Decen	nber 31,	Dece	mber 25,	Septe	mber 26,				
	20	)23	2	2023	2022		2022		2022		2	2021		2021
Interest expense, net	\$	-	\$	-	\$	-	\$	-	\$	2,687				
Labor and benefits		14		(248)		-		-		-				
Other instructional and operating expenses		75		56		28		1,792		-				
Related party expenses (reimbursements)	\$	89	\$	(192)	\$	28	\$	1,792	\$	2,687				

As of December 30, 2023 and December 31, 2022, Sylvan owed the Franchise Group \$438 and \$483, respectively. Amounts owed consist of allocations, shared services, acquisition costs and management fees.

Prior to the acquisition by the Franchise Group, the Company's debt was held by its equity owners. Amounts paid related to the interest expense correlated with this debt and are outlined in the chart above. See Note 7 for information on the debt.

### 14. SUBSEQUENT EVENTS

Sylvan has evaluated subsequent events that have occurred for recognition or disclosure through March 8, 2024 the date the financial statements were available to be issued and determined that there were no events requiring recognition or disclosure, except as disclosed below.

On February 15, 2024, Unleashed Brands made a strategic investment through the acquisition of Sylvan Learning under the terms of a private transaction. Unleashed Brands owns a portfolio of franchised companies that provide a safe, fun, and enriching environment to help kids learn, play and grow. Acquisition fees of \$123 and \$33 were incurred and recorded in other instructional and operating expenses on the accompanying consolidated statements of operations and member's equity (deficit) for the period from August 21, 2023 through December 30, 2023 and the period from January 1, 2023 through August 20, 2023, respectively. The Company is currently evaluating the impact of the acquisition on its financial position.



March 8, 2024

To whom it may concern:

SC&H Attest Services, P.C. consents to the use in the Franchise Disclosure Document issued by Sylvan Learning, LLC ("Franchisor") on March 8, 2024, as it may be amended, of its report dated March 8, 2024, relating to the consolidated financial statements of Sylvan Learning, LLC as of December 31 of each of 2023, 2022, and 2021.

Sincerely,

Angela Lynch

Sparks, MD

for SC&H Attest Services, P.C.

Argel Lynch

### EXHIBIT H TO THE FRANCHISE DISCLOSURE DOCUMENT

### FRANCHISES IN OPERATION, FRANCHISEES WHO HAVE SIGNED BUT NOT YET OPENED, FRANCHISEES WHO HAVE LEFT THE SYSTEM, AND AFFILIATE OWNED LOCATIONS

### Franchisees in Operation as of February 15, 2024

State	Center Number / Address	Contact Details	Licensee	Franchisee Name(s)
Alabama	Center #2623 3659 Lorna Road Lorna Brook Village, Suite 137 Birmingham, Alabama 35216	Phone: (205) 987-9802 Fax: (205) 989-3588 Email: birmingham2623.al@sylvanlearning.com	MKZ Management Group LLC	Ira Sullivan
Alabama	Center #2603 4240 Lomac Street Montgomery, Alabama 36106	Phone: (334) 262-0043 Fax: (334) 262-7060 Email: cd@montgomerysylvan.com	Ateam, Inc.	Frank Stewart
Alabama	Center #2631 2046 Fairview Avenue Prattville, Alabama 36066	Phone: (334) 212-8448 Email: Prattville.AL@sylvanlearning.com	Extending Palms LLC	Laquayna Henley Robert Henley
Alabama	Center #2624 1110 N. Chalkville Rd # 144 Trussville, Alabama 35173	Phone: (205) 508-5474 Fax: (205) 833-9577 Email: Birmingham2624.al@sylvanlearning.com	MKZ Management Group LLC	Ira Sullivan
Alabama	Center #2629 3421 S. Shades Crest Rd., Ste. 109 West Hoover, Alabama 35244	Phone: (205) 730-5330 Email: Info@westhooversylvan.com	Euell Ventures LLC	William Harrison
Alabama	Center #2633 617 S. Main Street Wetumpka, Alabama 36092	Phone: (334) 478-3047 Email: wetumpka.al@sylvanlearning.com	Extending Palms LLC	Laquayna Henley Robert Henley
Alaska	Center #2700 2217 E. Tudor Road Suite 1A Anchorage, Alaska 99507	Phone: (907) 561-2229 Fax: (907) 646-9206 Email: anchorage@sylvanalaska.com	Learning Systems, LLC	Jennifer Sierra Donald Petersen David Cottrell

Alaska	Center #2701 281 N Main St. Wasilla, Alaska 99654	Phone: (907) 357-7323 Fax: (907) 357-7322 Email: wasilla@sylvanalaska.com	Learning Systems,	Jennifer Sierra Donald Petersen David Cottrell
Arizona	Center #410 2160 W. Chandler Blvd. #16 Chandler, Arizona 85224	Phone: (480) 897-3222 Fax: (480) 897-3230 Email: chandler@sylvanaz.com	Wootton Educators, Inc.	Rebecca Wright
Arizona	Center #428 2335 S. Lindsay Road, Suite 114 Gilbert, Arizona 85295	Phone: (480) 963-5900 Fax: (480) 963-5915 Email: sylvan@sylvan-az.com	Vertex Education Services, LLC	David Satter Robert Satter Jaclyn Satter
Arizona	Center #418 8190 W. Union Hills Drive Suite 120 Glendale, Arizona 85308	Phone: (623) 537-3611 Fax: (623) 572-7089 Email: sylvan0418@sylvanarrowhead.com	West East Educate, LLC	Rodney Sherrard Victoria Sherrard
Arizona	Center #417 13770 West Van Buren Street Suite 100 Goodyear, Arizona 85338	Phone: (623) 932-5714 Fax: (623) 932-3336 Email: sylvan417@aol.com	Success Makers, Inc.	Claudia Honsberger Joe Honsberger
Arizona	Center #434 9110 N Silverbell Road, Suite 170 Marana, Arizona 85743	Phone: (520) 344-9328 Email: Tucson.az@sylvanlearning.com	AJ Squared, LLC	Catherine Oberholzer
Arizona	Center #413 16838 N. 7th Street Suite 6 Phoenix, Arizona 85022	Phone: (602) 548-2600 Fax: (602) 548-8360 Email: Sylvan0413@sylvanphoenix.com	SLC Educators AZ, Inc.	Steven Jones Tamara Jones
Arizona	Center #401 6969 East Shea Blvd. Suite 185 Scottsdale, Arizona 85254	Phone: (602) 953-3070 Fax: (602) 922-9390 Email: sylvan401@scottsdalesylvan.com	SLC Educators AZ, Inc.	Steven Jones Tamara Jones
Arizona	Center #435 4500 E Speedway Blvd, Suite 15 Tucson, Arizona 85712	Phone: (520) 333-5445 Email: CentralTucson.AZ@sylvanlearning.com	Saguaro Learning, LLC	Theodore Parker Lori Parker George Parker

Arkansas	Center #2801 2409 South 56th Street Suite 117 Fort Smith, Arkansas 72903	Phone: (479) 484-6799 Fax: (479) 484-0702 Email: sylvan@sylvanfs.com	Midwest Education Consultants, Inc.	Courtney Wright Kevin Wright
Arkansas	Center #2810 4408 West Walnut Suite 3 Rogers, Arkansas 72756	Phone: (479) 903-7210 Fax: (479) 903-7211 Email: info@sylvannwa.com	Wright Education Consultants, Inc.	Courtney Wright Kevin Wright
California	Center #525 2720 Calloway Drive Suite A Bakersfield, California 93312	Phone: (661) 387-6060 Email: delia@bakersfieldsylvan.com	Aspire Higher Educational Services	Delia Navarro
California	Center #509 160 N. El Camino Real Encinitas, California 92024	Phone: (760) 942-5454 Fax: (760) 942-1871 Email: tutor@sylvanlearningsd.com	PQ Academia Corporation	Brandon Khieu
California	Center #531 7365 Carnelian Street #133 Rancho Cucamonga, California 91730	Phone: (909) 944-9763 Fax: (909) 466-5135 Email: carlid123@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #501 11398 Kenyon Way Ste A Rancho Cucamonga, California 91701	Phone: (909) 941-6863 Fax: (909) 941-8277 Email: sylvan501@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #520 1465 North Davis Road Salinas, California 93907	Phone: (831) 449-7551 Fax: (831) 449-7640 Email: salinas.sylvan@kiwischolarinc.com	Kiwi Scholar, Inc.	Cary Swensen
California	Center #515 1421 Guerneville Rd., Ste. 112 Santa Rosa, California 95403	Phone: (707) 528-6000 Fax: (707) 595-5072 Email: slc0515@sonic.net	PGLR, Inc.	Laura Rogers Jim Rogers
California	Center #504 1252 Madera Rd, Suite A1 Simi Valley California, USA, 93065	Phone: (805) 413-4453 Email: SimiValley.CA@sylvanlearning.com	Pivotal Learning Centers, Inc.	Venetya Evans

California	Center #516 23326 Hawthorne Blvd. Suite 170 Torrance, California 90505	Phone: (310) 373-0073 Fax: (310) 373-0805 Email: torrance@sylvansocal.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #533 1241 Grand Ave, Ste F Diamond Bar, California 91765	Phone: (909) 861-1146 Fax: (909) 861-5359 Email: carlid123@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #534 12887 Mountain Ave. Chino, California 91710	Phone: (909) 548-3569 Fax: (909) 548-0920 Email: carlid123@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #535 42311 10th Street West Lancaster, California 93534	Phone: (661) 723-6400 Fax: (661) 723-6422 Email: sylvan535@gmail.com	DKM Learning LLC	Nate Tapia Carrie Tapia Nosrat Maghsoudi
California	Center #546 971 W Foothill Boulevard Claremont, California 91711	Phone: (909) 621-2483 Fax: (909) 624-6045 Email: carlid123@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #559 3100 Mowry Ave. Suite 103 Fremont, California 94538	Phone: (510) 745-7800 Fax: (510) 745-7898 Email: sylvanld@pacbell.net	S5 Global, Inc.	Venkat Bussa
California	Center #562 2493 Solano Avenue Napa, California 94558	Phone: (707) 258-9000 Fax: (707) 258-9001 Email: learning@sylvannapa.com	Accelerated Learning Academy LLC	Zhenus Wahidi
California	Center #565 1730 W. Cameron Avenue, Suite 125 West Covina, California 91790	Phone: (626) 338-2800 Fax: (626) 338-2828 Email: westcovina.ca@sylvanlearning.com	33 Services & Consulting Inc.	Jessica Agopian
California	Center #569 5089 Business Center Dr, Ste 102 Fairfield, California 94534	Phone: (707) 864-1800 Fax: (707) 864-1889 Email: learning@sylvanfairfield.com	Bay Area Education Support System Inc.	Zhenus Wahidi
California	Center #572 1162 South Main St Manteca, California 95337	Phone: (209) 823-9252 Fax: (209) 599-9663 Email: enroll@sylvanmanteca.com	Illuminate Edge, Inc.	Sachin Ambulkar Swati Ambulkar

California	Center #578 39400 Murrieta Hot Springs Road, Suite 114-A Murrieta, California 92562	Phone: (951) 506-0300 Fax: (951) 506-0337 Email: Temecula.CA@SylvanLearning.com	Blue Horizon Learning Center LLC	Chastity Cordova- Heyward Spencer Heyward
California	Center #580 Westgate Mall 4701 Hamilton Avenue San Jose, California 95130	Phone: (408) 257-2740 Fax: (408) 257-2744 Email: sylvansanj@sbcglobal.net	GKS Education LLC	Abhishek Srivastava
California	Center #583 3288-B South Mooney Blvd. Visalia, California 93277	Phone: (559) 636-1502 Fax: (559) 636-1532 Email: slc583@sbcglobal.net	DND Learning, Inc.	Elizabeth Dyer Robert Dyer Richard Navarro
California	Center #585 3200 Castro Valley Blvd Castro Valley, California 94546	Phone: (510) 856-9000 Fax: (510) 856-9005 Email: learning@sylvancastrovalley.com	Bay Area Education Support System Inc.	Bill Kramer Jack McAboy
California	Center #587 1345 Grand Ave. Piedmont, California 94610	Phone: (510) 428-4000 Fax: (510) 428-4030 Email: learning@sylvanpiedmont.com	Bay Area Education Support System Inc.	Bill Kramer Jack McAboy
California	Center #588 10810 San Pablo Avenue El Cerrito, California 94530	Phone: (510) 559-1400 Fax: (510) 559-1404 Email: learning@sylvanelcerrito.com	Bay Area Education Support System Inc.	Bill Kramer Jack McAboy
California	Center #589 2031 C. Harbison Drive Vacaville, California 95687	Phone: (707) 448-7000 Fax: (707) 455-3463 Email: learning@sylvanvacaville.com	Accelerated Learning Academy LLC	Zhenus Wahidi
California	Center #590 2970 Huntington Drive San Marino, California 91108	Phone: (626) 544-5109 Fax: (626) 285-9614 Email: sanmarino.ca@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #592 1180 El Camino Ave, #100 Corona, California 92879	Phone: (951) 272-0727 Fax: (951) 272-6824 Email: sylvan592@sbcglobal.net	JKG Learning LLC	Kulvinder Griffin Jason Griffin

California	Center #596 1045 W Redondo Beach Blvd #138, Gardena, CA 90247 Gardena, California 90247	Phone: (310) 329-1866 Fax: (310) 329-3544 Email: gardena.ca@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #8504 2050 N. Broadway Walnut Creek, California 94596	Phone: (925) 934-3000 Fax: (925) 934-3620 Email: learning@sylvanwalnutcreek.com	Accelerated Learning Academy LLC	Zhenus Wahidi
California	Center #8508 1532 Standiford Ave, Suite B Modesto, California 95350	Phone: (209) 491-5406 Fax: (209) 491-5409 Email: modestodirector@sylvancenter.com	D and D Learning Solutions LLC	Lynn Dusa James Donnelly
California	Center #8509 13553 Ventura Boulevard Sherman Oaks, California 91423	Phone: (818) 906-1505 Fax: (818) 906-1509 Email: sylvanshermanoaks@sylvanhms.com	Dream Island Learning, LLC	Zhuo Chen
California	Center #8511 2025 West El Camino Real Mountain View, California 94040	Phone: (650) 938-0484 Fax: (650) 694-7476 Email: Vincent@sylvanmv.com	True Vision Academy, Inc.	Yule Zhang Yingjiu Pan
California	Center #8514 1315 S. Grand Ave #130 Glendora, California 91740	Phone: (626) 914-7700 Fax: (626) 914-3834 Email: carlid123@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #8516 5415 Camden Avenue San Jose, California 95124	Phone: (408) 448-8100 Fax: (408) 448-4736 Email: sylvancamden@sbcglobal.net	GKS Education LLC	Abhishek Srivastava
California	Center #8520 600 Hampshire Road Suite 120 Westlake Village, California 91361	Phone: (805) 497-3216 Fax: (805) 497-0586 Email: sylvan.westlakevillage@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree

California	Center #8529 27762 Vista Del Lago Suite A-10 Mission Viejo, California 92691	Phone: (949) 462-7160 Fax: (480) 247-5185 Email: sylvan.missionviejo@sylvanlearning.com	JKG Learning LLC	Kulvinder Griffin Jason Griffin
California	Center #8532 4330 Barranca Parkway Stonecreek Plaza, Suite 205 Irvine, California 92604	Phone: (949) 552-3025 Fax: (480) 247-5185 Email: irvine.ca@sylvanlearning.com	LUD, Inc.	Joe Liu
California	Center #8542 8366 Parkway Drive La Mesa, California 91942	Phone: (619) 668-0145 Fax: (619) 668-4937 Email: sylvanlamesa@gmail.com	Hilbert LLC	Katherine Hilbert Jordan Hilbert Christopher Purcell Ruth Purcell
California	Center #8555 379 West Portal Avenue San Francisco, California 94127	Phone: (415) 681-7323 Fax: (415) 681-2404 Email: sylvan@sylvanwestportal.com	HL Learning Solutions LLC	Lori Livingston Donald Livingston
California	Center #8562 12441 Hesperia Rd Suite A1 Victorville, California 92395	Phone: (760) 245-8091 Fax: (760) 843-1821 Email: carlid123@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #8563 1505 W. Florida Ave, Ste A Hemet, California 92543	Phone: (951) 765-1171 Fax: (909) 765-0374 Email: sylvan.hemet@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #8564 7000 Indiana Ave #122 Riverside, California 92506	Phone: (951) 353-8600 Fax: (951) 353-1424 Email: sylvan.riverside@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #8565 101 E Redlands Blvd #140 Redlands, California 92373	Phone: (909) 792-9669 Fax: (909) 793-5435 Email: sylvan.redlands@aol.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #8568 8471 Beverly Boulevard, Suite 201 Los Angeles, California 90048	Phone: (310) 843-0111 Fax: (310) 843-0248 Email: sylvanbeverlyhills@sylvanhms.com	Soaring High LLC	Thomas Jung

California	Center #8569 8984 Tampa Avenue Northridge, California 91324	Phone: (818) 885-0283 Fax: (818) 885-0242 Email: northridge@sylvansocal.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #8570 5731 W Slauson Ave. Ste 150 Culver City, California 90230	Phone: (310) 256-3830 Fax: (310) 337-3755 Email: sylvanculvercity@sylvanhms.com	Soaring High LLC	Thomas Jung
California	Center #8571 16052 Beach Boulevard Suite 105 Huntington Beach, California 92647	Phone: (714) 848-6072 Fax: (714) 843-0543 Email: sylvanhuntingtonbeach@sylvanhms.com	G&L Investments Inc.	Linda Trinh Pham Ha Nhu Thanh Le
California	Center #8573 5510 Britton Drive Long Beach, California 90815	Phone: (562) 429-8320 Fax: (562) 420-1185 Email: sylvanlongbeach@sylvanhms.com	Temple Corp	Kiran Patel
California	Center #8575 18840 Ventura Blvd, Suite 219 Tarzana, California 91356	Phone: (818) 742-9000 Email: tarzana.ca@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #8582 11982 Los Osos Valley Rd. San Luis Obispo, California 93405	Phone: (805) 540-9986 Email: matt.shea@sylvanlearning.com	Central Coast Scholars, LLC	Matthew Shea
California	Center #8584 1137 Aviation Blvd. Hermosa Beach, California 90254	Phone: (310) 341-3137 Email: hermosasylvan@gmail.com	Open Minds, LLC	Amber Castillo
California	Center #8585 527 S Lake Avenue, Suite 160 Pasadena, California 91101	Phone: (626) 689-4214 Email: pasadena.ca@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #8587 19613 Yorba Linda Blvd. Yorba Linda, California 92886	Phone: (714) 340-0218 Email: yorbalinda.ca@sylvanlearning.com	SANPACT, LLC	Sandeep Malik

California	Center #8588 8832 Imperial Hwy Downey, California 90242	Phone: (562) 622-9210 Fax: (562) 622-9022 Email: adriene@sylvandowney.com	Student Advance Tutoring & Testing LLC	Adriene White
California	Center #8589 1910 Lincoln Blvd. Santa Monica, California 90405	Phone: (424) 395-9555 Email: SantaMonica.CA@sylvanlearning.com	EMM Investments LLC	Michael Guirguis Emad Guirguis Mina Guirguis
California	Center #8590 7273 Citrus Ave. #300 Fontana, California 92336	Phone: (909) 941-6863 Email: Fontana.CA@sylvanlearning.com	D.D. & S. Learning Systems, Inc.	Carli Ganet James Dyer Frances Dyer
California	Center #8591 741 Glenvia Street, Suite 201 Glendale, California 91206	Phone: (626) 423-6323 Email: Glendale.CA@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #8593 4653 Carmel Mountain Rd, Suite 304 San Diego, California 92130	Phone: (858) 413-9090 Email: DelMar.CA@sylvanlearning.com	Denham Enterprises, LLC	Chad Denham
California	Center #8594 601 Glenoaks Blvd, Suite 306 Burbank, California 91502	Phone: (818) 697-0114 Email: glendale.ca@sylvanlearning.com	The Southern California Learning Corporation	Todd Crabtree
California	Center #8595 11740 San Vicente Blvd. Suite 106 Los Angeles, California 90049	Phone: (424) 256-2640 Email: BrentwoodLA.CA@sylvanlearning.com	Coker Family Education Solutions Inc.	Bianca Coker Jerry Coker
California	Center #8596 720-C S Allied Way El Segundo, California 90245	Phone: (310) 730-8083 Email: ElSegundo.CA@sylvanlearning.com	Open Minds, LLC	Amber Castillo
Colorado	Center #1211 10143 West Chatfield Ave, Unit 13 Littleton, Colorado 80127	Phone: (303) 948-1870 Fax: (303) 948-1871 Email: littleton@sylvancolorado.com	mspedu, llc	Mishon Paul

Colorado	Center #1212 9552 Park Meadows Drive, Suite 300 Lone Tree, Colorado 80124	Phone: (303) 220-8607 Fax: (303) 770-8486 Email: learn@sylvancolorado.com	mspedu, llc	Mishon Paul
Colorado	Center #1221 12001-122 Tejon St Westminster, Colorado 80234	Phone: (303) 920-7424 Fax: (303) 920-7452 Email: broomfield@sylvancolorado.com	AJEXA Corporation	Amber Peoples Allen Peoples
Connecticut	Center #2408 220 Main Street South, Suite 201 Southbury, Connecticut 06488	Phone: (203) 262-6464 Fax: (203) 262-1481 Email: sby.teach@gmail.com	Bartoli Educational Solutions, LLC	Peter Bartoli
Connecticut	Center #2418 95 Fenn Road Newington, Connecticut 06111	Phone: (860) 231-2429 Fax: (860) 231-2431 Email: central.ct@sylvanlearning.com	Bartoli Educational Solutions, LLC	Peter Bartoli
Connecticut	Center #2426 243 Federal Road Brookfield, Connecticut 06804	Phone: (203) 775-9611 Email: brookfield.ct@sylvanlearning.com	AMG Ventures, LLC	Arlene Garcia
Delaware	Center #2900 258 Chapman Road, Suite 101 Newark, Delaware 19702	Phone: (302) 722-6445 Email: Newark.DE@sylvanlearning.com	Enwisen Consulting Services, LLC	Meenakshi Tayal
Delaware	Center #2901 260 Dove Run Center Drive Middletown, Delaware 19709	Phone: (302) 722-8455 Fax: (302) 722-8455 Email: Middletown.DE@sylvanlearning.com	Enwisen Consulting Services, LLC	Meenakshi Tayal
Delaware	Center #2902 19409 Plantations Road, Suite 2 Rehoboth Beach, Delaware 19971	Phone: (302) 841-9637 Email: Lewes.DE@sylvanlearning.com	Blanshine Party of 5 L.L.C.	Jason Blanshine

	1		I	
Florida	Center #602 1414 Kingsley Ave, Ste 4 Orange Park, Florida 32073	Phone: (904) 269-2000 Fax: (904) 269-2960 Email: hilary.pharies@firstcoastsylvan.com	Quizhark LLC	Rafael Urquiza Stacey Harkins- Urquiza
Florida	Center #604 3802 Ehrlich Road Suite 204 Tampa, Florida 33624	Phone: (813) 968-4497 Fax: (813) 968-8526 Email: sylvan_cwd@tampabay.rr.com	Intellex Enterprises, Inc.	Judy Johnson
Florida	Center #605 3231 SE Maricamp Road Ocala, Florida 34471	Phone: (352) 622-6284 Fax: (352) 624-2313 Email: sethpatricksylvan@gmail.com	GT Learning Group, LLC	Gilley Spaude Tod Spaude
Florida	Center #607 3730 Cleveland Heights Blvd. Suite 3 Lakeland, Florida 33813	Phone: (863) 647-1922 Fax: (863) 648-5427 Email: sylvandirectorfl@gmail.com	Achieve Learning Goals LLC	Deena Porter Andrew Porter
Florida	Center #609 2323 Curlew Road Suite 6C Dunedin, Florida 34698	Phone: (727) 787-7323 Fax: (727) 786-1407 Email: director@sylvansp.com	EduStation LLC	Sultan Jan
Florida	Center #611 2401 Executive Plaza #8 Pensacola, Florida 32504	Phone: (850) 474-9022 Fax: (850) 474-9401 Email: pensacolasylvan@cox.net	Remarkably You in Education, LLC	Gloria Kane Sally Figueroa
Florida	Center #617 2811 Capital Circle, NE Suite 3 Tallahassee, Florida 32308	Phone: (850) 385-8696 Fax: (850) 385-8670 Email: dmredinger@aol.com	Florida Learning Centers, Inc.	Diane Redinger William Amt
Florida	Center #621 13720 Ben C Pratt Six Mile Cypress Pkwy, Suite 302 Fort Myers, Florida 33912	Phone: (239) 275-1130 Fax: (239) 275-3264 Email: caoilfhionn@sylvanswfl.com	Hiduke Learning Centers, Inc.	Terrence Hiduke Joanne Bell
Florida	Center #627 6153 Lake Worth Rd Greenacres, Florida 33463	Phone: (561) 432-0445 Fax: (561) 432-0446 Email: info@sylvan-lakeworth.com	Learn Time LLC	Michael Gray Sheri Gray

Florida	Center #629 4961 NW 8th Ave Suite A Gainesville, Florida 32605	Phone: (352) 371-6891 Fax: (352) 371-1848 Email: sethpatricksylvan@gmail.com	GT Learning Group, LLC	Tod Spaude Gilley Spaude
Florida	Center #633 3275 66th Street N. Ste # 5 St. Petersburg, Florida 33710	Phone: (727) 347-6284 Fax: (727) 347-1207 Email: Director@Sylvansp.com	EduStation LLC	Sultan Jan
Florida	Center #651 2701 Michigan Avenue, Suite D Kissimmee, Florida 34744	Phone: (407) 944-9661 Fax: (407) 944-9627 Email: kissimmeesylvan@gmail.com	KAK Management, LLC	Kent Kolbow
Florida	Center #660 2255 Dunn Ave, Ste 501 Jacksonville, Florida 32218	Phone: (904) 757-2220 Fax: (904) 757-2292 Email: meshell.harris@firstcoastsylvan.com	Quizhark LLC	Rafael Urquiza Stacey Harkins- Urquiza
Florida	Center #663 38420 5th Ave Zephyrhills, Florida 33542	Phone: (813) 782-8219 Fax: (813) 782-8471 Email: sylvanzhpr@gmail.com	Intellex Enterprises, Inc.	Judy Johnson
Florida	Center #685 1200 Oakley Seaver Dr, Ste. 201 Clermont, Florida 34711	Phone: (352) 241-4141 Fax: (352) 241-6131 Email: Clermont.FL@sylvanlearning.com	KAK Management, LLC	Kent Kolbow
Florida	Center #698 3712 N. Wickham Road Suite A Melbourne, Florida 32934	Phone: (321) 253-3463 Fax: (321) 255-9896 Email: melbournesylvan@gmail.com	Leitch Learning II, LLC	Jane Leitch
Florida	Center #8609 136 N. Fourth Street, Suite 1270 Lake Mary, Florida 32746	Phone: (407) 688-2017 Fax: (407) 710-3001 Email: william.superstar@hotmail.com	Galaxy Star Tutoring LLC	William Ramos Da Silva
Florida	Center #8611 3097 NE 163rd Street North Miami Beach, Florida 33160	Phone: (305) 936-0370 Fax: (305) 936-2441 Email: Aventura.FL@sylvanlearning.com	ENLIGHTEN EDUCATION SOLUTIONS, LLC.	Manjunath Yerragunta

Florida	Center #8612 7154 Beracasa Way Boca Raton, Florida 33433	Phone: (561) 391-0707 Fax: (561) 391-5459 Email: BocaRaton.FL@sylvanlearning.com	ENLIGHTEN EDUCATION SOLUTIONS, LLC.	Manjunath Yerragunta
Florida	Center #8613 12592 Pine Blvd., Suite 101 Pembroke Pines, Florida 33027	Phone: (954) 517-1802 Fax: (954) 517-1810 Email: PembrokePines.FL@sylvanlearning.com	ENLIGHTEN EDUCATION SOLUTIONS, LLC.	Manjunath Yerragunta
Florida	Center #8614 12588A SW 88th Street Miami, Florida 33186	Phone: (305) 273-9366 Fax: (305) 596-0763 Email: Kendall.FL@sylvanlearning.com	ENLIGHTEN EDUCATION SOLUTIONS, LLC.	Manjunath Yerragunta
Florida	Center #8615 Coral Palm Plaza 2141 N. University Drive Coral Springs, Florida 33071	Phone: (954) 575-7410 Fax: (954) 575-7411 Email: CoralSprings.FL@sylvanlearning.com	ENLIGHTEN EDUCATION SOLUTIONS, LLC.	Manjunath Yerragunta
Florida	Center #8620 11509 Palmbrush Trail, Suite 106 Lakewood Ranch, Florida 34202	Phone: (941) 920-0950 Email: LakewoodRanch.FL@sylvanlearning.com	CLM Education, LLC	Christina Marziale
Florida	Center #8622 5549 Palmer Crossing Circle, Unit 7 Sarasota, Florida 34233	Phone: (941) 260-5840 Email: Sarasota.FL@sylvanlearning.com	CLM Education, LLC	Christina Marziale
Florida	Center #8623 8945 West Colonial Drive Ocoee, Florida 34761	Phone: (407) 984-4269 Email: Ocoee.FL@sylvanlearning.com	KAK Management, LLC	Kent Kolbow
Florida	Center #8625 6 11th Ave, H3 Shalimar, Florida 32579	Phone: (850) 613-2424 Email: emeraldcoast.fl@sylvanlearning.com	New Beginnings in Education LLC	Kativa Perkins Fred Perkins
Florida	Center #8626 405 Alexandria Blvd, Suite 100 Oviedo, Florida 32765	Phone: (407) 986-3175 Email: Oviedo.FL@SylvanLearning.com	Cultivating Shalom, LLC	Jason McCoskey Ana McCoskey

Florida	Center #8628 7606 West Sand Lake Road Orlando, Florida 32819	Phone: (407) 250-4030 Email: DrPhillips.FL@sylvanlearning.com	KAK Management, LLC	Kent Kolbow
Florida	Center #8629 7130 Big Bend Road, Suite 110 Gibsonton, Florida 33534	Phone: (813) 535-6426 Email: ApolloBeach.FL@sylvanlearning.com	SUNSHINE EDUCATION SOLUTIONS LLC	Vanity Barr-Little
Florida	Center #8630 1364 S Narcoossee Road Saint Cloud, Florida 34771	Phone: (407) 498-4009 Email: lakenona.fl@sylvanlearning.com	Stine Global LLC	Kathleen Eversole Blake Stine
Florida	Center #8632 2708 Santa Barbara Blvd. Unit 105 Cape Coral, Florida 33914	Phone: (239) 800-5003 Email: bradhansen@sylvanswfl.com	Hiduke Learning Centers, Inc.	Terrence Hiduke Joanne Bell
Florida	Center #8633 10011 Estero Town Commons Place, Unit 107 Estero, Florida 33928	Phone: (239) 319-5573 Email: caoilfhionn@sylvanswfl.com	Hiduke Learning Centers, Inc.	Terrence Hiduke Joanne Bell
Florida	Center #8634 22655 Bayshore Rd, Ste 140 Port Charlotte, Florida 33980	Phone: (941) 249-4983 Email: PortCharlotte.FL@sylvanlearning.com	Eminent Learning LLC	Catherine Frisch Dr. Mary Rowley
Florida	Center #8635 13444 Boyette Rd Riverview, Florida 33569	Phone: (813) 438-4346 Email: Riverview.FL@sylvanlearning.com	SUNSHINE EDUCATION SOLUTIONS LLC	Vanity Barr-Little
Georgia	Center #3006 5256 Armour Rd. Columbus, Georgia 31904	Phone: (706) 563-9864 Fax: (706) 243-4681 Email: Kimberly.voltz@sylvanlearning.com	KVDM, Inc.	Blake Voltz Kimberly Voltz
Georgia	Center #3007 340 Eisenhower Dr. Building 100 Savannah, Georgia 31406	Phone: (912) 355-2267 Fax: (912) 629-3095 Email: dinopapa@sylvan3007.com	Empire Coaching LLC	Dino Papadopoulos

Georgia	Center #3010 229 Furys Ferry Road, Suite 119 Augusta, Georgia 30907	Phone: (706) 868-1888 Fax: (706) 868-7117 Email: scsylvan@comcast.net	Ga Learning Ventures, Inc	Kelly Conway
Georgia	Center #3023 201 Turner McCall Blvd. Rome, Georgia 30165	Phone: (706) 295-3484 Fax: (706) 291-0742 Email: sylvanofrome@yahoo.com	Learned Success, Inc.	Angela Baron
Georgia	Center #3024 1709 River Street Valdosta, Georgia 31601	Phone: (229) 245-0381 Fax: (229) 245-0878 Email: dmredinger@aol.com	Florida Learning Centers, Inc.	Diane Redinger William Amt
Georgia	Center #3076 927 Joe Frank Harris Parkway Cartersville, Georgia 30120	Phone: (678) 721-6661 Email: sylvanofcartersville@yahoo.com	Learned Success, Inc.	Angela Baron
Georgia	Center #3079 149 Reinhardt College Pkwy, Suite 7 Canton, Georgia 30114	Phone: (470) 761-9128 Email: Canton.GA@sylvanlearning.com	Prestige Academics LLC	Sarah Whatley Sharon Stephens
Hawaii	Center #3102 4211 Waialae Avenue Kahala Mall, Suite 30 Honolulu, Hawaii 96816	Phone: (808) 735-8811 Fax: (808) 737-6960 Email: info@sylvankahala.org	Kahala Academic Center, Inc.	Sam Kim Pamela Lee
Hawaii	Center #3104 95-1249 Meheula Pkwy, Ste. 198 Mililani, Hawaii 96789	Phone: (808) 623-0808 Fax: (808) 627-9887 Email: info@sylvanmililani.com	Inspired Education, LLC	Tobin Yoder Kai Lee Awaya- Yoder
Idaho	Center #2018 30 S. Cole Road Boise, Idaho 83709	Phone: (208) 918-8150 Fax: (208) 918-8140 Email: sylvan.learning.boise@gmail.com	Daksh Academy LLC	Sudhaker Chhabra
Idaho	Center #2019 2845 E Overland Rd, Suite 170 Meridian, Idaho 83642	Phone: (208) 912-0010 Email: Sudhaker.chhabra@sylvanboise.com	Daksh Gurukul LLC	Sudhaker Chhabra

Illinois	Center #3203 211 Landmark, Building D Suite 5, 6, 7 Normal, Illinois 61761	Phone: (309) 662-8537 Fax: (309) 662-8537 Email: contact@sylvannormal.com	F&O Ventures Co.	Vanessa Ross Phillip Ross
Illinois	Center #3204 2735 South 6th Street Springfield, Illinois 62703	Phone: (217) 546-0009 Fax: (217) 546-0533 Email: Directors@SpringfieldSylvan.com	F&O Ventures Co.	Vanessa Ross Phillip Ross
Illinois	Center #3207 7501 N. University Suite 218 Peoria, Illinois 61614	Phone: (309) 683-3601 Fax: (309) 683-9055 Email: contact@sylvanpeoria.com	F&O Ventures Co.	Vanessa Ross Phillip Ross
Illinois	Center #3224 124 Springfield Court Suite 200 O'Fallon, Illinois 62269	Phone: (618) 628-0018 Fax: (618) 628-0164 Email: ofallon.il@sylvanlearning.com	J&N Education Services, LLC	Judy Cornell
Illinois	Center #3235 6A Glen Ed Professional Park Glen Carbon, Illinois 62034	Phone: (618) 656-0500 Fax: (618) 656-0555 Email: sabrinasylvan@sbcglobal.net	Grand Education, LLC	Leonard Rakers Misty Menossi
Illinois	Center #3237 24012 W. Main Street, Suite 104 Plainfield, Illinois 60544	Phone: (815) 609-7111 Fax: (815) 609-7333 Email: plainfield.il@sylvanlearning.com	Learning Solutions, Inc.	Daniel Callahan Margaret Callahan
Illinois	Center #3248 1292 Rickert Drive Suite 148 Naperville, Illinois 60540	Phone: (630) 778-7200 Fax: (630) 778-7222 Email: sylvan.naperville@comcast.net	Assured Educational Services, Inc.	Kevin Smith Marjorie Smith
Illinois	Center #3251 9501 W 171st Street Tinley Park, Illinois 60487	Phone: (708) 429-1072 Fax: (708) 429-1055 Email: tinleysylvan@sbcglobal.net	JAKD, LLC	Andrea Cronin James Cronin

Illinois	Center #3254 25 N. Roselle Rd. Schaumburg, Illinois 60194	Phone: (847) 301-2700 Fax: (847) 891-6440 Email: schaumburg.il@sylvanlearning.com	Ellkendan Education LLC	Kenneth Frysztak
Illinois	Center #3256 7217 W. Madison Street Forest Park, Illinois 60130	Phone: (708) 488-0161 Fax: (708) 771-1716 Email: forestpark.il@sylvanlearning.com	Tree of Knowledge, Inc.	Robert Keller
Illinois	Center #3258 745 W State Rte 22 Lake Zurich, Illinois 60047	Phone: (847) 382-7755 Fax: (847) 842-2929 Email: sylvan.lakezurich@sylvanlearning.com	Ellkendan Education LLC	Kenneth Frysztak
Illinois	Center #3259 495 N. Riverside Drive #113 Gurnee, Illinois 60031	Phone: (847) 625-2660 Fax: (847) 625-2663 Email: sylvan.gurnee@sylvanlearning.com	NIAA LLC	Chhaya Dave Sandip Chatterjee
Illinois	Center #3262 7136 N. Carpenter Road Skokie, Illinois 60077	Phone: (847) 329-5267 Fax: (773) 478-1236 Email: michellem@sylvanacademics.com	Bauer James Inc.	Elizabeth Samartzis
Illinois	Center #3263 10338 South Cicero Avenue Oaklawn, Illinois 60453	Phone: (708) 425-3114 Fax: (708) 425-0422 Email: dinad@sylvanacademics.com	Bauer James Inc.	Elizabeth Samartzis
Illinois	Center #3265 1609 S. Halsted Street Chicago, Illinois 60608	Phone: (312) 226-3306 Fax: (312) 226-3308 Email: educateuv@sylvank12.com	K12 Inc.	Nawaz Malik
Illinois	Center #3292 945 Lakeview #180 Vernon Hills, Illinois 60061	Phone: (224) 543-1120 Email: vernonhills.il@sylvanlearning.com	NIAA LLC	Chhaya Dave Sandip Chatterjee
Illinois	Center #3293 1107 Waukegan Road Glenview, Illinois 60025	Phone: (312) 208-3184 Email: Glenview.IL@sylvanlearning.com	Bauer James Inc.	Elizabeth Samartzis

Indiana	Center #1800 1400 S Guilford Road Suite 100 Carmel, Indiana 46032	Phone: (317) 257-6703 Fax: (317) 257-7582 Email: sylvan1800@comcast.net	KIK Management, LLC	Kent Kolbow
Indiana	Center #1801 862 South State Road 135, Suite A Greenwood, Indiana 46143	Phone: (317) 888-0438 Fax: (317) 888-0332 Email: sylvan1801@comcast.net	KIK Management, LLC	Kent Kolbow
Indiana	Center #1802 302 North Dan Jones Road Suite 115 Plainfield, Indiana 46168	Phone: (317) 247-7664 Fax: (317) 244-0650 Email: sylvan1802@comcast.net	KIK Management, LLC	Kent Kolbow
Indiana	Check #1805 10016 Auburn Park Dr Fort Wayne, Indiana 46825	Phone: (260) 485-1330 Fax: (260) 485-1886 Email: woblok@earthlink.net	KCK LLC	Kent Kolbow
Indiana	Center #1806 6227 Covington Road Fort Wayne, Indiana 46804	Phone: (260) 436-2710 Fax: (260) 436-2909 Email: sylvansouth@sylvanfw.com	KCK LLC	Kent Kolbow
Indiana	Center #1807 923 South Kenmore Drive Hebron Office Plaza Evansville, Indiana 47714	Phone: (812) 479-1150 Fax: (812) 479-0480 Email: SLC1807@aol.com	Charlotte Jolly and Janet Crocker	Charlotte Jolly Janet Crocker
Indiana	Center #1808 1001 Parkway Avenue, Suite 4 Elkhart, Indiana 46516	Phone: (574) 264-2223 Fax: (574) 262-1160 Email: Elkhart.in@sylvanlearning.com	KLC Management, Inc.	Kent Kolbow
Indiana	Center #1810 4050 Britt Farm Drive Suite A Lafayette, Indiana 47905	Phone: (765) 447-5996 Fax: (765) 449-7614 Email: anna@lafayettesylvan.com	Educare, Inc.	Carol Galbraith
Indiana	Center #1811 3005 Grape Road Suite A Mishawaka, Indiana 46545	Phone: (574) 254-1055 Fax: (574) 254-1063 Email: mishawaka.in@sylvanlearning.com	KLC Management, Inc.	Kent Kolbow
Indiana	Center #1812 958 W. US Highway 30 Schererville, Indiana 46375	Phone: (219) 322-5535 Fax: (219) 322-6229 Email: sylvan1812@sbcglobal.net	KSK Management LLC	Kent Kolbow

Indiana	Center #1814 3126 North National Road Columbus, Indiana 47201	Phone: (812) 372-2104 Fax: (812) 372-6079 Email: columbus@gosylvan.com	ExNil, LLP	Greg Moore Tamayo Fukumoto
Indiana	Center #1821 11740 Brooks School Rd Suite A Fishers, Indiana 46038	Phone: (317) 913-0963 Fax: (317) 913-0965 Email: fishers.in@sylvanlearning.com	DCSK Management, LLC	Kent Kolbow Sarah Miller
Indiana	Center #1832 2088 Liberty Drive Liberty Crossing Shoppes, Suite 109 Bloomington, Indiana 47403	Phone: (812) 330-0900 Fax: (812) 339-0955 Email: bloomington@gosylvan.com	ExNil, LLP	Greg Moore Tamayo Fukumoto
Indiana	Center #1833 735 Commerce Drive Franklin, Indiana 46131	Phone: (317) 346-6604 Fax: (317) 346-6634 Email: franklin@gosylvan.com	ExNil, LLP	Greg Moore Tamayo Fukumoto
Indiana	Center #1836 1070 W. 84th Drive Merrillville, Indiana 46410	Phone: (219) 736-1113 Fax: (219) 736-1118 Email: sylvan1836@yahoo.com	KSK Management LLC	Kent Kolbow
Indiana	Center #1837 2500 Calumet Avenue Valparaiso, Indiana 46383	Phone: (219) 707-5074 Fax: (219) 707-5693 Email: valparaiso.in@sylvanlearning.com	KSK Management LLC	Kent Kolbow
Iowa	Center #3307 4820 Mortensen Road, Suite 101 Ames, Iowa 50014	Phone: (515) 956-3366 Fax: (515) 956-3368 Email: ames@iowa-sylvan.com	FCH, LLC	April Samp Jeff Hohensee
Iowa	Center #3312 8707 N Park Court Johnston, Iowa 50131	Phone: (515) 867-8000 Fax: (515) 867-8800 Email: dsm@iowa-sylvan.com	FCH, LLC	April Samp Jeff Hohensee
Iowa	Center #3313 1854 Boyson Road Hiawatha, Iowa 52233	Phone: (319) 393-0000 Fax: (319) 393-5316 Email: cedarrapids@iowa-sylvan.com	FCH, LLC	April Samp Jeff Hohensee

Iowa	Center #3314 1100 5th Street, Suite 105	Phone: (319) 351-9343 Fax: (319) 341-3312	FCH, LLC	April Samp Jeff Hohensee
	Coralville, Iowa 52241	Email: coralville@iowa-sylvan.com		Jeff Hoffensee
	Center #3318 301 Center Place, Suite C-	Phone: (515) 304-5280	Jones Educational	Angela Jones
Iowa	1	Email: Altoona.IA@sylvanlearning.com	Services LLC	Nicholas Jones
	Altoona, Iowa 50009			
Kansas	Center #1500 22046 West 66th Street	Phone: (913) 543-6085 Fax: (913) 381-1322	Spearhead	Peter Anderberg
Kansas	Shawnee, Kansas 66226	Email: pamela.webb@sylvanks.com	Investments, LLC	Peter Anderberg III
	Center #1504	Phone: (785) 272-6284		
Kansas	2800 Wanamaker Suite 140	Fax: (785) 272-4450	Spearhead Investments, LLC	Peter Anderberg Peter Anderberg III
	Topeka, Kansas 66614	Email: Kirstin.palermo@sylvanks.com	investments, EEC	Teter Anderberg III
	Center #1505	Phone: (785) 842-6284		
Kansas	721 Wakarusa Drive, Suite 104	Fax: (785) 838-3825	Spearhead Investments, LLC	Peter Anderberg Peter Anderberg III
	Lawrence, Kansas 66049	Email: kirstin.palermo@sylvanks.com	mvestments, EEC	Teter Anderseig in
	Center #1506	Phone: (785) 587-8800		
Kansas	3208 Kimball Ave. Candlewood Center	Fax: (785) 587-8942	Spearhead Investments, LLC	Peter Anderberg Peter Anderberg III
	Manhattan, Kansas 66503	Email: Kirstin.palermo@sylvanks.com	investments, EEC	Teter Anderberg III
	Center #1509	Phone: (913) 681-1141		
Kansas	9165 W 133rd Street Overland Park, Kansas	Fax: (913) 681-1193	Spearhead Investments, LLC	Peter Anderberg Peter Anderberg III
	66213	Email: Joan.Anderberg@sylvanks.com	investments, LLC	1 Cur Anderverg III
	Center #1510	Phone: (316) 260-6811		
Kansas	8343 E. 32nd St. N. Ste. 125A	Fax: (613) 260-3121	A+ Tutoring of Witchita LLC	Hazel Darrow
	Wichita, Kansas 67226	Email: Wichita.ks@sylvanlearning.com	Witchia LLC	
	Center #1106	Phone: (270) 554-4111		
Kentucky	2856 Lone Oak Road	Fax: (270) 554-0079	Steve Bright	Steve Bright
	Paducah, Kentucky 42003	Email: sylvan@sylvanpaducah.com		

Kentucky	Center #1132 576 Centre View Blvd Crestview Hills, Kentucky 41017	Phone: (859) 344-5080 Fax: (859) 344-5085 Email: crestviewhills.ky@sylvanlearning.com	ITnet Consulting Group, LLC	Supraja Sanakari Madhu Sanakari
Kentucky	Center #1133 6413 Bardstown Road Louisville, Kentucky 40291	Phone: (502) 231-1717 Fax: (502) 231-1818 Email: ferncreek.ky@sylvanlearning.com	Coalescent Ventures, LLC	Jason Mucci
Louisiana	Center #1301 4449 N Interstate 10 Service Rd. W. Metairie, Louisiana 70006	Phone: (504) 885-5327 Fax: (504) 885-9127 Email: sylvan1301@gmail.com	Cusanza Enterprises II, LLC	Sal Cusanza
Louisiana	Center #1303 221 North College Road Lafayette, Louisiana 70506	Phone: (337) 232-7323 Fax: (337) 232-7675 Email: sylvanlafayette@sylvanacadiana.com	Academic Technology Corp., LLC	Troy Sharon Christy Sharon
Louisiana	Center #1304 7584 Bluebonnet Boulevard Suite D Baton Rouge, Louisiana 70810	Phone: (225) 291-7323 Fax: (225) 291-9502 Email: sylvan1304@gmail.com	Cusanza Enterprises IV, LLC	Sal Cusanza
Louisiana	Center #1305 1901 Manhattan Blvd, B- 102 Harvey, Louisiana 70058	Phone: (504) 324-7717 Email: sylvan1305@gmail.com	Cusanza Enterprises V, LLC	Sal Cusanza
Louisiana	Center #1307 1264 Canterbury Drive Alexandria, Louisiana 71303	Phone: (318) 443-0949 Fax: (318) 443-0091 Email: milesmaffett@yahoo.com	Miles Maffett Recordings LLC	Michael Maffett Margaret Maffett Stephen Maffett
Louisiana	Center #1312 58465 Tyler Street Suite 7 Slidell, Louisiana 70461	Phone: (985) 641-0343 Fax: (985) 641-0431 Email: sylvan1312@gmail.com	Cusanza Enterprises, LLC	Sal Cusanza

Louisiana	Center #1314 1000 Parkview Drive Suite 10 New Iberia, Louisiana 70560	Phone: (337) 369-1543 Fax: (337) 369-6076 Email: sylvannewiberia@sylvanacadiana.com	Academic Technology Corp., LLC	Troy Sharon Christy Sharon
Louisiana	Center #1320 657 Corporate Drive, Suite A Houma, Louisiana 70360	Phone: (985) 223-3798 Fax: (985) 223-3932 Email: annemarie@sylvanhouma.com	Anne Marie Naquin and John Cinnater	Anne Marie Naquin John Cinnater
Louisiana	Center #1322 2113 South Burnside (Hwy. 44) Gonzales, Louisiana 70737	Phone: (225) 644-7323 Fax: (225) 644-6202 Email: sylvan1322@gmail.com	Cusanza Enterprises, IV, LLC	Sal Cusanza
Maryland	Center #3545 23789 Garrett Highway, Suite 3 McHenry, Maryland 21541	Phone: (240) 442-5873 Email: Garrettcounty.md@sylvanlearning.com	Lakeside Learning, LLC	Katelyn O'Brien Suzan O'Brien
Massachusetts	Center #2538 85 Andover Street Danvers, Massachusetts 01923	Phone: (978) 208-2010 Fax: (978) 769-3088 Email: learn@northshoresylvan.com	Rosina Marie, Inc.	Rosina LeBlanc
Massachusetts	Center #2539 400 Tradecenter Dr, Suite 1910 Woburn, Massachusetts 01801	Phone: (781) 933-9480 Email: learn@burlingtonsylvan.com	Burlington Learning, Inc.	Rosina LeBlanc
Massachusetts	Center #2541 646 Middle Street Weymouth, Massachusetts 02189	Phone: (781) 247-5402 Fax: (781) 380-3879 Email: sylvan.learning@capstoneeducation.info	Unlocked Learning LLC	Eui Choi Eun Kyung Song
Massachusetts	Center #2542 44 Washington Street, Suite 105A Brookline, Massachusetts 02445	Phone: (617) 264-4150 Fax: (617) 264-4154 Email: sylvan.learning@capstoneeducation.info	Unlocked Learning LLC	Eui Choi Eun Kyung Song

Michigan	Center #2300 1601 Briarwood Circle Suite 300 Ann Arbor, Michigan 48108	Phone: (734) 665-7323 Fax: (734) 665-7333 Email: annarbor.mi@sylvanlearning.com	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2301 3910 Burton Street S.E. Suite 102 Grand Rapids, Michigan 49546	Phone: (616) 957-4617 Fax: (616) 957-1728 Email: GrandRapids.MI@sylvanlearning.com	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2303 1350 West Centre Avenue Suite 120 Portage, Michigan 49024	Phone: (269) 327-5237 Fax: (269) 327-8742 Email: Portage.MI@sylvanlearning.com	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2304 4330 44th Street Suite 101 Grandville, Michigan 49418	Phone: (616) 530-8488 Fax: (616) 530-4910 Email: Grandville.MI@sylvanlearning.com	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2305 3290 West Big Beaver Suite 160 Troy, Michigan 48084	Phone: (248) 822-7188 Fax: (248) 822-7191 Email: sylvanoftroy@gmail.com	K. J. & J. Ltd.	Linda Hoagland
Michigan	Center #2306 2500 Kerry Street Suite 100 Lansing, Michigan 48912	Phone: (517) 372-7410 Fax: (517) 372-9925 Email: sylvanlansing@sbcglobal.net	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2307 5829 West Maple, Suite 127 West Bloomfield, Michigan 48322	Phone: (248) 737-2880 Fax: (248) 737-6098 Email: info@sylvanwb.com	Guardian Angel Learning Centers, LLC	Eddie Bahoura Nidal Yasso
Michigan	Center #2308 22034 Novi Road Oakpoint Plaza Novi, Michigan 48375	Phone: (248) 344-1474 Fax: (248) 344-9802 Email: novi.mi@sylvanlearning.com	COT, LLC	Jessica Rollins George Rollins

Michigan	Center #2309 3343 Spring Arbor Road Suite 100 Jackson, Michigan 49203	Phone: (517) 787-9180 Fax: (517) 787-9983 Email: sylvanjackson@sbcglobal.net	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2312 37727 Professional Center Dr, Ste 130 Livonia, Michigan 48154	Phone: (734) 462-1132 Fax: (734) 462-2825 Email: livonia.mi@sylvanlearning.com	Guardian Angel Learning Centers, LLC	Eddie Bahoura Nidal Yasso
Michigan	Center #2313 5890 S. Harvey Muskegon, Michigan 49441	Phone: (231) 799-0613 Fax: (231) 799-0619 Email: Muskegon.MI@sylvanlearning.com	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2314 11080 Hall Road Suite B Sterling Heights, Michigan 48314	Phone: (586) 739-0270 Fax: (586) 739-2436 Email: sylvanofmacomb@gmail.com	K. J. & J. Ltd.	Linda Hoagland
Michigan	Center #2315 10407 East Grand River Suite 800 Brighton, Michigan 48116	Phone: (810) 227-1800 Fax: (810) 227-3111 Email: sylvanbrighton@sbcglobal.net	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2317 3913 Bay Road Saginaw, Michigan 48603	Phone: (989) 791-0088 Fax: (989) 791-9183 Email: saginaw.mi@sylvanlearning.com	Awaya-Yoder Edvantage Tutoring, Inc.	Tobin Yoder Kai Lee Awaya- Yoder
Michigan	Center #2319 8185 Holly Road, Suite 3 Grand Blanc, Michigan 48439	Phone: (810) 233-4600 Fax: (810) 233-7698 Email: sylvanflint@gmail.com	COT, LLC	Jessica Rollins George Rollins
Michigan	Center #2322 2040 Monroe Street Dearborn, Michigan 48124	Phone: (313) 724-1500 Fax: (313) 724-8317 Email: dearborn.mi@sylvanlearning.com	Guardian Angel Learning Centers, LLC	Eddie Bahoura Nidal Yasso
Michigan	Center #2334 7752 Canton Center Road Canton, Michigan 48187	Phone: (734) 451-3461 Email: canton.mi@sylvanlearning.com	Yasso Guardian Angel Learning Centers, LLC	Eddie Bahoura Nidal Yasso

Michigan	Center #2336 108 A West Wackerly Midland, Michigan 48640	Phone: (989) 633-9515 Fax: (989) 633-9517 Email: midland.mi@sylvanlearning.com	Edvantage Tutoring, Inc.	Tobin Yoder Kai Lee Awaya- Yoder
Michigan	Center #2349 6840 Northway Dr. NE Rockford, Michigan 49341	Phone: (616) 863-8835 Fax: (616) 863-8894 Email: Rockford.MI@sylvanlearning.com	Rusthoven Management Company	Jessica Rollins George Rollins
Minnesota	Center #801 10580 Wayzata Blvd, Suite 250 Minnetonka, Minnesota 55305	Phone: (952) 544-6253 Fax: (952) 544-0536 Email: info@sylvanmtka.com	Briannca, Inc.	Gail Bregi
Minnesota	Center #817 172 Pioneer Trail Chaska, Minnesota 55318	Phone: (952) 466-3838 Fax: (952) 466-3833 Email: info@sylvanchaska.com	Briannca, Inc.	Gail Bregi
Minnesota	Center #832 15035 Crestone Avenue W Rosemount, Minnesota 55068	Phone: (952) 236-9090 Fax: (651) 681-4485 Email: annetteb@sylvaneagan.com	AB Learning, Inc.	Annette Braaten
Minnesota	Center #833 154 17th Ave. NW Rochester, Minnesota 55901	Phone: (507) 258-6036 Email: learn@sylvanrochester.com	Bliss Education Resources, LLC	Jennifer Blissenbach David Blissenbach
Minnesota	Center #835 3495 Northdale Blvd NW, Suite 160 Minneapolis, Minnesota 55448	Phone: (763) 712-1118 Fax: (763) 712-1190 Email: CoonRapids.MN@sylvanlearning.com	Alldaymk LLC	Michael Kelley
Minnesota	Center #836 170 Cobblestone Lane Burnsville, Minnesota 55337	Phone: (952) 435-6603 Fax: (952) 435-7782 Email: Burnsville.MN@sylvanlearning.com	Alldaymk LLC	Michael Kelley
Minnesota	Center #837 9900 Valley Creek Road, Suite 150 Woodbury, Minnesota 55125	Phone: (651) 702-6690 Fax: (651) 702-6690 Email: Woodbury.MN@sylvanlearning.com	Alldaymk LLC	Michael Kelley

Minnesota	Center #838 1913 MN-36 Roseville, Minnesota 55113	Phone: (651) 639-1350 Fax: (651) 639-4261 Email: Roseville.MN@sylvanlearning.com	Alldaymk LLC	Michael Kelley
Minnesota	Center #839 7928 Main Street North Maple Grove, Minnesota 55369	Phone: (763) 494-9676 Fax: (763) 494-9678 Email: MapleGrove.MN@sylvanlearning.com	Alldaymk, LLC	Michael Kelley
Minnesota	Center #840 614 Snelling Avenue South Saint Paul, Minnesota 55116	Phone: (651) 583-5500 Email: SaintPaul.MN@sylvanlearning.com	Alldaymk LLC	Michael Kelley
Minnesota	Center #843 6550 York Ave S, Suite 119 Edina, Minnesota 55435	Phone: (651) 374-4396 Email: edina.mn@sylvanlearning.com	Alldaymk LLC	Michael Kelley
Minnesota	Center #844 14675 Martin Drive Eden Prairie, Minnesota 55344	Phone: (651) 374-4383 Email: edenprairie.mn@sylvanlearning.com	Alldaymk LLC	Michael Kelley Alldaymk LLC
Minnesota	Center #845 5585 La Centre Avenue, Suite 200 Albertville, Minnesota 55301	Phone: (763) 276-1006 Email: Albertville.MN@sylvanlearning.com	M4 Learning, LLC	Suzanne Flottmeier Morgan Flottmeier
Minnesota	Center #846 21395 John Milless Dr, Suite 500 Rogers, Minnesota 55374	Phone: (763) 710-5524 Email: Rogers.MN@sylvanlearning.com	Lana Education L. L. C.	Lana Eid
Mississippi	Center #3600 4915 I-55 North, Suite 106A P.O. Box 13275 Jackson, Mississippi 39206	Phone: (601) 366-6400 Fax: (601) 366-6293 Email: jacksonsylvan@gmail.com	Euell Ventures LLC	William Harrison
Mississippi	Center #3913 28 Pass Road Gulfport, Mississippi 39507	Phone: (228) 207-0331 Email: info@gulfportsylvan.com	Euell Ventures LLC	William Harrison

Mississippi	Center #3615 775 East Goodman Rd, Ste 3 Southaven, Mississippi 38671	Phone: (662) 548-3225 Email: ddefrehn@southavensylvan.com	Euell Ventures LLC	William Harrison
Mississippi	Center #3616 6059 US Hwy 98, Ste 10 Hattiesburg, Mississippi 39402	Phone: (601) 402-7869 Email: Hattiesburg.MS@sylvanlearning.com	The Kemebray Group, LLC	Ms. Brandian Bufford
Missouri	Center #700 1830 E Independence St. Springfield, Missouri 65804	Phone: (417) 268-9392 Fax: (417) 882-0979 Email: kristensylvan@sbcglobal.net	Educating Tomorrow's Future, LLC	Kristen Fischer Chris Fischer
Missouri	Center #702 14248 G Manchester Rd. Ballwin, Missouri 63011	Phone: (636) 394-3104 Fax: (636) 394-3648 Email: ballwin@sylvantutors.net	Guide & Grow LLC	Navdeep Batra Anil Batra
Missouri	Center #703 9338 Olive Boulevard, Suite 110 St. Louis, Missouri 63132	Phone: (314) 997-3035 Fax: (314) 997-1555 Email: robbie@robbiewalls.com	The Walls Speak LLC	Cardies (Dino) Walls Robbie Walls
Missouri	Center #705 1184 Douglas Street Lee's Summit, Missouri 64086	Phone: (816) 525-4500 Fax: (816) 525-9677 Email: tricia.sylvan.ls@gmail.com	SLC of Lee's Summit, Inc.	Larry Finkemeier
Missouri	Center #706 4504 Mid Rivers Mall Dr. St. Peters, Missouri 63376	Phone: (636) 441-1212 Fax: (636) 441-2319 Email: stpeters@sylvantutors.net	Guide & Grow LLC	Navdeep Batra Anil Batra
Missouri	Center #707 8409 N. Main (Barry Rd and N. Main) Kansas City, Missouri 64155	Phone: (816) 468-7900 Fax: (816) 468-6647 Email: barbara.sylvan@gmail.com	Educating Tomorrow's Future, LLC	Kristen Fischer Chris Fischer
Missouri	Center #733 141 Civic Center Drive Lake St. Louis, Missouri 63367	Phone: (636) 265-2580 Email: lakestl@sylvantutors.net	Guide & Grow LLC	Navdeep Batra Anil Batra

Missouri	Center #735 3301 Broadway Business Park, Suite B Columbia, Missouri 65203	Phone: (573) 446-0310 Email: Columbia.MO@sylvanlearning.com	KMB Solutions LLC	Melanie Bowers Kyle Bowers
Montana	Center #3700 2103 Central Ave. Billings, Montana 59102	Phone: (406) 656-4646 Fax: (406) 655-9434 Email: olga.prather@sylvanlearning.com	Valor Learning, LLC	Darcell Stracke Olga Prather
Montana	Center #3701 1941 Harrison Ave. Suite D Butte, Montana 59701	Phone: (406) 782-5700 Fax: (406) 782-5780 Email: sylvan@sylvanbutte.com	Ronald & Charlene Kautzman, Kristel Callahan	Ronald Kautzman Charlene Kautzman Kristel Callahan
Montana	Center #3702 1075 N. Rodney Suite 111 Helena, Montana 59601	Phone: (406) 443-9205 Fax: (406) 443-9209 Email: sylvanhelena@montana.com	Ronald & Charlene Kautzman, Kristel Callahan	Ronald Kautzman Charlene Kautzman Kristel Callahan
Nebraska	Center #3800 16613 Polk Street Omaha, Nebraska 68135	Phone: (402) 884-8731 Fax: (402) 934-8444 Email: bmcdermott@nesylvan.com	Partners in Learning, Inc.	Marissa Roh Gregory Roh
Nebraska	Center #3804 3900 Yankee Hill Road, #113 Lincoln, Nebraska 68516	Phone: (402) 423-8833 Fax: (402) 423-8853 Email: bcorey@nesylvan.com	Partners in Learning, Inc.	Marissa Roh Gregory Roh
Nebraska	Center #3808 3615 Summit Plaza Dr. #103 Bellevue, Nebraska 68123	Phone: (402) 403-0255 Fax: (402) 885-7820 Email: ameyer@nesylvan.com	Partners in Learning, Inc.	Marissa Roh Gregory Roh
Nebraska	Center #3815 15475 Ruggles Street #113 Omaha, Nebraska 68116	Phone: (402) 268-0585 Email: khirschman@nesylvan.com	Partners in Learning, Inc.	Marissa Roh Gregory Roh
Nevada	Center #2209 10895 S. Eastern Ave. #160 Henderson, Nevada 89052	Phone: (702) 795-7323 Fax: (702) 795-0970 Email: sylvangreenvalley@gmail.com	Grant Learning LV, LLC	Carole Grant Frank Grant IV Karen Grant
Nevada	Center #2210 9330 W Sahara Avenue Suite 130 Las Vegas, Nevada 89117	Phone: (702) 876-4090 Fax: (702) 876-4355 Email: sylvanlasvegas@gmail.com	Grant Learning LV, LLC	Carole Grant Frank Grant IV Karen Grant

Nevada	Center #2216 6070 S. Rainbow Blvd, Ste 4 Las Vegas, Nevada 89118	Phone: (725) 205-3533 Email: LasVegas- SW.NV@sylvanlearning.com	BergLynn LLC	Joann Berg-Lynn Terrence Lynn II
Nevada	Center #2217 8180 Blue Diamond Road Ste. 140 Blue Diamond, Nevada 89178	Phone: (702) 410-5140 Email: BlueDiamond.NV@Sylvanlearning.com	JTBIUS INVESTMENTS, NV LLC	Tina Herrera
New Hampshire	Center #4004 72 South River Road Bedford, New Hampshire 03110	Phone: (603) 622-7772 Fax: (603) 622-7822 Email: denisetompkins@comcast.net	DJS Educational Consultants, Inc.	Denise Tompkins James Tompkins
New Hampshire	Center #4009 800 Islington Street, Unit 10-B Portsmouth, New Hampshire 03801	Phone: (603) 819-4509 Fax: (603) 433-1848 Email: sylvan.learning@capstoneeducation.info	Unlocked Learning LLC	Eui Choi Eun Kyung Song
New Hampshire	Center #4010 240 North Broadway, Suite E Salem, New Hampshire 03079	Phone: (603) 952-4014 Fax: (603) 257-3111 Email: sylvan.learning@capstoneeducation.info	Unlocked Learning LLC	Eui Choi Eun Kyung Song
New Jersey	Center #4104 3635 Quakerbridge Road University Office Plaza Hamilton Township, New Jersey 08619-1247	Phone: (609) 588-9037 Fax: (609) 588-8285 Email: mkance@hamiltonsylvan.com	True Potential Group, LLC	Mark Kance
New Jersey	Center #4106 1086 Teaneck Road Teaneck, New Jersey 07666	Phone: (201) 837-0049 Fax: (201) 837-9640 Email: yoursylvan@gmail.com	Thunderstruck, Inc.	Anthony DiGiacomo
New Jersey	Center #4108 754 Route 18 Suite 103 East Brunswick, New Jersey 08816	Phone: (732) 390-4040 Fax: (732) 390-0515 Email: sylvan.eastbrunswick@gmail.com	Nancy Velisek & Frank Velisek	Frank Velisek Nancy Velisek

New Jersey	Center #4110 380 Bloomfield Avenue Caldwell, New Jersey 07006	Phone: (973) 228-8693 Fax: (973) 228-8691 Email: sylvancaldwell@gmail.com	Lorocco & Levine LLC	Brian LoRocco Edward Levine
New Jersey	Center #4119 33-00 Broadway, Ste 201 Fair Lawn, New Jersey 07410	Phone: (201) 794-6798 Fax: (201) 794-3604 Email: fairlawnsylvan@optonline.net	Brain Shake, Inc.	Anthony DiGiacomo
New Jersey	Center #4128 123 North Union Ave. Cranford, New Jersey 07016	Phone: (908) 709-0202 Fax: (908) 709-3260 Email: cranfordslvn@gmail.com	SDP Learning, Inc.	Lainia Bohen William Bohen
New Jersey	Center #4138 240 Franklin Avenue Nutley, New Jersey 07110	Phone: (973) 667-4580 Email: sylvannutley@yahoo.com	SDP Learning, Inc.	Lainia Bohen William Bohen
New Jersey	Center #4141 751 Route 73 North Marlton, New Jersey 08053	Phone: (856) 596-8455 Fax: (856) 596-8025 Email: sylvanexec@slcmarlton.com	Balaji Laxmi Inc.	Sumit Anand
New Jersey	Center #4150 1009 St. Georges Avenue #3 Woodbridge, New Jersey 07067	Phone: (908) 463-7207 Email: woodbridge.nj@sylvanlearning.com	Bright Future Global Consulting and Education Services L.L.C.	Wayne Davis Kwame Davis
New Jersey	Center #4151 1110 Route 1 N Edison, New Jersey 08817	Phone: (848) 229-2024 Email: Edison.NJ@sylvanlearning.com	Elite Youth Educational Services LLC	Nadia Hernandez Rick Hernandez Rachel Hernandez
New Jersey	Center #4152 740 South 18th St Newark, New Jersey 07103	Phone: (973) 830-1007 Email: CentralNewark.NJ@sylvanlearning.com	Paramount Family Services Limited Liability Company	LaRhonda Lewis
New Mexico	Center #1900 10555 Montgomery NE, Suite 70 Albuquerque, New Mexico 87111	Phone: (505) 296-1155 Fax: (505) 296-3084 Email: maria@sylvanisgreat.com	Navlys LLC	Maria Romero

New Mexico	Center #1901 10200 Golf Course Road NW Suite G Albuquerque, New Mexico 87114	Phone: (505) 899-3061 Fax: (505) 899-1492 Email: Jennifer@sylvanisgreat.com	Navlys LLC	Maria Romero
New Mexico	Center #1903 3050 North Roadrunner Pkwy Suite C, Unit 2 Las Cruces, New Mexico 88011	Phone: (575) 532-9005 Fax: (505) 532-9603 Email: sylvan.lascruces@rydinc.com	Rydholm, Inc.	Ian Rydholm Richard Rydholm
New Mexico	Center #1905 500 Tenth Street, Suite 304 Alamogordo, New Mexico 88310	Phone: (575) 491-0545 Email: Alamogordo.NM@sylvanlearning.com	Global Tutoring Network LLC	Israel Cordero
New York	Center #4218 3300 Monroe Avenue Rochester, New York 14618	Phone: (585) 385-9480 Fax: (585) 385-9481 Email: steron@frontiernet.net	Personal Educational Trainers, Inc.	Carol Steron
New York	Center #4219 8180 Main Street Williamsville, New York 14221	Phone: (716) 580-3566 Fax: (716) 565-0574 Email: amherst@sylvanwny.com	KD LionStar, LLC	Kristen Maines
New York	Center #4226 Washington Avenue Extension Pine West Plaza, Building 1A Albany, New York 12205	Phone: (518) 869-6005 Fax: (518) 869-4878 Email: Albany@sylvanlearningcenter.org	Transforming Lives, Inc.	Cathlene Schwartzbeck
New York	Center #4231 83-02 Broadway Elmhurst, New York 11373	Phone: (718) 997-8640 Fax: (718) 997-8650 Email: regopark.info@sylvannyc.com	Good Apple Achievement LLC	Jay Lin Grace Wen
New York	Center #4241 1600 West Ridge Road Rochester, New York 14615	Phone: (585) 663-6226 Fax: (585) 663-6407 Email: greece@sylvanlearningny.com	Personal Educational Trainers of Greece, Inc.	Carol Steron Scott Steron Susan Steron- Herbison

New York	Center #4243 850 Bronx River Road Suite 104 Bronxville, New York 10708-7095	Phone: (914) 237-4396 Fax: (914) 237-6178 Email: bronxville.ny@sylvanlearning.com	Niche Learning LLC	Gregory Caruso
New York	Center #4246 10 Maxwell Drive, Suite 201 Clifton Park, New York 12065	Phone: (518) 373-7995 Fax: (518) 373-7991 Email: pmcnamara@sylvanlearningcenter.org	Transforming Lives, Inc.	Cathlene Schwartzbeck
New York	Center #4275 23 Main Street Port Washington, New York 11050	Phone: (516) 303-9955 Email: sylvan.portwashington@sylvanlearning.com	Excellence and Beyond, Inc.	Lisa Yeung Wing Ng
New York	Center #4277 393 Jericho Turnpike Mineola, New York 11501	Phone: (516) 743-9943 Fax: (516) 743-9946 Email: sylvanofmineola@optonline.net	PMFRUSSO Learning, Inc.	Patricia Russo
New York	Center #4278 57 Wheeler Avenue Pleasantville, New York 10570	Phone: (914) 579-2584 Email: Pleasantville.NY@sylvanlearning.com	Ingenious Strategies Corporation	Natalie DiPaolo Gianni DiPaolo
New York	Center #4279 4845 Transit Road Depew, New York 14043	Phone: (716) 671-2036 Email: lancaster.ny@sylvanlearning.com	SAVI Learning Inc.	Sabrina Ghotra Harjeet Ghotra
New York	Center #4280 115 Dreiser Loop Bronx, New York 10475	Phone: (347) 275-1825 Email: BronxNE.NY@sylvanlearning.com	Teach Don't Tell, LLC	Juliet Duggins Guy Duggins
New York	Center #4281 14 Mountain Ledge Drive, Suite 2 Wilton, New York 12831	Phone: (518) 886-1330 Email: wilton.ny@sylvanlearning.com	Skills for Success, LLC	John Waterfield Audrey Waterfield
North Carolina	Center #1600 11220 Elm Lane, Suite 104 Charlotte, North Carolina 28277	Phone: (704) 542-7600 Fax: (704) 542-7400 Email: sylvanlearningnc@gmail.com	Sarrison, Inc.	Debbie Rosenbaum David Rosenbaum

North Carolina	Center #1601 8310 Medical Plaza Drive Mallard Creek Professional Park, Suite F Charlotte, North Carolina 28262	Phone: (704) 548-3910 Fax: (704) 548-3912 Email: charlotte1601.nc@sylvanlearning.com	MereCo Holdings LLC	Meredith Freeman
North Carolina	Center #1602 7209 Creedmoor Road Suite 111 Raleigh, North Carolina 27613	Phone: (919) 846-9441 Fax: (919) 954-1286 Email: raleigh.nc@sylvanlearning.com	RiverWise, LLC	Julie Hoyes Jonathan Hoyes
North Carolina	Center #1603 2921-A Battleground Ave. Greensboro, North Carolina 27408	Phone: (336) 218-8115 Fax: (336) 218-0808 Email: greensboro.nc@sylvanlearning.com	Maj-One LLC	Meredith Freeman
North Carolina	Center #1604 3360 Robinhood Road Winston-Salem, North Carolina 27106	Phone: (336) 724-5644 Fax: (336) 760-2314 Email: winstonsalem.nc@sylvanlearning.com	Maj-One LLC	Meredith Freeman
North Carolina	Center #1605 1040 Darrington Drive Cary, North Carolina 27513	Phone: (919) 380-1201 Fax: (919) 380-7904 Email: mariam@carolinasylvan.com	David Rosenbaum	David Rosenbaum
North Carolina	Center #1607 710 F Red Banks Road Greenville, North Carolina 27858	Phone: (252) 756-9383 Fax: (252) 756-7761 Email: britner.sylvan@gmail.com	Desler Inc.	Melissa Britner
North Carolina	Center #1608 1070 Tunnel Road #30 Asheville, North Carolina 28805	Phone: (828) 237-3900 Fax: (828) 252-2358 Email: Sylvan.Asheville@glc-sylvan.com	Gardner Learning Centers Inc.	Amy Gardner
North Carolina	Center #1609 4130 Oleander Drive, Ste. 101 Wilmington, North Carolina 28403	Phone: (910) 392-6284 Fax: (910) 350-2576 Email: wilmingtoninquiry@gmail.com	Successful Learning, Inc.	Amanda Cochran

North Carolina	Center #1612 2915 Raeford Road Suite 103 Fayetteville, North Carolina 28303	Phone: (910) 339-2909 Fax: (910) 323-2250 Email: sylvanlearningnc@gmail.com	David Rosenbaum	David Rosenbaum
North Carolina	Center #1613 3000 Henderson Drive Jacksonville, North Carolina 28546	Phone: (910) 455-3136 Fax: (910) 455-6928 Email: jacksonville.nc@sylvanlearning.com	Successful Learning, Inc.	Amanda Cochran
North Carolina	Center #1614 3931 Tinsley Dr, Suite 102 High Point, North Carolina 27265	Phone: (336) 841-5522 Fax: (336) 841-5524 Email: SylvanLearningNC@gmail.com	David Rosenbaum	David Rosenbaum
North Carolina	Center #1615 491 Williamson Road Suite 204 Mooresville, North Carolina 28117	Phone: (704) 660-3994 Fax: (704) 660-3996 Email: sylvanlearningnc@gmail.com	David Rosenbaum	David Rosenbaum
North Carolina	Center #1616 1411 Aversboro Road, Suite 315 Garner, North Carolina 27529	Phone: (919) 779-2229 Fax: (919) 779-0059 Email: garner.nc@sylvanlearning.com	RiverWise, LLC	Julie Hoyes Jonathan Hoyes
North Carolina	Center #1619 1008 - E Big Oak Court Knightdale, North Carolina 27545	Phone: (919) 217-0600 Fax: (919) 217-3620 Email: knightdale.nc@sylvanlearning.com	RiverWise, LLC	Julie Hoyes Jonathan Hoyes
North Carolina	Center #1623 11459 US Hwy 70 Business West Clayton Crossing Shopping Center Clayton, North Carolina 27520	Phone: (919) 553-8009 Fax: (919) 550-8360 Email: claytonsylvan@nc.rr.com	J&M Learning Systems, Inc.	James Hewitt

North Carolina	Center #1626 795 Boone Station Drive Burlington, North Carolina 27215	Phone: (336) 570-4999 Fax: (336) 586-9122 Email: burlington.nc@sylvanlearning.com	MereCo Holdings LLC	Meredith Freeman
North Carolina	Center #1627 416A Pollock St. New Bern, North Carolina 28562	Phone: (252) 497-2083 Fax: (252) 633-7804 Email: amanda@educateonslow.com	Successful Learning, Inc.	Amanda Cochran
North Carolina	Center #1639 1241 S. Main Street Suite 20 Wake Forest, North Carolina 27587	Phone: (919) 263-9184 Fax: (919) 562-8150 Email: wakeforest.nc@sylvanlearning.com	RiverWise, LLC	Julie Hoyes Jonathan Hoyes
North Carolina	Center #1654 1775 Weeksville Road Suite N Elizabeth City, North Carolina 27909	Phone: (252) 334-9700 Fax: (252) 334-9751 Email: ecsylvan1654@gmail.com	Northeast Education Foundation, Inc.	Michelle Calhoun Tynch
North Carolina	Center #1657 9111 Monroe Road Suite 160 Charlotte, North Carolina 28270	Phone: (704) 846-0660 Fax: (704) 846-0680 Email: sylvanlearningnc@gmail.com	Sarrison, Inc.	Debbie Rosenbaum David Rosenbaum
North Carolina	Center #1658 720-4 NC Highway 24-27 Bypass E Albemarle, North Carolina 28001	Phone: (704) 982-2288 Fax: (704) 982-2281 Email: sylvanalb@ctc.net	Bright Learning Center, LLC	Marianne Bright
North Carolina	Center #1659 5117 Highgate Drive Suite 100 Durham, North Carolina 27713	Phone: (919) 937-9363 Fax: (919) 937-9796 Email: crystal@sylvancarolina.com	Saha, Inc.	David Rosenbaum
North Carolina	Center #1660 15905 Brookway Dr., Suite 4104 Huntersville, North Carolina 28078	Phone: (704) 997-6515 Fax: (704) 542-7400 Email: sarrison16@gmail.com	Sarrison, Inc.	Debbie Rosenbaum David Rosenbaum

North Carolina	Center #1662 374 Raleigh Street Holly Springs, North Carolina 27540	Phone: (919) 388-3283 Email: sylvanlearningnc@gmail.com	David Rosenbaum	David Rosenbaum
North Carolina	Center #1663 50101 Governors Dr. Suite 130 Chapel Hill, North Carolina 27517	Phone: (919) 495-2335 Email: sylvanlearningNC@gmail.com	Career Partners, Inc.	David Rosenbaum
North Carolina	Center #1664 336 Rockwood Rd., Suite 106 Arden, North Carolina 28704	Phone: (828) 237-3900 Email: Sylvan.Arden@glc-sylvan.com	Gardner Learning Centers Inc.	Amy Gardner
North Carolina	Center #1667 140B North Main St Belmont, North Carolina 28012	Phone: (704) 829-3056 Email: Education@SylvanSchoolSuccess.com	Carpenter-Rhodes Ventures Inc	Brandi Rhodes Joe Carpenter Angela Carpenter
North Dakota	Center #4300 3033 Brandt Drive Suite A Fargo, North Dakota 58104	Phone: (701) 499-4300 Fax: (701) 476-2958 Email: sylvan@sylvanfargo.com	Educational Excellence, Inc.	Kari Weigel Ryan Weigel
North Dakota	Center #4301 212 West Century Avenue Century Office Plaza Park Bismarck, North Dakota 58503	Phone: (701) 223-0010 Fax: (701) 223-1239 Email: sylvan@sylvanbismarck.com	Educational Achievements, Inc.	Kari Weigel
Ohio	Center #1419 441 Miamisburg- Centerville Road Centerville, Ohio 45459	Phone: (937) 435-0222 Fax: (937) 435-5496 Email: sylvanc@sbcglobal.net	M & D Greenisen LLC	Sandra Keily Marcie Greenisen David Greenisen
Ohio	Center #1420 950 Youngstown-Warren Road Square One Centre - Suite D Niles, Ohio 44446	Phone: (330) 652-1886 Fax: (330) 652-2876 Email: nilessylvan@neo.rr.com	NBM Management, Inc.	Barry Miller

Ohio	Center #1425 237 S Dixie Drive. Suite 15 Vandalia, Ohio 45377	Phone: (937) 898-6686 Fax: (937) 898-8573 Email: vandalia@sylvaninfo.com	Miami Valley Learning Systems, Inc.	Deborah Slavens
Ohio	Center #1427 6270 Center Street Mentor, Ohio 44060	Phone: (440) 255-0055 Fax: (440) 255-0797 Email: mentor.oh@sylvanlearning.com	Mentor Center LLC	Melissa Fryer
Ohio	Center #1434 1566 Spring Meadows Dr. Holland, Ohio 43528	Phone: (419) 867-1331 Fax: (419) 867-1339 Email: Toledo.OH@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
Ohio	Center #1435 2365 Dayton-Xenia Road Beavercreek, Ohio 45434	Phone: (937) 427-8808 Fax: (937) 427-8809 Email: bcsylvan@gmail.com	M & D Greenisen LLC	Sandra Keily Marcie Greenisen David Greenisen
Ohio	Center #1440 2357 Elida Road Lima, Ohio 45805	Phone: (419) 331-7323 Fax: (419) 331-7882 Email: lima.oh@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
Ohio	Center #1441 1781 Tiffin Avenue Findlay, Ohio 45840	Phone: (419) 427-0975 Fax: (419) 427-0976 Email: findlay.oh@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
Ohio	Center #1442 8933 S Old State Rd. Lewis Center, Ohio 43035	Phone: (740) 548-5663 Fax: (614) 505-7429 Email: LewisCenter.OH@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
Ohio	Center #1451 1039 Haskins Road Suite D Bowling Green, Ohio 43402	Phone: (419) 353-7608 Fax: (419) 353-7529 Email: bowlinggreen.oh@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
Ohio	Center #1459 1018 Ralston Ave Suite 110 Defiance, Ohio 43512	Phone: (419) 782-8100 Fax: (419) 782-0003 Email: defiance.oh@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala

Ohio	Center #1476 5721 N. Hamilton Road Columbus, Ohio 43230	Phone: (614) 944-9510 Email: Columbus.OH@sylvanlearning.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
Ohio	Center #1480 2855 West Market Street Fairlawn, Ohio 44333	Phone: (330) 836-7031 Fax: (330) 864-1629 Email: fairlawn.oh@sylvanlearning.com	KNA Growth Opportunities, LLC	Krystle Nichols
Ohio	Center #1481 8900 Darrow Road Suite H106 Twinsburg, Ohio 44087	Phone: (330) 487-5820 Fax: (330) 487-5822 Email: twinsburg.oh@sylvanlearning.com	Academic Ventures MBR, LLC	Melinda Wood Roger Racey Bretaigne Phillips
Ohio	Center #1482 26040 Detroit Road #5 Westlake, Ohio 44145	Phone: (440) 835-5050 Fax: (440) 835-9696 Email: Westlake.oh@sylvanlearning.com	White Caps Stewardship LLC	Amy Tipton-Fletcher
Ohio	Center #1483 14240 Pearl Road Strongsville, Ohio 44136	Phone: (440) 238-0530 Fax: (440) 238-4277 Email: Strongsville.oh@sylvanlearning.com	White Caps Stewardship LLC	Amy Tipton-Fletcher
Ohio	Center #1485 6180 Tylersville Road Mason, Ohio 45040	Phone: (513) 204-5647 Fax: (513) 755-4945 Email: Westchester.oh@sylvanlearning.com	ITnet Consulting Group, LLC	Supraja Sanakari Madhu Sanakari
Oklahoma	Center #903 6808 South Memorial Suite 222 Tulsa, Oklahoma 74133	Phone: (918) 250-7323 Fax: (918) 250-8400 Email: ntrainor@sylvanok.com	Tulsa Learning Centers, Inc.	John Doak James Berryhill Glenn Haring
Oklahoma	Center #908 2603 NW Cache Road Suite 2 Lawton, Oklahoma 73505	Phone: (580) 351-9100 Fax: (580) 351-9416 Email: sylvanlc908@sbcglobal.net	Howard & Jo Jean Johnson	Howard Johnson Jo Jean Johnson
Oklahoma	Center #910 11632 E. 86th Street N. Owasso, Oklahoma 74055	Phone: (918) 274-0700 Fax: (918) 274-4848 Email: owasso.ok@sylvanlearning.com	Oklahoma Education Systems, Inc.	Joshua Ploch Rebekah Ploch

Oklahoma	Center #913 9634 N. May Ave. Oklahoma City, Oklahoma 73120	Phone: (405) 842-7323 Fax: (405) 843-2665 Email: Sylvan.OklahomaCity@sylvanlearning.com	Oklahoma Education Systems, Inc.	Joshua Ploch Rebekah Ploch
Oregon	Center #100 12225 SW 2nd Street, Suite 100 Beaverton, Oregon 97005	Phone: (503) 626-7466 Fax: (503) 643-5696 Email: sylvanbeaverton@gmail.com	AAEE LLC	Audra Bohn Amy Paradis Mitchell Bohn Eric Paradis
Oregon	Center #101 4300 NE Fremont Columbia Plaza, Suite 150 Portland, Oregon 97213	Phone: (503) 249-2855 Fax: (503) 249-8061 Email: slc101@comcast.net	Advantage Education, Inc.	Cliff Crawford
Oregon	Center #104 2150 Northeast Studio Road Suite 10 Bend, Oregon 97701	Phone: (541) 389-9252 Fax: (541) 389-1262 Email: Bend.OR@sylvanlearning.com	Cascade Learning, Inc.	Audra Bohn Mitchell Bohn
Oregon	Center #111 1672 Willamette Falls Dr, Ste C West Linn, Oregon 97068	Phone: (503) 451-5815 Email: twl.or@sylvanlearning.com	AAEE LLC	Audra Bohn Amy Paradis Mitchell Bohn Eric Paradis
Oregon	Center #112 4555 Liberty Rd. South Sunnyslope Center, Suite 180 Salem, Oregon 97302	Phone: (503) 363-2996 Fax: (503) 362-6624 Email: sylvan0112@hotmail.com	Kenzie Adam, Inc.	Kimberli Dorn Michael Dorn
Pennsylvania	Center #1730 7200 Peach Street Summit Towne Centre - Suite 440 Erie, Pennsylvania 16509- 4709	Phone: (814) 864-6100 Fax: (814) 864-7326 Email: erie.pa@sylvanlearning.com	Erdman Enterprises LLC	Brianna Erdman James Erdman Lavonne Steves
Pennsylvania	Center #1731 1695 Oregon Pike Lancaster, Pennsylvania 17601	Phone: (717) 293-0300 Fax: (717) 435-9692 Email: Lancaster.PA@sylvanlearning.com	Webster Associates, Inc.	Michael Webster Lauren Webster

Pennsylvania	Center #1734 6537 Roosevelt Blvd. Roosevelt Plaza Philadelphia, Pennsylvania 19149	Phone: (215) 743-6200 Fax: (215) 743-8674 Email: SylvanPhila@gmail.com	Philadelphia Learning, LLC	Lynne Zayas
Pennsylvania	Center #1757 4 N Londonderry Square Palmyra, Pennsylvania 17078	Phone: (717) 838-2650 Fax: (717) 838-0930 Email: Elizabeth@sylvanpa.com	Webster Associates, Inc.	Michael Webster Lauren Webster
Pennsylvania	Center #1773 2058 S Queen St, Suite B York, Pennsylvania 17403	Phone: (717) 293-0300 Email: lauren@sylvanpa.com	Webster Associates, Inc.	Michael Webster Lauren Webster
Pennsylvania	Center #1776 3150 Washington Rd McMurray, Pennsylvania 15317	Phone: (412) 409-2593 Email: Pittsburgh.PA@sylvanlearning.com	Hometown Learning, LLC	Suzan O'Brien Katelyn O'Brien
Pennsylvania	Center #1779 850 Cranberry Woods Dr, Suite 2223 Cranberry Township, Pennsylvania 16066	Phone: (724) 741-1046 Email: Cranberry.PA@sylvanlearning.com	Hometown Learning, LLC	Suzan O'Brien Katelyn O'Brien
Pennsylvania	Center #1782 4910 Carlisle Pike, Suite B5 Mechanicsburg, Pennsylvania 17050	Phone: (717) 769-2323 Email: Mechanicsburg.PA@sylvanlearning.com	Webster Associates, Inc.	Michael Webster Lauren Webster
Pennsylvania	Center #1784 148 Eagleview Blvd. Exton, Pennsylvania 19341	Phone: (248) 346-6307 Email: exton.pa@sylvanlearning.com	Crystal Hub Learning LLC	Sangeetha Kondaveeti Sridhar Kondaveeti
Rhode Island	Center #3902 2145 Diamond Hill Rd. #2 Cumberland, Rhode Island 02864	Phone: (401) 334-9309 Fax: (401) 334-4054 Email: Rjohnson@sylvanne.com	J & G Enterprises, Inc.	Rebecca Johnson Matthew Johnson
Rhode Island	Center #3904 65 Sandy Bottom Road Coventry, Rhode Island 02816	Phone: (401) 823-0011 Fax: (401) 823-0088 Email: rjohnson@sylvanne.com	J & G Enterprises, Inc.	Rebecca Johnson Matthew Johnson

South Carolina	Center #2101 1172 Oakland Market Road Suite 12 Mt. Pleasant, South Carolina 29466	Phone: (843) 352-7364 Fax: (843) 766-7324 Email: mpattimcgrath@gmail.com	MTC Learning, LLC	Mary Patricia McGrath Ray McGrath
South Carolina	Center #2106 119 E. 4th North Street. Unit A Summerville, South Carolina 29483	Phone: (843) 486-0081 Fax: (843) 569-3602 Email: summerville.sc@sylvanlearning.com	SLSC Group, Inc.	Susan Fairbairn
South Carolina	Center #2120 150 Tanner Road Greenville, South Carolina 29607	Phone: (864) 676-0011 Fax: (864) 676-9505 Email: Sylvan.Greenville@glc-sylvan.com	Gardner Learning Centers, Inc.	Amy Gardner
South Carolina	Center #2121 1270 Bower Parkway Suite C-10 Columbia, South Carolina 29212	Phone: (803) 781-7323 Fax: (803) 741-7700 Email: sylvan.irmo@glc-sylvan.com	Gardner Learning Centers, Inc.	Amy Gardner
South Carolina	Center #2122 130-9 Forum Drive Village at Sandhill Columbia, South Carolina 29229	Phone: (803) 741-0999 Fax: (803) 781-9984 Email: Sylvan.Columbia@glc-sylvan.com	Gardner Learning Centers, Inc.	Amy Gardner
South Carolina	Center #2123 110 Miracle Mile Rd. Suite D Anderson, South Carolina 29621	Phone: (864) 676-0011 Email: Sylvan.Anderson@glc-sylvan.com	Gardner Learning Centers, Inc.	Amy Gardner
South Carolina	Center #2124 100 Burton St. Suite A Spartanburg, South Carolina 29301	Phone: (864) 676-0011 Email: sylvan.spartanburg@glc-sylvan.com	Gardner Learning Centers, Inc.	Amy Gardner
South Carolina	Center #2125 856 Gold Hill Rd., Unit 108	Phone: (803) 431-7199 Email: FortMill.SC@Sylvanlearning.com	Carpenter-Rhodes Ventures Inc	Brandi Rhodes Joe Carpenter Angela Carpenter

	Fort Mill, South Carolina 29708			
South Carolina	Center #2126 8356 Charlotte Hwy, Suite 102 Indian Land, South Carolina 29707	Phone: (803) 547-1777 Email: IndianLand.SC@Sylvanlearning.com	Hopewell Learning Centers, LLC	Marty Conner
South Carolina	Center #2127 4607 Charlotte Hwy, Suite 110 Clover, South Carolina 29710	Phone: (803) 631-1313 Email: LakeWylie.SC@sylvanlearning.com	NJS Ventures LLC	Carl Shrewsbury Nancy Shrewsbury
South Carolina	Center #2128 625 Crown Pointe Ln, Ste 106 Rock Hill, South Carolina 29730	Phone: (803) 961-4752 Email: RockHill.SC@sylvanlearning.com	NJS Ventures LLC	Carl Shrewsbury Nancy Shrewsbury
South Dakota	Center #4400 4816 S. Technopolis Drive Sioux Falls, South Dakota 57106	Phone: (605) 362-4885 Fax: (605) 362-4882 Email: michellem@sylvansf.org	Dakota East Learning, Inc.	Michelle McGuckin William McGuckin Colleen Halbur Scott Halbur
South Dakota	Center #4402 1620 Mt. Rushmore Rd, Suite 100 Rapid City, South Dakota 57702	Phone: (605) 791-4544 Email: BlackHillsSylvan@gmail.com	Appletec SD, LLC	David Satter Robert Satter Jaclyn Satter
Tennessee	Center #1001 10969 Kingston Pike Knoxville, Tennessee 37922	Phone: (865) 675-7323 Fax: (865) 690-7600 Email: sylvanknoxville@yahoo.com	Full Circle, Inc.	Gary Pierce John Staten
Tennessee	Center #1010 7610 Hamilton Park Dr, Ste. 6 Chattanooga, Tennessee 37421	Phone: (423) 894-8333 Fax: (423) 894-1891 Email: sylvanchattanooga@yahoo.com	Learn to Learn, Inc.	Gary Pierce John Staten

Tennessee	Center #1011 220 Forbes Avenue Suite A Clarksville, Tennessee 37040	Phone: (931) 647-5811 Fax: (931) 647-8396 Email: sylvan1011@sylvanclarksville.com	Charlotte Jolly and Janet Crocker	Charlotte Jolly Janet Crocker
Tennessee	Center #1014 810 Medical Center Parkway Suite C Murfreesboro, Tennessee 37129	Phone: (615) 893-3542 Fax: (615) 893-6279 Email: sylvan1014@outlook.com	Success in Learning, Inc.	Ann Thomas Murphy Thomas Jr.
Tennessee	Center #1017 4260 Fort Henry Drive Suite 8 Kingsport, Tennessee 37663	Phone: (423) 239-9215 Fax: (423) 239-5269 Email: slc1017@earthlink.net	Educational Attainment Systems LLC	Janet Crocker Jeff Phillips
Tennessee	Center #1026 728 South Jefferson Suite 9 Cookeville, Tennessee 38501	Phone: (931) 528-2543 Fax: (931) 520-6411 Email: info@cookevillesylvan.com	Cumberland Ventures, Inc.	John Keisling
Texas	Center #310 19535 Champion Forest Dr, Ste A Spring, Texas 77379	Phone: (281) 586-8069 Fax: (281) 586-7473 Email: champions@sylvansw.com	JCP Ascenision, LLC	Christopher Pittner Jano Pittner Conlen Pittner
Texas	Center #313 913 W. Loop 281 Suite 206 Longview, Texas 75604	Phone: (903) 297-3300 Fax: (903) 297-3262 Email: carol.sylvan@att.net	TX Educate, LLC	Carol Pope
Texas	Center #318 4601 South Loop 289 Suite 23 Lubbock, Texas 79424	Phone: (806) 785-4400 Fax: (806) 785-4421 Email: matthew@sylvanabilene.com	South Plains Learning, Inc.	Michael Sullivan Cindy Sullivan Matthew Sullivan Michael C. Sullivan Tyler Backus
Texas	Center #319 4410 North Midkiff Suite D 201 Midland, Texas 79705	Phone: (432) 520-9411 Fax: (432) 689-4692 Email: sylvan0319@sbcglobal.net	Learning Enhancement, Inc.	Martha Stuart

Texas	Center #324 4019 South Broadway Tyler, Texas 75701	Phone: (903) 597-2535 Fax: (903) 597-7240 Email: sylvan_tyler@suddenlinkmail.com	TEW Educate, LLC	Carol Pope
Texas	Center #325 2412 Old North Rd #103 Denton, Texas 76209	Phone: (940) 484-7050 Fax: (972) 315-3568 Email: sylvan325@aol.com	AHM Ventures, Inc.	Holly Martin
Texas	Center #329 6001 S. Staples Suite A Corpus Christi, Texas 78413	Phone: (361) 993-3793 Fax: (361) 992-0760 Email: corpuschristi@sylvantexas.com	Jack & Sue Ann Pitts	Sue Ann Pitts Jack Pitts
Texas	Center #331 3100 Wildflower Drive, Suite 310 Bryan, Texas 77802	Phone: (979) 846-4988 Fax: (979) 691-2904 Email: sylvanbcs.tx@sylvanlearning.com	JEN 2000 Enterprises LLC	Jennifer Nelson Eric Nelson
Texas	Center #335 3233-B South Willis Abilene, Texas 79605	Phone: (325) 691-0646 Fax: (325) 690-0933 Email: mike@sylvanabilene.com	MJ Sullivan, Inc.	Mike Sullivan Cindy Sullivan Matthew Sullivan Michael C. Sullivan
Texas	Center #339 16648 San Pedro Avenue #101 San Antonio, Texas 78232	Phone: (210) 494-4357 Fax: (210) 496-8044 Email: sylvan0339@sylvantx.com	Lovin SLC, Inc.	Joan Lovin
Texas	Center #341 5832 Southwest Loop 820 Fort Worth, Texas 76132	Phone: (817) 377-0632 Email: trey@sylvanfw1.com	Watten LLC	Trey Watten
Texas	Center #344 5811 N. Mesa Street El Paso, Texas 79912	Phone: (915) 587-7323 Fax: (915) 587-7325 Email: sylvan.elpasowest@rydinc.com	Rydholm, Inc.	Ian Rydholm Richard Rydholm
Texas	Center #348 1757 George Dieter #114 El Paso, Texas 79936	Phone: (915) 629-8774 Fax: (915) 629-8730 Email: sylvan.elpasoeast@rydinc.com	Rydholm, Inc.	Ian Rydholm Richard Rydholm

Texas	Center #352 4240 Heritage Trace Parkway, Suite 500 Fort Worth, Texas 76244	Phone: (817) 431-6000 Fax: (817) 788-4882 Email: matthew@sylvanabilene.com	P&A Learning Services Inc	Andrea Uquillas Quila
Texas	Center #353 8910 Bandera Rd. San Antonio, Texas 78250	Phone: (210) 509-7323 Fax: (210) 509-0961 Email: sylvan0353@sylvantx.com	NWSA Educators, Inc.	Joan Lovin
Texas	Center #358 1420 Schertz Parkway Ste. 280 Schertz, Texas 78154	Phone: (210) 651-5477 Fax: (201) 651-4286 Email: Sylvan0358@sylvantx.com	Lovin SLC, Inc.	Joan Lovin
Texas	Center #363 5723 N. Foster Road, Suite 108 San Antonio, Texas 78244	Phone: (210) 661-3171 Fax: (210) 661-6992 Email: sylvan0363@sylvantx.com	Lovin SLC, Inc.	Joan Lovin
Texas	Center #366 1902 John Stockbauer Drive, Ste 200 Victoria, Texas 77901	Phone: (361) 578-3224 Fax: (361) 582-0040 Email: victoria@Sylvan-VictoriaTx.com	SWS Texas LLC	Beverly Wood-Stubbs
Texas	Center #368 4245 Kemp Blvd., Suite 709 Wichita Falls, Texas 76308	Phone: (940) 761-9000 Fax: (940) 761-9005 Email: kathy@sylvan-wichita-falls.com	K. Stearns Enterprises, LLC	Kathy Stearns
Texas	Center #370 901 Dove Avenue McAllen, Texas 78504	Phone: (956) 682-9800 Fax: (956) 682-9811 Email: McAllen.TX@sylvanlearning.com	RGV Tutor, LLC	Anjanette Garza
Texas	Center #371 3230 Pablo Kisel Blvd., Suite 103 Brownsville, Texas 78520	Phone: (956) 550-0050 Fax: (956) 550-0055 Email: betty.mendoza@sylvanlearning.com	RGV Tutor, LLC	Anjanette Garza
Texas	Center #373 10228 Broadway #156 Pearland, Texas 77584	Phone: (713) 340-0333 Fax: (713) 436-1837 Email: pearland@sylvancentral.com	Student Empowerment Centers, LLC	Rodney Barnes

45

Texas	Center #379 1930 SW Military Drive San Antonio, Texas 78221	Phone: (210) 923-8700 Fax: (210) 333-0298 Email: Sylvan0379@sylvantx.com	Lovin SLC, Inc.	Joan Lovin
Texas	Center #381 4804 East Freeway Baytown, Texas 77521	Phone: (281) 421-4169 Fax: (281) 421-3672 Email: liz.rrexpress@gmail.com	Quila A&A Inc.	Andrea Uquillas Quila
Texas	Center #382 15150 Badger Ranch Blvd., Ste. 202 Waco, Texas 76712	Phone: (254) 772-8811 Fax: (254) 772-6105 Email: Waco.TX@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #383 440 E. Central Texas Expressway, Suite 107 Harker Heights, Texas 76548	Phone: (254) 690-6669 Fax: (254) 690-7197 Email: killeen@sylvan-are.com	Quila A&A Inc.	Christopher Pittner Jano Pittner
Texas	Center #389 601 E Southlake Blvd. #200 Southlake, Texas 76092	Phone: (817) 416-6827 Fax: (817) 416-6569 Email: Southlake.TX@SylvanLearning.com	Diamond Kean Learning, LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #390 8863 Anderson Mill Road, Suite 120 Austin, Texas 78729	Phone: (512) 336-1818 Fax: (512) 336-1902 Email: CedarPark.TX@sylvanlearning.com	Nasus Group, Inc.	Susan Fairbairn
Texas	Center #391 4036 South Lamar Blvd. Suite 150 Austin, Texas 78704	Phone: (512) 441-3898 Fax: (512) 441-4333 Email: a.bennett@sylvanhmj.com	Nasus Group, Inc.	Susan Fairbairn
Texas	Center #392 305 N. Heatherwilde, Ste. 100 Pflugerville, Texas 78660	Phone: (737) 243-9600 Fax: (512) 310-8091 Email: RoundRock.TX@sylvanlearning.com	Nasus Group, Inc.	Susan Fairbairn
Texas	Center #394 8787 N. MacArthur Blvd. Suite 140 Irving, Texas 75063	Phone: (972) 480-1100 Fax: (972) 401-1880 Email: coppellirving.tx@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani

Texas	Center #395 8765 Stockard Drive Suite 801 Frisco, Texas 75034	Phone: (214) 618-2955 Email: Frisco.TX@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #398 10455 N. Central Expressway Suite 111 Dallas, Texas 75231	Phone: (214) 342-3305 Fax: (214) 342-1418 Email: Dallas.TX@SylvanLearning.com	Diamond Kean Learning, LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #399 6501 S. Cooper Street, #141 Arlington, Texas 76001	Phone: (817) 557-6333 Fax: (817) 557-4494 Email: Arlington.tx@sylvanlearning.com	Diamond Kean Learning, LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #8300 2750 Churchill Drive, Suite 130 Flower Mound, Texas 75022	Phone: (972) 355-0222 Fax: (972) 355-0202 Email: FlowerMound.TX@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #8309 20814 Gulf Freeway, Suite 50 Webster, Texas 77598	Phone: (281) 557-5558 Fax: (281) 557-5552 Email: Clearlake.TX@sylvanlearning.com	Helminiak Ascenision, LLC	Andrew Helminiak
Texas	Center #310 32731 Egypt Lane, Suite 503 Magnolia, Texas 77354	Phone: (281) 789-7597 Fax: (281) 367-7668 Email: woodlands@sylvansw.com	Venia Enterprises, LLC	Christopher Pittner Jano Pittner Conlen Pittner
Texas	Center #8311 4323 Kingwood Drive Kingwood, Texas 77339	Phone: (281) 361-9253 Fax: (281) 361-9263 Email: kingwood@sylvansw.com	Venia Enterprises, LLC	Christopher Pittner Jano Pittner Conlen Pittner
Texas	Center #8312 20033 Katy Freeway Houston, Texas 77450	Phone: (281) 492-7084 Fax: (281) 398-5349 Email: info@sylvankaty.com	Korpal & Anwar Learning LLC	Paras Korpal Anum Shabbir
Texas	Center #8315 15219 Southwest Freeway Sugar Land, Texas 77478	Phone: (281) 980-1199 Fax: (281) 980-2226 Email: sugarland.tx@sylvanlearning.com	Inwood Learning, LLC	Ashish Kothari Hemali Kothari

47

V.2024.051024

Texas	Center #8316 12250 Queenston Blvd., Ste. D Houston, Texas 77095	Phone: (281) 256-0200 Email: cypress@sylvansw.com	JCP Ascenision, LLC	Christopher Pittner Jano Pittner Conlen Pittner
Texas	Center #8317 9119 Highway 6, Suite 240 Missouri City, Texas 77459	Phone: (281) 778-9185 Fax: (281) 778-9187 Email: sylvanlearning@entouch.net	4 Ever Clever Learning, LLC	Deirdre Williams Johnny Williams
Texas	Center #8324 210 Central Expressway Suite 98 Allen, Texas 75013	Phone: (469) 675-3238 Fax: (972) 584-1154 Email: Allen.TX@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #8325 1016 E. Hebron Parkway Suite 175 Carrollton, Texas 75010	Phone: (972) 939-9000 Fax: (972) 957-2769 Email: Carrollton.TX@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #8326 6840 Virginia Parkway Suite 105 McKinney, Texas 75071	Phone: (214) 504-9200 Fax: (972) 957-2770 Email: McKinney.TX@sylvanlearning.com	Diamond Kean DFW LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #8327 1300 Dacy Lane, Suite #250 Kyle, Texas 78640	Phone: (512) 262-7035 Email: kyle.tx@sylvanlearning.com	Nasus Group, Inc.	Susan Fairbairn
Texas	Center #8329 415 E. Pleasant Run Road, Suite 125 Cedar Hill, Texas 75104	Phone: (469) 575-9980 Fax: (214) 750-1339 Email: Cedarhill.tx@sylvanlearning.com	Diamond Kean Learning, LLC	Fayzan Keshwani Nizar Bhulani
Texas	Center #8330 1310 North Ed Carey Drive Harlingen, Texas 78550	Phone: (956) 230-4410 Email: betty.mendoza@sylvanlearning.com	RGV Tutor, LLC	Anjanette Garza
Texas	Center #8333 2113 Harwood Road, Suite 365 Bedford, Texas 76021	Phone: (682) 738-3634 Email: Bedford.TX@sylvanlearning.com	Ohana DFW Learn 4, LLC	Jennifer Dempsey Mark Dempsey

Texas	Center #8334 5316 Bissonnet Bellaire, Texas 77401	Phone: (281) 501-2055 Email: Bellaire.TX@Sylvanlearning.com	JNJ Learning LLC	Yvette Johnson
Texas	Center #8335 150 Brand Road, Unit 400 Murphy, Texas 75094	Phone: (469) 454-0110 Email: Murphy.TX@sylvanlearning.com	AmeriTutors Inc.	Roy Vargis Pulikottil
Texas	Center #8336 110 South Preston Rd, Ste 30 Prosper, Texas 75078	Phone: (972) 782-5151 Email: director@sylvanprosper.com	AmeriTutors Inc.	Roy Vargis Pulikottil
Texas	Center #8337 6702 Dalrock Rd, Ste. 136 Rowlett, Texas 75088	Phone: (469) 573-5773 Email: Rowlett.TX@sylvanlearning.com	Laith Teaching Center Inc	Nizar Tayem
Texas	Center #8338 1333 W. Campbell Rd. Richardson, Texas 75080	Phone: (469) 277-6557 Email: WestRichardson.TX@SylvanLearning.com	Ohana DFW Learn 1, LLC	Jennifer Dempsey Mark Dempsey
Texas	Center #8339 120 El Chico Trail, Suite D Willow Park, Texas 76087	Phone: (817) 415-1313 Email: aledo.tx@sylvanlearning.com	Ohana DFW Learn 2, LLC	Jennifer Dempsey Mark Dempsey
Texas	Center #8340 6537 Horizon Road, Suite C Rockwall, Texas 75032	Phone: (972) 930-1230 Email: Rockwall- Heath.TX@sylvanlearning.com	Ohana DFW Learn 3, LLC	Jennifer Dempsey Mark Dempsey
Texas	Center #8344 6619 West Cross Creek Bend Lane, Ste 100 Fulshear, Texas 77406	Phone: (346) 707-8328 Email: info@sylvan.education	Daksh Vidhyapeeth LLC	Sudhaker Chhabra
Texas	Center #8345 16402 W. Lake Houston Parkway Suite 400 Houston, Texas 77044	Phone: (832) 406-7371 Email: atascocita@sylvansw.com	Venia Enterprises, LLC	Christopher Pittner Jano Pittner Conlen Pittner
Texas	Center #8346 1215 S. MacArthur Blvd. Irving, Texas 75060	Phone: (972) 399-0393 Email: Irving.TX@sylvanlearning.com	Diamond Kean LLC	Fayzan Keshwani Nizar Bhulani

Texas	Center #8348 3917 Woodson's Reserve Parkway, Suite 300 Spring, Texas 77386	Phone: (832) 585-1345 Email: eastspring@sylvansw.com	Venia Enterprises, LLC	Christopher Pittner Jano Pittner Conlen Pittner
Texas	Center #8349 24123 Boerne Stage Road, Suite 104 San Antonio, Texas 78255	Phone: (210) 367-7652 Email: sylvan0353@sylvantx.com	NWSA Ascenision, Inc.	Joan Lovin
Texas	Center #8350 1201 N Mechanic El Campo, Texas 77437	Phone: (979) 275-1301 Email: ElCampo.TX@sylvanlearning.com	Jen2000 Enterprises, LLC	Eric Nelson Jennifer Nelson
Texas	Center #8357 205 South Main Street, Suite 10-10 Red Oak, Texas 75154	Phone: (469) 820-9911 Email: RedOak.TX@sylvanlearning.com	Global Tutoring Network LLC	Israel Cordero
Texas	Center #8359 1050 S. Preston Rd #116 Celina, Texas 75009	Phone: (972) 598-9561 Email: Celina.TX@sylvanlearning.com	CSDT Ventures LLC	Shanna Miller Corey Miller
Texas	Center #8361 10450 FM 1464 Road, Suite 700 Richmond, Texas 77407	Phone: (281) 302-5684 Email: Richmond.TX@sylvanlearning.com	4 Ever Clever Learning, LLC	Johnny Williams Deirdre Williams
Texas	Center #8363 6370 N. Eldridge Pkwy, Ste C Houston, Texas 77041	Phone: (346) 635-4102 Email: Addicks.TX@sylvanlearning.com	A Team Learning LLC	Arthur Fonteneaux Nina Fonteneaux
Utah	Center #4501 8184 S Highland Drive, Suite C8 Sandy, Utah 84093	Phone: (801) 233-1700 Fax: (801) 569-8200 Email: sandy@sylvan-utah.com	DH2 GROUP LLC	Doug Dingman Lynne Hudson
Utah	Center #4509 57 North 500 East St. George, Utah 84770	Phone: (435) 656-1700 Fax: (435) 656-2600 Email: sylvan@sginet.com	J.C.H., Inc.	Clark Hatfield Doyle Hatfield Judy Hatfield
Utah	Center #4511 785 East 200 South, Suite 11 Lehi, Utah 84043	Phone: (801) 763-4880 Email: alpine@sylvan-utah.com	DH2 GROUP LLC	Doug Dingman Lynne Hudson

Utah	Center #4514 575 North Main Street Layton, Utah 84041	Phone: (801) 543-4400 Fax: (801) 543-4471 Email: Layton.ut@sylvanlearning.com	Utah Educate Inc.	Rosaelia Romero
Utah	Center #4515 499 West 1400 North Orem, Utah 84057	Phone: (801) 235-1800 Fax: (801) 705-4068 Email: Orem.UT@sylvanlearning.com	DH2 GROUP LLC	Doug Dingman Lynne Hudson
Utah	Center #4518 12447 S. Crossing Drive Suite 11 Riverton, Utah 84096	Phone: (801) 365-6500 Email: Jordan@sylvan-utah.com	DH2 GROUP LLC	Doug Dingman Lynne Hudson
Vermont	Center #4601 1 Market Place, Unit 10 Essex Junction, Vermont 05452	Phone: (802) 871-5230 Fax: (802) 864-5655 Email: vtsylvan@gmail.com	14th State Learning, LLC	Leah LaClaire
Virginia	Center #4702 1807 Huguenot Rd, Suite 101 Midlothian, Virginia 23113	Phone: (804) 240-8408 Fax: (804) 219-3319 Email: melissa.britner@sylvanlearning.com	Successful Learning, Inc.	Amanda Cochran
Virginia	Center #4709 5007-G Victory Boulevard Yorktown, Virginia 23693	Phone: (757) 873-2181 Fax: (757) 873-6613 Email: newportnews.va@sylvanlearning.com	WISE Learning Solutions, LLC	Deborah DeJesus Juan DeJesus
Virginia	Center #4723 8116 Timberlake Road Suite 200 Lynchburg, Virginia 24502	Phone: (434) 832-1000 Fax: (434) 832-0281 Email: sylvaninfo@sylvanlynchburg.com	Michael Bowen Enterprises LLC	Michael Bowen
Virginia	Center #4725 701-P North Battlefield Blvd. Chesapeake, Virginia 23320	Phone: (757) 548-1247 Fax: (757) 549-4532 Email: success@sylvanchesapeake.com	Anowem Educate LLC	Kimberly Teixeira

Virginia	Center #4751 6521 Arlington Blvd, Ste. 306 Falls Church, Virginia 22042	Phone: (703) 734-1234 Fax: (703) 734-1234 Email: center@sylvanfallschurch.com	Educate Virginia, LLC	R. Scott Lindsey Daniel Wildman Tasha Browne
Virginia	Center #4758 3575 Bridge Road #13 Suffolk, Virginia 23435	Phone: (757) 548-1247 Email: success@sylvanchesapeake.com	Anowem Educate LLC	Kimberly Teixeira
Virginia	Center #4759 25 Clement Drive Suite 115 Stafford, Virginia 22556	Phone: (540) 602-2360 Email: center@sylvanstafford.com	Educate Virginia, LLC	R. Scott Lindsey Daniel Wildman Tasha Browne
Virginia	Center #4763 1608 Village Market Blvd. SE, #100. Leesburg, Virginia 20175	Phone: (571) 366-4754 Fax: (571) 526-4586 Email: leesburg.va@sylvanlearning.com	learning force LLC	Shehryar Mohiuddin
Virginia	Center #4764 2579 John Milton Dr, Suite 120 Herndon, Virginia 20171	Phone: (703) 382-6300 Email: Reston.VA@sylvanlearning.com	RMS Enterprise, LLC	Ashish Gandhi Sapna Chordia
Virginia	Center #4768 2305 Mount Vernon Ave. Alexandria, Virginia 22301	Phone: (571) 309-0838 Email: Alexandria.VA@sylvanlearning.com	Ireedui LLC	Batireedui Batsaikhan
Virginia	Center #4769 949 Piney Forest Rd, Ste. B Danville, Virginia 24540	Phone: (434) 832-1000 Email: sylvaninfo@sylvandanville.com	Michael Bowen Enterprises LLC	Michael Bowen
Washington	Center #201 1510 N. Argonne Road Spokane Valley, Washington 99206	Phone: (509) 924-7656 Fax: (509) 924-9528 Email: Sarah.Doll@sylvanlearning.com	Advance Education Solutions NW, LLC	Heather Hamasaki Chad Hamasaki
Washington	Center #202 3333 184th Street SW Suite M Lynnwood, Washington 98037	Phone: (425) 774-3922 Fax: (425) 775-6475 Email: Lynnwood.WA@sylvanlearning.com	GKD Enterprises Limited Liability Company	Hiral Doshi

Washington	Center #205 4140 Factoria Boulevard S.E., Suite 2A Bellevue, Washington 98006	Phone: (425) 641-7609 Fax: (425) 641-6123 Email: kelsey@essylvan.com	RKC Education LLC	Mansi Garg Hemant Garg
Washington	Center #206 615 SE Chkalov Drive Suite 4 Vancouver, Washington 98683	Phone: (360) 859-0007 Fax: (360) 892-3210 Email: admissions@sylvan-pnw.com	Cascadia Learning Centers, LLC	Jamie Tate Patrick Tate
Washington	Center #208 24020 132nd Avenue SE, Unit C Kent, Washington 98042	Phone: (253) 854-7111 Fax: (253) 854-5095 Email: admin@kentsylvan.com	Stem King LLC	Kevin Moen
Washington	Center #211 12020 Meridian East Suite D Puyallup, Washington 98373	Phone: (253) 848-0771 Fax: (253) 845-9256 Email: admissions.sylvan@comcast.net	Puget Sound Learning Centers, LLC	Gail Smith David Smith, III
Washington	Center #212 1614 South Mildred St., Ste. 1614-E Tacoma, Washington 98465	Phone: (253) 987-8565 Fax: (253) 565-0923 Email: admissions.sylvan@comcast.net	Puget Sound Learning Centers, LLC	Gail Smith David Smith, III
Washington	Center #213 3825 Martin Way East Suite 103 Olympia, Washington 98506-5278	Phone: (360) 438-1833 Fax: (360) 438-3424 Email: admissions.sylvan@comcast.net	Puget Sound Learning Centers, LLC	Gail Smith David Smith, III
Washington	Center #214 32020 1st Avenue S Suite 109 Federal Way, Washington 98003	Phone: (253) 838-0507 Fax: (253) 838-6711 Email: darcy@fwsylvan.com	Darcy Webb	Darcy Webb
Washington	Center #216 3212 NW Byron St. Silverdale, Washington 98383	Phone: (360) 255-7290 Fax: (360) 698-2202 Email: susie.petersen@withsylvan.com	Puget Sound Learning Centers, LLC	Gail Smith David Smith, III

Washington	Center #217 12305 120th Avenue, NE Suite F Kirkland, Washington 98034	Phone: (425) 823-6727 Fax: (425) 814-0924 Email: Kirkland.WA@SylvanLearning.com	A1 Education Services, Inc.	Mansi Garg Hemant Garg
Washington	Center #218 1327 112th Street SE Suite A2 Everett, Washington 98208	Phone: (425) 348-6767 Fax: (866) 302-7506 Email: Everett.WA@SylvanLearning.com	A1 Education Services, Inc.	Mansi Garg Hemant Garg
Washington	Center #219 4152 Meridan Street Suite #207 Bellingham, Washington 98226	Phone: (360) 738-9590 Fax: (360) 738-0700 Email: bellingham@sylvan-wa.com	Havanah, LLC	Brehana Fregoso Robert Harvey
Washington	Center #223 723 Ave D, Suite F Snohomish, Washington 98290	Phone: (360) 568-2600 Fax: (360) 568-5056 Email: Snohomish.WA@SylvanLearning.com	A1 Education Services, Inc.	Mansi Garg Hemant Garg
Washington	Center #226 81 Keene Road Richland, Washington 99352	Phone: (509) 736-1509 Fax: (509) 736-1302 Email: info@tcsylvan.com	Sugarplum Associates, LLC	Carol Currie Carol Jo Rodriguez
Washington	Center #235 8710 271st St. NW PO Box 133 Stanwood, Washington 98292	Phone: (360) 629-9989 Fax: (360) 939-2645 Email: sylvan235@sylvannw.com	A1 Education Services, Inc.	Mansi Garg Hemant Garg
Washington	Center #236 380 SE Pioneer Way, Suite 101 Oak Harbor, Washington 98277	Phone: (360) 675-8010 Fax: (360) 675-5492 Email: ohsylvan@gmail.com	The Wilson Learning Academy LLC	Kevin Wilson Michelle Wilson
Washington	Center #239 22141 SE 237th Street, Suite 206 Maple Valley, Washington 98038	Phone: (425) 413-2625 Fax: (425) 413-0306 Email: admin@mvsylvan.com	Stem King LLC	Kevin Moen

Washington	Center #240 234 E Main Street Auburn, Washington 98002	Phone: (253) 218-0590 Fax: (253) 218-0593 Email: admissions.sylvan@comcast.net	Puget Sound Learning Centers, LLC	Gail Smith David Smith, III
Washington	Center #241 203 N Blakeley Street, Unit 103 Monroe, Washington 98272	Phone: (360) 794-5554 Email: sylvan241@sylvannw.com	A1 Education Services, Inc.	Mansi Garg Hemant Garg
Washington	Center #242 12660-B 164th Avenue SE Renton, Washington 98059	Phone: (425) 282-6595 Fax: (425) 282-6596 Email: cdrenton@essylvan.com	RKC Education LLC	Mansi Garg Hemant Garg
Wisconsin	Center #4903 15060 W. Greenfield Suite 104 Brookfield, Wisconsin 53005	Phone: (262) 796-0808 Fax: (262) 796-0804 Email: Sylvan4903@gmail.com	Van Milligen, Kathleen	Kathleen Van Milligen
Wisconsin	Center #4908 7863 Green Bay Rd. Kenosha, Wisconsin 53142	Phone: (262) 694-6286 Fax: (262) 694-6289 Email: learning@sylvankenosha.com	S B Associates, Inc.	John Sykora Joseph Sykora Mary Sykora
Wisconsin	Center #4916 3409 Golf Road Eau Claire, Wisconsin 54701	Phone: (715) 858-1234 Email: kayann@sylvancenter.org	Advantage Learning, Inc.	Corey Bock Kayann Bock
Wisconsin	Center #4928 904 Carmichael Rd Hudson, Wisconsin 54016	Phone: (715) 690-2722 Email: Hudson@SylvanCenter.org	Advantage Learning, Inc.	Corey Bock Kayann Bock

### Franchisees Who Have Signed But Not Yet Opened as of February 15, 2024

Territory Number	City	State	Contact Details	Licensee	Franchisee Name(s)
521	Bay Park	California	Phone: 619-572-3444 Email: rieley17@gmail.com	Denham Enterprises, LLC	Chad Denham
500	Folsom	California	Phone: 202-887-0447 Email: sara.qadir@gmail.com	Inspired Learning LLC	Shabnum Yang
502	Granite Bay	California	Phone: 202-887-0447 Email: sara.qadir@gmail.com	Inspired Learning LLC	Shabnum Yang
522	Orange	California	Phone: 310-467-1366 Email: sandeep@sylvanofyorbalinda.com	SANPACT, LLC	Sandeep Malik
8598	Pico Rivera	California	8		Adriene White Kali Wade
523	South Whittier	California	Phone: 818-639-9692 Email: toddcrabtree@sylvansocal.com	The Southern California Learning Corporation	Todd Crabtree
2903	North Wilmington	Delaware	Phone: 610-800-7763 / 302-858-7242 Email: adriabondanza@yahoo.com	Achievement Awaits, LLC	Adria Bondanza Michael Walters
8637	Apopka	Florida	Phone: 321-332-5312 Email: megan.a.cabrera@gmail.com	LearnBerry, LLC	Maria Cabrera
8638	Sumter	Florida	Phone: 352-603-2106 / 352-603-3085 Email: gilleyspaude@gmail.com / gnicarst@gmail.com	GT Learning Group, LLC	Gilley Spaude Tod Spaude
8636	Tavares	Florida	Phone: 321-332-5312 Email: megan.a.cabrera@gmail.com	LearnBerry, LLC	Maria Cabrera
2021	Coeur D'Alene	Idaho	Phone: 208-918-1674 Email: daksh.academy.boise@gmail.com	Daksh Education LLC	Sudhaker Chhabra

2020	Twin Falls	Idaho	Phone: 208-918-1674 Email: daksh.academy.boise@gmail.com	Daksh Institute LLC	Sudhaker Chhabra
3319	Ankeny	Iowa	Phone: 515-988-9955 / 641-849-0912 Email: jeffhohensee@mchsi.com / aprilsamp@gmail.com		Jeff Hohensee April Samp
3321	Des Moines	Iowa	Phone: 515-988-9955 / 641-849-0912 Email: jeffhohensee@mchsi.com / aprilsamp@gmail.com	Email: jeffhohensee@mchsi.com /	
3320	Saylorville	Iowa	Phone: 515-988-9955 / 641-849-0912 Email: jeffhohensee@mchsi.com / aprilsamp@gmail.com	FCH L.L.C.	Jeff Hohensee April Samp
3404	Topsham	Maine	Phone: 802-373-8611 Email: leahlaclaire@gmail.com	23rd State Learning, LLC	Leah LaClaire
4155	Closter	New Jersey	Phone: 201-675-7480 Email: mike56ft@gmail.com	Sirius Innovations Inc.	Michael Hazen
4153	Newark	New Jersey	Phone: 908-868-1162 Email: paramountfamilyservices@gmail.com	The Wilson Learning Academy LLC  LaRhonda Lew	
4154	Park Ridge	New Jersey	Phone: 201-675-7480 Email: mike56ft@gmail.com	Sirius Innovations Inc.	Michael Hazen
4156	Sparta	New Jersey	Phone: 973-964-4154 / 973-390-1620 Email: gjc@trueprod.com / cristinemercer@gmail.com	Sylvan Sparta LLC	George Crescione Cristine Crescione
4282	Glenville	New York	Phone: 518-423-3266 / 518-688-7955 Email: Wilton.NY@sylvanlearning.com / jwaterfile@gmail.com	Skills for Success, LLC	Audrey Waterfield John Waterfield
1668	Lexington	North Carolina	Phone: 336-766-0460 Email: Lexington.NC@sylvanlearning.com	Spartan Akademia LLC	Michelle Varoutsos John Varoutsos

1665	Stallings	North Carolina	Phone: 704-299-8088 Email: Hopewelleduinc@yahoo.com	Hopewell Educational Enterprises, Inc.	Marty Conner
1488	Dublin	Ohio	Phone: 609-902-2037 / 609-865-9989 / 650-814-5291 Email: prabhakar.kasu@gmail.com / ashok1sanjose@gmail.com / jyothsna.kasu@gmail.com	Educare Learning Solutions LLC	Prabhakar Kasu Jyothsna Kasu Ashok Vuppala
1785	Harrisburg	Pennsylvania	Phone: 717-443-7725 / 413-557-4830 Email: mayamdulal@gmail.com / adhikari.naradm@gmail.com	Pro Tutoring Services LLC	Maya Dulal Narad Adhikari
1786	Media	Pennsylvania	Phone: 484-431-5548 Email: david.neill@yahoo.com	DJ Neill Management, LLC	David Neill
1041	Memphis	Tennessee	Phone: 850-377-0728 / 850-377-0719 Email: cartezone@gmail.com / kativasimone@hotmail.com	New Beginnings in Education LLC	Fredrick Perkins Kativa Perkins
8364	Bulverde	Texas	Phone: 419-234-1014 Email: j.lovin@sylvantx.com	NWSA Educators, Inc.	Joan Lovin
8368	Cinco Ranch	Texas	Phone: 713-410-7021 Email: zibaagowani@gmail.com	AZG Ventures LLC	Zibaa Pirani
8360	Conroe	Texas	Phone: 469-774-1862 / 469-744-5571 Email: thomasc01@hotmail.com / thomasrl07@hotmail.com	CR Thomas, LLC	Clarence Thomas Renitta Thomas
8369	Greatwood	Texas	Phone: 713-410-7021 Email: zibaagowani@gmail.com	AZG Ventures LLC	Zibaa Pirani
8370	Hunters Creek	Texas	Phone: 713-410-7021 Email: zibaagowani@gmail.com	AZG Ventures LLC	Zibaa Pirani
8367	Lake Jackson	Texas	Phone: 305-942-7243 Email: jwilliamsmsed@gmail.com	JWilliamsMSED LLC	Jasmine Williams
8366	Liberty	Texas	Phone: 713-703-9879 / 404-821-5685 Email: amfonteneaux@yahoo.com / blairnpb84@gmail.com	A Team Learning LLC	Arthur Fonteneaux Nina Fonteneaux

8365	Spring Branch	Texas	Phone: 713-703-9879 / 404-821-5685 Email: amfonteneaux@yahoo.com / blairnpb84@gmail.com	A Team Learning LLC	Arthur Fonteneaux Nina Fonteneaux
8362	Westbranch	Texas	Phone: 713-703-9879 / 404-821-5685 Email: amfonteneaux@yahoo.com / blairnpb84@gmail.com	A Team Learning LLC	Arthur Fonteneaux Nina Fonteneaux
4806	Morgantown	West Virginia	Phone: 724986-3132 / 412-889-9450 Email: sobrien157@mac.com / katelyn0101@me.com	Lakeside Learning, LLC	Suzan O'Brien Katelyn O'Brien

## Franchisees Which Have Been Terminated, Canceled, Not Renewed, or Otherwise Ceased to do Business or Which Have Not Communicated as of December 31, 2023

Previous Owner	Home City	State	Last Phone No. Recorded
Jason McLaughlin	Hoover	AL	205-908-4606
Carol Wettig	Prattville	AL	804-690-0557
Terry Wettig	Prattville	AL	804-690-0557
Venkat Bussa	Danville	CA	510-745-7800
Rand Bleimeister	Encino	CA	310-773-7555
Todd Crabtree	Grenada Hills	CA	310-373-0073
Bill Kramer	Oakland	CA	510-856-9000
Jack McAboy	Piedmont	CA	510-856-9000
Christine Vallen	Arvada	CO	303-920-7424
Mark Bellestri	Castle Rock	CO	(612) 481-0406
Jennifer Bellestri	Castle Rock	CO	(612) 242-0875
Dave Sekula	Jacksonville	FL	904-757-2220
Ginny Sekula	Jacksonville	FL	904-757-2220
Sultan Jan	Oldsmar	FL	205-514-6768
William Ramos Da Silva	Orlando	FL	407-688-2017
Gloria Swardenski	Pensacola	FL	850-474-9022
An Hong Tran	Alpharetta	GA	404-547-6722
Robert Keller	Bloomington	IL	708-488-0161
Nontaporn Konkaeow	Elgin	IL	312-404-9811
Sally Figueroa	Carmel	IN	850-474-9022
Smita Thakkar	Tyngsboro	MA	978-799-9035
Rick Rusthoven	Byron Center	MI	616-516-4520
Christopher Sesi	West Bloomfield	MI	248-756-6778
Christopher Sesi	West Bloomfield	MI	248-756-6778
Christopher Sesi	West Bloomfield	MI	248-756-6778
Christopher Sesi	West Bloomfield	MI	248-756-6778
Caroline Lindeman	Minnetonka	MN	303-920-7424
Mike Kelley	St. Cloud	MN	763-712-1118
Kelly Conway	Greenville	NC	706-868-1888
Krista Martin	East Wakefield	NH	717-333-4233
Tristan Plummer	East Wakefield	NH	603-871-8575

Patrick McNamara Estate	Albany	NY	999-999-9999
Patrick McNamara Estate	Albany	NY	999-999-9999
Patricia Russo	Albertson	NY	516-743-9943
Greg Caruso	White Plains	NY	914-237-4396
Mital Rathod	Olmsted Township	ОН	330-523-0396
Cliff Crawford	Portland	OR	503-249-8061
Kimberli Dorn	Salem	OR	503-930-8972
Janet Crocker	Clarksville	TN	931-647-5811
Jeffrey Phillips	Nashville	TN	423-239-9215
Mike Sullivan	Abilene	TX	806-785-4400
Cindy Sullivan	Abilene	TX	806-785-4400
Matthew Sullivan	Abilene	TX	806-785-4400
Susan Fairbairn	Austin	TX	512-336-1818
Robert Jones	Austin	TX	512-336-1818
Rita McKissick	Forney	TX	214-850-2787
Cathryn Butts	Katy	TX	713-291-5565
Dennis Butts	Katy	TX	713-397-1127
Mikhael Mitchell	Missouri City	TX	832-646-5195
O.Steven Jones	San Antonio	TX	602-953-3070
Tamara Jones	San Antonio	TX	602-953-3070
Prabhjeet Singh	San Antonio	TX	210-367-7652
Sapna Chordia	Sterling	VA	703-382-6300
Ashish Gandhi	Sterling	VA	703-382-6300
David Smith III	Lakewood	WA	253-848-0771
Gail Smith	Lakewood	WA	253-848-0771
Jason Jensen	Gilette	WY	307-299-2802
Jamie Sanders	Gilette	WY	307-660-7950

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

## LIST OF CORPORATE OR AFFILIATE-OWNED OUTLETS AS OF DECEMBER 31, 2023

NAME	ADDRESS	PHONE
Maryland Learning Centers, LLC	1203 Baltimore Pike Suite E, Bel Air, MD 21014	(410) 656-8832
Maryland Learning Centers, LLC	405 Frederick Rd #5, Catonsville, MD 21228	(443) 334-6156
Maryland Learning Centers, LLC	8801 Centre Park Dr #5, Columbia, MD 21045	(410) 656-8859
Maryland Learning Centers, LLC	2151 Defense Hwy #3, Crofton, MD 21114	(410) 656-8421
Maryland Learning Centers, LLC	1840 York Rd unit d, Timonium, MD 21093	(410) 656-8368

62

## EXHIBIT I TO THE FRANCHISE DISCLOSURE DOCUMENT

#### SAMPLE FORM OF GENERAL RELEASE

#### **GENERAL RELEASE**

This general release (the "<u>General Release</u>") is made and entered into on by and between Sylvan Learning, LLC ("<u>Licensor</u>"), ("<u>Licensee</u>"), and (each a "<u>Guarantor</u>," together with the Licensee, the "<u>Licensee Parties</u>"). For good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Licensee Parties agree as follows.

- 1. To the maximum extent permitted by applicable law, the Licensee Parties on behalf of themselves and each of their past and present parents, subsidiaries, affiliates, predecessors, successors, assigns, officers, directors, shareholders, partners, owners, members, managers, agents, attorneys, employees, and representatives (together with the Licensee Parties, the "Releasing Parties") do remise, release, waive, and forever discharge Sylvan Learning, LLC, Sylvan In-Home, LLC, Unleashed Brands, LLC, Unleashed Services, LLC and each of their respective past and present parents, subsidiaries, affiliates, predecessors, successors, assigns, officers, directors, shareholders, partners, owners, members, managers, agents, attorneys, employees, and representatives (collectively, the "Licensor Parties") from any and all claims, demands, obligations, liabilities, actions, proceedings, agreements, debts, demands, damages, accounts, charges, invoices, discounts, incentives, allowances, controversies, expenses, attorneys' fees, suits, arbitrations, and causes of action whatsoever, in law or equity, whether known or unknown, past, present, or future, which the Releasing Parties have, have had, claim to have, or may have against the Licensor Parties including, but not limited to, any and all claims and damages in any way arising out of or related to (1) that franchise or license agreement between Licensor and Licensee dated regarding the operation of a Sylvan franchise located at , as amended; (2) any other License Agreement, license agreement, or any other contract between any Releasing Party and any Licensor Party; (3) the offer and sale of any Sylvan franchise opportunity, (4) the disclosure requirements under the FTC Franchise Rule (16 CFR et seq); (5) any other state franchise law, (6) any alleged misrepresentations made by the Licensor Parties in the sale of a franchise to the Releasing Parties or otherwise; (7) any and all claims arising under local, state, and federal laws, rules, and ordinances, whether statutory or under common law; (8) Sylvan franchise located at ; and (9) any relationship between the Releasing Parties and the Licensor Parties.
- 2. The Releasing Parties acknowledge this General Release extends to all claims the Releasing Parties do not know or suspect to exist in their favor at the time of executing this General Release, which if were known to exist may have materially affected the decision to enter into this General Release. The Releasing Parties understand the facts in respect of which this General Release is given may hereafter turn out to be other than or different from the facts known or believed to be true and agree this General Release shall be in all respects effective and not subject to termination or rescission by any such difference in facts. By executing this General Release, the Releasing Parties expressly assume the risk of the facts turning out to be different and agree this General Release shall be in all respects effective and not subject to termination or rescission by any such difference in facts. The Releasing Parties acknowledge and agree that they have had the opportunity to seek the advice of and are represented by independent legal counsel and have read and understood all the terms and provisions of this General Release. The Releasing Parties, jointly and individually, covenant and agree that none of them will commence, maintain, participate in, or prosecute any claim, demand, suit, action, or cause of action against the Licensor Parties concerning the claims released in this General Release.
- 3. This General Release represents the entire agreement of the parties regarding the subject matter hereof and may only be modified in writing.

[If Releasor is domiciled or has his or her principal place of business in the State of California]

#### WAIVER OF SECTION 1542 OF THE CALIFORNIA CIVIL CODE.

("Releasor") for myself and on behalf of all persons acting by or through me, acknowledge that I am familiar with Section 1542 of the California Civil Code, which reads as follows:

1

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASING PARTY.

Releasor hereby waives and relinquishes every right or benefit which I have under Section 1542 of the Civil Code of the State of California, and any similar statute under any other state or federal law, to the fullest extent that they may lawfully waive such right or benefit. In connection with this waiver and relinquishment, Releasor acknowledges he or she may hereafter discover facts in addition to or different from those which he or she now knows or believes to be true with respect to the claims herein released, but that it is the parties' intention, subject to the terms and conditions of this General Release, to fully, finally and forever settle and release all such claims, known or unknown, suspected or unsuspected, which now exist, may exist or did exist. In furtherance of such intention, the releases given in this General Release shall be and remain in effect as full and complete releases, notwithstanding the discovery or existence of any such additional or different facts.

Releasor warrants and represents the release set forth above is a complete defense to any claim encompassed by its terms, and covenants not to initiate, prosecute, or otherwise participate in any action or proceeding in any court, agency, or other forum, either affirmatively or by way of cross-claim, defense, or counterclaim, against any person or entity released under this General Release with respect to any claim or cause of action released under this General Release..

[If Releasor is domiciled or has his or her principal place of business in the State of Washington or if the franchised business is located in the State of Washington]

Notwithstanding anything to the contrary, this General Release does not apply to any claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

**IN WITNESS WHEREOF**, the parties hereto have executed this General Release as of the dates set forth below.

# 

[This General Release will be modified as necessary for consistency with any state law regulating franchising.]

## EXHIBIT J TO THE FRANCHISE DISCLOSURE DOCUMENT

#### FRANCHISE DISCLOSURE QUESTIONNAIRE

(THIS FRANCHISEE DISCLOSURE QUESTIONNAIRE WILL NOT BE USED IF THE FRANCHISE IS TO BE OPERATED IN, OR YOU ARE A RESIDENT OF, CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, RHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR WISCONSIN).

(THIS FRANCHISEE DISCLOSURE QUESTIONNAIRE WILL NOT BE USED IF THE FRANCHISE IS TO BE OPERATED IN, OR YOU ARE A RESIDENT OF, CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, RHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR WISCONSIN).

#### FRANCHISE DISCLOSURE QUESTIONNAIRE

As you know, SYLVAN LEARNING, LLC ("we" or "us") and you are preparing to enter into a License Agreement for the operation of a SYLVAN franchise. The purpose of this Questionnaire is to determine whether any statements or promises were made to you that we have not authorized or that may be untrue, inaccurate or misleading, to be certain that you have been properly represented in this transaction, and to be certain that you understand the limitations on claims you may make by reason of the purchase and operation of your franchise. You cannot sign or date this Questionnaire the same day as the Receipt for the disclosure document, but you must sign and date it the same day you sign the License Agreement and pay your franchise fee. Please review each of the following questions carefully and provide honest responses to each question. If you answer "No" to any of the questions below, please explain your answer by emailing Franchising@UnleashedBrands.com.

Yes	_ No	1.	Have you received and personally reviewed SYLVAN License Agreement and each attachment, exhibit, or schedule attached to it?
Yes	_ No	2.	Have you received and personally reviewed SYLVAN disclosure document we provided?
Yes	_ No	3.	Did you sign a receipt for SYLVAN disclosure document indicating the date you received it?
Yes	_ No	4.	Do you understand all the information contained in SYLVAN disclosure document and License Agreement?
Yes	_ No	5.	A) Have you had ample time and the opportunity to review SYLVAN disclosure document and SYLVAN License Agreement with a lawyer, accountant or other professional advisor?
Yes	_ No		B) Have you had the opportunity to discuss the benefits and risks of operating a SYLVAN franchise with your professional advisor?
Yes	_ No		C) Did you discuss the benefits and risks of operating a SYLVAN franchise with an existing SYLVAN franchisee?
Yes	_ No	6.	Do you understand the risks of operating a SYLVAN franchise?
Yes	_ No	7.	Do you understand the success or failure of your SYLVAN franchise will depend in large part upon your skills, abilities and efforts and those of the person you employ, as well as many factors beyond your control such as weather, competition, interest rates, the economy, inflation, labor and supply costs, lease terms and the marketplace?
Yes	_ No	8.	Do you understand we are not obligated to provide assistance to you in finding and securing a location for your SYLVAN Franchised Business?
Yes	_ No	9.	A) Do you understand all disputes or claims you may have arising out of or relating to SYLVAN License Agreement must be brought in the judicial district in which our principal place of business is located, if not resolved informally?
Yes	_ No		B) Do you understand SYLVAN License Agreement provides you can only collect compensatory damages on any claim under or relating to SYLVAN License Agreement, and not any punitive, special, consequential or exemplary damages?

<u>NORTH DAKOTA, R</u> WISCONSIN) <u>.</u>	CHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR		
. <u> </u>	. Do you understand that your Designated Manager must successfully complete		
10 10	our initial training program?		
Yes No 11	. Do you understand we do not have to sell you a franchise or additional franchises or consent to your purchase of existing franchises?		
Yes No 12	. Is it true that, except as provided in Item 19 of our FDD, we and our affiliates have made no representation, warranty, promise, guaranty, prediction projection, or other statement, and given no information, as to the future, past likely, or possible income, sales volume, or profitability, expected or otherwise of SYLVAN franchise or any other business?		
Yes No 13	. Do you understand that actual results vary from unit to unit and from time period to time period, and we cannot estimate, project, or predict the results of any particular SYLVAN business?		
Yes No 14	. Do you acknowledge that you are an independent contractor and responsible for running your own SYLVAN business and that we do not have any authority to control, hire, or fire your employees?		
Yes No 15	. Is it true that neither we or our affiliates, or any of our or our affiliates employees, have provided you with services or advice that is legal, accounting or other professional services or advice?		
Yes No 16	. A) Do you understand that the U.S. Government has enacted anti-terrorist legislation that prevents us from carrying on business with any suspected terrorist or anyone associated directly or indirectly with terrorist activities?		
Yes No	B) Is it true that you have never been a suspected terrorist or associated directly or indirectly with terrorist activities?		
Yes No	C) Do you understand that we will not approve your purchase of a SYLVAN franchise if you are a suspected terrorist or associated directly or indirectly with terrorist activity?		
Yes No	D) Is it true that you are not purchasing a SYLVAN franchise with the intent of purpose of violating any anti-terrorism law, or for obtaining money to be contributed to a terrorist organization?		
	Only: Such representations are not intended to nor will they act as a release liability incurred under the Maryland Franchise Registration and Disclosure Law		
	ts Only: Such representations are not intended to nor will they act as a waiver of er the Washington Investment Protection Act, RCW 19.100, and the rules adopted		
Print Name			
Signature	 Date		
If signing on behalf of a	corporation or other entity, please complete the following:		
Name	of Entity Title		

(THIS FRANCHISEE DISCLOSURE QUESTIONNAIRE WILL NOT BE USED IF THE FRANCHISE IS TO BE OPERATED IN, OR YOU ARE A RESIDENT OF, CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK,

## EXHIBIT K TO THE FRANCHISE DISCLOSURE DOCUMENT

#### STATE EFFECTIVE DATES

#### **State Effective Dates**

The following states require that the disclosure document be registered or filed with the state or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington and Wisconsin.

This disclosure document is registered, on file, or exempt from registration in the following states having franchise registration and disclosure laws, with the following effective dates:

State	Effective Date
California	March 14, 2024, amended May 10, 2024
Illinois	March 14, 2024, amended May 10, 2024
Indiana	March 14, 2024, amended May 10, 2024
Maryland	March 14, 2024, amended May 10, 2024
Michigan	March 14, 2024, amended May 10, 2024
Minnesota	Pending
New York	March 14, 2024, amended May 10, 2024
North Dakota	March 14, 2024, amended May 10, 2024
Rhode Island	March 14, 2024, amended May 10, 2024
South Dakota	March 14, 2024, amended May 10, 2024
Virginia	Pending
Washington	March 14, 2024, amended May 10, 2024
Wisconsin	March 14, 2024, amended May 10, 2024

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

## EXHIBIT L TO THE FRANCHISE DISCLOSURE DOCUMENT

#### **RECEIPTS**

#### RECEIPT

This disclosure document summarizes certain provisions of the License Agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Sylvan offers you a franchise, it must provide this disclosure document to you 14 calendar days before you sign a binding agreement or make a payment with franchisor or an affiliate in connection with the proposed franchise sale, or sooner if required by applicable law.

New York requires that we give you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first. Iowa requires that we give you this disclosure document at the earlier of the first personal meeting or 14 calendar days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship.

If Sylvan does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state regulatory agency listed in <u>Exhibit F</u>. Franchisor authorizes the respective state agencies identified on <u>Exhibit F</u> to receive service of process for it in a particular state.

The name, principal business address, and telephone number of the franchise seller offering the franchise is:

Name	Principal Business Address	Telephone Number
Joshua Wall	2350 Airport Freeway, Suite 505, Bedford, TX 76022	877.958.9716

Issuance date: March 8, 2024, as amended May 10, 2024.

I received a disclosure document dated March 8, 2024, as amended May 10, 2024 (and with effective dates of state registration as listed on the State Effective Dates page) that included the following Exhibits:

	Keep this copy for your	r records.		
Company Nan	ne Author	ized Signatory and Signature		
If signing on be	half of a company in addition to individually, plea	se complete the following:		
Print Name	Signature	Date		
Exhibit I Exhibit J Exhibit K Exhibit L	Who Have Left the System Sample Form of General Release Franchise Disclosure Questionnaire State Effective Dates Receipts			
Exhibit G Exhibit H	Financial Statements Franchises in Operation, Franchisees Who Have Signed But Not Yet Opened, and Franchisees			
Exhibit F	List of State Administrators and Agents for Service of Process			
Exhibit C Exhibit D Exhibit E	ACE IT! Tutoring System License Agreement Multi-Territory Incentive Plan Amendment Operations Manual Table of Contents			
Exhibit A Exhibit B	License Agreement for Sylvan and SylvanSyno Sylvan Edge Package License Agreement	e, Attachments, and State-Specific Amendments		

#### RECEIPT

This disclosure document summarizes certain provisions of the License Agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Sylvan offers you a franchise, it must provide this disclosure document to you 14 calendar days before you sign a binding agreement or make a payment with franchisor or an affiliate in connection with the proposed franchise sale, or sooner if required by applicable law.

New York requires that we give you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first. Iowa requires that we give you this disclosure document at the earlier of the first personal meeting or 14 calendar days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship.

If Sylvan does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state regulatory agency listed in <u>Exhibit F</u>. Franchisor authorizes the respective state agencies identified on <u>Exhibit F</u> to receive service of process for it in a particular state.

The name, principal business address, and telephone number of the franchise seller offering the franchise is:

Name	Principal Business Address	Telephone Number
Joshua Wall	2350 Airport Freeway, Suite 505, Bedford, TX 76022	877.958.9716

Issuance date: March 8, 2024, as amended May 10, 2024.

I received a disclosure document dated March 8, 2024, as amended May 10, 2024 (and with effective dates of state registration as listed on the State Effective Dates page) that included the following Exhibits:

Exhibit A	License Agreement for Sylvan and Sylvan	ylvanSync, Attachments, and Sta	te-Specific Amendments	
Exhibit B	Sylvan Edge Package License Agreen	ment		
Exhibit C	ACE IT! Tutoring System License A	greement		
Exhibit D	Multi-Territory Incentive Plan Amendment			
Exhibit E	Operations Manual Table of Contents			
Exhibit F	•			
Exhibit G	Financial Statements			
Exhibit H	Franchises in Operation, Franchisees Who Have Signed But Not Yet Opened, and Franchisees			
	Who Have Left the System			
Exhibit I	Sample Form of General Release			
Exhibit J	Franchise Disclosure Questionnaire			
Exhibit K	State Effective Dates			
Exhibit L	Receipts			
Print Name	Signature		Date	
If signing on bel	alf of a company in addition to individu	nally, please complete the follows	ing:	
Company Name		Authorized Signatory and Sign	nature	

Please sign this copy of the receipt, date your signature, and return it by mail or email to Joshua Wall, 2350 Airport Freeway, Suite 505, Bedford, TX 76022.